

# **OpenAIR@RGU**

# The Open Access Institutional Repository at Robert Gordon University

http://openair.rgu.ac.uk

**Citation Details** 

# Citation for the version of the work held in 'OpenAIR@RGU':

WARASIHA, B., 2008. Cytochrome P450 mRNA profile in human breast cancer cell lines. Available from *OpenAIR@RGU*. [online]. Available from: http://openair.rgu.ac.uk

### Copyright

Items in 'OpenAIR@RGU', Robert Gordon University Open Access Institutional Repository, are protected by copyright and intellectual property law. If you believe that any material held in 'OpenAIR@RGU' infringes copyright, please contact <u>openair-help@rgu.ac.uk</u> with details. The item will be removed from the repository while the claim is investigated.

# CYTOCHROME P450 mRNA PROFILE IN HUMAN BREAST CANCER CELL LINES

**BENJAMART WARASIHA** 

A thesis submitted in partial fulfilment of the requirements of The Robert Gordon University for the degree of Doctor of Philosophy

> School of Pharmacy and Life Sciences, The Robert Gordon University, Aberdeen. United Kingdom.

> > May 2008

## Abstract

Cytochrome P450 enzymes (P450s) are involved in cancer development and treatment due to their roles in the oxidative metabolism of various endogenous (e.g. oestrogen) and exogenous (e.g. tamoxifen) compounds. It is well-known that intermediate P450 metabolites derived from oestrogen metabolism are associated with breast carcinogenesis. The main aim of this project was to profile the cytochrome P450 and P450-regulatory nuclear receptor mRNAs in a series of breast cancer cell lines (BCCs) and compare this profile with normal breast cells. This study used the qualitative reverse transcriptase-polymerase chain reaction (RT-PCR) to detect mRNA expression of target genes. Results showed CYP1B1, CYP2D6, CYP2J2, CYP2R1, CYP2U1 and CYP4X1 mRNA to be present in all cell lines. CYP2A6, CYP2C8, CYP2C18, CYP2F1 and CYP4Z1 mRNA were expressed in oestrogen receptor (ER)-positive Caucasian and ER-negative Afro-Caribbean BCCs. Although no differences in P450 mRNA were observed between the different ethnic groups, these preliminary findings suggest potential similarities in the ER-positive Caucasian and ER-negative Afro-Caribbean BCCs which warrant further investigation

The CYP4Z1 PCR product was identified as two distinct bands. Specific primer sets were used to demonstrate potential intron retention in CYP4Z1. Using established *in vitro* models for the study of regulatory mechanisms of CYP4Z1, T47D and ZR-75-1 breast cancer cell lines were used to determine the appropriate nuclear receptors (i.e. progesterone receptor, glucocorticoid receptor or peroxisome proliferator-activated receptor alpha ). These findings suggest that there may be an alternative receptor mechanism involved in CYP4Z1 mRNA induction in these cells. In conjunction, pre-treatment of these two cell lines with the RNA synthesis inhibitor actinomycin D followed by the agonists showed a significant reduction (p < 0.05) of CYP4Z1 mRNA levels and inhibited CYP4Z1 induction by either progesterone, dexamethasone or pirinixic acid, indicating that these agonists have effects on CYP4Z1 mRNA transcription or stability. In contrast, cycloheximide differentially affected the level of CYP4Z1 mRNA induction by these agonists. Taken together, these results suggest that CYP4Z1 mRNA induction in T47D and ZR-75-1 is mediated through differential cell type specific regulatory mechanisms and there is evidence for differential regulation of the splice variants.

**Keywords:** Cytochrome P450 mRNA, cytochrome P450-regulatory nuclear receptor mRNA, Oestrogen receptor (ER)-positive and negative breast cancer cell lines (BCCs), Caucasian and Afro-Caribbean BCCs, CYP4Z1 alternative splicing, regulation of CYP4Z1

# Acknowledgements

I wish to express my sincere gratitude to the following people for their help, support, guidance and encouragement during the course of my studies.

My Director of Studies, Dr Morag McFadyen, for introducing me to the world of cytochrome P450s, her valuable suggestions, patience and encouragement throughout my PhD studies. Dr Rachel Knott, for her constant guidance, support and participating as my supervisor. Professor Angela Drake-Holland, for her helpful suggestions and discussion on the project. Mrs Dorothy Moir, for helping me out with practical support in the molecular biology laboratory, her wonderful friendship and motivation. Many thanks to Professor Mark Noble, Dr Graeme Kay and Dr Kennedy Nelson for taking the time to critically evaluate my thesis.

Professor Terry Healey, Head of the School of Pharmacy and Life Sciences, for giving me the opportunity to pursue my studies. Dr Susanne Boyle, for her support and advice in the first year of my studies. Mr Brian De Jonckheere, for information technology assistance. All of the technicians in the School of Pharmacy who have helped with my project and been wonderful friends. Mrs Elizabeth Tomchak, my English tutor, for helping me improve my English skills. Mr Martin Simpson, the Research Degrees officer, for his support and assistance. Associate Professor Veerapol Kukongwiriyapan from the Department of Pharmacology, Faculty of Medicine, Khon Kaen University, Thailand, for his support and guidance. Sarah Garry, Michelle Frazier, Eimear Molloy and Ian Lonergan, Honours students at the School of Pharmacy and Life Sciences, for the results of the preliminary work in Chapter 4.

Dr Colin Thompson, Mr Ray Mulrooney, Dr Bridgeen McCaughan, Mrs Julie Young, Miss Anita Weidman, Miss Nicola Silva, Mr Vibhu Paudyal and all other colleagues for sharing their time and friendship both in and out of the laboratory. To Anusorn Thiangsungnoen, Pattarin Kittiboonyakun, Sansanee Wangvoralak, Sirichai Mongkolkiatsri and all of the other Thai students in Aberdeen for their wonderful friendship and encouragement and for making me feel less homesick. To everyone in the Taekwondo and Badminton clubs at The Robert Gordon University, for their friendship. Special thanks to my Taekwondo instructor Miss Lyndzie Jefferey, whose classes always helped me to unwind and relax, no matter how stressful my day in the lab had been. To my wonderful parents, Samruay and Sudjai Warasiha who have taught me never to give up, for their constant care, unconditional love, continuous support and encouragement throughout my life. My two brothers, Taweesit and Supakit Warasiha, and two sisters Somjit Nampha and Jurairat Sriwicha including their families for their understanding, encouragement, support and love. My boyfriend, Tim Cushnie, for his help and encouragement and for sharing my frustration and happy moments during the studies-you have helped keep me smiling.

The Royal Thai Government Scholarship, for financial support throughout my studies. The School of Pharmacy and Life Sciences at The Robert Gordon University, for financial aid during the writing up period. The British Association for Cancer Research (BACR), for the travel grants to attend and present my research in the 2006 and 2007 National Cancer Research Institute (NCRI) Conference in Birmingham.

# **List of Publications**

### Abstract:

 Warasiha B, Knott R, <u>Boyle S.</u> Investigation of the role of xenobiotic metabolising enzymes in breast cancer development. *The Journal of Nutrition*. 2004 December; 134 (12 Suppl): 3525S. (Poster presented as part of the International Research Conference on Food, Nutrition and Cancer held in Washington, DC, USA, July, 2004).

#### Conference proceedings:

- Warasiha B, Knott R, McFadyen M. (2007) Comparative analysis of Cytochrome P450 mRNA Expression between Human Caucasian and Afro-Caribbean Breast Cancer Cell Lines. In the proceedings of National Cancer Research Institute (NCRI) Cancer Conference October 2007 at the ICC in Birmingham, UK. (Oral and poster presentation)
- Warasiha B, Knott R, McFadyen M. (2007) Differential Cytochrome P450 mRNA Profile between Non-invasive and Invasive Breast Cancer Phenotypes. In the proceedings of National Cancer Research Institute (NCRI) Cancer Conference October 2007 at the ICC in Birmingham, UK. (Poster presentation)
- Kay G, Cairns D, McCaughan B, Warasiha B. (2007) The Design, Synthesis and Biological Evaluation of Novel Prodrugs for the Treatment of Cystinosis. In the proceedings of British Conference Pharmaceutical (BCP) September 2007 at the Manchester central in Manchester, UK. (Poster presentation)
- Kay G, Warasiha B, Melville M, Mincher DJ. (2007) Design, Synthesis and Biological Evaluation of Novel DNA Binding Agents. In the proceedings of British Conference Pharmaceutical (BCP) September 2007 at the Manchester central in Manchester, UK. (Poster presentation)
- 5. Warasiha B, Knott R, McFadyen M. (2006) Profile of cytochrome P450 mRNA expression in oestrogen positive and negative breast cancer cell lines. In the Proceedings of National Cancer Research Institute (NCRI) Cancer Conference October 2006 at the ICC in Birmingham, UK. (Poster presentation)

# **Table of Contents**

Title pag	2	i			
Abstract ii					
Acknow	edgements	iv			
List of F	List of Publications				
Table of	Contents	-vii			
List of A	obreviations	xi			
Chapte	1: General Introduction	1			
1.1	Introduction	-1			
1.2	Breast cancer	.4			
	1.2.1 Overview of breast cancer	4			
	1.2.2 The treatment of breast cancer	6			
1.3	The use of breast cancer cell lines as models	·10			
1.4	Cytochrome P450 enzymes (P450s)	13			
	1.4.1 Introduction to cytochrome P450s	-13			
	1.4.2 Nomenclature	15			
	1.4.3 Biochemical characteristic features	15			
1.5	Cytochrome P450 expression and function	16			
	1.5.1 CYP1 family	-17			
	1.5.2 CYP2 family	-19			
	1.5.3 CYP3 family	-28			
	1.5.4 CYP4 family	-30			
	1.5.5 CYP5 family	-34			
	1.5.6 CYP7 family	-34			
	1.5.7 CYP8 family	-35			
	L5.8 CYP11 family	36			
	59 CYP17 family	37			
	5 10 CYP19 family	38			
	5 11 CYP20 family	38			
	5 12 CYP21 family	39			
	5 13 CYP24 family	39			
	1.5.13 CYP26 family	40			
	1.5.11 CYP27 family	41			
	1.5.16 CYP39 family	42			
	1.5.17 CYP46 family	43			
	1.5.18 CVP51 family	43			
16	Expression of cytochrome P450s in breast cancer cell lines				
1.0	Suclear recentors involved in regulation of cytochrome P450s	_44			
1.7	7.1 Introduction	-45			
	1.7.2 Aryl hydrocarbon recentor (AhP)	18			
	1.7.2 Aryl hydrocarbon receptor (Ank)	/10			
	1.7.4 Dragnana X recentor (DVD)	49 50			
	1.7.5 Vitamin D recentor (VDP)	50			
	1.7.5  vitallill D ICCCPIOI (VDK)	51			
	1.7.0 PEROXISON PROMIETATOR - ACTIVATED RECEPTOR AIPHA (PPAR $\alpha$ )	52			
	1.7.7 Keunoic acid receptor (KAK)	· 52			
	1.7.0 Farnesold A receptor (FAK)	55			
1.0	1.7.9 Crosstalk between nuclear receptors P450	-33			
1.8	ancer treatment approaches based on cytochrome P450s	50			
1.9	A1ms	-03			

napte	r 2: Ma	aterials and Methods	64
2.1	Introd	uction	64
2.2	Cell c	ulture and cell lines	65
	2.2.1	Growth media	66
		2.2.1.1 GlutaMax <sup>TM</sup> I	66
		2.2.1.2 Roswell Park Memorial Institute medium (RPMI1640)	67
		2.2.1.3 Leibovitz's medium (L15)	67
		2.2.1.4 Dulbecco's Modified Eagle's Medium (DMEM)	68
		2.2.1.5 DMEM/Ham's F12 medium	68
	2.2.2	Cell lines	68
	2.2.3	Growth and maintenance of human normal breast and breast can	cer
		cell lines	69
	2.2.4	Resuscitation of frozen cells	70
	2.2.5	Trypsinisation	73
	2.2.6	Freezing cell cultures	74
<b>~</b> ~	2.2.7	Cell viability determination by trypan blue stain	75
2.3	Purific	cation of total RNA	/6
2.4	Rever	se transcription and first strand cDNA synthesis	· /9 01
2.5	Quant	ative reverse transcriptase-polymerase chain reaction $(RI-PCR)$ -	81
2.6	Design	n of synthesis of oligonucleotide primers	83
2.7	Ineste	a PCK	92
2.8	Agaro	ste ger electrophoresis of DNA	93 05
2.9	Restri	ction digestion	95 04
2.10	Purino		90 10
3.2	Mater	ials and Methods	10
0.2	3.2.1	RNA isolation	10
	3.2.2	Reverse transcription and cDNA synthesis	10
	3.2.3	Optimisation of PCR analysis	10
		3.2.3.1 Initial PCR amplification	10
		3.2.3.2 Optimisation of template cDNA and primer concentration	ns10
		3.2.3.3 Oligonucleotide primer design	10
	3.2.4	Nested PCR	10
	3.2.5	Restriction enzyme digestion of DNA	10
	3.2.6	DNA purification and sequencing	10
3.3	Result	ts	10
	3.3.1	RNA isolation	10
	3.3.2	β <sub>2</sub> -microglobulin RT-PCR	10
	3.3.3	Cytochrome P450 enzyme RT-PCR	10
		3.3.3.1 CYP1 family RT-PCR	10
		3.3.3.2 CYP2 family RT-PCR	11
		3.3.3.3 CYP3 family RT-PCR	11
		3.3.3.4 CYP4 family RT-PCR	11
		3.3.3.5 CYP24 family RT-PCR	11
	<b>a a</b> i	3.3.3.6 CYP26 family RT-PCR	12
	3.3.4	Cytochrome P450-regulatory nuclear receptor RT-PCR	12
		3.3.4.1 Oestrogen receptor (ER) RT-PCR	12
		5.5.4.2 Constitutive androstane receptor (CAR) KT-PCR	- 12
		3.3.4.3 Pregnane X receptor (PXK) KT-PCR	12

3.4	Discus	ssion	- 129
Chante	or 4. Cv	tochrome P450 expression in breast cancer cell lines	- 135
2 1	Introdu	uction	135
4.2	Materi	als and Methods	136
1.2	4 2 1	Cell line and cell culture	136
	422	RNA isolation and cDNA synthesis	136
	4.2.2	Qualitative PCR amplification	136
	4.2.3	Oligonucleotide primer design	- 137
43	Result	S	138
1.5	4.3.1	RNA isolation	138
	4.3.2	β <sub>2</sub> -microglobulin RT-PCR	138
	4.3.3	CYP1 family RT-PCR	- 138
	4.3.4	CYP2 family RT-PCR	- 140
	4.3.5	CYP3 family RT-PCR	- 144
	4.3.6	CYP4 family RT-PCR	- 146
	4.3.7	CYP24 family RT-PCR	148
	4.3.8	CYP26 family RT-PCR	148
	4.3.9	Cytochrome P450-regulatory nuclear receptors RT-PCR	150
	4.3.10	Preliminary work	156
4.4	Discus	ssion	- 164
			100
Cnapte 5 1	r 5: Cn Introdu	aracterisation of CYP4Z1 mKNA in breast cancer cell lines	- 188 
5.1	Matori	action	102
5.2	5 2 1	Qualitative and nested PCR of CVP/71 mRNA	192
	5.2.1 5.2.2	Purification DNA and DNA sequencing	- 194
53	Result	s	196
5.5	5.3.1	Identification of CYP4Z1 splice variants in the cell lines	- 196
5.4	Discus	ssion	202
	<		• • • •
Chapte	er 6: Re	gulation of CYP4Z1 expression in breast cancer cell lines	- 209
6.1	Introd		209
6.2	Materi	als and Methods	212
	6.2.1	Cell culture and CYP4Z1 induction in cultured cells	212
		6.2.1.1 Induction of CYP4Z1 mRNA by the nuclear receptor	212
		6.2.1.2 Evaluation of transcriptional regulation of CVP471	-212
	677	<b>BNA</b> extraction and cDNA synthesis	212
	623	Auglitative RT PCR analysis	213
	62.3	Semi-quantitative PCR amplification	213
	625	Statistical analysis	-214
63	Result	S	215
0.5	631	Profile of constitutive mRNA of the candidate nuclear recentors	215
	0.5.1	6 3 1 1 Progesterone recentor (PGR) RT-PCR	- 215
		6 3 1 2 Glucocorticoid receptor (GCR) RT-PCR	216
		6.3.1.3 Perovisome proliferator-activated recentor alpha (PPAR)	v)
		RT-PCR	~) 217
	632	Induction of CYP471 mRNA by the agonists of PGR GCP and	<u>~1</u> /
	0.3.2	PPARα	- 218
	6.3.3	Evaluation of transcriptional and translational regulation	210
		of CYP4Z1	225

6.4	Discussion	236	
Chapte	r 7: General Discussion	243	
7.1	Introduction	243	
7.2	Profile of cytochrome P450 mRNA in breast cancer cell lines	243	
7.3	Characterisation of CYP4Z1	245	
7.4	Regulation of CYP4Z1	246	
7.5	Conclusion	248	
7.6	Future work	249	
Referei	nces	252	
Append	lix A List of chemicals and reagents	283	
Append	lix B Preparation of RNase-free solution	284	
Append	Appendix C Buffer for restriction enzymes		
Appendix D DNA sequencing results			

# List of Abbreviations

A	adenine
ACTD	actinomycin D
AhR	aryl hydrocarbon receptor
AhRR	aryl hydrocarbon receptor repressor
AMV	avian myeloblastosis virus
ARNT	aryl hydrocarbon receptor nuclear translocator
ATCC	American Type Culture Collection
BLAST	Basic Local Alignment Search Tool
B2M	$\beta_2$ -microglobulin
bp	base pair
С	cytosine
CAR	constitutive androstane receptor
cDNA	complementary deoxyribonucleic acid
CHX	cycloheximide
$CO_2$	carbon dioxide gas
СҮР	cytochrome P450
dATP	deoxyadenosine triphosphate
dCTP	deoxycytidine triphosphate
DEPC	diethyl pyrocarbonate
dGTP	deoxyguanosine triphosphate
DMEM	Dulbecco's modified Eagle's medium
DMEM/F12	a 1:1 (v/v) mixture of DMEM and Ham's F12
DMSO	dimethyl sulphoxide
DNA	deoxyribonucleic acid
dNTP	dinucleotide triphosphate
dT	deoxythymidylic acid
dTTP	deoxythymidine triphosphate
DTT	dithiothreitol
ECACC	European Collection of Cell Cultures
EDTA	ethylenediaminetetraacetic acid
ER	oestrogen receptor
FBS	foetal bovine serum
G	guanine

GCR	glucocorticoid receptor	
HIF-1α	hypoxia-inducible factor-1 alpha	
L15	Liebovitz's medium	
LMP	low melting point	
MEM	Minimal Essential Medium	
M-MLV	moloney-murine leukemia virus	
mRNA	messenger ribonucleic acid	
NCBI	National Centre for Biotechnology Information	
NNK	4-(methylnitrosamino)-1(3-pyridyl)-1-butanone	
$O_2$	oxygen gas	
OD	optical density	
PBS	phosphate-buffered saline	
PCR	polymerase chain reaction	
PGR	progesterone receptor	
PPARα	peroxisome proliferator-activated receptor alpha	
PXR	pregnane X receptor	
RAR	retinoic acid receptor	
RNA	ribonucleic acid	
RPMI	Rosewell Park Memorial Institute	
rRNA	ribosomal ribonucleic acid	
RT-PCR	reverse transcriptase-polymerase chain reaction	
RXR	retinoid X receptor	
Т	thymine	
T <sub>a</sub>	annealing temperature	
TAE	Tris-acetate-EDTA	
TBE	Tris-borate-EDTA	
T <sub>m</sub>	melting temperature	
U	Unit	
UV	ultraviolet	
V	volt	
VDR	vitamin D receptor	

# Chapter 1

# **General Introduction**

## **1.1 Introduction**

According to a recent report by Cancer Research UK, breast cancer is the second most common cause of cancer mortality in women (<u>http://www.cancerresearchuk.org</u>). In 2005, it was responsible for more than 12,000 deaths in women and accounted for 17% of all female cancer deaths. The most important prognostic factor related to survival and tumour recurrence following surgery is node-positive breast cancer patients. The prognosis of these patients compared to patients with primary breast cancer is poor, despite improvements in chemotherapy (Wapnir et al. 2006). In addition, ethnic differences in breast cancer survival between Caucasian and Afro-Caribbean women have been well documented. A poorer prognosis and more aggressive breast cancer are found in Afro-Caribbean compared to Caucasian women (Curtis et al. 2008). However, it is unclear why, and there is an urgent need to identify the main factors involved in these differences. Clinically, the critical aspects concerning the use of chemotherapy associated with the poor prognosis of patients are tumour resistance and drug toxicity. Some of this variability may be due to the clearance and metabolism of the anticancer drugs used in breast cancer treatment.

Although antioestrogen therapy and aromatase inhibitors have long been shown to be effective in reducing the risk of recurrence and death in patients with oestrogen receptor (ER) positive tumours, the use of these treatments has been limited in patients with ER negative tumours (Chia et al. 2005). ER-negative breast tumours are more aggressive and have a poorer prognosis when compared with ER-positive tumours and it is therefore, clear that targeted therapies for this patient population are needed (Rochefort et al. 2003).

At the present time, the progression and invasion of breast cancer is not fully understood. However, lifetime exposure to oestrogen correlates with increased risk of breast cancer in women, suggesting that oestrogen plays a crucial role in breast carcinogenesis (Henderson and Feigelson 2000). Two main hypotheses have been discussed concerning oestrogen and breast cancer. Firstly, binding of oestrogen to ER increases cell division and DNA synthesis, resulting in DNA mutations in mammary cells (Yager and Davidson 2006). Secondly, intermediate metabolites derived from oestrogen metabolism cause DNA damage by producing free radicals, leading to initiation and promotion of breast cancer (Deroo and Korach 2006).

For the latter hypothesis, a better understanding of the mechanisms involved in biotransformation of endogenous and/or xenobiotic compounds by breast cancer cells is very important as it contributes to the development of new target therapies as well as the knowledge of breast cancer progression and invasion. These compounds are usually detoxified into inactive forms by enzymes existing in local cells. Cytochrome P450 enzymes (P450s) are one of the key enzymes involved in the oxidative metabolism of a wide range of endogenous (e.g. oestrogen, vitamin A and vitamin D) and exogenous (e.g. tamoxifen, cyclophosphamide, paclitaxel and some carcinogens) compounds (Nebert and Nelson 2005). It is important to note that certain chemicals which are not carcinogens (procarcinogens) can be transformed into carcinogens by cytochrome P450s (Guengerich 2004).

Currently, 57 functional cytochrome P450 enzymes have been classified in humans. The majority of these enzymes are predominantly expressed in the liver, but some cytochrome P450s are preferentially expressed in extrahepatic tissues such as the breast (Nebert and Nelson 2005). Previous studies demonstrated a differential expression profile of cytochrome P450s between tumours and their corresponding normal tissues, and some cytochrome P450s have been identified as having tumour-specific expression. For instance, overexpression of CYP1B1 has been detected in various tumours including breast (Murray et al. 1997, McFadyen et al. 1999, Murray et al. 2001) and a novel member of the cytochrome P450 family 4, CYP4Z1 mRNA is preferentially expressed in breast carcinoma and associated with poor prognosis in primary ovarian cancer (Downie et al. 2005). More recently, CYP2W1 expression has been shown to be highly expressed in colon tumours (Karlgren et al. 2006). The existence of tumour specific cytochrome P450 expression suggests that these enzymes could be targets for cancer therapy, and a better understanding of their roles in tumour progression and invasion may be beneficial to cancer research as a whole.

Several chemotherapeutic drugs used in breast cancer therapy (e.g. anthracyclines, taxane, cyclophosphamide, tamoxifen and aromatase inhibitors) are extensively biotransformed in the liver, thereby reducing the release of toxic metabolites present in the tumour cells. Previous studies investigated the expression of cytochrome P450s in breast cancer cells from patients-derived samples (Albin et al. 1993, Huang et al. 1996, Murray et al. 1997, Iscan et al. 2001, Modugno et al. 2003, Haas et al. 2006), *in vitro* [i.e. cell lines (Sonneveld et al. 1998, Spink et al. 1998, Soulez and Parker 2001, McFadyen et al. 2003, Rieger et al. 2004, Savas et al. 2005, Fischer et al. 2007) and from xenografts (Smith et al. 1993)]. However, only a few cytochrome P450s (i.e. CYP1A1, CYP1B1, CYP2B, CYP2C and CYP3A) have been identified in breast cancer cells and the current knowledge of cytochrome P450 expression related to tumour characteristics in breast cancer is extremely limited. Therefore, this study was designed to identify the cytochrome P450 mRNA profile in a series of breast cancer cell lines derived from different tumour characteristics (i.e. ER status, invasive property and ethnicity) and compare this profile with a normal breast cell line.

### **1.2** Breast cancer

#### 1.2.1 Overview of breast cancer

In the UK, breast cancer is the commonest cancer and the second most common cause of cancer death (after lung cancer) in women. According to a recent report by Cancer Research UK, breast cancer accounted for 29% of all female cancers. Each year more than 40,000 women are diagnosed with breast cancer in the UK compared to 324 new cases of breast cancer annually in men. In 2005, there were more than 12,000 deaths from breast cancer in the UK. Risk of developing breast cancer in women significantly increases with age particularly in women older than 50 (<u>http://www.cancerresearchuk.org</u>). The lifetime risk of being diagnosed with breast cancer in women is 1 in 9 (O'Halloran et al. 2004).

Over the last two decades, the number of new cases of breast cancer diagnosed has increased from 75 (in 1975) to 116 (in 2003) per 100,000 in the UK. Five year survival has also improved from 52% (1971-1975) to 80% (2001-2003). Some of this improvement is as a consequence of the National Health Service Breast Screening Programme (NHSBSP) which targets the female population aged between 50 and 64 for routine mammography. The service was set up in 1988 (Reddy and Given-Wilson 2004).

In addition, the fourth worldwide meta-analysis (2000-2005) by the Early Breast Cancer Trialists' Collaborative Group (EBCTCG) reported that interethnic variation in the incidence and mortality rate of breast cancer in women is widening. Caucasian women have a higher incidence of breast cancer whereas African women are more likely to die from breast cancer (Chia et al. 2005). Recently, Morris and colleagues (Morris et al. 2007) demonstrated that African-American patients with breast cancer appear to have more highgrade, more ER-negative and more aggressive tumours than Caucasian women. These observations suggest that differences in survival by race/ethnicity may be due to breast carcinoma characteristics. Further molecular studies are required to investigate this

correlation with survival and perhaps identify new therapeutic targets for this patient population.

The aetiology of breast cancer remains complex with multiple risk factors including hormonal receptor status (positive/negative), duration of oestrogen exposure (early menarche/late menopause), age (pre-/post-menopausal status), lifestyle (high fat/low fibre diet), racial/ethnic groups and familial history of breast cancer (Dumitrescu and Cotarla 2005). Oestrogen mediates its effects on growth, differentiation and function of target tissues particularly reproductive organs by binding to ER found in the nucleus of those cells that are targets for oestrogen action. The ER exists in two subtypes, ER $\alpha$  and ER $\beta$ which are highly homologous and have distinct target tissue. ER $\alpha$  protein has been shown to be associated with breast cancer by several clinical and experimental studies, while the role of ER $\beta$  in relation to breast cancer remains controversial (Deroo and Korach 2006). In addition, *in vitro* studies indicated that ER $\beta$  appears to inhibit ER $\alpha$  activity (McDonnell 2004) and its expression is associated with the response of antioestrogen tamoxifen in ERnegative breast cancer (Gruvberger-Saal et al. 2007). Henceforth, the term 'ER status' in this thesis will refer to only ER $\alpha$  protein expression.

The nuclear oestrogen-ER complex binds to oestrogen response elements (EREs), resulting in transcriptional activation of oestrogen-responsive genes and this leads to the production of protein and a physiological response (Deroo and Korach 2006). In breast cancer cells, the ER signalling pathway is involved in increasing cell growth, cell division and DNA synthesis, and these can trigger DNA mutations in cells. Mutant cells can lead to uncontrolled proliferation in response to oestrogen stimulation and increase the risk of developing cancer in breast tissue (Yager and Davidson 2006). Therefore, targeting of the ER using antioestrogen therapy becomes an important therapeutic option in the treatment of breast cancer patients and this is described in the next section (McDonnell 2004).

The link between the metabolism of oestrogen and breast carcinogenesis is well documented. It is known that the carcinogenic effect of oestrogen can be attributed to the initiation of oestrogen metabolism by cytochrome P450s. A metabolite of 2-hydroxy-oestradiol is primarily catalysed by CYP1A2 and CYP3A4 in the liver, and by CYP1A1 in extrahepatic tissues including the breast. CYP1B1 which is mainly expressed in the breast, ovary and uterus catalyses the 4-hydroxylation of oestradiol. The quinone metabolites of 4-hydroxyoestradiol have been shown to act through the mechanisms that initiate and/or promote breast cancer by producing free radicals such as superoxides, and more reactive semiquinone-quinone intermediates, leading to the formation of DNA adducts (Tsuchiya et al. 2005, Yager and Davidson 2006).

### **1.2.2** The treatment of breast cancer

In general, breast cancer begins as a localised microscopic lesion which grows and spreads into the regional lymph nodes and metastases to other organs of the body, particularly the liver and lung, and may eventually lead to death. Traditional strategies for breast cancer therapy aim to reduce the risk of distant metastasis prevent a recurrence of tumour and improve overall survival. Clinically, a two-stage treatment strategy is commonly used to control the risk of local tumour recurrence. The first stage is surgery to control the regional tumour and this is usually supplemented with radiotherapy. Secondly, adjuvant hormonal therapy or chemotherapy is used to reduce risk of recurrence and improve longterm survival among women with breast cancer (Goldhirsch et al. 2007).

Guidelines for the use of adjuvant therapy in the management of early stage and metastases in breast cancer have developed over time. Two categories were recommended for patients with node-positive breast cancer. Adjuvant hormonal therapy (i.e. antioestrogen and aromatase inhibitors) is used for most patients who have ER-positive tumours but not recommended for women with ER-negative tumours. Another category, adjuvant

chemotherapy is advised for women with oestrogen-independent tumours (Harlan et al. 2006, Goldhirsch et al. 2007).

In hormonal therapy, a better understanding of the role of the ER signalling pathway and the action of oestrogen in breast carcinogenesis is allowing the development of therapeutic inventions, particularly in hormonal therapy which has been used to treat advanced and localised breast cancer in patients with ER-positive tumours. Antioestrogen therapy (e.g. tamoxifen and raloxifene) is commonly used as first-line therapy in pre-menopausal women. They exhibit antagonist properties by competitively inhibiting the interaction between oestrogens and ER blocking oestrogen action at the ER and reducing ER levels (Moulder and Hortobagyi 2008). However, Vogel and co-workers reported that the use of tamoxifen has been associated with a high risk of thromboembolism and ovarian cancer due to the similarity of its clinical efficacy to ovarian ablation. Raloxifene is as effective as tamoxifen in the prevention of invasive breast cancer and has a lower risk of tamoxifen's serious adverse effects (Vogel et al. 2006).

Aromatase (CYP19A1) is the key enzyme that converts androgen to oestrogen. Many aromatase inhibitors have been developed for the treatment of breast cancer and details of these are provided in Section 1.8. They are an important approach for reducing growthstimulatory effects of oestrogen in post-menopausal oestrogen-dependent early stage breast cancer patients (Brueggemeier et al. 2005, Perez 2007). Long-term treatment of aromatase inhibitors in breast tumours have, however, resulted in resistance to tamoxifen, which may be due to hypersensitivity of ER to oestrogen, mutations in ER or alteration of ER expression. Cross-resistance between tamoxifen and aromatase inhibitor has also been reported in some breast cancer patients (Shao and Brown 2004).

7

Introduction

In adjuvant chemotherapy, anthracycline based treatment such as the regimens of docetaxel and doxorubicin plus cyclophosphamide (TAC) are widely used as first-line therapy for node-positive breast cancer with ER-negative tumours. These regimens have been shown to have better antitumour activity than the classical CMF regimen (cyclophosphamide plus methotrexate and 5-fluorouracil) as adjuvant chemotherapy with TAC significantly improves the overall survival of breast cancer patients (Ahluwalia et al. 2005).

Doxorubicin (Adriamycin<sup>®</sup>) is an anthracycline antibiotic that interacts with DNA by intercalation and is widely used in the treatment of various types of cancers such as breast and ovary (O'Shaughnessy 2005). Cyclophosphamide (Cytoxan<sup>®</sup>, Neosar<sup>®</sup>) is an alkylating agent from oxazaphosphorine prodrug which is used against a wide range of human tumours such as breast and ovary (Moulder and Hortobagyi 2008). Docetaxel (Taxotere<sup>®</sup>) is a widely used taxane and the most active microtubule-interfering agents for breast cancer. It is not cross-resistant with anthracycline and does not interfere with the action of doxurubicin (Ahluwalia et al. 2005).

In addition, advances in adjuvant chemotherapy in patients who have ER-negative and node-positive breast cancer have been reported where biweekly doxorubicin and cyclophophamide combined with paclitaxel results in a lower rate of recurrence and improved survival compared to low dose of the CAF regimen (cyclophosphamide and doxorubicin plus 5-fluorouracil) (Berry et al. 2006). However, these anticancer drugs used in the management of breast cancer have been shown to be associated with acute and long-term cardiovascular complications (Jones et al. 2007).

Indeed, these anticancer drugs offer an advantage of presenting a lethal threat to the tumour cell by killing rapidly proliferating cells. At the same time, they also have a disadvantage to the body as they affect normal cells undergoing rapid proliferation, for

Resistance to chemotherapeutics in breast cancer patients is another common phenomenon and a serious clinical problem, leading to reductions in the overall survival period. Drug resistance can be generated via a number of mechanisms, none of which are fully understood. However, tumour-specific physiological changes such as very low oxygen concentrations (tumour hypoxia) may facilitate the development of drugs that target the local gene expression responsible for malignant cells (McKeown et al. 2007). Tumour hypoxia in solid tumours has been demonstrated to be involved in tumour progression as well as resistance to chemotherapy and radiotherapy (Denny 2004). In addition, the presence of hypoxia is associated with aggressive phenotype and poor prognosis. This has led to the development of tumour-selective therapy such as hypoxia-activated prodrugs, hypoxia-activated gene therapy and treatments targeting the hypoxia-inducible factor-1 alpha (HIF-1 $\alpha$ ) (Nagasawa et al. 2006, Brown 2007).

Due to problems with high drug toxicity and incidence of chemotherapeutic resistance, selective therapies that target the clinical and/or molecular characteristics of tumour cells are urgently required. Anticancer drugs are usually biotransformed to active and/or inactive form by certain drug metabolising enzymes. It is well recognised that cytochrome P450 enzymes (P450s) play a crucial role in the metabolism of various anticancer drugs (Section 1.8). Many cytochrome P450s have been found in extrahepatic tissues and certain cytochrome P450 genes have been defined as having tumour-specific expression. Therefore, expression of individual cytochrome P450's in breast cancer cells is of interest as it may help identify a new target for cancer therapy resulting in more effective and less toxic treatments for breast cancer patients.

## **1.3** The use of breast cancer cell lines as models

In vitro systems such as cell culture have been widely established from several diseases including tumours, and have been characterised as models to represent the pathogenesis process in vivo (Weigelt et al. 2005). Tumour-derived cell lines have been shown to closely resemble patient tumour characteristics and reflect breast cancer cells in vivo (Burdall et al. 2003, Lacroix and Leclercq 2004). There are currently a limited number of breast cancer cell lines and only a few of these models (i.e. MCF7, T47D and MDA-MB-231) have been extensively used to investigate cell signalling pathways and genes involved in cancer progression (Burdall et al. 2003). Distinct features of available breast cancer cell lines have been previously detailed and classified by the origin of the cells, ethnic origin of patient, type of epithelial cells, oestrogen receptor status and *in vitro* invasive property. However, questions have been raised as to whether they are truly representative of the tissue of origin and cancer biology *in vivo*. The same type of breast cancer cell lines distributed to different environments (laboratories), grown under various culture conditions and after numerous passages could generate both genetic and phenotypic alterations in subsequent sub-populations. Serious consideration must be given to these factors when designing experiments, interpreting results and comparing findings with other studies (Lacroix and Leclercq 2004).

Table 1.1 shows a database of seven breast cancer cell lines MCF7, T47D, ZR-75-1, ZR-75-30, MDA-MB-157, MDA-MB-231 and MDA-MB-468 which were used in the present study. Of the available cell lines, most originated from pleural effusions, containing human breast adenocarcinoma cells from metastatic tumours. The advantages of using cells of this type are that they dissociate into single cells and generate viable tumour cells with low or no contamination by fibroblasts and tumour stroma cells (Lacroix and Leclercq 2004).

Cell line name	Age	Site of origin	Pathology	Epithelial-mesenchymal	ER protein	In vitro invasive properties
	(years)			transition (EMT)	status	
Caucasian						
MCF7	69	Pleural effusion	Adenocarcinoma	Luminal-epithelial like	+	Low
T47D	54	Pleural effusion	Ductal carcinoma	Luminal-epithelial like	+	Low
ZR-75-1	63	Ascitic effusion	Infiltration ductal carcinoma	Luminal-epithelial like	+	Low
MDA-MB-231	51	Pleural effusion	Adenocarcinoma	Mesenchymal-epithelial like	-	High
Afro-Caribbean						
ZR-75-30	47	Ascitic effusion	Infiltration ductal carcinoma	Luminal-epithelial like	+	Low
MDA-MB-157	44	Pleural effusion	Medulla carcinoma	Mesenchymal-epithelial like	-	High
MDA-MB-468	51	Pleural effusion	Adenocarcinoma	Weakly luminal-epithelial like	-	Low

**Table 1.1:** Database of human normal breast and breast cancer cell lines used in this study (Combined and adapted from Lacroix and Leclercq 2004 and the Breast Cancer Cell Line Database at The University of Texas MD Anderson Cancer Centre website <u>http://www.mdanderson.org/</u>).

Molecular classification of breast cancer cell lines using epithelial-mesenchymal transition (EMT) is very important as EMT is associated with increased aggressiveness, and the invasive and metastatic potential of cancer cells (Sarrio et al. 2008). EMT is essential for the cell developmental process; it is an important component of tumour transformation and it can be characterised by cell adhesion, cell mobility and expression of E-cadherin. Specific characteristics of the initiation of invasive/metastatic cells are lack of cell adhesion, enhanced cell mobility and decreased E-cadherin expression. The cell lines are classified by their invasive properties, which are tested using the chemo-invasion assay that detects the migratory activity associated with matrix degradation (Matrigel in a Boyden chamber).

The cell lines can be divided into three phenotypes (Lacroix and Leclercq 2004). The first group, luminal epithelial-like cells highly express the epitheliod markers (e.g. ER and E-cadherin) and they have weak *in vitro* invasiveness. This group consists of MCF7, T47D, ZR-75-1 and ZR-75-30 cell lines. The second phenotype is weakly luminal epithelial-like cells which is very similar to the first group as they are poorly invasive cells (i.e. MDA-MB-468). The last group, mesenchymal epithelial-like cells express high amount of vimentin but do not express the ER/E-cadherin gene. There are highly invasive breast cancer cell lines, such as MDA-MB-231 and MDA-MB-157 (Sommers et al. 1994, Pishvaian et al. 1999).

Breast cancer classification is based on standard histopathology which describes both the morphology and pathogenesis of the tumour. Normal breast is a bilayered structure that is composed of two types of epithelial cells, inner luminal secretory cells and outer contractile myoepithelial cells, and these cells are surrounded with basement membrane (Adriance et al. 2005). In breast tumour, it has been demonstrated that disruption of myoepithelial layer and degradation of basement membrane are required for tumour

invasion. In addition, tumour myoepithelial cells are associated with the loss of ER expression, a higher expression of invasion-related genes and a more aggressive cancer (Man and Sang 2005). Gordon and colleagues investigated myoepithelial markers (i.e. CK14 and  $\alpha 6\beta 4$  integrin) in breast cancer cell lines, and demonstrated that MDA-MB-231 and MDA-MB-468 cells are defined as myoepithelial phenotype whereas MCF7, T47D and ZR-75 cells are represented as luminal phenotype (Gordon et al. 2003).

## 1.4 Cytochrome P450 enzymes (P450s)

### 1.4.1 Introduction to cytochrome P450s

Cytochrome P450 enzymes (P450s) have only been discovered and characterised in the last 50 years. The name cytochrome P450 was derived from the presence of a carbon monoxide binding pigment in rat liver microsomes which was first detected by Klingenberg (Klingenberg 1958). Six years later, this enzyme was found to have an unusual spectrum which has a very strong optical absorption peak at 450 nm (Omura and Sato 1964). The inactive form of cytochrome P450 has an absorbance maximum of 420 nm, known as cytochrome P420 (Tian et al. 1995).

Cytochrome P450s constitute a large superfamily of haem-containing proteins and are widely distributed among many organisms (e.g. bacteria, fungi, plants and animals). To date, 57 functional genes and 58 pseudogenes have been discovered and identified in humans (Nelson et al. 2004). These enzymes play a key role in several important biosynthesis and catalysis pathways (e.g. cholesterol, steroid hormones, bile acids, vitamins and arachidonic acids) as well as in the metabolism of various xenobiotic substrates including carcinogens and anticancer drugs (Guengerich et al. 2005, Nebert and Nelson 2005). The major cytochrome P450 enzymes with physiological roles in the metabolism of endogenous and xenobiotic compounds are summarised in Figure 1.1. Details of cytochrome P450 expression and functions are described in Section 1.5.



NNK; 4-(methylnitrosamino)-1-(3-pyridyl)-1-butanone

Figure 1.1: Cytochrome P450 enzymes implicated in the metabolism of endogenous and xenobiotic substrates.

#### 1.4.2 Nomenclature

According to the standardised cytochrome P450 nomenclature, individual cytochrome P450s within a family are defined as having at least 40% amino acid sequence homology and the family can be further subdivided into subfamilies which generally have at least 55% amino acid sequence similarity. This is followed by an Arabic numeral designating the family, a capital letter for the subfamily and a sequential Arabic number for the individual gene (Nebert et al. 1991). CYP1A1, for example, has been categorised as belonging to the cytochrome P450 family 1 subfamily A and being the first individual gene identified in this subfamily (Figure 1.2).



Figure 1.2: An example of human cytochrome P450 enzyme nomenclature.

#### **1.4.3** Biochemical characteristic features

In eukaryotic cells, cytochrome P450 enzymes are primarily located in the endoplasmic reticulum or inner mitochondrial membrane. These enzymes act as the terminal oxidase in the mix-function oxidase system and are therefore also known as mono-oxygenases (Nelson et al. 1996). In a simplified form, cytochrome P450 function can be summarised as shown below (Montellano 1995).

#### **RH** (substrate) + $O_2$ + NADPH + H<sup>+</sup> $\rightarrow$ **ROH** (product) + NAD<sup>+</sup> + H<sub>2</sub>O

The catalytic cycle of microsomal cytochrome P450 is shown in a schematic view in Figure 1.3. A functional cytochrome P450 system includes the active site of the catalytic reaction which is a haem moiety containing the iron ion (Fe<sup>3+</sup>). The reaction begins when the substrate binds to the ferric form of cytochrome P450 in the active site (1). The

cytochrome P450-substrate complex is then reduced to the ferrous state (Fe<sup>2+</sup>) by nicotinamide adenine dinucleotide phosphate-oxidase (NADPH)-cytochrome P450 reductase which serves as an electron donor protein (2). This production binds to molecular oxygen (O<sub>2</sub>) (3) to generate asuperoxide complex which is unstable and easily auto-oxidised releasing  $O_2^{=}$  (4, 5). However, if the final step (the second reduction process) occurs this reaction can continue. This form converts to an activated oxygen atom which reacts with the substrate molecule (6), resulting in the hydroxylation of the substrate (7) (Coon 2003, Guengerich 2004).



Figure 1.3: The catalytic reaction of cytochrome P450 enzymes (adapted from Coon 2003).

### 1.5 Cytochrome P450 expression and function

The 57 active cytochrome P450 genes are classified into 18 families and further subdivided into 43 subfamilies and they are located on 16 different chromosomes in the human genome which are summarised in Table 1.2. Enzymes belonging to families 1, 2, 3 and some enzymes of family 4 are responsible for the deactivation and/or activation of xenobiotic compounds, while the members of families 4 to 51 play a central role in the biosynthesis and catabolism of endogenous substrates. In the following sections, the importance and current knowledge of the cytochrome P450 genes are briefly described in terms of their expression and physiological roles in humans.

Chromosome No.	Cytochrome P450s
1	CYP2J2, CYP4A11/22, CYP4B1, CYP4X1, CYP4Z1
2	CYP1B1, CYP20A1, CYP26B1, CYP27A1
3	CYP8B1
4	CYP2U1, CYP4V2
6	CYP21A2, CYP39A1
7	CYP2W1, CYP3A4/5/7/43, CYP5A1, CYP51A1
8	CYP7A1, CYP7B1, CYP11B1/2
10	CYP2C8/9/18/19, CYP2E1, CYP26A1, CYP26C1
11	CYP2R1
12	CYP27B1
14	CYP46A1
15	CYP1A1/2, CYP11A1, CYP19A1
17	CYP17A1
19	CYP2A6/13, CYP2B6, CYP2F1, CYP2S1, CYP4F2/3/8/11/12/22
20	CYP8A1, CYP24A1
22	CYP2D6

## 1.5.1 CYP1 family

Three members CYP1A1, CYP1A2 and CYP1B1 are categorised in the cytochrome P450 family 1 (Nebert and Jones 1989, Nelson et al. 1996). The CYP1A1 (Jaiswal et al. 1986) and CYP1A2 (Ikeya et al. 1989) genes are located on chromosome 15q22-q24 and contain seven exons whereas the CYP1B1 gene is located on 2p21 and contains three exons (Tang et al. 1996).

CYP1A1 mRNA is mainly expressed in extrahepatic tissues including the breast (Huang et al. 1996, Hellmold et al. 1998, Iscan et al. 2001, Modugno et al. 2003). CYP1A1 catalyses the 2-hydroxylation of oestradiol and is induced by several procarcinogens, especially polycyclic aromatic hydrocarbons (PAHs) into reactive carcinogens which lead to the formation of DNA adducts (Tsuchiya et al. 2003, Shimada 2006). The induction of CYP1A1 by PAHs (e.g. 2,3,7,8-tetrachlorodibenzo-*p*-dioxin, benzo(a)pyrene and 3-methylcholanthrene) is mediated via the pathway of ligand-activated aryl hydrocarbon

receptor (AhR) complex (Kim and Guengerich 2005) which is described in Section 1.7.2. Previous studies demonstrated that CYP1A1 mRNA appears to be expressed in both breast cancer tissues (Huang et al. 1996, Iscan et al. 2001, Modugno et al. 2003) and cell lines (Spink et al. 1998, McFadyen et al. 2003). These findings suggest that CYP1A1 may be involved in breast cancer initiation and progression because it has an important role in the metabolism of oestrogen and procarcinogens. The cancer chemotherapeutic prodrug ellipticine (an intercalative alkaloid) is activated by CYP1A1/1A2 to form DNA adducts and is currently in clinical trials (Aimova et al. 2007).

CYP1A2 is predominantly expressed in the liver but also at lower levels in extrahepatic tissues (Sterling and Cutroneo 2004). CYP1A2 mRNA has been detected in both normal human breast or breast tumours (Hellmold et al. 1998, Modugno et al. 2003). CYP1A2 can be induced by carcinogenic heterocyclic amines such as 2-amino-3,8 dimethylimidazo(4,5-f) quinoxaline (MeIQx) and 2-amino-1-methyl-6-phenylimidazo(4,5-b) pyridine (PhIP), which are produced in meat and fish during cooking (Guengerich and Shimada 1991, Kim and Guengerich 2005). It catalyses the 2-hydroxylation of oestrogen and this metabolite is believed to be associated with a reduced risk of breast cancer (Yager and Leihr 1996). Takata and colleagues investigated an association between genetic polymorphisms in enzymes involved in oestrogen metabolism and mammographic density in both breast cancer and healthy members of the Hawaiian population. Results showed an association between the CYP1A2 genotype and high levels of breast density, suggesting that CYP1A2 may be involved in the development of breast cancer (Takata et al. 2007). In addition, CYP1A2 is involved in metabolism of the anticancer drug flutamide (Chimax<sup>®</sup>, Drogenil<sup>®</sup>), which is commonly used for the treatment of advanced prostate cancer (Goda et al. 2006).

CYP1B1 metabolises a wide range of procarcinogens (e.g. PAHs) and oestrogen (Shimada et al. 1996). Like CYP1A1, the induction of CYP1B1 by PAHs is mediated through the AhR/ARNT complex (Section 1.7.2). In addition, Hanna and co-workers demonstrated that inter-individual differences in breast cancer risk associated with oestrogen-mediated carcinogenicity may be related to CYP1B1 polymorphisms (Hanna et al. 2000). Recently, Yang and co-workers showed that high levels of CYP1B1 expression are associated with constitutive expression of AhR in human breast cancer but not with CYP1A1, suggesting interactions between CYP1B1 and AhR may be important in breast cancer initiation and progression (Yang et al. 2007).

CYP1B1 mRNA is expressed in several normal tissues but its protein expression has been detected primarily in tumours (i.e. breast, colon, lung, oesophagus, skin, lymph node, brain and testes) (Murray et al. 1997). In addition, Gibson and co-workers confirmed that overexpression of CYP1B1 protein was found in human colorectal tumours but lower levels were also found in normal colon cells (Gibson et al. 2003). Most importantly, CYP1B1 protein was not detected in human liver (Edwards et al. 1998). CYP1B1 overexpression in tumour cells is now recognised as a biomarker of the malignant phenotype (McKay et al. 1995, Murray et al. 1997, McFadyen et al. 2001, Roos and Bolt 2005, Barnett et al. 2007, Yang et al. 2007). Therefore, CYP1B1 is considered to be a potential novel cancer therapeutic target in a therapeutic setting, and details of how this enzyme is being targeted are provided in Section 1.8.

#### 1.5.2 CYP2 family

CYP2 is the largest cytochrome P450 family in humans containing 13 subfamilies and 16 individual genes (Table 1.3). There are three members in the CYP2A subfamily (i.e. CYP2A6, CYP2A7 and CYP2A13) but one of these CYP2A7, has been proven to be a pseudogene (Ding et al. 1995). These genes are located on chromosome 19q13.2 and

contain nine exons (Fernandez-Salguero et al. 1995). CYP2A6 is mainly expressed in the liver but lower levels are also expressed in extrahepatic tissues including the breast (Hellmold et al. 1998). CYP2A6 plays a role in the oxidation of nicotine, coumarin (Yun et al. 1991) and anticancer drugs (i.e. tegafur and letrozole) (Rodriguez-Antona and Ingelman-Sundberg 2006). In addition, this enzyme is involved in the metabolic activation of procarcinogens, such as a tobacco-specific carcinogen 4-(methylnitrosamino)-1(3-pyridyl)-1-butanone (NNK) and the food mutagen aflatoxin B<sub>1</sub> (Gonzalez and Gelboin 1994). CYP2A6 mRNA and protein have been shown to be induced in human liver cells by dexamethasone and the induction by this compound is suppressed by the antagonist of glucocorticoid receptor (RU468), suggesting that the glucocorticoid receptor may play a role in the regulation of CYP2A6 (Onica et al. 2008).

Subfamily of CYP2	Number of genes	Members
А	3	CYP2A6/7/13
В	1	CYP2B6
С	4	CYP2C8/9/18/19
D	1	CYP2D6
E	1	CYP2E1
F	1	CYP2F1
J	1	CYP2J2
R	1	CYP2R1
S	1	CYP2S1
U	1	CYP2U1
W	1	CYP2W1

Table 1.3: List of human CYP2 genes.

In breast cancer, Bieche and co-workers investigated xenobiotic metabolising enzymes in breast tumours compared to normal breast. Results showed that higher levels of CYP2A6 mRNA were found in ER-positive tumours than ER-negative tumours or normal breast, suggesting that CYP2A6 could be used as a biomarker to predict an antioestrogen responsiveness (Bièche et al. 2004). It is important to note that CYP2A6 plays a role in the metabolism of a third-generation aromatase inhibitor letrozole which is widely used in

Introduction

postmenopausal patients with oestrogen-dependent tumours (Ingelman-Sundberg et al. 2007). In addition, a recent study examined the regulation of CYP2A6 in the ER-positive breast cancer MCF7 cells and found that CYP2A6 is induced by oestrogen through ER. These results suggest that an interaction between CYP2A6 and oestrogen/ER may play a role in breast cancer initiation and progression (Higashi et al. 2007).

CYP2A7 is also expressed in the liver and shares 94% amino acid sequence homology with CYP2A6. However, CYP2A7 is a pseudogene as it has lost the haem-binding site (Wang et al. 2006). CYP2A13 is predominantly expressed in the respiratory tract (i.e. lung and trachea) (Su et al. 2000). This gene shares 93% amino acid sequence identity to CYP2A6 (Su et al. 2000) and their substrate-specificity is overlapping (He et al. 2004). However, the procarcinogen NNK is activated much more efficiently by CYP2A13 than CYP2A6, suggesting that CYP2A13 could be the primary enzyme in the human respiratory tract responsible for local metabolic activation of NNK and susceptibility to lung cancer (Su et al. 2000). A study by Zhang and colleagues (Zhang et al. 2007) demonstrated that a high level of CYP2A13 expression is associated with an increased risk of lung cancer particularly in smokers.

The CYP2B6 gene is located on chromosome 19q13.2 and contains nine exons (Miles et al. 1988). CYP2B6 constitutes 1-2% of the total cytochrome P450 expression in the liver (Mimura et al. 1993) and can be found in extrahepatic tissues such as lung, kidney (Nelson et al. 1996), heart (Thum and Borlak 2000) and breast (Hellmold et al. 1998, Iscan et al. 2001). CYP2B6 is highly inducible by phenobarbital, and its induction is regulated by the nuclear receptors constitutively active receptor (CAR) (Honkakoski and Negishi 2000) as described in Section 1.7.3. Extensive interindividual variability of CYP2B6 expression has been observed and polymorphisms cause functional alterations (Lamba et al. 2003), which have been shown to influence the pharmacokinetics of several drugs. CYP2B6 activates

anticancer prodrugs cyclophosphamide and ifosphamide into an active toxic form (phosphoramide mustard and acrolein), so polymorphisms of this gene may impact on the activation of cyclophosphamide and the outcome of treatment (Zanger et al. 2007).

The human CYP2C subfamily contains four highly homologous genes CYP2C8, CYP2C9, CYP2C18 and CYP2C19 which are predominantly expressed in the liver (Shimada et al. 1994). These genes are located on chromosome 10q24.1-24.3 and contain nine exons (Gray et al. 1995). CYP2C8 plays a role in the metabolism of endogenous compounds such as arachidonic acids and retinoic acids (Zhao and Imig 2003). CYP2C8 is a major enzyme responsible for the inactivation of the anticancer drug paclitaxel which is an effective agent against a wide range of human cancers including breast cancer (Totah and Rettie 2005, Ingelman-Sundberg et al. 2007). Several research groups have investigated CYP2C mRNA in normal human breast and breast tumours but their results seem to be inconsistent. For example, Modugno and colleagues demonstrated that no CYP2C mRNA was detected in normal breast or breast cancer (Modugno et al. 2003) while two separate research groups found CYP2C mRNA in normal breast as well as breast tumour samples (Huang et al. 1996, Iscan et al. 2001). Such inconsistencies may be due to the use of matched samples, low number of samples or inter-individual variations of this enzyme.

CYP2C9, the major CYP2C form in human liver (~20%) metabolises several therapeutic drugs including the hypoglycemic agent tolbutamide, anticonvulsant phenytoin, anticoagulant warfarin, a number of nonsteroidal anti-inflammatory drugs flurbiprofen, diclofenac (Totah and Rettie 2005) and endogenous compounds arachidonic acids (Goldstein 2001). It has been shown to participate in the metabolism of anticancer prodrugs tamoxifen (Boruban et al. 2006) and ifosfamide (Schmidt et al. 2004). The genetic polymorphisms of the CYP2C9 gene have been demonstrated to have functional Introduction consequences for clinical drug response and adverse drug effects (Ingelman-Sundberg et al. 2007).

CYP2C18 mRNA is expressed in human liver at very low levels (Furuya et al. 1991) but is also found in extrahepatic tissues (i.e. trachea, small intestine, ovary and testes) (Bièche et al. 2007). CYP2C18 shares 89%, 93% and 93% amino acid sequence identity to CYP2C8, CYP2C9 and CYP2C19 respectively (Goldstein and de Morais 1994). Zhu and Yu reported that a variant allele of the CYP2C18 gene, which is associated with decreased enzyme activity, affects anticancer ifosfamide efficacy in human hepatoma HepG2 cells (Zhu-Ge and Yu 2004).

CYP2C19 is a clinically important enzyme that metabolises a wide range of therapeutic drugs including the anticancer drugs tamoxifen and cyclophosphamide (Rodriguez-Antona and Ingelman-Sundberg 2006). Polymorphisms in the CYP2C19 gene cause poor metabolism of these drugs, resulting in drug toxicity or a severe adverse drug response to specific drugs. Approximately 3-5% of Caucasians and 12-23% of Asians have been shown to have a poor metaboliser phenotype (Mathijssen and van Schaik 2006).

Recently, Wu and colleagues investigated CYP2C19 mRNA and protein in a wide range of tumours (i.e. breast, liver, colon, stomach, esophagus, lung, uterus, brain, pancreas, ovary and kidney) derived from the Chinese population using semi-quantitative RT-PCR and Western blotting methods respectively. Results showed that the highest level of CYP2C19 mRNA was detected in hepatocarcinoma tissues and their adjacent normal tissue. CYP2C19 protein was found only in hepatocarcinoma samples and their adjacent normal tissues, with no protein expression being detected in other types of tumour (Wu et al. 2006). These findings suggest that CYP2C19 may be a tumour-specific expression in human liver.
Introduction The CYP2D subfamily contains one functional gene CYP2D6 which is of great importance in the metabolism of several clinically used drugs (e.g. anti-arrhythmic, antidepressants and antipsychotic drugs) (Pelkonen et al. 1998). The CYP2D6 gene is located on chromosome 22q13.1 and contains nine exons (Gough et al. 1993). CYP2D6 primarily catalyses the 4-hydroxylation of tamoxifen which has potent antioestrogenic effects. Tamoxifen is commonly used to treat patients with oestrogen-dependent breast cancer (Ingelman-Sundberg 2005a). CYP2D6 mRNA and protein have been detected in the breast (Hellmold et al. 1998) and lung (Guidice et al. 1997). Furthermore, a study by Iscan and colleagues showed that CYP2D6 mRNA expression in breast tumour was equivalent to tumour-free breast tissue (Iscan et al. 2001).

Currently, more than eighty variant alleles of CYP2D6 have been identified. The CYP2D6 activity is extremely variable and interethnic variations in drug metabolism are well documented. Poor metabolisers (PMs) display low or absence of CYP2D6 activity and have been found in 7% of Caucasians, whereas ultra-rapid metabolisers (UMs) exhibit dramatically high CYP2D6 activity and have been observed in 29% of the population in North Africa (Ethiopians) and the Middle East (Saudi Arabians) (de Leon et al. 2006a). Recently, Schroth and colleagues investigated an association between breast cancer treatment outcome with adjuvant tamoxifen and genotypes of enzymes involved in tamoxifen metabolism. Findings demonstrated that CYP2D6 with low-activity alleles are unlikely to benefit from adjuvant tamoxifen treatment (Schroth et al. 2007). In light of this evidence, new clinical tools such as the AmpliChip CYP450 genotyping test are being used to analyse psychiatric patient genotypes for CYP2D6 prior to initiating treatment with tricyclic antidepressants and antipsychotics (de Leon et al. 2006b). This may be helpful for patients who take other CYP2D6 substrates including tamoxifen.

The CYP2E1 gene is located on chromosome 10q24.3-qter and contains nine exons (Kölble 1993). CYP2E1 is responsible for the oxidative metabolism of many drugs (e.g. acetaminophen, halothane and chlorzoxazone) as well as the activation of several lowmolecular-weight toxicants and cancer suspect agents (e.g. ethanol, benzene, toluene and nitrosamines) (Tanaka et al. 2000). CYP2E1 is inducible by ethanol as well as pathophysiological conditions (e.g. obesity, starvation and diabetes) (Johnsrud et al. 2003). It is widely recognised that CYP2E1 produces free radicals causing tissue damage particularly the hepatotoxicity of acetaminophen (Lee et al. 1996). In addition, Reszka and colleagues examined a relationship between variant alleles of key enzymes implicated in dietary components and cancer susceptibility. Results showed that CYP2E1 high-activity alleles appear to be associated with colorectal cancer risk (Reszka et al. 2006).

The CYP2F1 gene is located on chromosome 19q13.1-13.2 and contains nine exons (Trask et al. 1993). CYP2F1 is highly expressed in the lung but at very low levels or not at all in the liver (Ding and Kaminsky 2003). A study by Thomas and co-workers showed a low level of CYP2F1 mRNA expression in sub-confluent cultured MCF10A cells but not in confluent cultures (Thomas et al. 2006). This enzyme is responsible for metabolism of the prototypical pneumotoxicants 3-methyllindole and 1,1-dichloroethylene as well as the activation of carcinogens styrene, naphthalene and benzene which can produce carcinogenic epoxide in the lung tissues (Kartha and Yost 2008). In primary ovarian cancer, CYP2F1 protein has been found at higher levels than in adjacent normal ovary tissues (Downie et al. 2005).

The CYP2J2 gene is located on chromosome 1p31.2-p31.3 and contains nine exons (Ma et al. 1998). CYP2J2 is highly expressed in the coronary endothelial cells and cardiac myocytes but at lower levels in the liver, small intestine, colon and kidney (Wu et al. 1996). This enzyme is involved in the epoxidation of arachidonic acid in cardiac tissue,

Introduction

forming epoxyeicosatrienoic acids (EETs) (Wu et al. 1996) which are potent endogenous vasodilators and inhibitors of vascular inflammation (Node et al. 1999, Scarborough et al. 1999). A study by Jiang and colleagues demonstrated that higher levels of CYP2J2 mRNA and protein were found in a variety of tumours (i.e breast, oesophagus, lung, stomach, liver and colon) than in their corresponding normal tissues (Jiang et al. 2005). Like CYP1B1, CYP2J2 is a tumour-associated cytochrome P450 enzyme which could be a new target for designing cancer therapeutic prodrugs activated by CYP2J2.

The CYP2R1 gene is located on chromosome 11p15.2 and contains four exons (Cheng et al. 2003). CYP2R1 (vitamin D<sub>3</sub> 25-hydroxylase) has been shown to play a crucial role in the synthesis of bioactive vitamin D<sub>3</sub>. It has been long known that vitamin D<sub>3</sub> has potent anticancer properties because it can inhibit cell proliferation and induce cell apoptosis. Therefore, enzymes involved in vitamin D<sub>3</sub> signalling pathways have been developed and used as potential targets for anticancer therapeutics (Deeb et al. 2007). CYP2R1 is predominantly expressed in the testes, kidney and pancreas (Ingelman-Sundberg 2005b, Karlgren et al. 2005). Higher levels of CYP2R1 expression were detected in primary ovarian tumours than in their corresponding normal tissues (Downie et al. 2005). Like the oncogene CYP24A1 (Section 1.5.13), CYP2R1 may be of particular interest to maximise vitamin D<sub>3</sub> levels in tumour sites and could be used in cancer therapy.

The CYP2S1 gene is located on chromosome 19q13.1-13.2 and contains nine exons (Rylander-Rudqvist et al. 2003). CYP2S1 is highly expressed in human lung (Rivera et al. 2002, Smith et al. 2003) but at lower levels in the liver (Rylander et al. 2001). The expression of CYP2S1 mRNA and protein in human skin is induced by ultraviolet radiation, coal tar and all-*trans* retinoic acid (Smith et al. 2003). CYP2S1 is also inducible by dioxin and this induction is mediated by AhR/ARNT which is similar to a manner

In mouse models, the CYP2S1 gene is induced by hypoxia which is mediated via the HIF-1 $\alpha$ /ARNT heterodimer. A recent study demonstrated that induction of CYP2S1 by both dioxin and hypoxia is mediated in a large part by multiple overlapping xenobiotic and hypoxia response elements. In theory, AhR and HIF-1 $\alpha$  requires ARNT for dimerisation prior to binding with the DNA response element of their target genes. Therefore, an interaction between the AhR/ARNT and HIF-1 $\alpha$ /ARNT heterodimerisation may affect the transcriptional activity of CYP2S1 (Rivera et al. 2007). Higher levels of CYP2S1 protein expression was detected in ovarian cancer metastasis than in primary tumour cells (Downie et al. 2005). CYP2S1 protein has been shown to be associated with poor prognosis in primary colorectal cancer (Kumarakulasingham et al. 2005).

The CYP2U1 gene is located on chromosome 4q25 and contains five exons (Chuang et al. 2004). The human CYP2U1 mRNA and protein are predominantly expressed in the thymus and brain (Chuang et al. 2004, Karlgen et al. 2004). CYP2U1 converts arachidonic acid into 19- and 20-hydroxyeicosatetraenoic acid (HETE) which serves as an endogenous vasoconstrictor and plays a biologically important role in the modulation of blood flow in the brain (Chuang et al. 2004, Karlgren et al. 2005). The role of CYP2U1 in the metabolism of xenobiotic compounds has not been reported. In primary ovarian cancer, a markedly higher level of CYP2U1 protein was found compared with normal ovary samples (Downie et al. 2005).

The CYP2W1 gene is located on chromosome 7q22.3 and contains 9 exons (Nebert and Nelson 2005). CYP2W1 has been detected at very low levels or not at all in foetal and adult human normal tissues but with higher CYP2W1 mRNA levels occuring in human

tumours particularly in colon and adrenal tumours (Karlgren et al. 2006). This may suggest that CYP2W1 is specifically expressed in tumours and that it could potentially be targeted by a novel anticancer drug (Karlgren and Ingelman-Sundberg 2007). A study by Thomas and co-workers showed a low level of CYP2W1 mRNA in the human normal breast MCF10A cell line (Thomas et al. 2006) but no information on CYP2W1 expression in breast cancer. To date, little is known about the physiological function of CYP2W1 in humans.

### 1.5.3 CYP3 family

The human CYP3 family has a single subfamily CYP3A and four individual genes which account for approximately 30% of total liver cytochrome P450 content (Shimada et al. 1994). This family is the most important group of drug metabolising enzymes in humans, therefore, intra- and inter-individual variations of these enzymes may influence the efficacy of drug treatment. CYP3A4 (Inoue et al. 1992), CYP3A5 (Gellner et al. 2001), CYP3A7 (Gellner et al. 2001) and CYP3A43 (Gellner et al. 2001) genes are all located on chromosome 7q22.1 and contain thirteen exons.

CYP3A4 is the most abundant cytochrome P450s in human adult liver (Wrighton and Stevens 1992) and is also found in the small intestine (Kolars et al. 1992). CYP3A4 is responsible for the oxidative metabolism of a wide range of xenobiotic compounds including approximately 60% of all therapeutic drugs such as anticancer drugs paclitaxel, mitoxantrone, tamoxifen, cyclophosphamide and ifosphamide (Guengerich 1999). In addition, CYP3A4 is involved in the metabolism of several steroid hormones (i.e. oestrogen, progesterone and testosterone) (Wrighton and Stevens 1992) and activation of procarcinogens (e.g. aflatoxin B<sub>1</sub>) to carcinogens (Forrester et al. 1990). The induction of CYP3A4 by these compounds is mainly mediated by pregnane X receptor (PXR, Section 1.7.4). Studies on CYP3A mRNA expression in normal human breast and breast tumour samples appear to be generating inconsistent results. For example, Iscan and colleagues demonstrated that CYP3A mRNA was not detected in normal breast or breast tumours (Iscan et al. 2001), whereas Huang and co-workers reported that CYP3A mRNA was higher in non-tumour breast tissues than in breast tumour samples (Huang et al. 1996). In addition, Hass and colleagues reported that strong CYP3A4 protein expression was significantly associated with poor tumour differentiation (Haas et al. 2006).

CYP3A5 shares 84% amino acid sequence identity to CYP3A4 and its substrates are similar to those of CYP3A4. CYP3A5 is primarily expressed in the liver and extrahepatic tissues (e.g. kidney and intestine) (Blake et al. 2005). High levels of CYP3A5 protein appear to be associated with primary colorectal cancer (Kumarakulasingham et al. 2005) and ovarian cancer (Downie et al. 2005) compared with their corresponding normal tissue. Large inter-individual and ethnic differences exist in the expression of CYP3A5 (Gervasini et al. 2007). A recent study reported that no association between CYP3A5 polymorphisms, and cancer risk was observed in Caucasian patients with liver, gastric or colorectal cancer (Gervasini et al. 2007).

CYP3A7 is predominantly expressed in human foetal liver. CYP3A7 has similar catalytic properties to CYP3A4, and catalyses the  $6\beta$ -hydroxytestosterone and  $16\alpha$ -hydroxylation of oestriol but not of oestradiol. This enzyme is involved in oestriol biosynthesis which it is an important enzyme for foetal growth and development (Blake et al. 2005). However, CYP3A7 has been shown to be involved in the formation of the procarcinogenic  $16\alpha$ -hydroxytestrone in women (Lee et al. 2003). In addition, higher levels of CYP3A7 protein are found in primary ovarian cancer than in normal ovary tissue (Downie et al. 2005).

CYP3A43 is a novel cytochrome P450 belonging to subfamily CYP3A. CYP3A43 shares 76%, 76%, and 72% amino acid homology with CYP3A4, CYP3A5, and CYP3A7

29

Introduction respectively (Westlind et al. 2001). It is mainly expressed in the prostate (Domanski et al. 2001) but also at very low level in the liver which differs from other CYP3A subfamily enzymes (Westlind et al. 2001). Moreover, Zeigler-Johnson and colleagues examined an association between genotype of enzymes involved in testosterone metabolism and prostate cancer risk. Findings showed that a CYP3A43 variant seems to be associated with prostate cancer susceptibility (Zeigler-Johnson et al. 2004). In addition, CYP3A43 protein in primary ovarian cancer has been found to be higher than in normal ovary tissue (Downie et al. 2005).

## 1.5.4 CYP4 family

The CYP4 family consists of six subfamilies and twelve members (Table 1.4). The CYP4A11 (Bell et al. 1993) and CYP4A22 (Nebert and Nelson 2005) genes are located in a cluster of other cytochrome P450 genes (i.e. CYP4B1, CYP4Z1 and CYP4X1) on chromosome 1p33 (Bellamine et al. 2003) and contain twelve exons. CYP4A11 is predominantly expressed in the kidney and liver (Savas et al. 2003, Ito et al. 2007). It catalyses the  $\omega$ -hydroxylation of fatty acids (i.e. lauric acid, palmitic acid and arachadonic acid) (Sharma et al. 1989, Palmer et al. 1993). CYP4A mRNA expression is induced by the synthetic glucocorticoid dexamethasone (Savas et al. 2003) and several peroxisome proliferators including hypolipidemic drugs clofibrate ciprofibrate and Wy14643 (or pirinixic acid) through a nuclear receptor called the peroxisome proliferator activated receptor alpha (PPAR $\alpha$ ) (Okita and Okita 2001, Hsu et al. 2007) as detailed in Section 1.7.6.

CYP4A22 shares 96% amino acid sequence identity to CYP4A11. CYP4A22 mRNA is found to be expressed in the liver and kidney; however, the levels of CYP4A22 mRNA expression in human liver were lower than CYP4A11. Unlike CYP4A11, CYP4A22

The physiological role of human CYP4A22 is being investigated (Hsu et al. 2007).

Subfamily of CYP4	Number of genes	Members
А	2	CYP4A11/22
В	1	CYP4B1
F	6	CYP4F2/3/8/11/12/22
V	1	CYP4V2
Х	1	CYP4X1
Z	1	CYP4Z1

 Table 1.4: List of human CYP4 genes.

The CYP4B1 gene is located on chromosome 1p34-12 and contains twelve exons (Nhamburo et al. 1989). CYP4B1 is selectively expressed in human lung but at lower levels in the liver and gastrointestinal tissues (Poch et al. 2005). CYP4B1 is involved in the activation of the pulmonary toxicant 4-ipomeanol and the metabolism of various arylamaines such as 2-aminofluorene, to more reactive intermediates which can cause bladder cancer (Baer and Rettie 2007), suggesting that CYP4B1 may be of interest for designing an anticancer prodrug activated by CYP4B1. Unlike other CYP4 enzymes, the role of CYP4B1 in the metabolism of endogenous compounds has not been reported.

There are six enzymes in the human CYP4F subfamily (Table 1.4) which are located on chromosome 19p13 (Nebert and Nelson 2005). CYP4F2 encodes for a leukotriene B4 (LTB4) ω-hydroxylase which inactivates LTB4 by ω-hydroxylation (Hsu et al. 2007). A study by Zhang and Hardwick demonstrated that this enzyme is highly expressed in human hepatoma HepG2 cells and the CYP4F2 gene is modulated by all-*trans* retinoic acid (Zhang and Hardwick 2000).

CYP4F3 consists of two forms, CYP4F3A and CYP4F3B, which arise from different promoters (Christmas et al. 1999). CYP4F3A preferentially catalyses the ω-hydroxylation of LTB4 and is mainly expressed in human polymorphonuclear leukocytes whereas CYP4F3B mainly catalyses fatty acids especially long chain fatty acid (Hsu et al. 2007). CYP4F3B shares 92% amino acid identity with CYP4F2 and they are predominantly

expressed in the liver and kidney (Christmas et al. 2001).

CYP4F8 (prostaglandin 19-hydroxylase) shares 81.2% and 76.7% amino acid similarity to CYP4F2 and CYP4F3 respectively (Bylund et al. 1999). CYP4F8 mRNA and protein are detected in various normal tissues (i.e. seminal vesicle epithelium, epidermis, corneal epithelium and proximal renal tubules) (Stark et al. 2003). CYP4F8 is closely related to CYP4F12 as they exhibit an overlapping catalytic profile. They convert arachidonic acid to 18-HETE, and metabolise prostaglandin  $H_1$  and  $H_2$  to 19-hydroxyl compounds (Stark et al. 2005, Hsu et al. 2007).

CYP4F11 and CYP4F12 mRNA are mainly expressed in human liver but also at lower levels in the kidney (Kalsotra et al. 2004). These two enzymes have different catalytic profiles. CYP4F11 efficiently metabolises xenobiotic compounds such as erythromycin but does not efficiently catalyse endogenous compounds particularly eicosanoids. In contrast, CYP4F12 efficiently hydroxylates arachdonic acids to 18-HETE (Bylund et al. 2001, Hsu et al. 2007). Their physiological roles have not been characterised.

CYP4F22 mRNA has been demonstrated to be expressed in human keratinocytes, testes, liver, kidney, small intestine and brain cell lines. Little is known about human CYP4F22. A recent study showed that mutations of the CYP4F22 gene are associated with lamellar ichthyosis type 3, suggesting that this enzyme may play a role in the pathway of 12R-

Introduction

CYP4V2 is located on chromosome 4p35.1 and contains eleven exons (Li et al. 2004). CYP4V2 mRNA is highly expressed in the retina but also found in a wide range of human tissues (i.e. heart, brain, placenta, lung, liver, skeletal muscle, kidney, pancreas and lymphocytes) (Li et al. 2004). In human normal breast MCF10A cells, CYP4V2 mRNA was also detected (Thomas et al. 2006). In addition, CYP4V1 protein has been shown in colorectal cancer compared with adjacent normal colon samples. Results showed that higher levels of CYP4V1 were detected in colon cancer with lymph node metastasis than in their corresponding primary tumour cells (Kumarakulasingham et al. 2005). To date the catalytic properties and physiological function of CYP4V2 remain unknown.

The CYP4X1 gene is located on chromosome 1p33 and contains twelve exons (Nelson 2003). CYP4X1 mRNA is highly expressed in human trachea and aorta but at lower levels in the breast, ovary, uterus, liver and kidney (Savas et al. 2005). Recently, CYP4X1 protein has been shown to be specifically expressed in mouse brain (Al-Anizy et al. 2006). A study by Savas and co-workers indicated that CYP4X1 mRNA in human hepatoma HepG2 cells is regulated by PPARα (Savas et al. 2005). A physiological role of CYP4X1 is currently unknown.

CYP4Z1 is located on chromosome 1p33 and contains twelve exons (Rieger et al. 2004). CYP4Z1 mRNA is preferentially expressed in human mammary gland but lower levels in the liver, kidney, heart, brain, lung, prostate, testes and ovary. CYP4Z1 mRNA and protein are overexpressed in breast tumour compared to their corresponding normal tissue (Rieger et al. 2004). Higher levels of CYP4Z1 protein were found in primary ovarian cancer than in normal ovary tissue and its expression in primary ovarian cancer appears to be an independent marker of prognosis (Downie et al. 2005). A study by Savas and colleagues demonstrated conditional regulation of the human CYP4Z1 gene in breast cancer cell lines T47D and MCF7 using quantitative RT-PCR. Results suggested that CYP4Z1 mRNA is induced by the glucocorticoid and progesterone receptor agonists but not by PPARα agonist Wy14643 (Savas et al. 2005).

## 1.5.5 CYP5 family

The CYP5A1 gene is located on chomosome 7q34-q35 and contains thirteen exons (Chase et al. 1993). CYP5A1 (thromboxane A<sub>2</sub> synthase) has a role to play in the conversion of the prostaglandin endoperoxide into thromboxane A<sub>2</sub>. Thromboxane A<sub>2</sub> is a potent vasoconstrictor and inducer of platelet aggregation which is important in the hemodynamic status in humans (Schuster and Bernhardt 2007). In addition, Nie and co-workers reported finding higher levels of CYP5A1 expression in prostate cancer cells than in normal prostate epithelial cells (Nie et al. 2004).

## 1.5.6 CYP7 family

Currently, the CYP7 family consists of two subfamilies, CYP7A1 and CYP7B1. CYP7A1 shares 40% amino acid sequence homology to CYP7B1. Both genes are located on chromosome 8q21.13. The CYP7A1 and CYP7B1 genes contain six and five exons respectively (Setchell et al. 1998). CYP7A1 (cholesterol  $7\alpha$ -hydroxylase) directly converts cholesterol into  $7\alpha$ -hydroxycholesterol, resulting in the production of deoxycholic acid (Beigneux et al. 2002). CYP7A1 has been found to be expressed only in the liver (Wu et al. 1999). The down-regulation and up-regulation of CYP7A1 are mediated by farnesoid X receptor (FXR, Section 1.7.8) and liver X receptor alpha (LXR $\alpha$ ) respectively. A previous study showed that high levels of the deoxycholic acid in plasma are found in postmenopausal women with breast cancer. In addition, FXR has been found to be highly

expressed in breast tumours and involved in the induction of apoptosis (Swales et al. 2006).

CYP7B1 mRNA is found in the liver and extrahepatic tissues (i.e. brain, testes, prostate, ovary and kidney). CYP7B1 mainly catalyses  $7\alpha$ -hydroxylation of oxysterol (25- and 27-hydroxycholesterol) and neurosteroids [dehydroepiandrosterone (DHEA) and pregnenolone] (Wu et al. 1999). CYP7B1 expression is sexually dimorphic (male > female). A study by Martin and co-workers demonstrated that CYP7B1 generates  $7\alpha$ -hydroxy-DHEA to selectively activate oestrogen receptor beta in human prostate epithelium. This suggests that CYP7B1 may have a role in the regulation of androgen and/or oestrogen in prostrate tissue (Martin et al. 2004).

# 1.5.7 CYP8 family

The human CYP8 family encodes two subfamilies, CYP8A and CYP8B (Nelson et al. 1996). The CYP8A1 gene resides on chromosome 20q13.11-13.13 and contains ten exons (Yokoyama et al. 1996). CYP8A1 (prostacyclin synthase) is widely expressed in human tissues (i.e. ovaries, heart, skeletal muscle, lung and prostate) (Miyata et al. 1994). It converts prostaglandin H<sub>2</sub> to prostacyclin (prostaglandin I<sub>2</sub>) whose function is opposite to that of CYP5 (thromboxane A<sub>2</sub> synthase). Prostacyclin is a potent vasodilator and inhibitor of platelet aggregation. In addition, a stable prostacyclin analogue, cicaprost has been demonstrated to play a key role in the prevention of metastasis and reduce growth of breast tumours in rodent models (Schirner and Schneider 1997). Previous study showed that the levels of prostacyclin were reduced in colon cancer samples as compared with matched adjacent normal colon samples (Miyata et al. 1994). Recently, variants of CYP8A1 have been demonstrated to be associated with colorectal polyps risk (Poole et al. 2006).

Introduction

The CYP8B1 gene is located on chromosome 3p21.3-22 and contains only one exon (Gafvels et al. 1999). CYP8B1 (sterol 12α-hydroxylase) is found in the liver. It plays an important role in bile acid biosynthesis which is closely related to the CYP7 family (Eggertsen et al. 1996, Gafvels et al. 1999). CYP8B1 activity can be altered by fasting, cholestyramine, streptozotocin-induced diabetes mellitus and clofibrate (Eggertsen et al. 1996). It has been long known that soy isoflavones are promising agents for breast cancer prevention. Recently, Li and colleagues demonstrated that human hepatic CYP8B1 expression can be up-regulated by soy isoflavones, suggesting that these agents can modulate cholesterol metabolism through CYP8B1 (Li et al. 2007).

## 1.5.8 CYP11 family

The CYP11 family consists of two subfamilies containing three enzymes; CYP11A, CYP11B1 and CYP11B2 which are localised to the inner mitochondrial membrane. The CYP11A1 gene is located on chromosome 15q23-q24 and contains nine exons (Chung et al. 1986). This gene encodes the cholesterol side chain cleavage enzyme that catalyses the initial and rate-limiting step of steroid biosynthesis (Payne and Hales 2004). CYP11A1 is predominantly expressed in steroidogenic tissues such as adrenal cortex, ovary, testes, placenta and central/peripheral nervous systems (Compagnone et al. 1995). Yaspan and colleagues investigated the relationship between variation of genes involved in oestrogen biosynthesis and metabolism, in particular CYP11A1, and breast cancer susceptibility in a study of case-control from the Chinese population. Findings showed that the CYP11A1 polymorphisms may have a role in breast cancer (Yaspan et al. 2007).

The CYP11B1 gene shares 95% amino acid sequence homology with the CYP11B2 gene (Mornet et al. 1989) and these genes are mainly expressed in the adrenal cortex (Curnow et al. 1991). Both CYP11B1 and CYP11B2 genes reside on chromosome 8q21-q22 and contain nine exons (Mornet et al. 1989). CYP11B1 (11β-hydroxylase) catalyses the last

steps in cortisol and corticosterone biosynthesis. CYP11B2 (aldosterone synthase) specifically catalyse aldosterone biosynthesis (Hakki and Bernhardt 2006). Inhibition of steroid biosynthesis by members of the CYP11 family is one of the therapy options for patients with inoperable adrenocoticoid cancer and hypercortisolism. However, CYP11B1 and CYP11B2 deficiencies can cause severe side effects such as Cushing's syndrome which leads to chronic glucocorticoid excess. Further development of selective inhibitors for distinct enzymes in this family is required (Schuster and Bernhardt 2007).

### 1.5.9 CYP17 family

The CYP17A1 gene is located on chromosome 10q24.3 which consists of eight exons (Fan et al. 1992). CYP17 (17α-hydroxylase or 17,20-lyase) catalyses both 17α-hydroxylation and 17,20-lyase conversion of C21 steroids (progesterone or pregnolone) to C19 precursors of sex steroids (dehydroepiandrosterone or androstenedione) (Payne and Hales 2004). CYP17A1 is closely related to CYP21 and it shares almost 30% amino acid sequence identical to CYP21, however; CYP17A1 has lost two exons which are present in CYP21 (Picado-Leonard and Miller 1987). CYP17A1 is predominantly expressed in steroidogenic tissues such as the adrenal cortex and ovary (Nagamani and Urban 2003, Olson et al. 2007).

CYP17A1 is a key enzyme in the androgen biosynthesis. Androgen is an important hormone in the development and maintenance of sexual characteristics in human males; however, it is believed to play a role in androgen-dependent diseases, in particular prostate cancer (Bruno and Njar 2007, Schuster and Bernhardt 2007). Thus the use of potent CYP17A1 inhibitors such as ketoconazole have been introduced to treat androgendependent prostate cancer patients (Bruno and Njar 2007, Moreira et al. 2007) and details of this treatment are provided in Section 1.8. Recently, the interactions between gene-gene and gene-environment which are involved in hormone biosynthesis and metabolism pathways have been investigated by several research groups. One of these studies demonstrated that CYP17A1 polymorphism in combination with long-term hormonereplacement therapy use and high body mass index in postmenopausal women may contribute to breast cancer susceptibility (Chen et al. 2008).

# 1.5.10 CYP19 family

The CYP19A1 gene resides on chromosome 15q21.1 which contains eleven exons (Chen et al. 1988). CYP19A1 (aromatase or oestrogen synthase) has important physiological functions both in sex steroid hormone biosynthesis and growth/ differentiation in humans. It converts the C19 androgens, testosterone and 16α-hydroxyandrostenedione into the C18 oestrogens, oestradiol and oestrone respectively. This enzyme is widely expressed in several tissues (i.e. ovary, Leydig cells, placenta, adipose tissue and bone), however; tissue- and sex-specific expression in cellular metabolism have been demonstrated in these tissues as reviewed by Payne and Hales (Payne and Hales 2004). CYP19A1 activity appears to be increased in breast cancer tissues, thus several potent CYP19A1 inhibitors have become a powerful tool in the treatment of oestrogen-dependent diseases, in particular breast cancer (Brueggemeier et al. 2005, Gibson et al. 2007, Jonat et al. 2007, Perez 2007). Details of CYP19 inhibitors in cancer therapy are provided in Section 1.8.

#### 1.5.11 CYP20 family

The CYP20A1 gene is located on chromosome 2q33 which contains thirteen exons. Only one member has been found in this family and human CYP20A1 has been shown to be homologous in mouse, cow and fish but not in *Drosophila* or *Caenorhabditis elegans* (Nebert and Nelson 2005). A recent study suggests that a high level of CYP20A1 expression is found in immune system cell types (Thomas 2007) but its physiological function in humans is still unknown.

38

### 1.5.12 CYP21 family

The CYP21A2 is found on chromosome 6p21.3 which contains ten exons (Higashi et al. 1986). CYP21A2 (steroid 21-hydroxylase) is mainly expressed in the adrenal cortex (Wijesuriya et al. 1999) and converts the C21 of progesterone and 17 $\alpha$ -hydroxyprogesterone into 11-deoxycorticosterone and 11-deoxycortisol respectively in the zona fasciculata of the adrenal cortex. CYP21A2 plays an important role in the biosynthesis of sex hormones, glucocorticoids and mineralocorticoids, and is closely related to CYP17A1 (Payne and Hales 2004, Dragan et al. 2006). Dragan and colleagues determined the effect of IC<sub>50</sub> values of various known CYP17A1 inhibitors (i.e. ketoconazole, Sa40, YZ5ay and BW33) on CYP21A2 expression in a fission yeast strain expressing mammalian cytochrome P450 steroid hydroxylases. Results showed that CYP17A1 inhibitors appear to be moderate CYP21A2 inhibitors (Dragan et al. 2006).

### 1.5.13 CYP24 family

The CYP24A1 gene resides on chromosome 20q13.2-13.3 and contains twelve exons (Hahn et al. 1993). The mitochondrial enzyme CYP24A1 (1, 25-(OH)<sub>2</sub>D<sub>3</sub>-24-hydroxylase) plays a crucial role in the degradation of vitamin D<sub>3</sub> (1, 25-(OH)<sub>2</sub>D<sub>3</sub>) (Masuda et al. 2006). CYP24A1 mRNA is regulated either by calcium ions or excess amounts of vitamin D<sub>3</sub> and is mediated through it nuclear receptor vitamin D receptor (VDR). CYP24A1 is expressed in several normal tissues such as kidney, liver and gastrointestinal tract (Jones et al. 1998). However, Anderson and colleagues detected overexpression of CYP24A1 in a wide range of tumour cells (i.e. breast, colon, ovary and lung) compared with their matched adjacent normal cells, suggesting that CYP24A1 may be an oncogene (Anderson et al. 2006). Inhibition of CYP24A1 has been developed as a new strategy for cancer therapy as it has an antiproliferative effect (Sundaram et al. 2006) (Section 1.8).

### 1.5.14 CYP26 family

Three CYP26 isoforms have been identified in humans, CYP26A1, CYP26B1 and CYP26C1 (Ray et al. 1997, Nelson 1999, Taimi et al. 2004). CYP26 is induced by all*trans*-retinoic acid (ATRA) through its nuclear receptor (retinoic acid receptor, RAR) and retinoid X receptor (RXR), which then interacts with retinoic acid response elements within the promoter of CYP26 (Bastien and Rochette-Egly 2004). ATRA is an active metabolite of vitamin A or retinoid which play an important role in the cell differentiation and proliferation of epithelial cells, vision and reproduction (White et al. 2000). ATRA has been shown to induce apoptosis in various cancer cell lines, suggesting that it can be used as a chemopreventive agent (Osanai and Petkovich 2005)

The CYP26A1 gene is found on chromosome 10q23-q24 and contains seven exons (White et al. 1998). CYP26A1 is mainly expressed in the brain, liver and placenta (Ray et al. 1997). The metabolism of ATRA by CYP26A1 can produce two main metabolites, 4-hydroxy-ATRA and 4-oxo-ATRA which can bind to RAR but with low affinity, resulting in the inactivation of ATRA (Villani et al. 2004). This indicates that high levels of CYP26A1 decrease the effects of ATRA in target cells. A higher level of CYP26A1 protein was detected in primary ovarian cancer than in normal ovary tissues (Downie et al. 2005). In addition, overexpression of CYP26A1 has been shown to be associated with human breast and colon cancer cells (Sonneveld et al. 1998), suggesting that inhibition of CYP26A1 may be a new strategy to increase local ATRA and inhibit the catabolism of ATRA. CYP26A1 inhibitors (e.g. liarozole, R116010 and R115866) have entered clinical trials for the treatment of cancer patients (Schuster and Bernhardt 2007) and these are described in Section 1.8.

The CYP26B1 gene carries six exons located on chromosome 2p12 (Nebert and Nelson 2005). CYP26B1 is mainly expressed in the adult brain several regions (i.e. cerebellum

40

and pons) (White et al. 2000) and embryonic tissues (MacLean et al. 2007). CYP26B1 is involved in the metabolism of ATRA which is similar to CYP26A1. In animal models, it appears to have an essential role in the development of the embryo (MacLean et al. 2001).

The CYP26C1 is the newest member of the CYP26 family to be identified in humans. CYP26C1 expression is found in human keratinocyte cell lines. It can be induced by ATRA and 9-cis-retinoic acid which differs from CYP26A1 and CYP26B1 (Taimi et al. 2004, Njar et al. 2006). However, the physiological function of CYP26C1 relevant to diseases in humans remains unknown.

### 1.5.15 CYP27 family

The three subfamilies CYP27A1, CYP27B1 and CYP27C1 have been identified in the human CYP27 family. The CYP27A1 gene is located on chromosome 2q33-qter and contains eleven exons (Cali and Russell 1991). CYP27A1 is a multifunctional mitochondrial enzyme, sterol 27-hydroxylase, that is predominantly expressed in the liver and catalyses several oxidative reactions in the biosynthesis of bile acids and the bioactivation of vitamin D<sub>3</sub> (Gottfried et al. 2006). This enzyme is an initial enzyme in the pathway of bile acid biosynthesis which is related to CYP7A and CYP8B. Tang and co-workers demonstrated the regulation of CYP27A1 mRNA and activity in human hepatoma HepG2 and prostate cancer LNCaP cells. Results showed that CYP27A1 is regulated by hormones especially oestrogen and androgen compounds (Tang et al. 2007).

The CYP27B1 gene is located on chromosome 12q13.1-q13.3 and contains nine exons (St-Arnaud et al. 1997). CYP27B1 converts the vitamin D<sub>3</sub> precursor into the most active form of vitamin D<sub>3</sub> (1,25-(OH)<sub>2</sub>D<sub>3</sub>) in the kidney and is also called 25-OH-D<sub>3</sub>-1 $\alpha$ -hydroxylase. This enzyme has a crucial role in physiological function such as cell differentiation and proliferation (Takeyama et al. 1997). In the proximal tubules,

CYP27B1 mRNA and protein have been shown to be repressed in the presence of  $(1,25-(OH)_2D_3)$ . CYP27B1 was found in non-renal tissues (i.e. skin, lymph nodes, colon, pancreas and brain) (Zehnder et al. 2001) including human mammary epithelial cells (Kemmis et al. 2006). In addition, several cancer cell lines have shown CYP27B1 expression (Diesel et al. 2005, Becker et al. 2007, Cordes et al. 2007, Turunen et al. 2007). A study by de Lyra and colleagues showed that decreased levels of  $1,25-(OH)_2D_3$  are associated with breast cancer (de Lyra et al. 2006). In addition, maximising  $1,25-(OH)_2D_3$  by using the combination of a CYP24A1 inhibitor (genetein) and a CYP27B1 inducer (trichotatin A) has been demonstrated to increase the antiproliferative effect of  $1,25-(OH)_2D_3$  in prostate cancer cells (Farhan et al. 2003, Khorchide et al. 2005). Currently, the CYP27C1 gene has been sequenced and identified in humans but it was not detected in rodent (Nelson et al. 2004). The physiological function of CYP27C1 remains unknown.

## 1.5.16 CYP39 family

The CYP39A1 gene is located on chromosome 6p21-p11.2 and carries twelve exons (Li-Hawkins et al. 2000). CYP39A1 is predominantly found in the liver and its expression in females is higher than in males, which is opposite to that of CYP7B1 expression. In the liver,  $7\alpha$ -hydroxylation of cholesterol is the rate-limiting step of bile acid synthesis and metabolic elimination, and this is mediated by CYP7A1 and CYP39A1. CYP39A1 is specific for 24S-hydroxycholesterol as a substrate. Dietary cholesterol, bile acids or a bile acid-binding resin does not affect hepatic CYP39A1 mRNA level (Li-Hawkins et al. 2000). Higher levels of CYP39A1 protein were found in colorectal cancer patients with lymph node metastasis than in their corresponding primary tumours (Kumarakulasingham et al. 2005).

Introduction

### 1.5.17 CYP46 family

The CYP46A1 gene resides on chromosome 14q32.1 and contains fifteen exons (Lund et al. 1999, Nebert and Nelson 2005). This gene encodes cholesterol 24*S* hydroxylase which is a rate-limiting enzyme for cholesterol removal from the brain. CYP46 mRNA and protein are selectively expressed in human brain particularly in neurons and some astrocytes. It modulates the turnover of cholesterol in the brain by converting excess cholesterol into 24*S*-hydroxycholesterol (Bjorkhem et al. 1998). High levels of CYP46A1 have been shown to be associated with degenerating neuritic plaques in Alzheimer's disease. These observations suggest that CYP46 expression may lead to cholesterol being abundant in plasma, resulting in greater loss of synapses in the brain and the development of Alzheimer's disease (Bogdanovic et al. 2001, Desai et al. 2002, Brown et al. 2004). Recently, Bretillon and colleagues examined CYP46A1 expression in mammalian neural retina using RT-PCR and immunohistochemical analysis. Results showed that CYP46A1 mRNA and protein were specifically expressed in bovine and rat samples (Bretillon et al. 2007).

#### **1.5.18 CYP51 family**

The CYP51A1 gene is located on chromosome 7q21.2-21.3 and contains ten exons (Nebert and Nelson 2005). CYP51A1 (lanosterol 14 $\alpha$ -demethylase) is found in both prokaryotes and eukaryotes (Aoyama et al. 1996, Waterman and Lepesheva 2005). CYP51A1 mRNA is expressed in most tissues in humans but the highest expression has been detected in Leydig cells in the testes (Stromstedt et al. 1998). CYP51A1 plays an important role in the key step of cholesterol biosynthesis as it catalyses the oxidative removal of the 14 $\alpha$ -methyl group from lanosterol (Rozman et al. 1996). Inhibitors of human CYP51A1 have been attracting interest in antimicrobial and anticholesterolemic drug research (Waterman and Lepesheva 2005, Schuster and Bernhardt 2007). A high expression of CYP51 protein has been shown to be associated with poor prognosis and is an independent marker of

43

prognosis in colorectal cancer (Kumarakulasingham et al. 2005). In addition, a higher level of CYP51A1 protein was detected in primary ovarian cancer than in normal ovary tissues (Downie et al. 2005).

# **1.6** Expression of cytochrome P450s in breast cancer cell lines

As mentioned in Section 1.3, *in vitro* cell culture models in breast cancer cells have been extensively studied in preclinical models with respect to the investigation of molecular mechanisms involved in breast carcinogenesis, mechanisms of drug resistance/sensitivity in tumour cells and the identification of novel targets for cancer therapeutics (Lacroix and Leclercq 2004). Knowledge of cytochrome P450 enzymes derived from preclinical models, including their expression profile can be representative of the malignant cells in relation to clinical findings. MCF7 is the most popular of the oestrogen-dependent breast cancer cells and this cell line has been shown to stably express cytochrome P450s (i.e. CYP1A1) in a similar manner to their tissue of origin (Rekha and Sladek 1997, Michael and Doherty 2005).

However, there remains controversy regarding mRNA/protein expression profile in breast cancer tissues and cell lines. For instance, expression of CYP1B1 mRNA and protein appeared to be high in human breast cancer tissues (McKay et al. 1995, Murray et al. 1997, McFadyen et al. 1999), but MCF7 cells expressed CYP1B1 mRNA but not protein (McFadyen et al. 2003). This may be due to the expression alteration of key transcription factors involved in the translation of cytochrome P450. It reflects individual variation in the expression profiles. Nevertheless, cell line models are useful tools, especially in human breast cancer research where animal models may not truly represent cytochrome P450 expression found in humans as species and strain differences in expression of these enzymes are well recognised (Miles et al. 1990, Michael and Doherty 2005). No model is perfect and it is important to keep these limitations in mind.

44

Previous studies have demonstrated the presence of cytochrome P450 mRNA in breast cancer cell lines. However, a very few of these models (i.e. MCF7, T47D and MDA-MB-231) have been widely used and only a limited number of individual cytochrome P450 enzymes are currently available in these models. Also, there appears to be a lack of data on the association between cytochrome P450 expression and breast tumour characteristics (i.e. oestrogen receptor status, invasiveness and ethnicity).

# 1.7 Nuclear receptors involved in regulation of cytochrome P450s

#### **1.7.1 Introduction**

Members of the nuclear receptor superfamily fundamentally function as ligand-regulated transcription factors by transforming the extracellular and intracellular signals, and controlling the transcription of their target genes. Nuclear receptors can be found within the nucleus and they are responsible for sensing the presence of certain ligands such as lipophilic hormones, vitamins, dietary lipids and other xenobiotic compounds (Zhang et al. 2004). These receptors are highly conserved across species and play a role in the growth and development of organisms. They are also important in several signalling pathways (i.e. cell differentiation, reproduction, metabolism of lipids, drugs and energy) (Sonoda et al. 2008). Currently, 48 nuclear receptors have been identified in the human genome (Zhang et al. 2004).

In principle, nuclear receptors share a common structural domain (Figure 1.4). The Nterminal regulatory domain contains the activation function-1 (AF-1) whose action is independent on the presence of bound ligand, and this domain is highly variable in sequence between various nuclear receptors. The DNA-binding domain (DBD) is the most conserved region which comprises of two zinc finger (ZF-1 and ZF-2) complexes and is involved in dimerisation (homo- or heterodimerisation) of nuclear receptors. The hinge region is responsible for connecting the DBD and ligand binding domain (LBD), and this Introduction is a flexible domain containing the nuclear localisation signal. LBD, which is the largest domain, contributes to the dimerisation interface of the receptor and binds to coactivator/ corepressor protein. The LBD contains the activation function-2 (AF-2) (Sonoda et al. 2008). Nuclear receptors bind to each other by either homo- or heterodimerisation, resulting in transcriptional activation or repression of the target genes, which mediates physiological significances (Figure 1.5).



Figure 1.4: Schematic view of structural organisation of nuclear receptors (Adapted from Sonoda et al. 2008).



**Figure 1.5:** Schematic view of general mechanism for nuclear receptor activation and/or repression (Adapted from Xie and Evans 2001).

As mentioned above, nuclear receptors are usually activated by ligand binding, and many cytochrome P450 enzymes are involved in the metabolism of these ligands (Akiyama and Gonzalez 2003). The molecular mechanism of cytochrome P450 induction by its ligands was discovered almost 50 years ago, and this knowledge is essential for both the study of drug metabolism and the carcinogenic process in humans (Meyer 2007). Many ligand-activated nuclear receptors have been demonstrated to be key mediators in the induction of

cytochrome P450s. The transcriptional activation of some members of the cytochrome P450 family (i.e. families 1, 2, 3, 4, 7, 24 and 26) that are involved in the metabolism of endogenous and xenobiotic compounds have been identified (Honkakoski and Negishi 2000). The review in this section summarises the roles of cytochrome P450-associated nuclear receptors such as aryl hydrocarbon receptor (AhR), pregnane X receptor (PXR), constitutive androstane receptor (CAR), PPARα, vitamin D receptor (VDR), retinoic acid receptor (RAR) and farnesoid X receptor (FXR) including crosstalk in the regulation of cytochrome P450 expression. The regulation of cytochrome P450 genes by nuclear receptors and their heterodimerisation with a partner is shown in Figure 1.6.





Abbreviations: DRE; DNA response element, AhR; aryl hydrocarbon receptor, ARNT; AhR nuclear translocator, CAR; constitutive androstane receptor, RXR; retinoid X receptor, PXR; pregnane X receptor, PPAR $\alpha$ ; peroxisome proliferator-activated receptor alpha, FXR; farnesoid X receptor, LXR $\alpha$ ; liver X receptor, VDR; vitamin D receptor, RAR; retinoic acid receptor.

Introduction

#### **1.7.2** Aryl hydrocarbon receptor (AhR)

Aryl hydrocarbon receptor (AhR) is a member of the basic helix-loop-helix (bHLH) protein which belongs to the Per-Arnt-Sim (PAS) family of transcription factors. AhR plays a role in xenobiotic metabolism and in mediating the carcinogenic effects of dioxinlike compounds. A number of studies have demonstrated that the induction of CYP1 family member expression is controlled by AhR signalling pathway. AhR ligands include polycyclic aromatic hydrocarbons (PAHs, e.g. 3-methylcholanthrene) and 2,3,7,8tetrachlorodibenzo-p-dioxin (TCDD) (Kim et al. 2005). In the absence of ligand, AhR is a cytoplasmic protein complex consisting of chaperone proteins such as heat shock protein 90 (Hsp90), p23 and ARA9. In the presence of ligand, AhR ligand-bound is dissociated from the chaperone proteins in the cytoplasm and is translocated into the nucleus where it dimerises with AhR nulcear translocator (ARNT, also called hypoxia inducible factor-1 beta; HIF-1 $\beta$ ) protein. This complex then binds to DNA response elements (DRE) in regulatory regions of AhR responsive genes (AhREs), leading to the induction of CYP1 genes in tissue-specific targets (Kawajiri and Fujii-Kuriyama 2007). In addition, Kim and co-workers demonstrated a significant positive correlation between hepatic AhR and CYP1A1 expression levels (Kim et al. 2005).

Another key protein in the PAS family is the AhR repressor (AhRR), which is constitutively expressed in the nucleus and is inducible in an AhR-dependent manner. AhRR needs to form a dimer with ARNT and the AhRR/ARNT complex binds to DREs. It was believed that AhRR can inhibit the AhR signalling pathway by inhibition of the AhR/ARNT complex and binding to DREs (Kawajiri and Fujii-Kuriyama 2007). However, Evans and colleagues have recently shown that repression of AhR activity by AhRR is through protein-protein interactions rather than the previously proposed mechanism (Evans et al. 2008). In addition to the induction of CYP1 genes by AhR, hypoxia-inducible factor-1alpha (HIF-1 $\alpha$ ) requires dimerisation with ARNT to begin

48

transcription of its target genes which is similar to either AhR or AhRR expression. In the breast cancer ZR-75-1 cells grown under hypoxic conditions for 24 hours, AhR-mediated transactivation was significantly decreased and this contributed to loss of Ah-responsiveness, possibly due to decreased expression of AhR or competetion for ARNT or their bindings with CYP1A1 promoter (Khan et al. 2007). These observations suggest that expression of AhR, ARNT, AhRR or HIF-1 $\alpha$  affects the regulation of CYP1 genes.

A novel chemotherapeutic prodrug aminoflavone (an AhR ligand) leads to a translocation of AhR from the cytoplasm to the nucleus, resulting in the interaction between AhR and DRE, and CYP1A1 transcription. This was observed in the aminoflavone-sensitive MCF7 cells but not in the resistant MDA-MB-435 cells, suggesting that the toxicity of aminoflavone prodrug was selectively activated through the engagement of AhR-mediated signal transduction only in a sensitive breast tumour cell line (Loaiza-Perez et al. 2004). This is an essential consideration for the use of aminoflavone in breast cancer patients in clinical studies.

## **1.7.3** Constitutive and rostane receptor (CAR)

Constitutive androstane receptor (CAR), also called constitutive active receptor, is the most closely related to PXR. CAR is highly expressed in the liver and at lower levels in the intestine. It is called a 'constitutive' receptor because of its ability to transactivate in the absence of ligand-binding (Akiyama and Gonzalez 2003). CAR plays an important role in the transcription of CYP2B6, CYP3A4 and CYP2C (Pascussi et al. 2003), and its role in xenobiotic regulation of CYP2B6 genes has been widely studied in humans. The CYP2B6 gene is inducible by phenobarbital-like inducers (e.g. phynetoin) through CAR, which bind to the phenobarbital-response element (PBRE) sites as a heterodimer with retinoid X receptor (RXR) in the nucleus. The CAR/RXR heterodimer results in the transcription of CYP2B6 mRNA (Wang et al. 2004). Chang and co-workers

Introduction showed a significant positive relationship between CYP2B6 and CAR expression in human liver (Chang et al. 2003).

In addition to CAR ligands, the PXR activator clotrimazole can bind to CAR and subsequently suppresses CYP2B6 transcription. The antifungal antibiotic clotrimazole and steroid hormone androstanol are classified as potent deactivators of CAR and inverse agonists, suggesting that CAR and PXR might share xenobiotic and steroid ligands (Moore et al. 2000). However, CAR has also been shown to be involved in the detoxification of some endogenous ligands (i.e. bile acids and bilirubin) (Moore et al. 2006) and biosynthesis of sterols and steroids in human liver (Meyer 2007) which alter the induction of CYP2B6 gene.

# **1.7.4** Pregnane X receptor (PXR)

Pregnane X receptor (PXR) also termed steroid X receptor (SXR) is predominantly expressed in the liver and small intestine (Moore et al. 2006) and mediates induction of CYP3A4, CYP2B6, CYP2C8 and CYP2C9 (Pascussi et al. 2003). PXR functions as a sensor for steroids (e.g. oestrogen, progesterone and corticosteroid) and xenobiotics (e.g. phenobarbital, clotrimazole, rifampicin and nifedipine). PXR mainly activates CYP3A4, which is a primary target gene of PXR. Therefore, compounds that activate PXR also induce transcriptional activity of CYP3A4. The ligand-activated PXR forms a heterodimer with RXR, and the heterodimer PXR/RXR binds with an everted repeat separated by six base pairs (ER6) from the CYP3A4 proximal promoter, resulting in an increase in CYP3A4 induction (Vyhlidal et al. 2004). PXR and CAR are closely related nuclear receptors, sharing ligands and sharing a heterodimer partner (RXR). In addition, CAR and PXR are also involved in biosynthesis and catabolism of steroid hormones, lipid metabolism, glucose homeostasis and inflammation. This could explain crosstalk between these receptors (Section 1.7.9), which may affect cytochrome P450 expression and physiological functions (Moreau et al. 2007b).

Miki and colleagues investigated the expression of PXR mRNA and protein using realtime RT-PCR and immunohistochemical methods in 60 human breast cancer tissues. They reported that PXR was detected in breast tumours but not in matched adjacent normal breast cells. Positive correlations between PXR and both the histological grade and lymph node status of the tumours were noted in this study (Miki et al. 2006), suggesting that PXR may be involved in human breast carcinogenesis. These findings are in disagreement with a study by Dotzlaw and co-workers (Dotzlaw et al. 1999) though, who demonstrated PXR mRNA in both breast tumour and their corresponding normal breast tissues. Higher levels of PXR mRNA were detected in ER-negative tumours than in ER-positive tumours. They showed that PXR is expressed in breast cancer cell lines T47D, MCF7 and MDA-MB-231.

#### **1.7.5** Vitamin D receptor (VDR)

1,25-(OH)<sub>2</sub>D<sub>3</sub> and its analogues have been shown to inhibit proliferation, induce apoptosis and immunomodulation of human cancer cells mediated by vitamin D receptor (VDR). Ligand-activated VDR forms a heterodimer with RXR and binds to vitamin D-response elements (VDREs) on the target genes, causing either activation or repression of target gene expression (Moreau et al. 2007a). VDR is mainly expressed in the kidney, intestine and thyroid (Moore et al. 2006). CYP24A1 is one of the VDR target genes which is directly regulated by VDR, and plays a role in the inactivation of vitamin D in the kidney (Zhou et al. 2006). Anderson and colleagues demonstrated overexpression of CYP24A1 and down-regulation of VDR mRNA in various cancer cell lines including breast cancer MCF7 cells (Anderson et al. 2006), suggesting that the inverse relationship between CYP24A1 and VDR is associated with carcinogenesis. In addition, Yee and co-workers showed that a combination of ketoconazole and tetralone inhibits CYP24A1 mRNA and Introduction enhances VDR signalling in androgen-independent prostate cancer DU-145cells (Yee et al. 2006).

# **1.7.6** Peroxisome proliferator-activated receptor alpha (PPARα)

Peroxisome proliferators (e.g. fibrate class of hypolipidemic drugs including WY-14643 or pirinic acid) are a group of chemicals that lead to an increase in size and number of hepatic peroxisomes when administered to rodents. PPAR $\alpha$  is highly expressed in the liver, kidney and vascular wall and is activated by a variety of fatty acids. PPAR $\alpha$  plays a role in the transcriptional regulation of enzymes involved in the  $\omega$ - or  $\beta$ -fatty acid oxidative pathway (Bishop-Bailey and Wray 2003). In the presence of peroxisome proliferators, PPAR $\alpha$  can heterodimerise with RXR, and the PPAR $\alpha$ /RXR heterodimer complex binds to peroxisome proliferator-response elements (PPREs), leading to the transcriptional activity of target genes such as CYP4A (Ito et al. 2007). High levels of PPAR $\alpha$  mRNA have been demonstrated in MCF7 breast cancer cells, and this may suggest a crucial role for PPAR $\alpha$  in human breast carcinogenesis (Suchanek et al. 2002)

# 1.7.7 Retinoic acid receptor (RAR)

All-*trans*-retinoic acid (ATRA), the most active metabolite of retinoids (vitamin A) plays an important role in cell differentiation and proliferation of epithelial cells mediated by retinoic acid receptor alpha (RAR $\alpha$ ) (Njar et al. 2006). The binding of ligand to RAR $\alpha$ results in a heterodimer with RXR, and the RAR $\alpha$ /RXR heterodimer binds to retinoic acid response elements (RAREs) or retinoid X response elements (RXREs), causing transcriptional suppression or activation of target genes such as CYP26A1. In human breast cancer, a positive relationship was found between the retinoid growth-inhibitory effects and RAR $\alpha$  expression (Fitzgerald et al. 1997). Expression of RAR and RXR has been demonstrated in breast cancer cells (Titcomb et al. 1994). In addition, Mu and colleagues investigated the effects of various nuclear receptor ligands on CYP19A1 activity in the breast cancer MCF7 cells. Findings showed that RAR and RXR selective ligands increased CYP19A1 activity, suggesting that these compounds may be involved in the regulation of oestrogen production in breast cancer cells (Mu et al. 2000).

## 1.7.8 Farnesoid X receptor (FXR)

Bile acids are the main product of cholesterol metabolism in the liver and are important for lipid absorption in the intestine. Farnesoid X receptor (FXR) has been characterised as a bile acid receptor. In the presence of the most potent bile acid (chenodeoxycholic acid; CDCA), FXR forms a heterodimer with RXR and the FXR/RXR heterodimer binds to the bile acid response elements (BAREs), resulting in the suppressive effects of bile acid and a transcriptional repression of target genes such as CYP7A1 (Akiyama and Gonzalez 2003). FXR is found to be expressed in breast cancer tissue, suggesting that FXR expression in association with high levels of bile acid may play a role in breast cancer treatment (Swales et al. 2006).

# 1.7.9 Crosstalk between nuclear receptors

Crosstalk is defined as a functional interaction between two (or more) signalling pathways of members of the nuclear receptor superfamily. Crosstalk between nuclear receptors has been extensively studied (Keller et al. 1995, Zhou et al. 2006, Meyer 2007, Moreau et al. 2007b, Roth et al. 2008) and their interactions have been demonstrated to be involved in several intracellular signalling pathways (i.e. growth factors, hormones and cytokines) and xenobiotic metabolism (Figure 1.7). Regulation of target gene expression by nuclear receptors is complex. A better understanding of the associated molecular mechanism is likely to explain biological phenomena in the body. As shown in Figure 1.5, transcriptional activation of nuclear receptors' target genes requires a number of different steps (i.e. ligand binding, coactivator/corepressor protein, dimerisation with a partner and DNA binding within their target genes) and each signalling pathway is usually connected with other pathways (Reményi et al. 2004). Crosstalk may lead to alterations in other genes' target gene expression or activity (Waxman 1999, Honkakoski and Negishi 2000). This section provides some examples of crosstalk between nuclear receptors which have been shown to affect cytochrome P450 expression.



**Figure 1.7:** Crosstalk interactions between the regulation of xenobiotic and endogenous compounds. Abbreviations: AhR; arylhydrocarbon receptor, PXR; pregnane X receptor, CAR; constitutive androstane receptor, PPAR $\alpha$ ; peroxisome proliferator-activated receptor alpha VDR; vitamin D receptor, RAR; retinoic acid receptor, ER; oestrogen receptor, P450; cytochrome P450, cpds; compounds, met; metabolism.

The negative crosstalk between ER and AhR signalling pathways has been well documented by several research groups using breast tumour tissue and cell lines (Safe et al. 1998, Safe et al. 2000, Safe and Wormke 2003, Matthews et al. 2005, Matthews and Gustafsson 2006). ER and AhR are ligand-activated transcription factors. As mentioned in Section 1.7.2, AhR mediates the toxic effects of dioxin-like compounds (e.g. TCDD) and is involved in the induction of CYP1 genes. However, the AhR signalling pathway can be linked to the ER signalling pathway as CYP1 genes play a key role in oestrogen metabolism. Recently, mechanisms of inhibitory cross talk between ER and AhR have been proposed by Matthews and Gustafsson (Matthews and Gustafsson 2006). There are at least five mechanisms that are involved in the role of AhR inhibiting ER activity (1) direct inhibition via inhibitory xenobiotic response elements (iXREs), (2) squelching shared coactivators (i.e. ARNT), (3) synthesis of an inhibitory factor (unknown), (4) increased ER degradation and (5) increased expression of enzymes involved in oestrogen synthesis (i.e. CYP19) and metabolism (i.e. CYP1A1 and CYP1B1). In addition, these cytochrome P450s play an important role in the metabolism of anticancer drugs (Section 1.8), and this knowledge could improve the clinical approach to particular cancer therapies

Transcriptional crosstalk between ER and RAR has been demonstrated in human breast cancer cells (Rousseau et al. 2003). Retinoic acid has been shown to be associated with down-regulation of ER mRNA and inhibition of cell growth in oestrogen-dependent breast cancer MCF7 cells. By contrast, this was not observed in ER-negative breast cancer MDA-MB-231 cells. Higher levels of RAR were found in ER-positive breast cancer than in ER-negative breast cancer cell lines (Rubin et al. 1994), suggesting that expression of RAR and ER is essential for retinoid responsiveness. Furthermore, oestrogen has been shown to elevate RAR expression, whereas antioestrogen seems to be associated with down-regulation of RAR (Lu et al. 2005).

CAR and PXR are mediated by induction of CYP2B and CYP3A genes respectively, and share a common heterodimerisation partner i.e. RXRα, ligands and DNA response elements within their target genes (Handschin and Meyer 2003). The CAR and PXR genes are closely related members of the nuclear receptor family and have overlapping substrates. These nuclear receptors are involved in the regulation of various endogenous compounds such as steroid hormones, bile acids, eicosanoids and cholesterol (Echchgadda et al. 2007).

A recent study investigated CAR- and VDR-binding to specific response elements in human hepatocyte cells using electromobility shift assay, and determined CYP24

55

expression (Moreau et al. 2007a). Results showed that the CAR/RXR heterodimerisation binds to the promoter of CYP24, which is the same region as VDR/RXR, resulting in decreased CYP24 expression and response to vitamin D<sub>3</sub>, suggesting that CAR ligands interfere with VDR target genes. These findings are similar to a previous study that investigated crosstalk between PXR and VDR and showed that the PXR/RXR heterodimer binds to the promoter of CYP24 and competes with VDR (Pascussi et al. 2005). This suggests that PXR, CAR and VDR share DNA response elements in the promoter of their target genes, leading to altered expression of CYP24A1, CYP2B6 and CYP3A4.

It is important to note that RXR is a common heterodimerisation partner for several nuclear receptors (i.e. PXR, VDR, RAR, PPAR $\alpha$ , FXR and LXR $\alpha$ , Figure 1.6) and this may also cause crosstalk between other signalling pathways, not only vitamin D<sub>3</sub> metabolism. Crosstalk between PPAR $\alpha$ /RXR and ER has been reported by Keller and colleagues (Keller et al. 1995). A PPAR $\alpha$ /RXR heterodimer inhibits ER activity through competition for binding to oestrogen response elements.

# **1.8** Cancer treatment approaches based on cytochrome P450s

The role and importance of cytochrome P450s in cancer formation and progression has led to the development of cancer therapeutics-based on cytochrome P450 expression and metabolic pathways. Knowledge in the area of these enzymes' substrate and mechanism of action in tumour cells has become very useful in the design of new chemotherapeutics and/or development of more effective existing anticancer drugs. It can help reduce systemic toxicity and enhance specificity to tumour cells. To date, applications of cytochrome P450s in cancer therapy consist of (1) inhibiting enzymes involved in hormone-dependent cancer and vitamin metabolism, (2) designing prodrugs activated by the enzymes and (3) immunotherapy targeted at tumour-specific enzymes (McFadyen et al. 2004, Bruno and Njar 2007, Purnapatre et al. 2008) (Figure 1.8).

Introduction

In steroid hormone biosynthesis pathways, CYP17 and CYP19 are key enzymes that regulate the biosynthesis of oestrogen. CYP17 plays an important role in the conversion of pregnenolone and progesterone to androgen and oestrogen precursors, whereas CYP19 or



Figure 1.8: Implications of cytochrome P450s in cancer therapy.

aromatase is involved in the conversion of testosterone and androstenedione to oestradiol and oestrone. Several inhibitors of CYP19 such as letrozole, anastrazole and exemestane have been developed as anticancer agents for the management of breast cancer patients who have oestrogen-dependent tumours. These agents inhibit the production of oestrogen in target tissues and improve survival in patients with metastatic breast cancer (Gibson et al. 2007, Perez 2007a). Inhibitors of CYP17 such as ketoconazole and abriaterone are also used as anticancer drugs for the treatment of androgen-dependent cancers particularly prostate cancer, and these compounds inhibit androgen production and are more effective in advanced prostate cancer treatment (Moreira et al. 2007).

Vitamins A and D are believed to have a crucial role in cancer prevention and treatment. All-*trans*-retinoic acid (ATRA) is the most active metabolite of vitamin A, it plays a role in cell differentiation and proliferation of epithelial cells, and has been shown to be associated with tumour growth inhibition in various solid tumours (Njar 2002). CYP26A1 is involved in the metabolic inactivation of ATRA and it is related to the rapid ATRA metabolism in cancer patients. Inhibitors of CYP26A1 (e.g. liarozole, R115866 and R116010) inhibit both ATRA metabolism and CYP19 activity. However, liarozole lacks specificity, and has a relatively weak potency for CYP26A1, and is no longer being developed as an anticancer agent. R116010 is more potent and a selective inhibitor of CYP26A1 (Njar et al. 2006) which is currently in preclinical trial for patients with hormone-independent prostate and breast tumours (Njar 2002, Njar et al. 2006).

The biologically active metabolite of vitamin D<sub>3</sub> has been shown to play a role in the inhibition of cell proliferation, promotion of differentiation and induction of apoptosis and it has potential as a cancer therapeutic (González-Sancho et al. 2006, Deeb et al. 2007). CYP24A1 is the key enzyme involved in the catabolism of 1,25-(OH)<sub>2</sub>D<sub>3</sub> to the less active vitamin D<sub>3</sub> metabolite. Overexpression of CYP24A1 has been shown in a wide range of tumours (Mimori et al. 2004, Anderson et al. 2006, González-Sancho et al. 2006) and is known as an oncogene (Mimori et al. 2004). In addition, Townsend and colleagues demonstrated that higher levels of CYP24A1 were detected in breast tumour compared to normal breast tissue (Townsend et al. 2005). Inhibitors of CYP24 such as genestein, QW-1624F2-2 and EB1089 are vitamin D<sub>3</sub> analogues and have ability to inhibit CYP24A1 expression (Bruno and Njar 2007). Recently, Sundaram and co-workers showed that QW-1624F2-2 inhibits cell proliferation and invasion of breast tumour and has been proposed for preclinical studies as an effective chemotherapeutic agent in breast cancer treatment (Sundaram et al. 2006).

As mentioned previously (Section 1.2.2), hypoxic conditions existing in the tumour microenvironment are associated with resistance to chemotherapy and radiotherapy in malignant cells. However, tumour hypoxia appears to be a specific property of cells in

many solid tumours, and it is a potential target for tumour-specific prodrugs activated by oxygen-sensitive reductase (bioreductive drugs) including cytochrome P450s (Denny 2004). The topoisomerase II inhibitor prodrug AQ4N (Banoxantrone) is designed to have anti-tumour activity following activation by CYP1A1, CYP1B1 and CYP3A4. Under hypoxic conditions, AQ4N is metabolised to the potent topoisomerase II inhibitor AQ4 this process is inhibited by oxygen.. AQ4N is currently in phase Ib/IIa clinical trials, and it has been reported that it is well tolerated and less toxic than another bioreductive prodrug tirapazamine (McKeown et al. 2007).

Recently, Albertella and colleagues demonstrated clinical results derived from a phase I study of AQ4N in thirty-two patients with advanced solid tumours (i.e. breast, cervix, glioblastoma, bladder, and head and neck). These patients received a single dose of AQ4N prior to surgery and results showed that AQ4N penetrated well within tumour cells, as high concentrations of the activated form AQ4 were detected in tumour samples (Albertella et al. 2008). This data indicates that the effect AQ4N has on solid tumours is hypoxia-selective and further evaluation in clinical trials is required.

A number of cytochrome P450 enzymes are involved in the activation and/or inactivation of several anticancer drugs. Paclitaxel which is widely used in cancer chemotherapy is inactivated in human liver by CYP2C8 forming a 6α-hydroxylation derivative with 30-fold less than the parent drug (Taniguchi et al. 2005). Another example of a taxane anticancer drug or stabilising-microtubule agent that is inactivated in the liver by CYP3A is docetaxel (Scripture et al. 2005). In addition, the tumour response to docetaxel in human breast tumours can be predicted by CYP3A4 (Miyoshi et al. 2005). Patients with CYP3A4 negative tumours showed a higher response rate to docetaxel than those with CYP3A4 positive tumours. However, a previous study reported that the coadminstration of CYP3A4 inhibitors (e.g. ketoconazole and midazolam) with docetaxel may cause a

Introduction
inflammatory status and risk of docetaxel-induced febrile neutropenia in patients with advanced cancer.

The anticancer prodrugs cyclophosphamide, doxorubicin and tamoxifen are bioactivated to active agents mainly by cytochrome P450s. Cyclophosphamide is an alkylating agent derived from the produrg oxazaphosphorine which is primarily activated in the liver by CYP2B6 to generate a cytotoxic activity of phosphoramide mustard and acrolein, then delivered to tumour cells via blood circulation (Stoff-Khalili et al. 2006). To prevent deleterious secondary effects and minimise damage to normal tissue, cytochrome P450 mediated gene-directed enzyme and prodrug therapy (GDEPT) have recently been developed with CYP2B6 for the treatment of cyclophophamide by delivery of exogenous cytochrome P450s to the target site (Roy and Waxman 2006). MetXia encodes human CYP2B6, it activates cyclophosphamide in tumour cells is designed to sensitise malignant cells and can be directly injected into tumours prior to cyclophosphamide treatment. MetXia is currently in phase I/II clinical trials for advanced breast cancer patients (Braybrooke et al. 2005).

The anthracycline antibiotic prodrug doxorubicin is one of the most widely used anticancer drugs. Doxorubicin is activated in human liver by CYP3A4 to form a potent active metabolite methoxymorpholinyl doxorubicin (MMDX) which interacts with DNA by intercalation and inhibition of topoisomerase II (Mathijssen and van Schaik 2006). In addition, a study by Lu and colleagues investigated anti-tumour activity of MMDX in various tumour cell lines (i.e. breast, lung, brain and colon). Results showed that very low levels of CYP3A4 expression and low anti-tumour activity of MMDX were found in these

Introduction cell lines (Lu and Waxman 2005), suggesting that increasing the content of CYP3A4 in the tumour site using gene-therapy vectors may be helpful when treating patients with doxorubicin.

Tamoxifen is currently used for the treatment of oestrogen-dependent breast cancer patients and is activated in the liver by several cytochrome P450s (i.e. CYP2D6, CYP3A, CYP2C9, CYP2C19 and CYP2B6). It is mainly activated by CYP2D6, forming 4hydroxytamoxifen which has 50-100-fold more potent antioestrogenic effects than tamoxifen (Dehal and Kupfer 1997). CYP2D6 is a highly polymorphic gene and this may contribute to variability of treatment outcome between individuals and/or ethnicities (Zanger et al. 2007). Other cytochrome P450s appear to have less important roles in the metabolism of tamoxifen (Scripture et al. 2005).

Novel synthetic prodrugs selectively designed to be activated by cytochrome P450s, especially CYP1A1 and CYP1B1, have been developed and show promise as new options for cancer therapeutics. The phytoestrogen resveratrol (3,5,4'-trihydroxy-*trans*-stilbene) is derived from natural sources (i.e. red grape skin). Resveratrol is mainly metabolised by CYP1B1 within tumour cells to form the anticancer agent piceatannol (Piver et al. 2004) which has antioestrogen effects. It has antioxidant and anti-inflammatory effects which may induce apoptosis and modulate the function of ER in breast cancer cells (Athar et al. 2007). Phase I and pharmacokinetic evaluations of oral resveratrol in healthy volunteers have recently been demonstrated through investigation of six metabolites of resveratrol in urine (Boocock et al. 2007). Results suggested that the highest dose of resveratrol did not cause severe adverse drug reaction and cancer chemopreventive effects of resveratrol metabolites in this study warrant further investigation by clinical studies. Another novel synthetic prodrug involving cytochrome P450s is a water-soluble fluorinated benzothiazole prodrug phortress that is a potent AhR agonist and can induce CYP1A1 expression. CYP1A1 converts phortress prodrug into an electrophilic reactive intermediate, that induces formation of DNA adducts resulting in DNA damage and cell death. Phortress is rapidly hydrolysed in the presence of cells to its lipophilic parent drug, which is then selectively taken up into sensitive cells, followed by AhR binding and translocation into the nucleus, leading to induction of CYP1A1 (Leong et al. 2003). Phortress has recently entered phase I clinical studies with breast cancer patients in the UK (Bradshaw and Westwell 2004). However, Brockdorff and colleagues demonstrated that increased expression of CYP1A1 and CYP1B1 is associated with antioestrogen resistance in breast cancer cell lines (Brockdorff et al. 2000), and further clinical studies are required to investigate CYP1A1 and CYP1B1 expression in tumour cells and avoid this problem in patients.

Gene expression of tumour-specific cytochrome P450s has introduced the concept of cancer immunotherapy using a DNA vaccine. CYP1B1 has been shown to be a tumour-associated antigen that is overexpressed in a wide range of tumour tissues and cell lines (e.g. breast, uterus and ovary) and therefore represents an attractive target for immune-based therapy. The CYP1B1-based DNA vaccine Zyc300 is designed to induce the immune system anti-CYP1B1-specific CD8<sup>+</sup> T-cell against CYP1B1-expressing tumour cells. Zyc300 is currently in phase I evaluation (Maecker et al. 2003) and has been shown to improve survival in cancer patients who have developed immunity to CYP1B1. In addition, a study by Gribben and co-workers has shown that increased anti-CYP1B1 immunity in patients with metastatic tumours (i.e. ovarian, colorectal, renal, prostate and breast cancer) dramatically improved their response to next salvage therapy (Gribben et al. 2005). Though unexpected, it was associated with a good clinical outcome and low toxicity.

A CYP3A4-based therapeutic antisense phosphorodiamidate morpholino oligomer AVI-4457 has been developed as CYP3A4 plays an important role in the deactivation of paclitaxel and cyclophosphamide which are important chemotherapeutic agents, but it is not a tumour-specific expression enzyme (McFadyen et al. 2004). A previous study demonstrated that inhibition of CYP3A4 expression was observed in primary human hepatocytes after 24 hours exposure to AVI-4557. Findings showed that the cytocidal activity of cyclophosphamide and paclitaxel are inhibited and induced with AVI-4557 treatment in these cells respectively, but no effect on the cytocidal activity of cisplatin (Arora et al. 2002). However, further information on clinical trials of this agent has not been reported.

# **1.9** Aims

Cytochrome P450 enzymes are involved in the carcinogenic process as well as cancer treatment. The current study aimed to detect the presence of these enzymes and their regulatory nuclear receptors in breast cancer cell lines as models, and to identify potential novel targets for breast cancer treatment based upon the expression profiles obtained. This study selected 24 cytochrome P450 genes and 10 P450-regulatory nuclear receptors for investigation (Table 1.5). Characterisation and regulation of CYP4Z1 expression were of particular interest.

Cytochrome P450 enzymes	Cytochrome P450-regulatory nuclear
	receptors
Family 1: CYP1A1, CYP1A2 and CYP1B1	AhR, ARNT, AhRR, HIF-1α, CAR, PXR,
Family 2: CYP2A6, CYP2B6, CYP2C8, CYP2C9,	VDR, RAR, RXR and ER
CYP2C18, CYP2C19, CYP2D6, CYP2F1,	
CYP2J2, CYP2R1, CYP2S1 and CYP2U1	
Family 3: CYP3A4, CYP3A5, CYP3A7, CYP3A43	
Family 4: CYP4X1 and CYP4Z1	
Family 24: CYP24A1	
Family 26: CYP26A1 and CYP26B1	

**Table 1.5:** List of candidate genes in this study.

# **Chapter 2**

# **Materials and Methods**

### 2.1 Introduction

As previously detailed (Section 1.3), the use of cell culture as an *in vitro* model system has become increasingly important in several aspects of biomedical sciences at the molecular and cellular levels in order to provide a better understanding of cell biology *in vivo* particularly in disease stages. Established human breast cancer cell lines representing characteristics of the primary breast tumours (e.g. ethnicity, ER $\alpha$  status, epithelialmesenchymal transition (EMT) and invasive phenotype) have been extensively used to determine target gene expression (Burdall et al. 2003).

Microbial contamination in cell culture is one of the most common problems that can lead to incorrect interpretation of gene expression as it may result in serious effects (i.e. alteration of cell growth pattern, cell morphology, and synthesis of DNA or RNA or protein). To minimise contamination, all cell culture procedures (e.g. maintenance of cell lines, resuscitation of frozen cells, trypsinisation and freezing cell cultures) in this study were carried out in a Class II safety cabinet (BioAir Instruments S.r.l., Siziano, Italy). Pipette tips and glass containers were autoclaved at 121°C and 15 lbs/in<sup>2</sup> for 15 minutes, and glassware was heated at 160°C for 2 hours. Penicillin/streptomycin solution is the most widely used combination of antibiotics in mammalian cell culture to prevent bacterial growth. In this study, a final concentration of 100 U/ml of penicillin and 100 µg/ml of streptomycin was used in growth media (Freshney 2005).

In addition, all cell lines, growth media and sterile solutions of reagent were tested for microbial contamination prior to use. Brain heart infusion (BHI) broth was used to test for

bacterial contamination by incubating at 37°C for 1-2 days. Sabouraud (SAB) broth was used to detect fungal and yeast contamination by incubating at 32°C for 7-10 days (Freshney 2005).

Over the last 50 years, molecular biologists have made advances in the isolation, manipulation, and characterisation of the genetic material in cells. The genetic material is transcribed into RNA and then translated into protein. Total RNA is isolated from the cell of interest, and a complementary DNA (cDNA) library is synthesised from messenger RNA (mRNA) using an oligonucleotide deoxythymidylic acid (oligo dT) primer which is annealed onto the polyadenylated (poly A) tail of the mRNA template. Thus, single stranded mRNA is converted into a cDNA copy of the message using a reverse transcriptase enzyme and deoxynucleotide triphosphates (dNTPs). The reverse transcription-polymerase chain reaction (RT-PCR) is commonly used in the determination of the abundance of mRNA within a cell as a measure of gene expression. RT-PCR is used to amplify specific regions of the cDNA template using sequence specific primers. The molecular weight size of the DNA fragment is estimated by semi-logarithmic plotting and compared to size prediction from known sequence. Restriction digestion and DNA sequencing are widely used to identify the gene products.

Figure 2.1 represents a flowchart of the process used in this thesis for the detection of gene-specific expression. The following sections in this chapter provide information on molecular biology techniques. A list of chemicals and reagents including their suppliers are presented in Appendix A.

# 2.2 Cell culture and cell lines

The growth media, cell lines and culture techniques employed in starting growth and maintaining cell lines, resuscitation of frozen cells, trypsinisation, freezing cell cultures

and cell viability determination are described below. All working steps in cell culture

procedure were performed in a Class II safety cabinet.



Figure 2.1: Flowchart of procedure for the detection of gene-specific expression.

### 2.2.1 Growth media

# 2.2.1.1 GlutaMax<sup>TM</sup>I

L-glutamine is an unstable essential amino acid required in most cell culture media formulations. It is used for energy production, protein synthesis and nucleic acid synthesis. However, L-glutamine spontaneously degrades in cell culture media or aqueous solution to the ammonium product. High ammonia concentrations are toxic to cells, lowering cell viability and protein production. Therefore, several cell culture media which can be purchased are now supplemented with L-glutamine in dipeptide forms, such as alanyl-lglutamine and glycyl-l-glutamine. GlutaMax<sup>TM</sup>I formulation is a dipeptide which is cleaved by aminopeptidase releasing, Lglutamine and L-alanine. GlutaMax<sup>TM</sup>I contains a stabilized form of L-glutamine, in equal molar concentrations as glutamine-containing media. Therefore, GlutaMax<sup>TM</sup>I-containing cell culture media can increase media stability, minimise toxic ammonia build-up and maximize cell performance (Christie and Butler 1994). Media formulations RPMI 1640, L15, DMEM and DMEM/F12 supplemented with GlutaMax<sup>TM</sup>I were employed for all cell culture experiments.

#### 2.2.1.2 Roswell Park Memorial Institute medium (RPMI1640)

This was developed by Moore et al. (Moore et al. 1967) at Roswell Park Memorial Institute (RPMI). The formulation utilizes a biocarbonate buffering system, and is based on the RPMI1630 series with alterations in the amounts of amino acids and vitamins. RPMI1640 medium has extensive applications for culturing growth of several types of cells, such as human normal and neoplastic leukocyte cell lines.

#### 2.2.1.3 Leibovitz's medium (L15)

This medium was developed by Leibovitz (Leibovitz 1963) in 1963 and was originally formulated for use in the culture of tumour cell lines in the absence of carbon dioxide (CO<sub>2</sub>) and without a sodium bicarbonate buffer system. L15 medium is buffered by a particularly high concentration of free base amino acids (L-arginine primarily, but also Lhistidine and L-cysteine) and sodium pyruvate. The sodium pyruvate can be utilized as a more readily accessible energy source for cell metabolism and can produce endogenous carbon dioxide. Production of acidic metabolites is reduced by replacing glucose with galactose in order to maintain physiological pH (Barngrover et al. 1985).

#### 2.2.1.4 Dulbecco's Modified Eagle's Medium (DMEM)

DMEM is one of the most commonly used variations of Basal Medium Eagle (BSE). This medium contains twice the amino acid concentrations of MEM and four times the vitamin concentrations, as well as additional components to facilitate better buffering (e.g. hydrogencarbonate and carbon dioxide concentrations). High and low glucose forms are available containing 4.5 g/L and 1 g/L of glucose respectively. This formulation is well suited for culturing a wide range of mammalian cell lines (Dulbecco and Freeman 1959).

#### 2.2.1.5 DMEM/Ham's F12 medium

Ham's F12 media was developed to clone Chinese Hamster Ovary (CHO) cells in low serum concentrations (Ham 1965) and is now extensively used to culture different kinds of mammalian cells. The mixture of DMEM and Ham's F12 (1:1 v/v) is suitable for some epithelial, endothelial, and granulose cell types.

#### 2.2.2 Cell lines

A database of the seven breast cancer cell lines used in this study is shown in Table 1.1. MDA-MB-157, MDA-MB-231, T47D, ZR-75-1 and ZR-75-30 cells were obtained from the European Collection of Cell Cultures (ECACC; Salisbury, Wiltshire, UK). MDA-MB-468 cells were a generous gift from Professor Andreas Gescher (Department of Cancer Studies and Molecular Medicine, Leicester Medical School, University of Leicester, UK). MCF7 cells were an established cell line within the School of Pharmacy and Life Sciences, The Robert Gordon University, Aberdeen, UK.

MCF10A, a non-transformed human breast epithelial cell line is derived from breast tissue of a 36 year old patient with fibrocystic disease (Soule and McGrath 1986). A previous study demonstrated that MCF10A contains a number of features of normal breast epithelium, for example, lack of carcinogenicity in nude mice, and dependence on hormones and growth factors for cell proliferation (Soule et al. 1990). In the present study, MCF10A was obtained from the American Type Culture Collection (ATCC; LGC Promochem, Middlesex, UK).

# **2.2.3** Growth and maintenance of human normal breast and breast cancer cells

All the media used (i.e. DMEM, RPMI1640, L15 and DMEM/Ham's F12) were supplemented with GlutaMax<sup>TM</sup>I and all cell lines were cultured under the recommended conditions. Stock cultures of MCF7, T47D and MDA-MB-157 cells were grown in DMEM. Cell stocks of ZR-75-30 and MDA-MB-468 cells were propagated in RPMI1640. Stock cultures of ZR-75-1 cells were grown in RPMI1640 containing 1 mM sodium pyruvate. The media above were supplemented with 10% (v/v) foetal bovine serum (FBS). Stock cultures of MDA-MB-231 cells were cultured in L15 supplemented with 15% (v/v) FBS. Cell stocks of MCF10A cells were grown in DMEM: Ham's F12 medium (1:1) supplemented with 20 ng/mL epidermal growth factor, 0.01 mg/mL insulin, 500 ng/mL hydrocortisone and 5% (v/v) horse serum (Thomas et al. 2006). All growth media were prepared containing penicillin (100 U/mL) and streptomycin (100  $\mu$ g/mL) in order to reduce the risk of microbial contamination. Cells were seeded at 2-5x10<sup>4</sup> cells/cm<sup>2</sup> and grown as a monolayer at 37°C in 5% CO<sub>2</sub> and 95% air. The cell culture media was changed regularly every 2-3 days to prevent a build-up of waste products as the cells grew and divided. Cells reached 70-80% confluence within 5-7 days.

As mentioned above, L15 medium (Section 2.2.1.3) must be used without carbon dioxide, therefore, MDA-MB-231 cells were cultured in flasks without vented filter caps and the flask caps were opened in the cell culture cabinet for 30 minutes daily to allow gaseous exchange to occur with atmospheric air and the replenishment of oxygen content as recommended by ECACC.

MDA-MB-157 cells were grown in L15 with GlutaMax<sup>TM</sup>I supplemented with 15% (v/v) FBS as recommended (ECACC), but they were extremely slow growing. Subsequently, ECACC's technical support suggested changing their growth media to DMEM with GlutaMax<sup>TM</sup>I supplemented with 10% (v/v) FBS. The MDA-MB-157 cells clearly grew better in DMEM medium than L15 medium. However, there was no difference in the morphological appearance between the cells cultured in L15 and in DMEM growth medium. The morphological appearance of all cell lines in this study is shown in Figures 2.2 and 2.3. All cells were photographed as a living monolayer (at high and low density) using an inverted microscope Leica AF6000 system (Germany) with Leica DFC300FX camera (20x magnification).

### 2.2.4 Resuscitation of frozen cells

Gentle handling and immediate removal of cryopreservative medium was required for the recovery of cells, as cryopreserved cells are extremely fragile. A cryovial containing frozen cells was transferred from liquid nitrogen storage to a Class II safety cabinet. The outside of the cryovial was wiped with 70% (v/v) ethanol to avoid microbial contamination, and the screw top was opened slightly for 2 seconds to reduce the pressure within the vial. The bottom half of the vial was then immediately immersed in a  $37^{\circ}$ C water bath until thawed. The cell suspension was gently transferred to a sterile universal tube containing 10 mL of an appropriate pre-warmed growth medium in order to remove or dilute residual dimethyl sulphoxide (DMSO), prior to centrifugation at 500 x g (Heraeus Biofuge Primo centrifuge, USA) for 5 minutes at room temperature. Increased speed and prolonged spin time has been suggested as decreasing viability of cells (Freshney 2005). The cell pellet was then re-suspended in 10 mL growth medium and the resulting suspension was transferred into a 25 cm<sup>2</sup> flask, before being placed in the incubator at  $37^{\circ}$ C in an atmosphere of 5% CO<sub>2</sub> and 95% air.



**Figure 2.2:** Morphology of oestrogen positive receptor breast cancer cell lines. Living monolayer, low and high density of MCF7 (A, B), T47D (C, D), ZR-75-1 (E, F) and ZR-75-30 (G, H) cells. Scale bar = 200  $\mu$ m.



**Figure 2.3:** Morphology of oestrogen negative receptor breast cancer cell lines. Living monolayer, low and high density of MDA-MB-231 (A, B), MDA-MB-157 (C, D), MDA-MB-468 (E, F), and normal breast cell line MCF10A (G, H) cells. Scale bar =  $200 \ \mu m$ .

#### 2.2.5 Trypsinisation

Trypsin is a proteolytic enzyme that cleaves proteins at the carboxyl side of the basic amino acids lysine and arginine. In a cell culture laboratory, trypsin solutions are commonly used to dissociate adherent cell monolayers during the process of harvesting cells. In this study, cells were harvested using trypsin-ethylenediaminetetraacetic acid (EDTA) (0.05% w/v trypsin with 0.53 mM EDTA.4Na) which had been derived from porcine pancreas. This solution was prepared in Hanks' Balanced Salt Solution (HBSS) without calcium and magnesium. EDTA promotes disaggregation of cells and is a chelator that binds divalent cations such as calcium and magnesium.

Cells were harvested at 70-80% confluence. Spent culture medium was aspirated from the flask, and cells were washed twice with 10 mL of calcium-magnesium free-phosphate buffered saline (pH 7.4) to remove residual medium containing foetal bovine serum, which is a rich source of trypsin inhibitors. Calcium and magnesium also inhibit trypsinisation, as these two cations play a critical role in cell-to-cell and cell-to-substrate attachment. Depending on the size of flask either 2 mL (for a 25 cm<sup>2</sup> flask) or 3 mL (for a 75 cm<sup>2</sup>) flask) of trypsin-EDTA was added, and incubated at 37°C in 5% CO<sub>2</sub> and 95% air. The progress of cell detachment was checked briefly under the inverted microscope every 5 minutes. When the cells were completely detached, trypsin was inactivated and EDTA removed by adding 10 mL of growth medium containing foetal bovine serum. The resulting cell suspension was collected and centrifuged at 500 x g for 5 minutes at room temperature. The cell pellet was re-suspended in 10 mL growth medium, and cell density and viability were determined using the trypan blue stain (Section 2.2.7). Cells were seeded at  $2-5 \times 10^4$  cells/cm<sup>2</sup> into a new flask. A total volume of 10 mL and 15 mL growth medium was used for 25  $\text{cm}^2$  and 75  $\text{cm}^2$  flasks respectively. The cell cultures were incubated at 37°C in an atmosphere of 5% CO<sub>2</sub> and 95% air.

#### 2.2.6 Freezing cell cultures

Freezing cells or cryopreservation is a technique where cells are stored at very low temperatures. Cells become metabolically inert and genetically stable and ice crystal formation is minimised. DMSO solution is the most commonly used cryoprotectant for the protection of cells from mechanical and physical stress, and reduction of water content within cells. The reagents are made into a solution with media or serum; cells are suspended in this solution, and are stored in liquid nitrogen at -196°C (Greene et al. 1967, Freshney 2005).

As indicated earlier, cells cultured following a high passage are more likely to change in cell characteristics (i.e. morphology, growth pattern and genetic stability) than cells derived from a low passage (Davis 2002). The procedure of freezing cells allows cells to be stored for long periods without alterations in viability or other characteristics (Freshney 2005). For this reason, the present study kept a stock of cells at a low passage level in liquid nitrogen with freezing medium. Cells that had been passaged between 5 and 15 times were used for experiments.

After trypsinisation (Section 2.2.5), the cell pellet was re-suspended in 10 mL growth medium, cell samples were counted and viability determined using trypan blue (Section 2.2.7). The suspended cells were centrifuged at 500 x g for 5 minutes at room temperature. The cell pellet was re-suspended in an appropriate volume of freezing medium containing 10% (v/v) DMSO and 90% (v/v) FBS (ECACC recommended) to achieve a cell density of  $1x10^6$  cells/mL and placed in cryovials. These cells were frozen by a controlled rate freezing device (isopropanol bath at -80°C) to slow the freezing process to ~ -1°C/minute. A slow cooling rate is important as it provides adequate time for proper adjustment of cells to the frozen environment and this limits cell death. Cells were placed in a -80°C freezer for 24 hours, prior to transfer to liquid nitrogen for storage.

#### 2.2.7 Cell viability determination by trypan blue stain

Trypan blue stain which is used to determine the viability of cells does not pass across the plasma membrane unless the cell is damaged or dead. Therefore, viable cells with intact cell membranes are unstained and non-viable cells are stained blue. Cells were counted using a haemocytometer and an inverted microscope (Freshney 2005).

After trypsinisation (Section 2.2.5), the cell pellet was re-suspended in 10 mL growth medium, and 20  $\mu$ L of the cell suspension was mixed with 20  $\mu$ L 0.4% (w/v) trypan blue stain in a sterile 1.5 mL microcentrifuge tube (giving a dilution factor of 2). This mixture was placed on a haemocytometer for cell counting. The hemacytometer has 2 chambers (Figure 2.4A) and there are 9 large squares with grid in each chamber (Figure 2.4B). Each outer quadrant is divided into 16 smaller squares and Figure 2.4B shows one chamber of a haemocytometer slide under 10x magnification. Only cells that overlap the top and left borders of squares should be counted to avoid overestimating the cell concentration (Davis 2002).

The four outer quadrants in the grid were used in both chambers to count the non-viable and viable cells at 10x magnification. The total count was divided by eight to determine the average number of cells in one quadrant, and this number was multiplied by 2 (the dilution factor) and then  $10^4$  in order to calculate the number of cells in one millilitre of suspension (the number  $10^4$  is derived from the total volume of each quadrant; 0.1 mm<sup>3</sup> or  $1 \times 10^{-4}$  mL). The percentage of viable cells was determined by dividing the number of unstained cells by the total number of cells and multiplying by 100.



**Figure 2.4:** Cell counting using a haemocytometer. (A) A haemocytometer slide. (B) Improved Neubauer pattern ruled haemocytometer (Davis 2002).

# 2.3 Purification of total RNA

Monolayer cells were washed twice with 10 mL sterile phosphate buffered saline. Total RNA was isolated from the cells using a Qiagen RNeasy Plus Mini kit according to the manufacturer's instructions as follows (Figure 2.5). In brief, lysis buffer which contained guanidine thiocyanate and 1% (v/v) beta-mercaptoethanol as a reducing reagent was added to the cells in order to disrupt the cell walls, and release the RNA contained in the sample. The lysed cells were manually scraped and transferred to a new 1.5 mL microcentrifuge tube. Cell lysate was passed though a 23-gauge needle attached to a sterile plastic syringe 20 times manually to reduce the viscosity of the cell lysates, and shear the high-molecular weight genomic DNA. The lysate was then transferred to a genomic DNA Eliminator spin column and centrifuged at 14,500 x g for 30 seconds (Eppendorf microcentrifuge 5415D, Germany). The column was discarded, and one volume of 70% (v/v) ethanol was added to

the flow-through and mixed well. The cell lysate was then placed in an RNeasy mini column and centrifuged at 14,500 x g for 15 seconds. The flow-through was discarded and this process repeated. An aliquot (700  $\mu$ L) of RW1 Buffer (provided in the Kit) was added to the RNeasy column and centrifuged at 14,500 x g for 15 seconds; the flow-through was then discarded. The RNeasy column was washed with 500  $\mu$ L RPE Buffer (provided in the Kit) and centrifuged at 14,500 x g for 15 seconds; again the flow-through was discarded. The RNeasy silica-gel membrane was dried by adding an additional 500  $\mu$ L of RPE Buffer prior to centrifugation at 14,500 x g for 2 minutes, the resulting flow-through was discarded and the tube recentrifuged once more at 25,000 x g for 1 minute. The RNeasy column was then placed on a new sterile 1.5 mL microcentrifuge collection tube. Total RNA was eluted from the column by the addition of 2 x 50  $\mu$ L RNase-free water, and the column was then centrifuged twice at 14,500 x g for 1 minute, using the same collection tube. All centrifugation steps were performed at room temperature. The RNA isolation procedures were performed in a Class II safety cabinet.



Figure 2.5: Flowchart of isolation of RNA using RNeasy Plus procedure.

The total RNA was precipitated by the addition of 0.1x volume of 3.0 M sodium acetate (pH 5.2) and 3x volume of ice cold ethanol. The RNA sample was subsequently mixed gently and stored at -20°C overnight. RNA was recovered by centrifugation at 15,000 x g (Heraeus Biofuge 28RS centrifuge, Germany) for 5 minutes at 4°C. The pellet was washed with 0.5 mL of 70% (v/v) ethanol and then centrifuged at 15,000 x g for 5 minutes at 4°C and this process repeated in order to remove residual salt. Evaporation of residual ethanol from the RNA pellet was performed by leaving the open tube on the lab bench at room temperature for 20 minutes. Care was taken not to overdry the RNA pellet, as this makes it more difficult to solubilise the pellet. The RNA pellet was re-suspended in 20  $\mu$ L 0.1% (v/v) diethyl pyrocarbonate (DEPC)-treated water (Appendix B) and stored at -80°C until required (Sambrook and Russell 2001). RNA samples solubilised in RNase-free water can be stored at -80°C without degradation for more than one year (Chomczynski 1992).

RNA concentration and purity were determined by spectrophotometry (Thermo Electron Spectrometry Biomate  $5^{TM}$ , UK). Ten µL of the total RNA was diluted with 490 µL of 0.1% (v/v) DEPC-treated water (a dilution factor of 50). The RNA concentration of a diluted sample was determined by absorbance at 260 nm (A<sub>260</sub>), with one absorbance unit of A<sub>260</sub> corresponding to 40 µg of RNA per mL. The A<sub>260</sub>/ A<sub>280</sub> ratio was calculated to give an indication of sample purity, with ranges between 1.8 and 2.1 considered acceptable for good quality RNA (Sambrook and Russell 2001).

Agarose gel electrophoresis is commonly used to judge the quality of the isolated total RNA sample by inspection of the 28S and 18S ribosomal RNA (rRNA) bands. A sample loading of 1  $\mu$ g RNA was verified visually by electrophoresis on a 1.0% (w/v) agarose gel dissolved in Tris-Borate-EDTA (TBE) buffer and containing a final concentration of 0.5  $\mu$ g/mL ethidium bromide. Electrophoresis was carried out for 60 minutes at 100 V in a Minicell EC370M (EC Apparatus Corporation, USA). With intact total RNA, the upper

Mw 1.9 kb) rRNA band, indicating non-degraded high quality RNA (Jones et al. 1994).

Additional information on agarose gel electrophoresis is provided in Section 2.8.

# 2.4 Reverse transcription and first strand cDNA synthesis

Since RNA does not serve as a template for DNA polymerase, a total RNA sample must firstly be reverse-transcribed into complementary DNA (cDNA) using a reverse transcriptase (RT) reaction. The subsequent cDNA can then be amplified by the polymerase chain reaction (PCR) method to detect the expression of target genes. The RT reaction consists of RNA template, reverse transcriptase (e.g. moloney murine leukemia virus; M-MLV or avian myeloblastosis virus; AMV), specific primers (e.g. oligo(dT)<sub>n</sub>, anchored oligo(dT)<sub>n</sub>, random hexamers, sequence-specific) and deoxynucleotide triphosphates (dATP, dCTP, dGTP and dTTP) (Figure 2.6).



**Figure 2.6:** Synthesis of the first strand cDNA. Abbreviations: cDNA; complementary deoxyribonucleic acid, A<sup>+</sup>; adenine dT; deoxythymidylic acid dATP; deoxyadenosine triphosphate, dCTP; deoxycytidine triphosphate, dGTP; deoxyguanosine triphosphate, dTTP; deoxythymidine triphosphate, mRNA; messenger ribonucleic acid, M-MLV; moloney murine leukemia virus; or AMV; avian myeloblastosis virus

M-MLV reverse transcriptase is most commonly used to synthesise full-length copies of

RNA molecules which can generate high copies of cDNA in samples containing low

amounts of RNA (Newton and Graham 1997). At the 3' end of mRNA, there is a long

sequence of adenine nucleotides known as 'the 3' poly (A) tail' which can be hybridised

In this study, 5  $\mu$ g of total RNA was added to a 5  $\mu$ M suspension of oligo (dT)<sub>13</sub> primers, and then made up to a reaction volume of  $12 \,\mu\text{L}$  with 0.1% (v/v) DEPC-treated water, gently mixed, and followed by a brief centrifuge at 4000 x g for 2 seconds to bring all components to bottom of tube. The reaction was incubated at 70°C for 10 minutes in the Biometra T3000 thermocycler (Thistle scientific Ltd., UK) to allow annealing of the oligo dT to the RNA template; the tubes were then placed on ice for 15 minutes to stop the reaction. Eight µL of RT reaction mixture consisted of 2 µL 10x RT reaction buffer, 1 µL 200 U M-MLV reverse transcriptase, 1 µL 10 mM dNTPs and 0.1% (v/v) DEPC-treated water to make up a total volume of 8  $\mu$ L. The RT reaction was prepared as presented in Table 2.1. The reaction mixture was incubated at  $42^{\circ}$ C for 50 minutes, prior to terminating the reaction at 95°C for 5 minutes and chilling at 4°C for 30 minutes in the thermocycler. The thermal profile for the first strand synthesis was set as shown in Table 2.2. Following the thermocycling reaction 80 µL 0.1% (v/v) DEPC-treated water was added to the samples. In order to standardise all subsequent procedures, the concentration of this is then represented as an equivalent cDNA concentration of 50 ng cDNA/µL which was stored at -80°C until further use.

Component	Concentration in reaction
Template RNA	5 µg
Oligo(dT) <sub>13</sub>	5 μΜ
10x RT reaction buffer	1x (containing 50 mM Tris-HCl, pH 8.3, with 50 mM KCl, 3 mM MgCl <sub>2</sub> , and 5 mM DTT)
dNTPs (10 mM)	0.5 mM each of dATP, dCTP, dGTP, dTTP
M-MLV RT	10 U
0.1% (v/v) DEPC-treated water	To make final volume 20 µL

**Table 2.1**: Components of first strand (RT) reaction.

**Table 2.2:** Thermocycling conditions for first strand (RT) cDNA synthesis.

Phase	Temperature	Duration
Oligo (dT) priming	70°C	10 minutes
Reverse transcription (RT)	42°C	50 minutes
RT inactivation	95°C	5 minutes

### 2.5 Qualitative reverse transcription-polymerase chain reaction

# (RT-PCR) analysis

Polymerase chain reaction (PCR), a technique originally invented by Kary Mullis in 1985, is an *in vitro* enzymatic reaction capable of amplifying the number of copies of a target DNA sequence. This method can be used for a variety of analyses in biomedical research (Newton and Graham 1997). Over two decades, PCR has been modified and developed for different applications. Reverse transcription-polymerase chain reaction (RT-PCR) is a highly sensitive technique and widely used in the study of gene expression at the RNA level, as a low copy number of RNA molecule can be detected. The first strand cDNA is synthesised (Section 2.4), followed by amplification of the specific parts of double-stranded DNA within cells using the PCR. PCR is carried out in a thin walled tube contained within a thermocycler using a specified programme that can control the precise temperature requirements for each step of the reaction.

The PCR requires basic components containing *Taq* DNA polymerase, reaction buffer (including magnesium chloride; MgCl<sub>2</sub>), dNTPs, primers, template DNA and water in a single tube. Determination of the optimal concentration of each component in a PCR reaction is important when DNA synthesis fidelity is critical. *Taq* DNA polymerase is a thermostable polymerase originally isolated from the thermophilic bacterium *Thermus aquaticus* and is the most widely used in PCR as it allows application of high annealing and extension temperatures during the amplification process. Under optimal conditions this enzyme can synthesise a new DNA strand at a rate of 35-100 nucleotides/second at 70-80°C and is stable during incubation at 95°C, allowing denaturation steps in the

amplification cycling (Newton and Graham 1997). The enzyme has exonuclease activity  $5' \rightarrow 3'$  but not on the  $3' \rightarrow 5'$  direction. The concentrations between 1.0 and 2.5 U of *Taq* DNA polymerase are usually used in 50 µL of PCR reaction mixture. An optimal concentration of *Taq* DNA polymerase is necessary to obtain a better yield of PCR products as a higher concentration of enzyme may result in non-specific products and lower amounts of *Taq* DNA polymerase may result in low yield of the desired product (Toranzos 1997).

The MgCl<sub>2</sub> concentration in the reaction is one of the most important factors for successful amplification of cDNA as it functions as a cofactor for DNA polymerase. However, Mg<sup>2+</sup> ions can interact with dNTPs, primers and DNA templates to form complexes so the MgCl<sub>2</sub> concentration has to be optimal for each experiment. Excessive Mg<sup>2+</sup> ions can lead to high levels of non-specific products, whereas inadequate Mg<sup>2+</sup> ions will inhibit the reaction, resulting in low yield of PCR products. The recommended range of MgCl<sub>2</sub> concentration in the final reaction mixture is between 0.5 and 5.0 mM for the standard reaction conditions (Newton and Graham 1997). All four dNTPs (i.e. dATP, dCTP, dGTP and dTTP) are required for the synthesis of DNA. Usually, the final concentration of each dNTP in the PCR reaction mixture is between 0.15 mM and 0.2 mM and these should be used at equal concentration to minimise misincorporation (Toranzos 1997).

The design of good synthetic oligonucleotide primers to anneal to target DNA is essential for the specificity of amplification product in order to have a successful PCR. Guidelines for primer selection in the current study are detailed in Section 2.6. In the PCR reaction mixture, the final concentration of each primer is in the range of 0.1-0.5  $\mu$ M and must be used in equal concentrations. Higher concentrations of primer usually generate nonspecific products and form primer-dimers (Toranzos 1997).

The working concentration of template DNA should be between 100 and 500 ng in a total PCR reaction volume of 50  $\mu$ L. Insufficient template DNA can lead to low yield of products while excessive template DNA usually may generate non-specific products and reduce efficiency (John and Stirling 2003).

Regarding the cycling conditions, a series of twenty-five to thirty-five cycles of PCR are usually carried out with each cycle of the PCR consisting of three steps (i.e. denaturation, annealing and extension of the DNA template, Figures 2.7A and 2.7B) (Newton and Graham 1997). During repeated cycles of PCR, the copy of new synthesised DNA strands increases exponentially and the target DNA replicates several million-fold. It is essential that the target DNA is analysed within the exponential phase of the PCR reaction, before the plateau phase when the reaction components are insufficient, leading to the low efficiency of amplification (Kainz 2000).

Initially, the double-stranded DNA is heated to 94-96°C in order to break apart the hydrogen bonds between the two DNA strands. The first denaturing step requires an extended time period to ensure that the template DNA is completely separated into single-strands. Denaturation for 0.5-2 minutes is usually sufficient. Following denaturation, the temperature is lowered to allow the primers to anneal to specific sequences on the single-stranded DNA in a step known as primer annealing. Identifying the appropriate annealing temperature ( $T_a$ ) for each primer pair is a critical factor in the PCR reaction and is based on the composition of the individual primer pairs (i.e. melting temperature;  $T_m$ ) in order to minimise non-specific amplification products.  $T_a$  should be set approximately 5°C below  $T_m$  (Newton and Graham 1997). Annealing of the oligonucleotides to the sequence of interest usually takes between 1 and 2 minutes. In the final elongation step the temperature is raised to the optimal temperature for the thermostable enzyme *Taq* DNA polymerase (70-80°C) as mentioned above.



**Figure 2.7:** *The polymerase chain reaction (PCR). (A) The PCR temperature cycling profile (B) The cycling reaction (Adapted from Newton and Graham 1997).* 

Housekeeping genes are constitutively expressed in all cells and can be used to ensure the efficiency of amplification reaction. Several housekeeping genes have been used as control genes in different model systems such as glyceraldehyde 3-phosphate dehydrogenase (GAPDH),  $\beta$ -actin and  $\beta_2$ -microglobulin (Schmittgen and Zakrajsek 2000, Lupberger et al. 2002).  $\beta_2$ -microglobulin is commonly used as an internal standard gene in gene expression studies as it shows low variation between different cell origins (Lupberger et al. 2002). This gene was used in the present study to verify the expression of target genes.

In this study, the standard PCR was performed in 50  $\mu$ L reaction mixtures. The PCR reaction mixture contained 5  $\mu$ L of 10x PCR reaction buffer, 1  $\mu$ L of dNTP mixture, 0.5  $\mu$ L of *Taq* DNA polymerase (5 U/ $\mu$ L), 5  $\mu$ L of the mixture of forward and reverse primers, template cDNA (50 ng/ $\mu$ L) and 0.1% (v/v) DEPC-treated water to make up a total volume of 50  $\mu$ L. A negative control (no template cDNA) was included in each PCR reaction. Initially, the standard PCR reaction was prepared as shown in Table 2.3. All components were mixed gently and followed by a brief centrifuge at 4000 x g for 2 seconds to collect them at the bottom of the microfuge tube. The reaction mixture was incubated in a

Biometra T3000 thermocycler (Germany) which was set as shown in Table 2.4.

Optimisation of the PCR procedure involved identification of the appropriate annealing temperature, the concentration of template cDNA and the appropriate primer sets in the PCR reaction as described in Chapter 3.

**Table 2.3:** Components of second strand (PCR) reaction.

Component	Concentration in reaction
10x reaction buffer	1x (containing 10 mM Tris-HCl, 1.5 mM, MgCl <sub>2</sub> , 50 mM KCl, pH 8.3)
Template cDNA	100 ng
Mixture of forward and reverse primers	$0.5 \mu M$ each of forward and reverse primers
dNTPs (10 mM)	0.2 mM each of dATP, dCTP, dGTP, dTTP
Taq DNA polymerase	2.5 U
0.1% (v/v) DEPC-treated water	To make final volume 50 µL

 Table 2.4: Thermocycling conditions for second strand (PCR) cDNA synthesis.

Phase	Temperature	Duration	Number of cycle
Initial denaturation	94°C	2 minutes	1
Template denaturation	94°C	50 seconds	
Primer annealing	$5^{\circ}$ C below $T_{m}$ of primers	1 minute	30
DNA synthesis	72°C	1 minute	
Elongation	72°C	5 minutes	1

# 2.6 Design of synthesis of oligonucleotide primers

A primer is a short synthetic oligonucleotide which is used to determine the DNA fragment to be amplified in applications of PCR or DNA sequencing. The primers are designed to be complementary to the target sequence of template DNA, which serves as a starting point for synthesising a new DNA strand or DNA replication by a DNA polymerase. Primer design is a critical issue in PCR-based detection methods as an inappropriate primer may produce internal secondary structures and primer-dimers. Internal secondary structure and primer-dimer formation occurs, if the forward and reverse primer can anneal to each other or have internal hybridisation. This reduces the available amount of primers, resulting in low effective amplification. Poor primer design may also cause non-specific products which appear as an incorrect product (Newton and Graham 1997, Lodge et al. 2007). Selection of efficient and specific primers remains empirical. There is no firm rule for the synthesis of effective or specific primer pairs for an amplification reaction but the primer should be designed according to some general guidelines. Usually, a suitable primer for the PCR process should be between 20 and 30 nucleotides in length with a melting temperature (T<sub>m</sub>) between 55°C and 65°C, and the PCR reaction works well if primer pairs have similar T<sub>m</sub> in order to set the same annealing temperature. Primers that are too short may lead to non-specific copies as they can anneal at many points on a DNA template. However, if primers are too long they are limited by high annealing temperature (Newton and Graham 1997). Currently, there are several free and commercially available webbased services or software provided for primer design, with Primer3 being the most widely used. The web-based application Primer3 programme is hosted by the Whitehead Institute for Biomedical Research (Rozen and Skaletsky 2000). Initially, primers in this study were designed to cross an exon-exon boundary in order to prevent co-amplification of genomic DNA. The designation was performed using design criteria as shown in Table 2.5.

<b>Table 2.5:</b> Criteria for primer picking conditions using the web-based Primer3 programme.	

Primer condition	Minimum	Optimum	Maximum
Primer size (bp)	18	20	27
Primer T <sub>m</sub> (°C)	57	60	63
Primer GC content (%)	20	50	80

Resulting primer sequences were then analysed for any cross-matching with the nonredundant database of GenBank using the standard nucleotide-nucleotide BLAST (The Basic Local Alignment Search Tool) search of the National Centre for Biotechnology Information (NCBI) available at <u>http://www.ncbi.nlm.nih.gov/blast/</u>. The Genbank accession number of each P450 and P450-NR genes used fro primer design is shown in Tables 2.6 and 2.7.Table 2.6 lists the primers for the candidate genes which were designed using programme Primer3. Table 2.7 shows the list of published primers which were considered for PCR experiments depending on particular objectives, and details will be provided in the subsequent chapters.

Gene name	Accession No	Primer	Sequence (5' to 3')	Length	$T_m (^{o}C)$	T <sub>a</sub> (°C)	Product size (bp)
$\beta_2$ -microglobulin	NM_004048	Forward	<sup>24</sup> CCTTGAGGCTATCCAGCGTACTCCAAA <sup>50</sup>	27	60.7	58	319
		Reverse	<sup>342</sup> TGATGCTGCTTACATGTCTC <sup>323</sup>	20	55.2		
CYP1A1	K03191	Forward	<sup>1019</sup> CTTGGACCTCTTTGGAGCTG <sup>1038</sup>	20	63.8	58	212
		Reverse	<sup>1230</sup> CGAAGGAAGAGTGTCGGAAG <sup>1211</sup>	20	63.8		
CYP1A2	Z00036	Forward	<sup>575</sup> CAATCAGGTGGTGGTGTCAG <sup>594</sup>	20	64.2	58	245
		Reverse	<sup>819</sup> GCTCCTGGACTGTTTTCTGC <sup>800</sup>	20	63.9		
CYP1B1	U03688	Forward	<sup>1132</sup> GCAGCTCAACCGCAACTTCA <sup>1151</sup>	20	68.8	60	397
		Reverse	<sup>1528</sup> AAAGCTGGAGAAGCGCATGG <sup>1509</sup>	20	68.6		
CYP2A6	X13930	Forward	<sup>1001</sup> TTGACAGAGTGATCGGCAAG <sup>1120</sup>	20	64.1	58	186
		Reverse	<sup>1186</sup> GGTACACTTCGGTGCCCTTA <sup>1167</sup>	20	63.7		
CYP2B6	M29874	Forward	<sup>2402</sup> CCCCATACCCCATTCTCTTT <sup>2421</sup>	20	63.6	58	194
		Reverse	<sup>2595</sup> GCAGAGGTTGCAGTGAACAA <sup>2576</sup>	20	64.1		
CYP2C8	Y00498	Forward	964TATGGACTCCTGCTCCTGCT983	20	63.9	58	230
		Reverse	<sup>1193</sup> CCCTTGGGGATGAGGTAGTT <sup>1174</sup>	20	63.9		
CYP2C9	M61855	Forward	905GGACAGAGACGACAAGCACA <sup>924</sup>	20	64.2	58	156
		Reverse	<sup>1060</sup> CATCTGTGTAGGGCATGTGG <sup>1041</sup>	20	64.1		
CYP2C18	M61856	Forward	<sup>1522</sup> CCGCATGGAGCTGTTTTTAT <sup>1541</sup>	20	63.7	58	199
		Reverse	<sup>1720</sup> GAGAATTGCAGGTGACAGCA <sup>1701</sup>	20	64.1		
CYP2C19	M61854	Forward	<sup>887</sup> ACTTGGAGCTGGGACAGAGA <sup>906</sup>	20	64.0	58	167
		Reverse	<sup>1053</sup> CATCTGTGTAGGGCATGTGG <sup>1034</sup>	20	64.1		

### Table 2.6(1): List of designed primers.

Gene name	Accession No	Primer	Sequence (5' to 3')	Length	$T_m (^{o}C)$	$T_{a}\left( ^{o}C\right)$	Product size (bp)
CYP2D6	X08006	Forward	<sup>830</sup> CAGAGATGGAGAAGGCCAAG <sup>849</sup>	20	63.8	58	191
		Reverse	<sup>1020</sup> CCCTATCACGTCGTCGATCT <sup>1001</sup>	20	64.0		
CYP2S1	AK027605	Forward	<sup>331</sup> AGGTCAGGCTGAGGAGTTCA <sup>350</sup>	20	64.0	58	216
		Reverse	<sup>546</sup> TCTGTCCCCTGGAATGTCTC <sup>527</sup>	20	64.1		
CYP3A4	J04449	Forward	666CAAGACCCCTTTGTGGAAAA685	20	63.7	58	184
		Reverse	<sup>849</sup> CGAGGCGACTTTCTTTCATC <sup>830</sup>	20	63.7		
CYP3A5	J04813	Forward	<sup>856</sup> AAGAAAAGTCGCCTCAACGA <sup>875</sup>	20	63.7	58	227
		Reverse	<sup>1082</sup> TGCAGTTTCTGCTGGACATC <sup>1063</sup>	20	64.1		
CYP3A7	D00408	Forward	<sup>776</sup> AAGGTCGCCTCAAAGAGACA <sup>795</sup>	20	63.9	58	223
		Reverse	998TGCACTTTCTGCTGGACATC979	20	64.1		
CYP3A43	AA417369	Forward	<sup>97</sup> AGGGAAAGTCAGGGTCCACT <sup>116</sup>	20	63.8	58	190
		Reverse	<sup>286</sup> GCATGAGGTTTGCTCTCACA <sup>267</sup>	20	64.1		
CYP4Z1	AY262056	Forward	<sup>575</sup> TCATGAAGTGTGCCTTCAGC <sup>594</sup>	20	64.1	58	159
		Reverse	<sup>733</sup> GGCCTTGAGAGCTGAATTTG <sup>714</sup>	20	63.7		
CYP24A1	L13286	Forward	<sup>1590</sup> GGCAACAGTTCTGGGTGAAT <sup>1609</sup>	20	63.9	58	249
		Reverse	<sup>1838</sup> TATTTGCGGACAATCCAACA <sup>1819</sup>	20	63.8		
CYP26A1	AF005418	Forward	<sup>1159</sup> TTCGGGTTGCTCTGAAGACT <sup>1178</sup>	20	63.9	58	191
		Reverse	<sup>1349</sup> TCCTCCAAATGGAATGAAGC <sup>1330</sup>	20	63.8		
CYP26B1	AF252297	Forward	<sup>383</sup> ACACGGTGTCCAATTCCATT <sup>402</sup>	20	64.0	58	172
		Reverse	<sup>554</sup> GCCTCCTGGTACACGTTGAT <sup>535</sup>	20	63.9		

### Table 2.6(2): List of designed primers.

Gene name	Accession No	Primer	Sequence (5' to 3')	Length	$T_m (^{o}C)$	$T_{a}\left( ^{o}C\right)$	Product size (bp)
HIF-1a	U29165	Forward	<sup>2035</sup> GAAAGCGCAAGTCCTCAAAG <sup>2054</sup>	20	63.8	58	167
		Reverse	<sup>2201</sup> TGGGTAGGAGATGGAGATGC <sup>2182</sup>	20	64.0		
AhR	NM_001621	Forward	<sup>1117</sup> GGACAGAAAAAGAAAGGGAAAG <sup>1138</sup>	22	61.8	55	262
		Reverse	<sup>1378</sup> CGGCACAATAAAGCATATCAG <sup>1358</sup>	21	61.9		
ARNT	NM_178427	Forward	<sup>2075</sup> ATCCTCCTTCAGCTCCATGTCC <sup>2096</sup>	22	67.6	60	387
		Reverse	<sup>2461</sup> TTTCTCCCCCACCCCTTATC <sup>2442</sup>	20	65.9		
AhRR	BM682431	Forward	<sup>120</sup> TTCAGGATAGGCTCCTGCAT <sup>139</sup>	20	63.6	58	171
		Reverse	<sup>290</sup> AGTGACCACCATCCAAGCTC <sup>271</sup>	20	64.1		
CAR*	Z30425	Forward	<sup>861</sup> GCAGCTGTGGAAATCTGTCA <sup>880</sup>	20	64.1	58	232
		Reverse	<sup>1092</sup> CAGGTCGGTCAGGAGAGAAG <sup>1073</sup>	20	64.0		
PXR	AJ009936	Forward	<sup>3958</sup> CTGTGTGGATGCTGAGCTGT <sup>3977</sup>	20	64.3	58	214
		Reverse	<sup>4171</sup> TGATTGTCAGCGTAGCCTTG <sup>4152</sup>	20	63.9		
VDR	J03258	Forward	<sup>1597</sup> GAGACCTCAGCCATGAGGAG <sup>1616</sup>	20	64.0	58	250
		Reverse	<sup>1846</sup> CGTGAGTAAGGCAGGAGAGG <sup>1827</sup>	20	63.8		
ER	X03635	Forward	<sup>1625</sup> TGGAGATCTTCGACATGCTG <sup>1644</sup>	20	64.1	58	208
		Reverse	<sup>1832</sup> GCCATCAGGTGGATCAAAGT <sup>1813</sup>	20	63.9		

 Table 2.6(3): List of designed primers.

Note: \*; This primer set was also used as nested primer for the published CAR.

Name	Accession No	Primer	Sequence (5' to 3')	Length	$T_m (^{\circ}C)$	$T_a (^{o}C)$	Product size (bp)	Reference
CYP1A1	K03091	Forward	967TGGATGAGAACGCCAATGTC986	20	66.0	58	394	Huang et al. 1996
		Reverse	<sup>1358</sup> TGGGTTGACCCATAGCTTCT <sup>1339</sup>	20	63.3			
CYP2D6	M20403	Forward	<sup>1033</sup> CCAGAGATGGGTGACCAG <sup>1050</sup>	18	61.8	55	261	Thomas et al. 2006
		Reverse	<sup>1293</sup> CTCCGGCTTCACAAAGTG <sup>1276</sup>	18	62.1			
CYP2F1	NM_000774	Forward	<sup>646</sup> CCATTATCCGCCTTATCAAT <sup>665</sup>	20	60.1	55	305	Thomas et al. 2006
		Reverse	<sup>950</sup> GCCAAAGAGCAGGTTATGT <sup>932</sup>	19	59.3			
CYP2J2	NM_000775	Forward	<sup>466</sup> GCTTAGAGGAACGCATTCA <sup>484</sup>	19	60.5	55	714	Thomas et al. 2006
		Reverse	<sup>1179</sup> CAGCCAAAGTGGTATCAACT <sup>1160</sup>	20	59.2			
CYP2R1	NM_024514	Forward	<sup>260</sup> TGGTTCTAAATGGCTATGATG <sup>280</sup>	21	59.0	52	308	Thomas et al. 2006
		Reverse	<sup>567</sup> GTTTGAAACAGCATTCGTTA <sup>548</sup>	20	57.8			
CYP2U1	NM_183075	Forward	<sup>545</sup> GCTCATCTCCATCGTGAC <sup>562</sup>	18	59.2	55	320	Thomas et al. 2006
		Reverse	<sup>864</sup> TTTCTAGGCCTCGTGACATA <sup>845</sup>	20	59.8			
CYP3A4	M14096	Forward	<sup>782</sup> CTGTGTGTTTCCAAGAGAAGTTAC <sup>805</sup>	24	60.6	55	742	Huang et al. 1996
		Reverse	<sup>1523</sup> TGGTTGAAGAAGTCCTCCTAAG <sup>1502</sup>	22	61.2			
CYP3A5	J04813	Forward	<sup>799</sup> GTCTCTCTGTTTCCAAAAGATACC <sup>822</sup>	24	60.9	55	737	Huang et al. 1996
		Reverse	<sup>1536</sup> TGAAGAAGTCCTTGCGTGTC <sup>1516</sup>	20	63.0			
CYP4X1	NM_178033	Forward	<sup>655</sup> GCATCGTCGCCTACTAACTC <sup>674</sup>	20	61.4	55	364	Thomas et al. 2006
		Reverse	<sup>1018</sup> CTGATTCAACACTCGGCTTA <sup>999</sup>	20	60.4			
CYP4Z1	NM_178134	Forward	<sup>1147</sup> ATATCCCGGTTACTCGACAA <sup>1166</sup>	20	60.7	55	257	Thomas et al. 2006
		Reverse	<sup>1403</sup> GCCACTGCCACTTTACACT <sup>1385</sup>	19	60.3			

### Table 2.7(1): List of published primers.

Table 2.7(2):	List of	published	primers.
---------------	---------	-----------	----------

Name	Accession No	Primer	Sequence (5' to 3')	Length	$T_m (^{o}C)$	$T_a (^{o}C)$	Product	Reference
							size (bp)	
CAR	Z30425	Forward	<sup>684</sup> CCAGCTCATCTGTTCATCCA <sup>703</sup>	20	63.9	58	417	Chang et al. 2003
		Reverse	<sup>1110</sup> GGTAACTCCAGGTCGGTCAG <sup>1081</sup>	20	63.5			
PXR	AF061056	Forward	<sup>678</sup> CAAGCGGAAGAAAAGTGAACG <sup>698</sup>	21	65.7	60	246	Chang et al. 2003
		Reverse	923CACAGATCTTTCCGGACCTG904	20	64.6			
PGR	M15716	Forward	<sup>1817</sup> GATTCAGAAGCCAGCCAGAG <sup>1836</sup>	20	64.0	58	533	Lau et al. 1999
		Reverse	<sup>2349</sup> TGCCTCTCGCCTAGTTGATT <sup>2330</sup>	20	63.7			
GCR	X03225	Forward	<sup>2038</sup> CTACCCTGCATGTACGACC <sup>2056</sup>	19	60.6	55	360	Gupta and Wagner 2003
		Reverse	<sup>2396</sup> TCAGCTAACATCTCGGGG <sup>2378</sup>	18	61.4			
PPARα	L02932	Forward	913GCCAGTAACAATCCACCTTTT933	21	61.7	55	735	Saidi et al. 2006
		Reverse	<sup>1647</sup> AAGGTGTGGCTGATCTGAAGG <sup>1627</sup>	21	65.4			
RXR	NM_002957	Forward	<sup>490</sup> GCTCCTCAGGCAAGCACTATGGAGTGTA <sup>517</sup>	28	73.1	68	521	Wang and Yen 2004
		Reverse	<sup>1010</sup> GGAGAAGGAGGCGATGAGCAGCTCATT <sup>984</sup>	27	75.9			
RAR	NM_000964	Forward	908GGGCATGTCCAAGGAGTCTGTGAGAA933	26	73.8	68	756	Wang and Yen 2004
		Reverse	<sup>1663</sup> GGGATCTCCATCTTCAGCGTGATCA <sup>1638</sup>	25	73.0			

Oligonucleotide primers were synthesised by Sigma-Genosys Ltd. Lyophilised primers were diluted to a stock concentration of 100  $\mu$ M by adding 0.1% (v/v) DEPC-treated water according to the manufacturers supplied OD readings, then adjusting this to a working concentration of 5  $\mu$ M.

# 2.7 Nested PCR

Nested PCR is used to enhance sensitivity and specificity of amplification as it can eliminate the occurrence of non-specific products generated from unexpected primer binding sites. Nested PCR technique is performed with two sets of primers one internal to the other. The primary amplification product is used as template in the nested PCR reaction with another primer set (Figure 2.8). The components for this reaction are listed in Table 2.8.



Figure 2.8: Illustration of nested PCR.

 Table 2.8: Components of the nested PCR.

Components	Concentration in reaction
10x reaction buffer	1x (1.5 mM MgCl <sub>2</sub> )
RT-PCR product from the primary amplification used as a template	5 μL
Mixture of forward and reverse nested primers	$0.5\ \mu M$ each of forward and reverse nested primers
dNTPs (10 mM)	0.2 mM each of dATP, dCTP, dGTP, dTTP
Taq DNA polymerase	0.025 U
0.1% (v/v) DEPC-treated water	To make final volume 50 µL

### 2.8 Agarose gel electrophoresis of DNA

Electrophoresis is a technique used to separate DNA, RNA or protein molecule by size. Nucleic acids (DNA and RNA) have a negative charge and when an electric current is applied they move through the agarose matrix migrating towards the positive electrode. Larger DNA fragments move at a slower rate than smaller fragments. Ethidium bromide is a fluorescent dye that intercalates between bases of nucleic acids. On exposure to UV light, ethidium bromide transmits the energy as visible orange light and allows detection of DNA fragments. A final concentration of  $0.5 \,\mu$ g/mL ethidium bromide is usually used and this can be added into the gel either before or after electrophoresis. The gel is prepared by dissolving the agarose in an appropriate volume of buffer following heating in a microwave oven. The dissolved agarose solution is then placed into a horizontal gel manifold and allowed to cool. Once the gel is set (approximately 30 minutes to 1 hour) it is then placed into the electrophoresis tank. The same buffer used to prepare the gel is then used to fill the tank. Samples (PCR products) are prepared as follows. Gel loading buffer must always be added to the samples prior to loading to increase the density of the samples and ensure that they sink into the well, and to add colour to the samples so they can be easily seen when loading. The gel is generally run at 1-10 V/cm until the dyes have migrated an appropriate distance which depends on the size of DNA to be analysed. Each gel electrophoresis must include at least one lane of molecular weight markers to enable analysis of DNA fragment size in the samples. The most commonly used buffers for DNA are TAE (Tris-acetate-EDTA) and TBE (Tris-borate-EDTA). TBE buffer is recommended for analysing fragments less than 1 kb as it gives better resolution and sharper bands. The concentration of agarose used is dependent on the size of product being determined and is varied accordingly for different fragment ranges (Table 2.9) in order to achieve good separation (Martin 1996, Sambrook and Russell 2001).

Agarose concentration [% (w/v)]	DNA fragment ranges (kb)
0.3	5-60
0.6	1-20
0.7	0.8-10
0.9	0.5-7
1.2	0.4-6
1.5	0.2-3
2.0	0.1-2

**Table 2.9:** Agarose concentration used for separating different DNA fragment sizes (Sambrook and Russell 2001).

Materials and Methods

For analysing the PCR product in this study, a 2.5  $\mu$ L aliquot of 6x loading dye (10mM Tris-HCl (pH 7.6), 0.03% bromophenol blue, 0.03% xylene cyanol FF and 60% glycerol, 60mM EDTA) was added to 10  $\mu$ L of the PCR product and mixed well. A 1  $\mu$ L aliquot of 6x loading dye was added to 5  $\mu$ L of PCR markers (1000, 750, 500, 300, 150 and 50 bp) and this mixture was loaded beside the samples as a reference for relative qualification of PCR products. The PCR marker and sample mixture were separated by gel electrophoresis using a 1.5% (w/v) agarose gel dissolved in TBE buffer (89 mM Tris-HCl pH 7.8, 89 mM Boric acid and 2 mM EDTA) containing ethidium bromide. Electrophoresis was carried out for 70 minutes at 150 V in a Sub-Cell GT system (BioRad, USA) at room temperature. The UV visualised photographs of gels were taken using a Polaroid camera (Hoya, Japan) in the roof of the transilluminator (UVI tech, Cambridge, England). To photograph a gel, the aperture (F-stop) at 5.6 was set with an exposure time (shutter speed) of 2 seconds.

Following visualisation of a band in a PCR product, the size of DNA fragments was determined using semi-logarithmic plotting to estimate molecular weight. To predict an unknown fragment, a logarithmic scale on the vertical (y axis) was plotted for all the molecular weight marker sizes in base pairs against the distance each of those bands had migrated from the wells in millimetres (horizontal; x-axis). The log-linear relationship between migration on the gel and fragment size of the markers provided a standard curve to determine the size of unknown fragments by measuring distance migrated. Figure 2.9

shows how DNA fragment size was estimated by semi-logarithm plotting using an example from an experiment in the current study. The semi-logarithm plot of distance migration based on the DNA markers used allows the size of an unknown fragment that migrated 43 mm to be estimated as approximately 275 bp in size.



**Figure 2.9:** *Example of semi-logarithm plot of distance migrate the DNA markers used (scanned from one of the results in this study).* 

### 2.9 Restriction digestion

Restriction digestion is the method of cutting double-stranded DNA molecules at specific sequences (known as recognition sites) into smaller fragments using restriction enzymes. This process is useful in molecular biology applications as it provides high sequence specificity and precision cutting. Figure 2.10 shows an example of the DNA nucleotide sequences recognised by the restriction enzyme known as *Eco*RI.



**Figure 2.10:** *Recognition sequence of the restriction endonuclease EcoRI (Adapted from Albert et al 2002).* 

The expression of an individual gene can be identified by restriction digestion. In this study, an appropriate restriction enzyme for each individual (presence) candidate gene was
selected and information on these enzymes is presented in Table 2.10. Table 2.11 highlights the conditions used for restriction digests in this research. All components were mixed gently by pipetting and followed by a brief centrifuge at 4000 x g for 2 seconds to collect them at the bottom of the tube. This mixture was kept on ice prior to the reaction. Restriction digests were performed in a thermocycler at 37°C for 1 hour to complete digestion, and then incubated at 70°C for 10 minutes to inactivate the restriction enzyme. The digested and undigested products were further analysed by 2% (w/v) agarose gel electrophoresis to separate the fragments by size or alternatively stored at -20°C until required.

# 2.10 Purification of DNA and DNA sequencing

DNA sequencing is used to determine nucleotide sequences of a specific DNA molecule in order to identify the gene product and predict its physiological functions. This requires the purification of DNA fragments from agarose gels or amplification reactions. Purification of DNA from low-melting point (LMP) agarose gel is commonly used as it is a quick procedure and produces high purity DNA. LMP agarose gel can be melted at 65-75°C and this will not dissociate the double-stranded nature of duplex DNA, whereas a standard agarose gel can only be melted at 80-90°C and this will disrupt duplex DNA molecules (Martin 1996).

A high Pure PCR product purification kit (Roche Diagnostics Ltd., Lewes, UK) was used for isolation of DNA, according to the manufacturer's instructions (Figure 2.11). In brief, PCR product was loaded on a 1.8% (w/v) low melting point (LMP) agarose gel containing a final concentration of 0.5  $\mu$ g/mL ethidium bromide and TAE (40 mM Tris acetate, 2 mM EDTA; pH 8.5) was used as running buffer. Electrophoresis was carried out for 60 minutes in a Minicell EC370M (EC Apparatus Corporation) at 100 V. To handle LMP

 Table 2.10: Recognition sequences for the restriction endonucleases used.

Name	<b>Restriction enzyme</b>	Recognition site	Size of fragment (bp)	Buffer**
$\beta_2$ -microglobulin	EcoRI	G/AATTC	215, 104	SH
CYP1B1	RsaI	GT/AC	174, 223	SL
CYP2A6	HaeIII	GG/CC	48, 72, 66	SM
CYP2B6	HaeIII	GG/CC	30, 164	SM
CYP2C8	MspI	C/CGG	170, 60	SL
CYP2C18	RsaI	GT/AC	125, 74	SL
CYP2S1	AluI	AG/CT	103, 63, 50	SA
CYP3A7	EcoRI	G/AATTC	65, 158	SH
CYP4Z1	RsaI	GT/AC	47, 112	SL
CYP24A1	BamHI	G/GATCC	68, 181	SB
CYP26A1	AluI	AG/CT	172, 19	SA
CYP26B1	AluI	AG/CT	104, 68	SA
HIF-1a	SspI	AAT/ATT	140, 27	SH
AhR	MspI	C/CGG	99, 163	SL
ARNT	PstI	CTGCA/G	118, 269	SH
AhRR	HaeIII	GG/CC	92, 79	SM
VDR	PstI	CTGCA/G	198, 52	SH
ER	RsaI	GT/AC	111, 97	SL
Nested CAR	PstI	CTGCA/G	170, 62	SH
CYP1A1*	HaeIII	GG/CC	188, 40, 164	SM
CYP2D6*	RsaI	GT/AC	119, 142	SL
CYP2F1*	MspI	C/CGG	195, 110	SL
CYP2J2*	AluI	AG/CT	331,170,159,54	SA
CYP2R1*	HaeIII	GG/CC	130, 118, 60	SM
CYP2U1*	HaeIII	GG/CC	252, 60, 8	SM
CYP4X1*	RsaI	GT/AC	278, 86	SL
CYP4Z1*	AluI	AG/CT	196, 61	SA
PXR*	MspI	C/CGG	142, 90, 14	SL
PGR*	AluI	AG/CT	32, 84, 417	SA
GCR*	AluI	AG/CT	148, 36, 172, 4	SA
PPARα*	RsaI	GT/AC	518, 188, 29	SL
RXR*	BamHI	G/GATCC	280,153,88	SB
RAR*	HaeIII	GG/CC	456, 193, 107	SM

**Note:** \*; Previously published primers, \*\* ; The digestion buffer was varied according to the enzyme of choice and was supplied along with the restriction enzyme from supplier. Composition of the 1x buffer is shown in Appendix C.

Stock solution	Undigested sample (µL)	Digested sample (µL)	
Re-amplified RT-PCR product	2	2	
10x digestion buffer	2	2	
Restriction enzyme (10 U/ $\mu$ L)	0	1	
0.1% (v/v) DEPC-treated water	16	15	

**Table 2.11:** Components of the restriction digest.

agarose gel, these procedures were performed on ice due to the fragile nature of gels at

higher temperatures (Martin 1996).



**Figure 2.11:** Flowchart of purification of DNA using Roche High Pure PCR product Purification procedure.

The desired DNA band was excised from the gel using a sterile scalpel blade and the DNA gel slice dissected into small fragments under UV light. A 100 mg portion of the agarose gel slice was then placed into a pre-weighed 1.5 mL microcentrifuge tube containing 300  $\mu$ L Binding Buffer provided by the manufacturer (3 M guanidine-thiocyanate, 10 mM Tris-HCl, 5% ethanol (v/v), pH 6.6) and vortexed for 30 seconds. To release the DNA, the agarose gel slice was dissolved in the binding buffer, incubated at 65°C for 10 minutes and vortexed for 15 seconds every 2 minutes during incubation. Isopropanol (150  $\mu$ L) was added to the dissolved gel and vortexed for 15 seconds prior to transfer to a High Pure

Filter Tube. The tube was centrifuged at 25,000 x g for 1 minute at room temperature and the flow-through was discarded. A 500  $\mu$ L aliquot of Washing Buffer (20 mM NaCl, 2 mM Tris-HCl, pH 7.5) was added to the filter tube and centrifuged at 25,000 x g for 1 minute with flow-through again being discarded. To ensure optimal purity of the sample, an additional wash with 200  $\mu$ L of Wash buffer provided by the manufacturer centrifuged at 25,000 x g for 1 minute was carried out. The filter tube was then placed into a new sterile 1.5 mL microcentrifuge tube and a 50  $\mu$ L aliquot of Elution Buffer provided by the manufacturer (10 mM Tris-HCl, pH 8.5) was added to the tube and centrifuged at 25,000 x g for 1 minute.

The purified DNA was precipitated by the addition of 0.1x volume of 3 M sodium acetate (pH 5.2) and 2x volume of ice cold ethanol to the DNA sample which was gently mixed, before storing at -20°C overnight. The DNA was recovered by centrifugation at 15,000 x g (Heraeus Biofuge 28RS centrifuge, Germany) for 5 minutes at 4°C. The pellet was washed with 0.5 mL of 70% (v/v) ethanol and then centrifuged twice at 15,000 x g for 5 minutes at 4°C to remove residual salt. Evaporation of residual ethanol from the DNA pellet was performed by leaving the open tube on the lab bench for 20 minutes. The DNA pellet was re-suspended in 20  $\mu$ L 0.1% (v/v) DEPC-treated water and stored at -80°C until required.

DNA amount and purity was determined by spectrophotometry (Thermo Electron Spectrometry Biomate  $5^{\text{TM}}$ , UK). Ten microlitres of the purified DNA was diluted with 490 µL of 0.1% (v/v) DEPC-treated water. Absorbance at 260 nm was measured to determine DNA concentration, with each unit of A<sub>260</sub> corresponding to 50 µg/mL double stranded DNA. The A<sub>260</sub> /A<sub>280</sub> ratio provided an estimate of the DNA purity. Pure preparations of DNA should have a value of 1.8 and 2.1. All purified DNA of interest was sent to the Rowett Research Institute (Bucksburn, Aberdeen UK) to identify the sequences of DNA. The method used was automated dideoxy DNA sequencing. The results of Materials and Methods sequencing were analysed using the standard nucleotide-nucleotide BLAST search and comparing with the non-redundant database of GenBank.

# 2.11 Summary

This chapter encompasses the methodological approach used in this thesis. Section 2.2 covers cell biology techniques, while molecular biology techniques are described in Sections 2.3 to 2.10. In Chapters 3, 4, 5 and 6, the techniques are described in more detailed terms and in relation to the particular objectives of the thesis. Chapter 3 describes validation of RT-PCR analysis. In Chapter 4, the conditions for RT-PCR analysis were optimised for MCF7 and MDA-MB-468 cell lines and these conditions were used to profile P450s in a panel of breast cancer cell lines compared to a normal breast cell line. Chapter 5 is concerned with characterisation of CYP4Z1 in PCR products, while Chapter 6 identifies the receptors that regulate the expression of the CYP4Z1 gene and evaluates whether the transcription or translation process regulates CYP4Z1 expression.

# Chapter 3

# Validation of RT-PCR analysis in breast cancer cell lines

# 3.1 Introduction

RT-PCR is a highly sensitive method so the risk of contamination and false-positive results is of concern. To reduce the risk of RT-PCR contamination, the preparation of total RNA including pre-PCR work was performed in a Class II safety cabinet and away from the PCR area which was set up in a separate designated laboratory. The PCR workstation was always cleaned with 70% (v/v) ethanol before use. Pipette tips and pipettors used in the amplification procedures were pre-treated with UV crosslinker for 15 minutes, and water used for the PCR reactions was treated with diethylpyrocar-bonate (DEPC) prior to use in order to inactivate RNase (Appendix B, Sambrook and Russell 2001). A set of dedicated pipettes was used for each separate laboratory procedure (i.e. pre-PCR, PCR and post-PCR work) and storage of all PCR reagents isolated from samples.

To detect DNA carryover contamination and false-positive products, the use of negative controls (no DNA template) is important and these must be routinely used when conducting PCR. No product formation should be seen (Hamadeh and Afshari 2004). In addition, a known housekeeping gene such as  $\beta_2$ -microglobulin which is expressed in most cells should be employed to verify the expression of DNA template (Section 2.5).

The validation process identifies aspects of a procedure that are critical and must be carefully controlled. To validate an RT-PCR assay, a demonstration of the accuracy, reliability and reproducibility of PCR-based technologies is required and realisation of the limitations of the method. As detailed in Sections 2.4 and 2.5, there are a number of considerations involved in amplifying a specific region of cDNA such as the quality of

RNA sample and the PCR reaction conditions which are detailed in the following sections. Quality of total RNA obtained is essential for the successful amplification of the gene of interest and this can be checked on a gel by examination of the 28S and 18S rRNA bands (Section 2.3).

As described in Section 2.8, agarose gel electrophoresis of DNA is most commonly used to determine the size of products obtained from PCR reaction. The appearance of desired DNA fragments under UV light is vital for identification of the PCR products which can be further confirmed by restriction digestion or purified for various purposes such as DNA sequencing. In the current study, both restriction digestion and DNA sequencing were used to identify the sequence specificity of PCR products. However, if PCR products appear as multiple fragments or non-specific products, nested PCR can be used to improve specificity of the PCR reaction.

The main aim of this chapter was to optimise the conditions for RT-PCR analysis for the target cytochrome P450 and P450-regulartory nuclear receptor genes using MCF7 and MDA-MB-468 cells as representative oestrogen receptor (ER) protein positive and negative breast cancer cell lines respectively.

# **3.2** Materials and Methods

## 3.2.1 RNA isolation

As discussed in Section 2.3, total RNA was isolated with the Qiagen RNeasy Plus Mini kit. The absorbance at 260 nm ( $A_{260}$ ) of a diluted RNA sample was used to assess RNA concentration and the  $A_{260}/A_{280}$  ratio was employed to determine RNA purity. Quality of the purified RNA was determined by 1% (w/v) agarose gel electrophoresis. Total RNA was precipitated by ethanol and sodium acetate to ensure stable and concentrated storage of the starting material. The resulting pellet was washed and re-centrifuged prior to storage or use.

# 3.2.2 Reverse transcription (RT) and cDNA synthesis

As described in Section 2.4, first strand cDNA was synthesised from 5  $\mu$ g of purified RNA samples using M-MLV reverse transcriptase. The components of the RT-reaction are summarised in Table 2.1. All reagents were thawed on ice prior to centrifugation. The reverse transcriptase assay was initiated by addition of M-MLV reverse transcriptase. The reaction conditions were as previously described in Table 2.2. Following first strand synthesis the RT product was placed on ice to inactivate the reaction before it was diluted to a concentration of 50 ng of template cDNA per mL and stored at -80°C until use.

## 3.2.3 Optimisation of PCR analysis

## **3.2.3.1 Initial PCR amplification**

Optimisation of each component (i.e. concentration of primer, amount of template cDNA, dNTPs, *Taq* DNA polymerase, MgCl<sub>2</sub> and thermocycling parameters) including thermocycling conditions in a PCR reaction is important to obtain a better yield of amplification products (Section 2.5). Using cDNA template derived from the MCF7 and MDA-MB-468 cells as representative of the test cell lines, it was possible to detect the presence/absence and optimise the RT-PCR conditions for the determination of the PCR products of 19 cytochrome P450 and 8 P450-regulatory nuclear receptor genes (Table 2.6). The routine amplification was developed as shown in Tables 2.3 and 2.4. In each PCR reaction in this investigation, negative controls and  $\beta_2$ -microglobulin were always included. Where a visible PCR product is evident, this is subsequently referred to as mRNA by extrapolation from the method described.

### 3.2.3.2 Optimisation of template cDNA and primer concentrations

In this research, when the routine amplification resulted in no product formation the concentration of template cDNA was first varied (i.e. 150, 200 and 250 ng) and the concentration of primers was kept constant at 0.5  $\mu$ M (Figure 3.1, stage 2A). If PCR product was still not seen, a high range of primer concentrations (i.e. 1.0 and 1.5  $\mu$ M) was then used (Figure 3.1, stage 2B). A low range of primer concentrations (i.e. 0.1, 0.2, 0.3 and 0.4  $\mu$ M) was employed when the presence of a desired band as well as non-specific products were detected (Figure 3.1, stage 2C). Optimisation of primer concentration, both low and high ranges, was performed with a fixed quantity of 100 ng template DNA. In addition, the recommended concentrations of *Taq* DNA polymerase, MgCl<sub>2</sub> and dNTPs were used for the RT-PCR assay (Table 2.3).

# 3.2.3.3 Oligonucleotide primer design

All designed primers (Table 2.6) were optimised using the procedures described in Sections 3.2.3.1 and 3.2.3.2. However, there were twelve primer sets used that failed to detect mRNA expression in the MCF7 and MDA-MB-468 cells. This may be because mRNA expression was below the limit of detection, and details will be provided in the result section. Five out of the twelve undetectable genes (i.e. CYP1A1, CYP3A4, CYP3A5, CAR and PXR) were considered for PCR experiments using previously published primer sequences (Table 2.7) as previous studies have provided evidence of these genes in breast cancer research. These primers were checked for specificity by BLAST analysis.



Figure 3.1: General guidelines used for optimisation of template cDNA and primer concentration.

#### 3.2.4 Nested PCR

As mentioned in Section 2.7, nested PCR is a simple technique which can be employed to improve specificity of multiple fragment products using the first PCR product as a template for the second PCR reaction. In this research, design of the internal primer sequences for the CAR used in the nested PCR was performed by importing the known sequence of the primary PCR product (417 bp in size) into the Primer3 programme. BLAST analysis confirmed the specificity of the primer used. Information on these primers is shown in Table 2.6 and the nested PCR reaction was performed under the conditions described in Table 2.8.

#### 3.2.5 Restriction enzyme digestion of DNA

As previously described (Section 2.9), digestion of PCR products was carried out with specific restriction enzymes for each individual PCR product (Table 2.10). All restriction digests were performed in the same manner as the example presented in Table 2.11.

#### **3.2.6 DNA purification and sequencing**

The desired DNA fragment was purified with the High Pure PCR product purification kit as detailed in Section 2.10. DNA concentration and purity were assessed by spectrophotometry and the acceptable  $A_{260}/A_{280}$  ratio was between 1.8 and 2.1. The purified DNA was sequenced by the method of automated dideoxy DNA sequencing.

# 3.3 Results

#### 3.3.1 RNA isolation

Initial purity of total RNA in this study was poor with a calculated RNA  $A_{260}/A_{280}$  ratio of between 1.3 and 1.6. On investigation, the number of lysed cells loaded onto the RNeasy spin column was shown to play a major role in determining the purity of RNA collected. The number of cells loaded onto individual columns was reduced from >  $2x10^6$  cells per purification to an optimal loading of between  $5 \times 10^5$  and  $2 \times 10^6$  cells. Additional measures were also employed to increase purity and recovery as explained in Section 2.3. Following these measures the recovery of total RNA was between 25 and 60 µg per purification. In addition, the ratio of RNA obtained ( $A_{260}/A_{280}$ ) was between 1.8 and 2.1. The quality of the purified RNA was confirmed by inspecting the rRNA bands on a 1% (w/v) agarose gel containing ethidium bromide. No smear or band below the lower 18S rRNA band, a common indication of degraded low-quality RNA, was found in any sample (Figure 3.2).



**Figure 3.2:** *Quality of RNA isolated from four independent RNA preparations. Lanes 1-2: RNA isolated from the MCF, Lanes 3-4: RNA isolated from the MDA-MB-468.* 

# 3.3.2 β<sub>2</sub>-microglobulin RT-PCR

 $\beta_2$ -microglobulin was identified as a band of 319 bp corresponding to the  $\beta_2$ -microglobulin cDNA product size. The intensity of  $\beta_2$ -microglobulin amplification product increased between 15 and 30 cycles. The optimal cycle number for  $\beta_2$ -microglobulin amplification was identified as 20 cycles because under these conditions the product appeared as a sharp band (Figure 3.3). This cycle number was used for all  $\beta_2$ -microglobulin RT-PCR experiments.

Little variation was observed between  $\beta_2$ -microglobulin mRNA in the MCF7 and MDA-MB-468 cells. Figure 3.4 highlights the level of reproducibility found in  $\beta_2$ -microglobulin expression in replicate studies of this research. Restriction digestion using *Eco*RI

confirmed the specificity of  $\beta_2$ -microglobulin with a total digest of the PCR product

resulting in fragments of the expected sizes 215 bp and 104 bp (Figure 3.5).



**Figure 3.3:** Determination of the optimal cycle number for amplification of  $\beta_2$ -microglobulin. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: 15 cycles, Lane 4: 20 cycles, Lane 5: 25 cycles, Lane 6: 30 cycles.



**Figure 3.4:**  $\beta_2$ -microglobulin RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lanes 3&4: cDNA samples from the MCF7, Lanes 5&6: cDNA samples from the MDA-MB-468.



**Figure 3.5:** Restriction digestion of  $\beta_2$ -microglobulin RT-PCR using EcoRI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

#### 3.3.3 Cytochrome P450 enzyme RT-PCR

#### 3.3.3.1 CYP1 family RT-PCR

Initially, the primer sequence designed for CYP1A1 by Primer3 (Table 2.6) failed to detect expression of CYP1A1 mRNA in either the MCF7 or MDA-MB-468 cells. These results did not correlate with the findings of earlier investigations which showed CYP1A1 mRNA to be present in MCF7 cells (Spink et al. 1998, Brockdorff et al. 2000, McFadyen et al. 2003). To further confirm the expression of CYP1A1 mRNA in the MCF7 cells, a published CYP1A1 primer set (Table 2.7) was used in this experiment. A band of 392 bp was observed for CYP1A1 mRNA expression. The amount of template cDNA derived from the MCF7 cells was varied (i.e. 100, 150, 200 and 250 ng) with 0.5 µM of CYP1A1 primer (Figure 3.6). One hundred and fifty ng of template cDNA was selected for CYP1A1 amplification in subsequent experiments as this was the minimum concentration of cDNA that generated a sharp and clear band. Expression of CYP1A1 mRNA in the MCF7 cells was confirmed following observation of 188 bp and 164 bp fragments in restriction enzyme *Hae*III digested samples (Figure 3.7). However, no CYP1A1 mRNA

The CYP1A2 primer sequence designed by Primer3 (Table 2.6) showed the expected amplified product (245 bp) with bands representing non-specific amplification in either the MCF7 or MDA-MB-468 cells (Figure 3.8) and the images were inverted from positive to negative images for better visualisation. As detailed in section 3.2.3.2, optimisation of primer concentration in a low range (Figure 3.1, stage 2C) was performed but no PCR product of the correct size was seen. This experiment failed to confirm the specificity of CYP1A2. However, further experiments screening for the presence/absence of CYP1A2 mRNA in the remaining cell lines were performed and results are presented in Chapter 4.



**Figure 3.6:** *CYP1A1 RT-PCR optimisation of template cDNA in MCF7. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: 100 ng, Lane 4: 150 ng, Lane 5: 200 ng, Lane 6: 250 ng.* 



Figure 3.7: Restriction digestion of CYP1A1 RT-PCR with HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.



**Figure 3.8:** *CYP1A2 RT-PCR (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.* 

The routine amplification by RT-PCR using CYP1B1 primers did not show a 397 bp

cDNA product corresponding to CYP1B1 mRNA. To further optimise the PCR

conditions, the primer concentration was set at 0.5  $\mu$ M and the amount of template cDNA varied (i.e. 100, 150, 200 and 250 ng). The optimal concentration of template cDNA was identified as 150 ng because this was the minimum concentration that appeared as a sharp band (Figure 3.9). CYP1B1 mRNA was found in both the MCF7 and MDA-MB-468 cells. The presence of CYP1B1 mRNA was confirmed by restriction digest with *Rsa*I, which generated a digested fragment of the correct size (174 bp and 223 bp) as shown in Figure 3.10.

### 3.3.3.2 CYP2 family RT-PCR

Initial studies failed to demonstrate the presence of CYP2A6 or CYP2D6 mRNA in either the MCF7 or MDA-MB-468 cell lines. Further experiments to verify the presence or absence of the CYP2A6 and CYP2D6 PCR product were performed. Figure 3.1 (stage 2A and 2B) indicates the primer and cDNA template concentrations used. CYP2A6 and CYP2D6 mRNA were below the levels of detection in both the MCF7 and MDA-MB-468 cells under the conditions used.

A faint amplification product of 194 bp was correctly determined for CYP2B6 RT-PCR in the MDA-MB-468 cells, though no corresponding product was observed in the MCF7 cells when analysis was performed using 250 ng of template cDNA (Figure 3.11). CYP2B6 mRNA expression in the MDA-MB-468 cells was confirmed by restriction digestion with *Hae*III which generated a fragment of 164 bp (Figure 3.12).

Differential expression of the CYP2C subfamily mRNA was observed in the MCF7 and MDA-MB-468 cell lines. CYP2C9 and CYP2C19 mRNA were below detectable limits in both cell lines under the conditions used although template cDNA and primer (a high range) concentrations have been optimised (Figure 3.1, stages 2A and 2B).

111



**Figure 3.9:** *CYP1B1 RT-PCR optimisation of template cDNA in MCF7. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: 100 ng, Lane 4: 150 ng, Lane 5: 200 ng, Lane 6: 250 ng.* 



**Figure 3.10:** *Restriction digestion of CYP1B1 RT-PCR using RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



**Figure 3.11:** CYP2B6 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.12:** *Restriction digestion of CYP2B6 RT-PCR using HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

Expression of CYP2C8 mRNA was only detectable in the MCF7 cells as a faint product at the correct size of 230 bp and a negative image of gel is presented in Figure 3.13 in order to improve the visualisation. To confirm CYP2C8 mRNA expression, the PCR product was digested with *Msp*I and the predicted digested fragments of 170 bp and 60 bp were observed (Figure 3.14).

CYP2C18 mRNA was detected in both cell lines at the correct cDNA product size of 199 bp. Similar levels of expression were observed in both the MCF7 and MDA-MB-468 cells (Figure 3.15). Restriction digestion of CYP2C18 mRNA by the enzyme *Rsa*I generated fragments (125 bp and 74 bp) of the correct size, though a partial digest product was seen (Figure 3.16) and this will be discussed in Section 3.4.

A band of 216 bp was observed at the correct size for CYP2S1 mRNA in both the MCF7 and MDA-MB-468 cells. It was noticed that a slightly higher level of CYP2S1 mRNA expression occurred in the MDA-MB-468 cells compared to the MCF7cells (Figure 3.17). To confirm the CYP2S1 mRNA expression, the PCR product was digested with restriction enzyme *Alu*I and the correct fragments were found at 103 bp and 63 bp (Figure 3.18). The purified DNA from the 216 bp fragment was confirmed by DNA sequencing (Section 2.10). The results of sequencing were analysed using the standard nucleotide-nucleotide BLAST search and compared with the non-redundant GenBank database. The findings were 100% matched (183/183) to *Homo sapiens* CYP2S1 mRNA (GenBank accession number: NM\_030622) (Appendix D, Figure D.1).



**Figure 3.13:** *CYP2C8 RT-PCR (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.* 



**Figure 3.14:** *Restriction digestion of CYP2C8 RT-PCR using MspI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



Figure 3.15: CYP2C18 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.16:** *Restriction digestion of CYP2C18 RT-PCR using RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



Figure 3.17: CYP2S1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.18:** *Restriction digestion of CYP2S1 RT-PCR using AluI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 3.3.3.3 CYP3A subfamily RT-PCR

Initially, the primer sequence designed for CYP3A4/5/7/43 by Primer3 (Table 2.6) failed to detect the presence of any of the four candidate genes in either the MCF7 or MDA-MB-468 cells. From the literature, it appears that many research groups have demonstrated the expression of CYP3A in human breast tumours, but these results are inconsistent (Albin et al. 1993, Huang et al. 1996, Iscan et al. 2001, Modugno et al. 2003, Miki et al. 2006). These inconsistencies could have arisen in several ways, for example the use of samples from non-matched individuals and method choices. However, CYP3A expression has not been previously reported in breast cancer cell lines. Furthermore, CYP3A plays a major role in the metabolism of oestrogen and several anticancer drugs such as tamoxifen, ifosphamide, taxol and vinblastine (Scripture and Figg 2006). In the present study, CYP3A4 and CYP3A5 mRNA expression were therefore examined using published primer sets (Table 2.7) which were analysed by the BLAST search. Optimisation of template cDNA and primer concentration was performed as described in Section 3.2.3 (Figure 3.1), but the expression of CYP3A4 mRNA and CYP3A5 mRNA were below detectable limits under the conditions used. In addition neither CYP3A7 nor CYP3A43 mRNA was detected in this study.

#### 3.3.3.4 CYP4 family RT-PCR

Expression of CYP4Z1 mRNA was detectable in the MDA-MB-468 cells as a product at the correct size of 159 bp but no corresponding product was observed in the MCF7 cells. Analysis was performed using 100 ng of template cDNA with 0.5 µM of CYP4Z1 primer set. The result obtained from the MCF7 cells was compared with data from previous studies (Rieger et al. 2004, Savas et al. 2005) which detected CYP4Z1 mRNA in the MCF7 cells. Template cDNA and primer concentrations were therefore optimised (Figure 3.1, stages 2A and 2B) but these studies failed to demonstrate the expression of CYP4Z1 mRNA in the MCF7 cells. Regarding the presence of CYP4Z1 mRNA in a cDNA product of MDA-MB-468 cells, it should be noted that an additional band of approximately 250 bp (~250 bp) was observed (Figure 3.19).

Template cDNA samples from the MDA-MB-468 cells were amplified at different annealing temperatures (i.e. 55, 56, 57, 58, 59, 60, 61, 62 and 63°C) under the same PCR conditions. The purpose of this was to investigate the effect of annealing temperature on the specificity of primers and the yield of PCR product of CYP4Z1 mRNA, and to understand the presence of the additional band. After varied annealing temperature, products of 159 bp and ~250 bp were still visualised though DNA bands appeared sharp and clear at annealing temperatures of 58 and 59°C (Figure 3.20). Therefore, an annealing temperature of 58°C was selected for CYP4Z1 amplification in all subsequent experiments.



**Figure 3.19:** CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.20:** Illustration of the effect of annealing temperature on CYP4Z1 RT-PCR products . Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: 55°C, Lane 4: 56°C, Lane 5: 57°C, Lane 6: 58°C, Lane 7: 59°C, Lane 8: 60°C, Lane 9: 61°C, Lane 10: 62°C, Lane 11: 63°C.

The 159 bp DNA fragments and the additional band of ~250 bp obtained from the MDA-MB-468 cells were purified by the method explained in Section 2.10, and the resulting DNA products were digested with *Rsa*I. Restriction digestion of the purified DNA product of the 159 bp fragment generated a fragment of the 112 bp (Figure 3.21A). By comparison, the purified DNA product of the ~250 bp band generated a fragment of 112 bp and an additional band at 120 bp (Figure 3.21B).

The purified DNA products from the 159 bp and ~250 bp fragments were further investigated by DNA sequencing. The standard nucleotide-nucleotide BLAST was used to analyse the results of DNA sequencing and compared with the non-redundant GenBank database. This indicated that purified DNA of the 159 bp and ~250 bp fragments were 94% (118/125) (Appendix D, Figure D.2) and 98% (77/78) (Appendix D, Figure D.3) identical to *Homo sapiens* CYP4Z1 mRNA (GenBank accession number: NM\_178134) respectively. The expression of CYP4Z1 mRNA in breast cancer cell lines will be further investigated in Chapter 4 and characterised in Chapter 5.



**Figure 3.21:** Restriction digestion of purified DNA products from CYP4Z1 RT-PCR using RsaI (negative images). (A) Purified DNA product of 159 bp fragment. (B) Purified DNA product of 250 bp fragment. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

## 3.3.3.5 CYP24 family RT-PCR

A band of the correct size at 249 bp was not observed for CYP24A1 mRNA expression after routine amplification. When the amount of template cDNA was varied (i.e. 100, 150, 200 and 250) with 0.5 μM of CYP24A1 primer, expression of CYP24A1 mRNA was found in the MCF7 cells but no amplification product was detected in the MDA-MB-468 cells. The optimal amount of template cDNA was identified as 150 ng for CYP24A1 amplification because this was the lowest amount that produced a sharp DNA band (Figure 3.22). This amount would be utilised in subsequent experiments. Expression of CYP24A1 mRNA in the MCF7 cells was confirmed by restriction digestion with BamHI and

observation of the correct digested fragments of 68 bp and 181 bp (Figure 3.23).



**Figure 3.22:** CYP24A1 RT-PCR optimisation of template cDNA in MCF7. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: 100 ng, Lane 4: 150 ng, Lane 5: 200 ng, Lane 6: 250 ng.



**Figure 3.23:** *Restriction digestion of CYP24A1 RT-PCR using BamHI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

#### 3.3.3.6 CYP26 family RT-PCR

The expression of CYP26A1 and CYP26B1 mRNA was detected in both the MCF7 and MDA-MB-468 cells as 191 bp and 172 bp cDNA products respectively (Figures 3.24A and 3.24B). It was noted that a slightly higher level of CYP26A1 mRNA expression was found in the MCF7 compared to the MDA-MB-468 cells, while a higher level of CYP26B1 mRNA expression was observed in the MDA-MB-468 cells compared to the MCF7 cell line. To confirm CYP26A1 mRNA expression, the PCR product was digested with *Alu*I and the correct digested fragment of 172 bp was observed (Figure 3.25A). The

presence of CYP26B1 mRNA was also confirmed by restriction digestion with *Alu*I and observation of digested fragments of the correct size (104 bp and 68 bp); as shown in Figure 3.25B). Partial digest products, observed in Figures 3.25A and 3.25B, will be discussed in Section 3.4.



**Figure 3.24:** *CYP26A1 and CYP26B1 RT-PCR. (A) CYP26A1. (B) CYP26B1. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.* 



**Figure 3.25:** *Restriction digestion of CYP26A1 and CYP26B1 RT-PCR using AluI. (A) CYP26A1. (B) CYP26B1. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 3.3.4 Cytochrome P450-regulatory nuclear receptors RT-PCR

The presence of hypoxia-inducible factor-1 alpha (HIF-1α), aryl hydrocarbon receptor (AhR), aryl hydrocarbon receptor nuclear translocator (ARNT) and aryl hydrocarbon receptor repressor (AhRR) mRNA was detected in both the MCF7 and MDA-MB-468 cells as 167 bp, 262 bp, 387 bp and 171 bp cDNA products respectively. A product of 250 bp was correctly determined for vitamin D receptor (VDR) in the MCF7 cells although no

corresponding product was observed in the MDA-MB-468 cells (Figure 3.26). All experiments were performed using 100 ng of template cDNA with 0.5  $\mu$ M of each primer set.

Stronger intensity of PCR product of HIF-1α mRNA expression was clearly observed in the MCF7 compared to the MDA-MB-468 cells, whereas slightly higher levels of AhR and ARNT mRNA expression were found in the MDA-MB-468 compared to the MCF7 cells. Similar levels of AhRR mRNA expression were detected in both the MCF7 and MDA-MB-468 cells.

Restriction digestion of HIF-1 $\alpha$  mRNA by the enzyme *Ssp*I generated a fragment of 142 bp (Figure 3.27). To confirm the AhR mRNA expression, the PCR product was restricted with the enzyme *Msp*I and the correct fragments of 163 bp and 99 bp were detected (Figure 3.28). The presence of ARNT and VDR mRNA was confirmed by restriction digestion with *Pst*I which generated digested fragments of 269 bp and 118 bp for ARNT mRNA (Figure 3.29) and a fragment of 198 bp for VDR mRNA (Figure 3.31). To confirm AhRR mRNA expression, the PCR product was digested with *Hae*III and the correct digested fragments of 92 bp and 79 bp were observed (Figure 3.30).



**Figure 3.26:** *RT-PCR products of HIF-1α, AhR, ARNT, AhRR and VDR. Lane 1: 1000 bp PCR markers, Lanes 2-3: HIF-1α, Lanes 4-5: AhR, Lanes 6-7: ARNT, Lanes 8-9: AhRR, Lanes 10-11: VDR Lanes number 2, 4, 6, 8, 10 represent MCF7 and lanes number 3, 5, 7, 9, 11 indicate MDA-MB-468.* 



**Figure 3.27:** Restriction digestion of HIF-1 $\alpha$  RT-PCR using SspI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.



**Figure 3.28:** *Restriction digestion of AhR RT-PCR using MspI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



**Figure 3.29:** *Restriction digestion of ARNT RT-PCR using PstI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



**Figure 3.30:** *Restriction digestion of AhRR RT-PCR using HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



**Figure 3.31:** *Restriction digestion of VDR RT-PCR using PstI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 3.3.4.1 Oestrogen receptor (ER) RT-PCR

A band of the correct size at 208 bp was observed for ER mRNA expression in both the MCF7 and MDA-MB-468 cells. However, an additional fainter band at 450 bp was also detected in this experiment (Figure 3.32). The DNA products of 208 bp and 450 bp were purified by the method described in Section 2.10, and the purified DNA products were digested with the restriction enzyme *Rsa*I. The correct fragments of 111 bp and 97 bp were observed only in the purified 208 bp DNA fragment (Figure 3.33), not in the purified 450 bp DNA product.

The purified DNA products of 208 bp and 450 bp fragments were further analysed by DNA sequencing. The product of 208 bp fragment was sequenced and found to be a 92% match (69/75) to *Homo sapiens* ER mRNA (GenBank accession number: NM\_000125) (Appendix D.4), whereas the product of the 450 bp fragment was sequenced and found to consist of unrelated ER sequences. This will be discussed in Section 3.4.



Figure 3.32: ER RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.33:** *Restriction digestion of ER RT-PCR using RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

# 3.3.4.2 Constitutive and rostane receptor (CAR) RT-PCR

An initial experiment using the primer sequence designed for CAR by Primer3 (Table 2.6) failed to detect CAR mRNA in either the MCF7 or MDA-MB-468 cells. As mentioned in Section 1.7.3, CAR plays an important role in the transcriptional regulation of certain P450s (e.g. CYP2B6) and a study by Chang and co-workers (Chang et al. 2003)

demonstrated a positive correlation between CAR and CYP2B6 mRNA level in human liver tissues. In this study, the expression of CYP2B6 mRNA was detected and confirmed in the MDA-MB-468 cells (Figures 3.11 and 3.12); therefore, CAR mRNA was considered for investigation in this experiment using the published CAR primer set (Table 2.7). Results showed the expected PCR product (417 bp) for both MCF7 and MDA-MB-468 cells, though non-specific products were detected in the MCF7 cells (Figure 3.34).

Following the non-specificity of the published primer above, the nested CAR primer set (Table 2.6) was employed to enhance specificity of the first round PCR product under the conditions shown in Table 2.8. To obtain optimal conditions for this experiment, the volume of the first PCR product was varied (i.e. 1, 2, 3, 4 and 5  $\mu$ L) with 0.5  $\mu$ M of the nested CAR primer (Figure 3.35A), and the nested CAR primer concentration was varied (i.e. 0.1, 0.2, 0.3, 0.4 and 0.5  $\mu$ M) with 5  $\mu$ L of the first PCR product (Figure 3.35B). The optimal conditions were found to be 0.2  $\mu$ M of the nested primers with 5  $\mu$ L of the first PCR product and this was used for the remainder of this study. Expression of CAR mRNA was found in both the MCF7 and MDA-MB-468 cells which was confirmed by restriction digestion with *Pst*I and observation of the correct fragments of 170 bp and 62 bp (Figure 3.36).



Figure 3.34: Nested CAR RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF7, Lane 4: MDA-MB-468.



**Figure 3.35:** Nested CAR RT-PCR optimisation of the volume of the first PCR product and concentration of nested CAR primers in the MCF7. Lane 1: 1000 bp PCR markers, Lane 2: negative control (A) Varied the volume of the first PCR products. Lane 3: 1  $\mu$ L, Lane 4: 2  $\mu$ L, Lane 5: 3  $\mu$ L, Lane 6: 4  $\mu$ L, Lane 7: 5  $\mu$ L. (B) Varied the concentration of nested CAR primers. Lane 3: 0.1  $\mu$ M, Lane 4: 0.2  $\mu$ M, Lane 5: 0.3  $\mu$ M, Lane 6: 0.4  $\mu$ M, Lane 7: 0.5  $\mu$ M.



**Figure 3.36:** Restriction digestion of nested CAR RT-PCR using PstI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

### 3.3.4.3 Pregnane X receptor (PXR) RT-PCR

Initially, the primer sequence designed for PXR by Primer3 (Table 2.6) failed to demonstrate PXR mRNA expression in either the MCF7 or MDA-MB-468 cells. However, the expression of PXR mRNA in MCF7 cells has been demonstrated in previous studies (Miki et al. 2006, Sakai et al. 2006). In this research, the published PXR primer set (Table 2.7) was therefore used to determine the expression of PXR mRNA. No PCR product was detected after routine amplification so the optimisation of PCR analysis was performed. Amount of template cDNA was varied (i.e. 100, 150, 200 and 250 ng) with 0.5  $\mu$ M of PXR primer (Figure 3.37), and 250 ng of template cDNA was chosen for PXR amplification because the product appeared as an intense DNA band when compared to either 150 ng or 200 ng. The expression of PXR mRNA was found in only the MCF7 cells, not in the MDA-MB-468 cells. The presence of PXR mRNA in the MCF7 cells was confirmed by digestion with restriction enzyme *Msp*I, which generated fragments of the correct size (142 bp and 90 bp; Figure 3.38).



**Figure 3.37:** PXR RT-PCR optimisation of template cDNA in MCF7. Lane 1: 1000 bp PCR markers as indicated, Lane 2: negative control, Lane 3: 100 ng, Lane 4: 150 ng, Lane 5: 200 ng, Lane 6: 250 ng.



**Figure 3.38:** *Restriction digestion of PXR RT-PCR using MspI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

# 3.4 Discussion

This chapter describes an RT-PCR protocol to optimise the detection of cytochrome P450 and P450-regulatory nuclear receptor mRNA. The advantages of using end-point PCR in this study are that it is a simple and rapid method for screening and detecting amplification products of high numbers of candidate genes in cell lines. This study assessed a number of variables affecting RT-PCR analysis including the methods of RNA isolation, cDNA verification, PCR and restriction digest reaction conditions, and DNA purification and sequencing.

Isolating total RNA of high purity was critical for validation of the PCR system as cellular contaminants such as genomic DNA (gDNA) can interfere with reaction efficiency and complicate interpretation of results. Initially the RNA purity in this study was low with an  $A_{260}/A_{280}$  ratio between 1.3 and 1.6 (instead of 1.8-2.1) and this seriously affected the specificity of the RT-PCR reactions. To overcome this problem several technical measures were implemented, in particular it was felt that there had been overloading of cells onto the RNeasy spin column and that the method of disrupting the cells was critical. The use of a cell number between  $5 \times 10^5$  and  $2 \times 10^6$  per purification, and an increased efficiency in the homogenisation process resulted in higher RNA purity, non-degraded RNA (Figure 3.2)

and less gDNA contamination in downstream applications. Upon successful validation, this methodology was utilised for the remainder of the study.

In this investigation,  $\beta_2$ -microglobulin was used as an internal control to detect false negative results by comparing mRNA expression with the gene of interest in different cell lines. Low variation in  $\beta_2$ -microglobulin mRNA between the MCF7 and MDA-MB-468 cells was observed (Figure 3.4), suggesting that this gene was an appropriate reference gene in this study. As mentioned in Section 3.1, the use of negative controls (no DNA template) is critical for conducting PCR as it can monitor false-positive results in the PCR assay. No product was detected in the negative controls in any of the experiments in this chapter, indicating that PCR reaction mixtures were free from DNA carryover contamination.

Specificity in primer sequences is one of the most important keys to success in the PCR assay. In this research, the candidate sequences designed by Primer3 programme were assessed via a direct BLAST search. Ten out of the nineteen cytochrome P450 enzymes (i.e. CYP1A1, CYP1A2, CYP2A6, CYP2C9, CYP2C19, CYP2D6, CYP3A4, CYP3A5, CYP3A7 and CYP3A43), and two out of the eight P450-regulartory nuclear receptors (i.e. CAR and PXR) primers designed by Primer3 programme failed to demonstrate their mRNA expression in either the MCF7 or MDA-MB-468 cells, although sufficient validation following general guidelines used in this research was performed.

However, five out of the twelve undetectable genes (i.e. CYP1A1, CYP3A4, CYP3A5, CAR and PXR) were considered for further investigation using the published primers (Table 2.7) because evidence regarding the presence or the relevance of them has been demonstrated in breast cancer cells and they are involved in the metabolism of oestrogen and anticancer drugs. A single specific PCR product of CYP1A1 and PXR mRNA was

successfully detected after varying the concentration of cDNA template (Figures 3.6 and 3.37). The published CAR primer set was performed under routine amplification, but it generated non-specific PCR product (Figure 3.34). Therefore, nested PCR was used to improve specificity of the primer, a specific product was detected and its sequence identity confirmed by restriction digestion. The observation of non-specific bands in the first round PCR may be due to the fact published PCR conditions were not used in this experiment. In an attempt to generate high quality data, experiments were performed using the procedure described in Section 3.2.3 as standard. Unfortunately, the published primers of CYP3A4 and CYP3A5 fail to demonstrate mRNA expression in the cell lines used despite their validation.

It is well-known that CYP2D6 catalyses the N-demesthylation and 4-hydroxylation of tamoxifen which have potent anti-oestrogen effects. Tamoxifen is widely used in the treatment of breast cancer (Schroth et al. 2007). Previous studies demonstrated that CYP2D6 mRNA was expressed in normal breast as well as in breast tumour samples (Huang et al. 1996, Hellmold et al. 1998, Iscan et al. 2001, Bièche et al. 2007). In addition, Hellmond and colleagues (Hellmold et al. 1998) demonstrated CYP2D6 protein expression in normal breast. These findings were in disagreement with those of Modugno and co-workers (Modugno et al. 2003), who investigated CYP2D6 mRNA expression in breast tumour and normal adjacent breast tissues using quantitative RT-PCR. The present study is the first to examine the expression of CYP2D6 mRNA in breast cancer MCF7 and MDA-MB-468 cell lines, but no CYP2D6 mRNA was observed. As mentioned above, the importance of CYP2D6 in the metabolism of tamoxifen has been emphasised in the treatment of breast cancer. For this reason, the current project sought to ascertain whether CYP2D6 is present or absent in the test cells using the published primers. Preliminary work with previously published CYP2D6 mRNA in normal breast and seven breast cancer cell lines will be discussed in the next chapter.
Validation of RT-PCR analysis

According to the general guidelines for the optimisation of PCR analysis in this study (Figure 3.1), the findings indicated that optimal cDNA template amount for the genes of interest was between 100 and 250 ng in a PCR reaction. The optimal primer concentration for each individual gene was generally 0.5  $\mu$ M because non-specific PCR products were seen at higher primer concentration of 1.0  $\mu$ M to 1.5  $\mu$ M. Regarding the thermocycling conditions, twenty-five to thirty-five cycles is generally standard for a PCR reaction. In the current investigation, thirty amplification cycles were selected and performed for all candidate genes, but not for  $\beta_2$ -microglobulin. Twenty cycles were sufficient for  $\beta_2$ -microglobulin amplification as the product generated a clear and sharp band (Figure 3.3).

This research aimed to detect the expression of a number of target mRNAs in a series of cell lines. Due to time constraints, a single repetition was performed in individual genes. However, in experiments where genes were undetectable after routine amplification, the procedure was repeated with varying amounts of cDNA template and/or primer and any detectable products were then confirmed by restriction digestion and/or DNA sequencing.

PCR products of detectable genes were confirmed by restriction digestion and observation of digested fragments of the correct size on gels. It should be noted that partial restriction digestion was observed for products of CYP2C18 (Figures 3.16), CYP26A1 (Figures 3.25A), CYP26B1 (Figures 3.25B), and AhR (Figures 3.28). These results may be due to excessive amounts of starting material being used, insufficient incubation time for reactions and/or low activity of restriction enzymes. This problem could be investigated by repeating experiments with lower DNA template concentrations, an increased incubation time and a new batch/fresh supply of restriction enzymes for further study.

In section 3.3.3.4, it was shown that MDA-MB-468 cells express CYP4Z1 mRNA (159 bp) with the presence of an additional band of ~250 bp. To ensure specificity of primers,

annealing temperature of PCR analysis was varied between 55°C and 63°C. However, two bands were still visualised on the gel, and this experiment showed that an annealing temperature of 58°C was optimal (Figure 3.20). Purified DNA of the two fragments was digested using restriction enzyme *Rsa*I and a restricted fragment size of 112 bp was found in both purified fragments. This finding suggested that these fragments may contain similar DNA and they were therefore sequenced. The results of DNA sequencing were accessed directly by a BLAST search, and this predicted that both fragments were human CYP4Z1 mRNA (Appendix D). However, more than fifty percent of CYP4Z1 (~250bp) sequences were unreadable, possibly due to poor quality of the purified DNA samples or an unknown factor involved in the expression of CYP4Z1 mRNA. Prior to further characterisation of the CYP4Z1 mRNA (Chapter 5), the previously published primers which have been demonstrated a single PCR product were used to confirm the presence of CYP4Z1 mRNA in the cDNA samples employed in this study. Results will be shown and discussed in Chapter 4.

Several novel extrahepatic P450 enzymes may be of particular interest in breast cancer cells in terms of potential targets for cancer therapy as they are preferentially expressed in extrahepatic tissues, and overexpression of some cytochrome P450s have been identified in certain types of tumours. A preliminary investigation demonstrated the expression of CYP2F1, CYP2J2, CYP2R1, CYP2U1 and CYP4X1 in a panel of breast cancer cell lines and compared to a normal breast cell line performed by the Honours students within the School of Pharmacy who employed the same cDNA samples and established methods used in the current investigation. This preliminary work also investigated the expression of retinoic acid receptor (RAR) and retinoid X receptor (RXR), because their physiological roles have been shown to be involved in the development of various types of cancer (Germain et al. 2003, Altucci et al. 2007) and the regulation of some cytochrome P450 enzymes (Meyer 2007). These results will be shown in Chapter 4.

133

Table 3.1 summarises the presence of cytochrome P450 and cytochrome P450-regulatory nuclear receptor mRNA in the MCF7 and MDA-MB-468 cells after optimisation and confirmation of their expression. In summary, this chapter provides information on validation of PCR conditions derived from the model cell lines, MCF7 and MDA-MB-468. These conditions will be used for the remaining breast cancer cell lines and compared to a normal breast cell line in the next chapter.

**Table 3.1:** The presence of cytochrome P450 and P450-regulatory nuclear receptor mRNA in the MCF7 andMDA-MB-468 cells

P450	MCF7	MDA-MB-468	NR	MCF7	MDA-MB-468
CYP1A1	+	-	HIF-1α	+	+
CYP 1A2	NS	NS	AhR	+	+
CYP1B1	+	+	ARNT	+	+
CYP2A6	-	-	AhRR	+	+
CYP2B6	-	+	CAR	+	+
CYP2C8	+	-	PXR	+	-
CYP2C18	+	+	VDR	+	-
CYP2C9	-	-	ER	+	+
CYP2C19	-	-			
CYP2D6	-	-			
CYP2S1	+	+			
CYP3A4	-	-			
CYP3A5	-	-			
CYP3A7	-	-			
CYP3A43	-	-			
CYP4Z1	-	+			
CYP24A1	+	-			
CYP26A1	+	+			
CYP26B1	+	+			

Abbreviations: P450; cytochrome P450 enzyme, NR; P450-regulatory nuclear receptor, NS; non-specific product by RT-PCR analysis, +; PCR product detected at predicted size and sequence specificity, -; not detected by visual inspection.

# Chapter 4

# Cytochrome P450 expression in breast cancer cell lines

## 4.1 Introduction

Clinically, the majority of breast tumours express oestrogen receptor (ER) and it is well established that local production of oestrogen is particularly high in breast tumours (Suzuki et al. 2005). Oestrogen is involved in the development and progression of breast cancer (Yager and Davidson 2006). Women with ER-positive breast tumours are more likely to benefit from endocrine therapy than patients with ER-negative tumours. Approximately 60-65% of patients with ER-positive tumours respond to anti-oestrogen therapy (i.e. tamoxifen) whereas in women with ER-negative tumours, only low levels of response are achieved. Moreover, ER-negative women have a poorer survival and prognosis than patients who have node positive and ER-positive tumours (Putti et al. 2005).

However, the overall cytochrome P450 profile in breast cancer and the role of these enzymes in breast carcinogenesis is yet to be defined. A better understanding of the differential expression of these enzymes in different oestrogen receptor status, invasive phenotypes and racial groups could potentially be of enormous value in the development of novel cancer therapeutics (targeted at these enzymes) and the design of more effective drug regimens. Therefore, this chapter was designed to identify the cytochrome P450 and P450regulatory nuclear receptor mRNA profile in a panel of breast cancer cell lines derived from different ethnic backgrounds (i.e. Caucasians and Afro-Caribbeans) and including different breast tumour characteristics (i.e. oestrogen receptor status and invasive phenotype). A normal breast cell line was also examined for comparative purposes.

### 4.2 Materials and Methods

#### 4.2.1 Cell lines and cell culture

The eight cell lines used were obtained from different sources (Section 2.2.2) and a summary of their characteristics is presented in Table 1.1. All cells were maintained under the growth conditions described in Section 2.2.3. All experiments were conducted using cells cultured for no more than 15 passages after resuscitation of the first cell stock. All culture techniques were performed as explained previously (Sections 2.2.4 to 2.2.7).

### 4.2.2 RNA isolation and cDNA sysnthesis

Total RNA was isolated from all cell lines using the RNeasy Plus Mini kit from Qiagen as detailed in Section 2.3. Approximately  $5 \times 10^5$  to  $2 \times 10^6$  cells were extracted per purification as discussed in Section 3.3.1. The quantity and quality of total RNA in each preparation were determined by  $A_{260}/A_{280}$  ratios and inspection of the 28S and 18S rRNA bands in 1% (w/v) agarose gels. As described in Section 2.4, reverse transcription (RT) reactions were conducted in a final volume of 20 µL (Table 2.1) using the reaction conditions summarised in Table 2.2.

#### 4.2.3 Qualitative PCR amplification

PCR was performed under optimal conditions (Section 3.3) in order to profile P450 and P450-regulatory nuclear receptor mRNA expression in the test cell lines. The PCR products obtained from the housekeeping gene  $\beta_2$ -microglobulin and the target genes from each cell line were visualised on 1.5% agarose gels containing ethidium bromide. Following visualisation of bands in the PCR products, the band intensity of photographed gel (positive) images was subjected to densitometric scanning using UVI-BandMap software (UVItec, Cambridge, UK). Densitometry is the quantitative measurement of optical density which can be expressed as the number of dark spots in a given volume of intensity. The sensitivity (maximum spot size) was set at 5 mm for all samples to measure the volume density of the amplified products. To demonstrate a relative expression of candidate genes' mRNA in each cell line tested, mRNA expression of each gene was normalised against  $\beta_2$ -microglobulin mRNA expression. In the result section, some gel images were inverted from positive to negative images for better visualisation.

In this chapter, the PCR assay was used qualitatively i.e. it was used to determine whether mRNA was present or absent in all individuals from the test cell lines. To facilitate comparisons between the cell lines, the volume density of the  $\beta_2$ -microglobulin RT-PCR and the relative intensities of target genes normalised against  $\beta_2$ -microglobulin were determined.

### 4.2.4 Oligonucleotide primer design

Prior to further elucidation of the CYP4Z1 sequence the present investigation confirmed the expression of CYP4Z1 mRNA in the eight cDNA samples using the previously published primers (Table 2.7, Thomas et al. 2006) which had been designed to cross between exon 7 and exon 8 boundaries (Figure 5.2). Thomas and colleagues found that this primer set generated a single band of the CYP4Z1 amplification product.

During the course of the current study, the expression of CYP2D6, CYP2F1, CYP2J2, CYP2R1, CYP2U1, CYP4X1, RAR and RXR was investigated using cDNA samples as mentioned in Section 3.4. These were subjected to routine amplification (Tables 2.3 and 2.4) with the previously published primers (Table 2.7) and the products obtained were digested with restriction enzymes (Table 2.10) under the conditions described in Table 2.11 to confirm the mRNA expression of candidate genes.

## 4.3 Results

## 4.3.1 RNA isolation

Total RNA isolated from the cell lines had  $A_{260}/A_{280}$  ratios between 1.8 and 2.1. One microgram of total RNA isolated from the eight cell lines was loaded onto a 1% (w/v) agarose gel to confirm RNA quality (Figure 4.1).



**Figure 4.1:** *Quality of total RNA isolated. Lane 1: MCF10A, Lane 2: MCF7, Lane 3: T47D, Lane 4: ZR-75-1, Lane 5: ZR-75-30, Lane 6: MDA-MB-231, Lane 7: MDA-MB-157, Lane 8: MDA-MB-468.* 

## 4.3.2 β<sub>2</sub>-microglobulin RT-PCR

A band of 319 bp corresponding to the cDNA product size for  $\beta_2$ -microglobulin was detected for all cell lines. There was no visible difference in the housekeeping gene  $\beta_2$ -microglobulin mRNA expression between the breast cancer cells and the normal breast cell line used. No band was detected in the negative controls (no template cDNA, Figure 4.2). This data was used for each subsequent analysis in this chapter.

## 4.3.3 CYP1 family RT-PCR

CYP1A1 mRNA was detected in five out of the seven breast cancer cell lines as the correct band at 392 bp and it was expressed at a very low level in the normal breast MCF10A cell line. No CYP1A1 mRNA expression was detected in the ZR-75-30 and MDA-MB-468 cells (Figure 4.3).



**Figure 4.2:**  $\beta_2$ -microglobulin RT-PCR. (A) Visualisation of the  $\beta_2$ -microglobulin amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Volume density of the  $\beta_2$ -microglobulin amplified products (pixels).



**Figure 4.3:** *CYP1A1 RT-PCR.* (A) Visualisation of the CYP1A1 amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP1A1 normalised against  $\beta_2$ -microglobulin.

As mentioned in Section 3.3.3.1, the CYP1A2 primer sequence designed by Primer3 failed to detect the presence of CYP1A2 mRNA in either the MCF7 or MDA-MB-468 cells. This primer set was also used for the remaining cell lines, but unfortunately the results still showed non-specific amplification in all cases.

A band of 397 bp was observed for CYP1B1 mRNA in both the normal breast (MCF10A) and breast cancer cell lines. MCF7, T47D, ZR-75-1, MDA-MB-231 and MDA-MB-157 demonstrated higher levels of expression than the ZR-75-30 cells. Only very weak expression was detected in the MDA-MB-468 cells (Figures 4.4).



**Figure 4.4:** *CYP1B1 RT-PCR.* (A) Visualisation of CYP1B1 amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP1B1 normalised against  $\beta_2$ -microglobulin.

## 4.3.4 CYP2 family RT-PCR

As mentioned in Section 3.3.3.2, both the MCF7 and MDA-MB-468 cell lines failed to show the presence of CYP2A6 mRNA during PCR validation. However, a band of 186 bp corresponding to the cDNA product size for CYP2A6 mRNA was detected in three of the cell lines. CYP2A6 mRNA was expressed in the ZR-75-1 and T47D cells at higher levels

than the MDA-MB-157 cells (Figures 4.5). The expression of CYP2A6 mRNA was confirmed by restriction digestion with *Hae*III, which generated digested fragments of the correct size (72 bp and 66 bp, Figure 4.6).



**Figure 4.5:** *CYP2A6 RT-PCR.* (A) Visualisation of CYP2A6 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP2A6 normalised against  $\beta_2$ -microglobulin.



**Figure 4.6:** *Restriction digestion of CYP2A6 RT-PCR with HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

A band of 194 bp corresponding to the cDNA product size for CYP2B6 mRNA was found in five out of the seven breast cancer cell lines but no product was visible in the normal breast cells. A high level of expression was observed in the T47D, ZR-75-1 and MDA-MB-157 cells, while very low levels were detected in the ZR-75-30 and MDA-MB-468 cells. CYP2B6 mRNA expression was not detected in the MCF7 and MDA-MB-231 cells (Figure 4.7).

CYP2C8 mRNA was detected in three out of the seven breast cancer cell lines as a cDNA product at 230 bp but no mRNA was observed in the normal breast cell line. A higher level of CYP2C8 expression was found in the MDA-MB-157 cells compared to the MCF7 and ZR-75-1 cells (Figure 4.8).

A band of 199 bp was observed for the presence of CYP2C18 mRNA in four out of seven breast cancer and the normal breast cell lines. It was expressed in the T47D cells at a high level, whereas very weak levels were detected in the normal breast cell line, and the breast cancer cell lines MCF7, MDA-MB-157 and MDA-MB-468 (Figure 4.9).

During PCR validation (Section 3.3.3.2) both the MCF7 and MDA-MB-468 cells failed to demonstrate expression of CYP2C9/19 and CYP2D6 mRNA. Similarly, no amplification product was detected in the cell lines investigated. However CYP2D6, an enzyme which plays a crucial role in the metabolism of antioestrogen tamoxifen and is associated with breast cancer was considered for PCR experiments using previously published sequences (Table 2.7, Thomas et al. 2006). Preliminary results are shown in Figures 4.27 and 4.28.



**Figure 4.7:** *CYP2B6 RT-PCR.* (A) Visualisation of CYP2B6 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP2B6 normalised against  $\beta_2$ -microglobulin.



**Figure 4.8:** *CYP2C8 RT-PCR.* (A) Visualisation of CYP2C8 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP2C8 normalised against  $\beta_2$ -microglobulin.



**Figure 4.9:** CYP2C18 RT-PCR. (A) Visualisation of CYP2C18 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP2C18 normalised against  $\beta_2$ -microglobulin.

A band of 216 bp corresponding to the cDNA product size for CYP2S1 mRNA was highly expressed in all of the cell lines except MCF10A and ZR-75-30. The highest level of expression was found in the T47D cells. Moderate expression was detected in the MCF7, ZR-75-1, MDA-MB-231, MDA-MB-157 and MDA-MB-468 cells (Figure 4.10).

### 4.3.5 CYP3 family RT-PCR

In the CYP3A subfamily, the primer sequences designed for CYP3A4/5/7/43 by Primer3 (Table 2.6) failed to detect CYP3A mRNA expression in the MCF7 and MDA-MB-468 cells during PCR validation (Section 3.3.3.3). When profiling these genes against a series of cell lines, only a very weak level of CYP3A7 mRNA expression was detected in the normal breast cell line MCF10A as a cDNA product of 223 bp (Figure 4.11). The expression of CYP3A7 mRNA was confirmed by restriction digestion with *Eco*RI which generated fragments of the correct size (158 bp and 65 bp) (Figure 4.12). However, no

CYP3A4/5/43 mRNA expression was detected in either the normal breast or breast cancer



cell lines under the conditions used in this study.

**Figure 4.10:** *CYP2S1 RT-PCR.* (A) Visualisation of CYP2S1 amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP2S1 normalised against  $\beta_2$ -microglobulin.



**Figure 4.11:** *CYP3A7 RT-PCR.* (*A*) *Visualisation of CYP3A7 amplified products (negative image).* Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (*B*) *Relative intensities of CYP3A7 normalised against*  $\beta_2$ -microglobulin.



**Figure 4.12:** *Restriction digestion of CYP3A7 RT-PCR with EcoRI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 4.3.6 CYP4 family RT-PCR

A cDNA product at 159 bp indicated CYP4Z1 mRNA was found in three out of the seven breast cancer cell lines and the normal breast cell line. CYP4Z1 was expressed in the MDA-MB-468 and T47D cells at higher levels than in the MCF10A and ZR-75-1 cells. An additional band of ~250 bp was observed with the presence of CYP4Z1 mRNA in all four cell lines (Figure 4.13).

A single band of 257 bp corresponding to the cDNA product size for the published CYP4Z1 mRNA was detected in four of the seven breast cancer cell lines and the normal breast cell line (Figure 4.14). There was no additional band in this primer set. The CYP4Z1 mRNA in the MDA-MB-468 cells seemed to be overexpressed compared with the MCF10A, T47D and ZR-75-1 cells. CYP4Z1 mRNA was expressed in the ZR-75-30 cells at a very low level. Expression of CYP4Z1 mRNA in the MDA-MB-468 cells was confirmed by restriction digest with *Alu*I, and the correct fragments of 196 bp and 61 bp were observed (Figure 4.15).



**Figure 4.13:** CYP4Z1 RT-PCR. (A) Visualisation of CYP4Z1 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP4Z1 (159 bp) normalised against  $\beta_2$ -microglobulin.



**Figure 4.14:** *CYP4Z1 RT-PCR using the previously published primer set. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468.* 



**Figure 4.15:** *Restriction digestion of CYP4Z1 RT-PCR (using the published primers) with AluI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

#### 4.3.7 CYP24 family RT-PCR

A band at 249 bp was observed for CYP24A1 mRNA expression and it was detected in almost all the cell lines tested with the exception of ZR-75-30 and MDA-MB-468. Very low levels of CYP24A1 were observed in the MCF10A and MDA-MB-231 cells, whereas moderate expression was found in the MCF7, T47D, ZR-75-1 and MDA-MB-157 cells (Figure 4.16).

#### 4.3.8 CYP26 family RT-PCR

CYP26A1 mRNA, at 191 bp, was detected in all cell lines except ZR-75-1 and the normal breast cell line. CYP26A1 was expressed in MDA-MB-231 and MDA-MB-468 at very low levels, while a higher level of expression was observed in the MDA-MB-157 cells (Figure 4.17).

A band indicating the presence of CYP26B1 mRNA at 172 bp was detected in the normal breast cell line and six out of the seven breast cancer cell lines. No CYP26B1 mRNA was expressed in the ZR-75-30 cells. Very weak levels of CYP26B1 mRNA were found in the MCF7, T47D and ZR-75-1 cells, whereas higher levels were detected in both the normal breast and the breast cancer cell lines (MDA-MB231, MDA-MB-157 and MDA-MB-468; Figure 4.18).



**Figure 4.16:** CYP24A1 RT-PCR. (A) Visualisation of CYP24A1 amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP24A1 normalised against  $\beta_2$ -microglobulin.



**Figure 4.17:** CYP26A1 RT-PCR. (A) Visualisation of CYP26A1 amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP26A1 normalised against  $\beta_2$ -microglobulin.



**Figure 4.18:** CYP26B1 RT-PCR. (A) Visualisation of CYP26B1 amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of CYP26B1 normalised against  $\beta_2$ -microglobulin.

#### 4.3.9 Cytochrome P450-regulatory nuclear receptor RT-PCR

The expression of HIF-1α, AhR and ARNT mRNA was detected in both the normal breast and breast cancer cell lines at the correct band size of 167 bp, 262 bp and 387 bp respectively (Figures 4.19, 4.20 and 4.21). HIF-1α mRNA was expressed at high levels in the normal breast cells MCF10A, and the breast cancer cell lines T47D and MDA-MB-231. It was detected at lower levels in the ZR-75-1, ZR-75-30 and MDA-MB-468 cells. AhR and ARNT mRNA were expressed at similar levels in the cell lines used, and low levels of both AhR and ARNT mRNA expression were observed in the ZR-75-30 cells.

A band for the expression of AhRR mRNA at 171 bp was detected in six out of the seven breast cancer cell lines and in the normal breast cell line. Very weak levels of AhRR mRNA were found in the normal breast cells and ZR-75-30. AhRR mRNA expression was not detected in the ZR-75-1 cells (Figure 4.22). A band of 250 bp corresponding to the cDNA product size for VDR mRNA was found in five out of the seven breast cancer cell lines and in the normal breast cell line but no product was detected in the ZR-75-30 and MDA-MB-468 cells (Figure 4.23).

ER mRNA was found in six out of the seven breast cancer cell lines as a correct band at 208 bp but it was not detected in the normal breast cell line or the breast cancer cell line MDA-MB-231. ER was expressed in the T47D, ZR-75-1, ZR-75-30 and MDA-MB-157 cells at higher levels than in the MCF7 and MDA-MB-468 cells. An additional fainter band at 450 bp was only observed in the MCF7 and MDA-MB-468 cells as mentioned in Section 3.3.4.1 (Figure 4.24).



**Figure 4.19:** *HIF-1* $\alpha$ *RT-PCR.* (A) *Visualisation of HIF-1* $\alpha$  *amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of HIF-1* $\alpha$  *normalised against*  $\beta_2$ *-microglobulin.* 



**Figure 4.20:** *AhR RT-PCR.* (*A*) *Visualisation of AhR amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of AhR normalised against \beta\_2-microglobulin.* 



**Figure 4.21:** ARNT RT-PCR. (A) Visualisation of ARNT amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of ARNT normalised against  $\beta_2$ -microglobulin.



**Figure 4.22:** AhRR RT-PCR. (A) Visualisation of AhRR amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of AhRR normalised against  $\beta_2$ -microglobulin.



**Figure 4.23:** VDR RT-PCR. (A) Visualisation of VDR amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of VDR normalised against  $\beta_2$ -microglobulin.



**Figure 4.24:** *ER RT-PCR.* (*A*) *Visualisation of ER amplified products. Lane 1: 1000 bp PCR markers,* Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of ER $\alpha$  normalised against  $\beta_2$ -microglobulin.

A band of 232 bp was observed for the CAR mRNA using nested PCR in the breast cancer cell lines but it was not detected in the normal breast cell line. CAR mRNA (nested) was expressed at lower levels in the ZR-75-1 and MDA-MB-468 cells than in the MCF7, T47D, ZR-75-1, MDA-MB-231 and MDA-MB-157 cells. It was noticed that there were multiple bands of nested CAR RT-PCR product in the T47D, ZR-75-30, MDA-MB-231 and MDA-MB-468 cells (Figure 4.25).

A band of 246 bp was observed at the correct size for PXR mRNA in four out of the seven breast cancer cell lines but it was not expressed in the normal breast cell lines. PXR mRNA was expressed at similar levels in the ZR-75-1 and MDA-MB-157 cells and at very low levels in the MCF7 and T47D cells (Figure 4.26).



**Figure 4.25:** Nested CAR RT-PCR. (A) Visualisation of Nested CAR amplified products. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of Nested CAR normalised against  $\beta_2$ -microglobulin.



**Figure 4.26:** *PXR RT-PCR.* (A) Visualisation of PXR amplified products (negative image). Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468. (B) Relative intensities of PXR normalised against  $\beta_2$ -microglobulin.

#### 4.3.10 Preliminary work

A band of 261 bp indicated the cDNA product size for the published CYP2D6 mRNA was detected in both the normal breast and breast cancer cell lines (Figure 4.27). CYP2D6 mRNA was expressed at a lower level in the MDA-MB-231 cells than in the MCF7, T47D, ZR-75-1 and MDA-MB-157 cells. It should be noted that the presence of a cDNA product at 261 bp in MCF10A (Lane 2), MDA-MB-468 (Lane 6) and ZR-75-30 (Lane 9) shows up as visible smear in the background between ~250 bp and ~800 bp. However, the expression of CYP2D6 mRNA in the T47D cells was confirmed by restriction digestion with *Rsa*I, which generated digested fragments of the correct size (142 bp and 119 bp, Figure 4.28).

The expression of CYP2F1 mRNA was detected in two out of the seven breast cancer cell lines (i.e. ZR-75-1 and MDA-MB-468) at the expected band size of 305 bp but it was not detected in the normal breast cell line (Figure 4.29). CYP2F1 was expressed in the MDA-MB-468 cells at a higher level than in the ZR-75-1 cells. An additional fainter band at ~900 bp was only observed in the ZR-75-1 cells. Expression of CYP2F1 mRNA in the MDA-MB-468 cells was confirmed by restriction digestion with *Msp*I and observation of the correct fragments of 195 bp and 110 bp (Figure 4.30).

A single band of 714 bp corresponding to the cDNA product size for the CYP2J2 mRNA was found in both the normal breast and breast cancer cell lines (Figure 4.31). CYP2J2 mRNA was expressed in the MCF10A and MDA-MB-231 cells at very low levels while high levels of CYP2F1 mRNA were observed in almost breast cancer cell lines. To confirm CYP2J2 mRNA expression, the PCR product fromMDA-MB-468 cells was restricted with the enzyme *Alu*I and the correct fragments of 330 bp and 170 bp were detected (Figure 4.32).



**Figure 4.27:** Visualisation of CYP2D6 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.28:** Restriction digestion of CYP2D6 RT-PCR with RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.



**Figure 4.29:** Visualisation of CYP2F1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.30:** Restriction digestion of CYP2F1 RT-PCR with MspI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.



Figure 4.31: Visualisation of CYP2J2 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.32:** *Restriction digestion of CYP2J2 RT-PCR with AluI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

During the validation procedure of CYP2R1 RT-PCR analysis, a primer concentration of 0.2 μM was used for RT-PCR experiments as a higher concentration of primer (from 0.3 to 0.5 μM) generated non-specific amplification products. A band of 308 bp was observed for the CYP2R1 mRNA in both the normal breast and breast cancer cell lines (Figure 4.33). CYP2R1 mRNA was expressed at very low levels in the ZR-75-30 and MDA-MB-468 cells whereas higher levels of CYP2R1 mRNA were observed in the normal breast MCF10A cells and the breast cancer MCF7, T47D, MDA-MB-157, MDA-MB-231 and ZR-75-1 cells. Restriction digestion using *Hae*III confirmed the expression of CYP2R1 mRNA in the ZR-75-1 cells, and the correct fragment of 130 bp was observed but not a restricted band of 118 bp (Figure 4.34). It is possible that there was an overlapping band of the two fragments as they are close in size.

A cDNA product at 320 bp indicated CYP2U1mRNA was found in both the normal breast and breast cancer cells (Figure 4.35). CYP2U1 was expressed in the breast cancer cell lines (except in the ZR-75-30 cells) at higher levels than in the normal breast MCF10A cells. Expression of CYP2U1 mRNA in the ZR-75-1 cells was confirmed by restriction digestion *Hae*III, and the correct fragments of 252 bp and 60 bp were detected (Figure 4.36).

CYP4X1 mRNA was detected in both the normal breast and breast cancer cell lines at the correct cDNA product size of 364 bp (Figure 4.37). Very weak levels of CYP4X1 mRNA were observed in the MCF7, MDA-MB-157 and MDA-MB-231 ells. High levels of expression were found in the normal breast MCF10A and the breast cancer MDA-MB-468 and T47D cells. Restriction digestion of CYP4X1 mRNA by the enzyme *Rsa*I generated fragments (278 bp and 86 bp) of the correct size (Figure 4.38).



Figure 4.33: Visualisation of CYP2R1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.34:** *Restriction digestion of CYP2R1 RT-PCR with HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



Figure 4.35: Visualisation of CYP2U1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.36:** *Restriction digestion of CYP2U1 RT-PCR with HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



Figure 4.37: Visualisation of CYP4X1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: MCF7, Lane 4: MDA-MB-157, Lane 5: MDA-MB-231, Lane 6: MDA-MB-468, Lane 7: T47D, Lane 8: ZR-75-1, Lane 9: ZR-75-30, Lane 10: negative control.



**Figure 4.38:** Restriction digestion of CYP4X1 RT-PCR with RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

RAR mRNA was detected in both the normal breast and breast cancer cell lines at the expected cDNA product size of 756 bp (Figure 4.39). Higher levels of RAR mRNA were observed in the T47D, ZR-75-1, MDA-MB-157 and MDA-MB-468 cells than in the MCF10A, MCF7, ZR-75-30 and MDA-MB-231 cells. To confirm the RAR mRNA expression, the PCR product was digested with restriction enzyme *Hae*III and the correct fragments were found at 456 bp, 193 bp and 107 bp (Figure 4.40).

A band of 521 bp corresponding to the cDNA product size for RXR mRNA was found in both the normal breast and breast cancer cell lines. The highest expression was detected in the ZR-75-1 cells, and RXR mRNA was detected at weak to moderate levels in the MCF10A, T47D and MDA-MB-157 cells. Very low mRNA levels were detected in the MCF7, ZR-75-30 MDA-MB-231 and MDA-MB-468 cells (Figure 4.41). To confirm the RXR mRNA expression, the PCR product was digested with restriction enzyme *Bam*HI and the correct fragments were found at 280 bp, 153 bp and 88 bp (Figure 4.42).



Figure 4.39: Visualisation of RAR RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: MCF10A, Lane 3: ZR-75-30, Lane 4: ZR-75-1, Lane 5: T47D, Lane 6: MCF7, Lane 7: MDA-MB-468, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: negative control.



**Figure 4.40:** *Restriction digestion of RAR RT-PCR with HaeIII. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 



**Figure 4.41:** *RXR RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468.* 



**Figure 4.42:** Restriction digestion of RXR RT-PCR using BamHI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

### 4.4 Discussion

Results presented in this chapter provide a comprehensive profile of cytochrome P450 mRNA expression in a series of seven breast cancer cell lines as compared with the expression profile in a normal breast cell line using RT-PCR. Cytochrome P450regulatory nuclear receptors were also investigated to explore the relationship between expression of cytochrome P450 mRNA and corresponding receptors. In research presented here, the qualitative comparative analysis of target gene expression between human normal breast and breast cancer cell lines was determined. The expression of CYP1A1, CYP1B1, CYP2C18, CYP2D6, CYP2R1, CYP2U1, CYP4X1, CYP4Z1 CYP24A1 and CYP26B1 mRNA was detected in both the normal breast and individual breast cancer cell lines. CYP3A7 mRNA was expressed only in the normal breast cell line, while CYP2A6, CYP2B6, CYP2C8, CYP2F1, CYP2J2, CYP2S1 and CYP26A1 mRNA were found only in individual breast cancer cell lines. The findings demonstrate differential mRNA expression of individual cytochrome P450s between normal breast and breast cancer cells, and some cell line-specific expression of individual cytochrome P450s. The findings showed seven out of the ten cytochrome P450 corresponding receptors (i.e. HIF-1 $\alpha$ , AhR, ARNT, AhRR, VDR, RAR and RXR) were expressed in both the normal breast and breast cancer cells while the expression of CAR, PXR and ER mRNA was detected only in individual breast cancer cell lines (Table 4.1).

The relationship between candidate mRNA profile and breast cancer cell lines from different ethnic groups (i.e. Caucasian and Afro-Caribbean) with oestrogen status (i.e. ER-positive and ER-negative) was also explored. CYP1B1, CYP2D6, CYP2J2, CYP2R1, CYP2U1, CYP4X1 and CYP26A1 mRNA were found in all cell lines. The expression of CYP1A1, CYP2S1, CYP24A1 and CYP26B1 mRNA was not detected in the ER-positive Afro-Caribbean breast cancer cell line (ZR-75-30) and no CYP2B6 expression was observed in the ER-negative Caucasian breast cancer cell line (MDA-MB-231). CYP2A6

164

CYP2C18 (MCF7, T47D, MDA-MB-468 and MDA-MB-157), CYP2F1 (ZR-75-1 and MDA-MB-468) and CYP4Z1 (T47D, ZR-75-1 and MDA-MB-468) mRNA were expressed in the ER-positive Caucasian and ER-negative Afro-Caribbean breast cancer cell lines. Although no differences in cytochrome P450 mRNA expression were observed between the different ethnic groups, these preliminary results suggest potential similarities in the ER-positive Caucasian and ER-negative Afro-Caribbean breast cancer cells which warrant further investigation.

In addition to comparing cytochrome P450 expression with ethnicity (i.e. Caucasian and Afro-Caribbean) and ER status, the relationship between ethnic difference and invasive phenotype was also examined. CYP1B1, CYP2D6, CYP2J2, CYP2S1, CYP2R1, CYP2U1, CYP4X1 mRNA were found in all cell lines. CYP1A1, CYP2A6, CYP2C8 and CYP24A1 mRNA were not detected in the non-invasive Afro-Caribbean breast cancer cell lines. CYP26A1 mRNA was below the levels of detectability in only the normal breast MCF10A and ZR-75-1 ER-positive breast cancer cell lines. CYP26B1 mRNA was below limits of detectability only in the ER-positive ZR-75-30 breast cancer cell line. No cytochrome P450 mRNA was specific for the invasive phenotype, but the expression of CYP4Z1 mRNA appears to be specific for the non-invasive phenotype. CYP2F1 was observed in only two out of the eight cell lines both of the non invasive phenotype. No CYP2B6 or CYP2C18 expression was observed in the invasive Caucasian breast cancer cells. Although of a preliminary nature these results suggest an inverse relationship of cytochrome P450 expression (i.e. CYP2A6 and CYP2C8 mRNA) between ethnic group and invasive phenotype, particularly the non-invasive Caucasian and invasive Afro-Caribbean breast cancer cell lines. These preliminary findings warrant further investigation potentially in relation to patient samples.

**Table 4.1:** The presence of cytochrome P450 and cytochrome P450-regulatory nuclear receptor mRNA inbreast cancer cell lines compared to the normal breast cell line.

ER status	ER-positive					ER-negative						
Ethnicity	Cau	Cau	Cau	Cau	Afro	Cau	Afro	Afro				
Invasive	NBC	Non-inv	Non-inv	Non-inv	Non-inv	Inv	Inv	Non-inv				
phenotype												
Cell lines	MC10A	MCF7	T47D	ZR-75-1	ZR-75-30	MDA-MB-	MDA-MB-	MDA-MB-				
						231	157	468				
Cytochrome P450 mRNA												
CYP1A1*	+	+	+	+	-	+	+					
CYP 1A2	NS	NS	NS	NS	NS	NS	NS	NS				
CYP1B1	+	+	+	+	+	+	+	+				
CYP2A6	-	-	+	+	-	-	+	-				
CYP2B6	-	-	+	+	+	-	+	+				
CYP2C8	-	+	-	+	-	-	+	-				
CYP2C18	+	+	+	-	-	-	+	+				
CYP2C9	-	-	-	-	-	-	-	-				
CYP2C19	-	-	-	-	-	-	-	-				
CYP2D6*	+	+	+	+	+	+	+	+				
CYP2F1*	-	-	-	+	-	-	-	+				
CYP2J2*	+	+	+	+	+	+	+	+				
CYP2R1*	+	+	+	+	+	+	+	+				
CYP2S1	-	+	+	+	-	+	+	+				
CYP2U1*	+	+	+	+	+	+	+	+				
CYP3A4*	-	-	-	-	-	-	-	-				
CYP3A5*	-	-	-	-	-	-	-	-				
CYP3A43	-	-	-	-	-	-	-	-				
CYP3A7	+	-	-	-	-	-	-	-				
CYP4X1*	+	+	+	+	+	+	+	+				
CYP4Z1	+	-	+	+	-	-	-	+				
CYP24A1	+	+	+	+	-	+	+	-				
CYP26A1	-	+	+	-	+	+	+	+				
CYP26B1	+	+	+	+	-	+	+	+				
Cytochrome P450-regulatory nuclear receptor mRNA												
HIF-1α	+	+	+	+	+	+	+	+				
AhR	+	+	+	+	+	+	+	+				
ARNT	+	+	+	+	+	+	+	+				
AhRR	+	+	+	-	+	+	+	+				
CAR*	-	+	+	+	+	+	+	+				
PXR*	-	+	+	+	-	-	+	-				
VDR	+	+	+	+	-	+	+	-				
ER	-	+	+	+	+	-	+	+				
KAK* DVD*	+	+	+	+	+	+	+	+				
КАК"	+	+	+	+	+	+	+	+				

Note: Blue and yellow labelled indicate the detected and undetected mRNA in all cell lines respectively.

Abbreviations: ER; oestrogen receptor, Cau; Caucasian, Afro; Afro-Caribbean, NS; non-specific product, \*; published primers, NBC; normal breast cell line, Inv; invasive phenotype

The CYP1 family comprises three enzymes, CYP1A1, CYP1A2 and CYP1B1, and they are involved in the metabolism of oestrogen as well as procarcinogens (e.g. PAHs). Research conducted by several groups, with both tissues and cell lines, suggests an association between the roles of these enzymes in oestrogen metabolism and breast carcinogenesis (Singh et al. 2005, Tsuchiya et al. 2005, Cavalieri et al. 2006, Fernandez et al. 2006, Yager and Davidson 2006).

The expression of CYP1A1 mRNA has been detected in normal breast tissue (Huang et al. 1996, Hellmold et al. 1998, Iscan et al. 2001, Modugno et al. 2003) and breast tumours (Huang et al. 1996, Iscan et al. 2001, Modugno et al. 2003). Hellmold and co-workers showed CYP1A1 protein to be present in (5/15) normal breast (Hellmold et al. 1998). El-Rayes and colleagues reported that CYP1A1 protein was significantly higher (P < 0.05) in normal adjacent breast tissues than in breast tumours by immunoblotting (El-Rayes et al. 2003). However, these findings are in disagreement with those of Albin and colleagues who did not detect CYP1A1 protein in breast tumours and normal adjacent breast tissues using the same method (Albin et al. 1993).

In the present study, CYP1A1 mRNA was found in the normal breast MCF10A cells which is in agreement with previous studies (Spink et al. 1998, Thomas et al. 2006). The expression of CYP1A1 mRNA was detected in all breast cancer cell lines except for the ER-positive Afro-Caribbean ZR-75-30 and ER-negative Afro-Caribbean MDA-MB-468 breast cancer cells. The finding that CYP1A1 mRNA is present in the MCF7, ZR-75-1, T47D and MDA-MB-231 cells correlates well with results from a study by Spink and colleagues who investigated CYP1A1 mRNA using Northern hybridisation (Spink et al. 1998). The present investigation also found CYP1A1 mRNA in the MDA-MB-157 cells, though this was not detected in the previous study (Spink et al. 1998). The lack of CYP1A1 expression in the MDA-MB-468 cells is in agreement with an earlier

167
investigation (Wang et al. 1997). However, the expression of CYP1A1 mRNA in the ZR-75-30 cells has not been previously reported. In addition, McFadyen and co-workers reported that the MCF7 cells express CYP1A1 mRNA but did not express the protein (McFadyen et al. 2003). These observations suggest that CYP1A1 protein expression in these cell lines warrants further investigation to clarify the functional significance of the CYP1A1 expression.

The association between the CYP1A1 polymorphisms and cancer susceptibility, lung cancer in particular, have been demonstrated to be dependent upon ethnicity of population. However, investigation into a possible relationship between the CYP1A1 alleles and breast cancer risk has generated inconsistent results and this remains unclear (Masson et al. 2005). Preliminary findings from the present study suggest that CYP1A1 is expressed in the majority of cell lines apart from Afro-Caribbean cell lines of the non-invasive phenotype.

Several breast cancer research groups have demonstrated the expression of CYP1B1 mRNA in normal breast (Huang et al. 1996, Hellmold et al. 1998, Iscan et al. 2001) and breast tumour tissues (Huang et al. 1996, Iscan et al. 2001). A higher level of CYP1B1 mRNA was found in breast tumours than in normal breast tissue (Modugno et al. 2003). In addition, previous studies demonstrated that CYP1B1 protein was overexpressed in breast tumour samples and undetectable in normal breast tissue (Murray et al. 1997, McFadyen et al. 1999). On this basis, tumour-specific CYP1B1 expression appears to be a potential target for the development of new anticancer treatments and a CYP1B1 inhibitor and CYP1B1 vaccine are currently in clinical trials (Section 1.8). In this research, CYP1B1 mRNA was detected in the normal breast MCF10A cells and the seven breast cancer cell lines. No difference in CYP1B1 mRNA expression was observed between the different ER status, ethnic groups or invasive phenotype. However, it is essential to confirm CYP1B1 protein expression in these cell lines before it can be verified as a target for the development of new anticancer drugs.

The CYP1A1 and CYP1B1 genes are regulated by the aryl hydrocarbon receptor (AhR) and induced by AhR agonists such as TCDD and benzo(a)pyrene (Iwanari et al. 2002). The AhR belongs to the basic helix-loop-helix/Per-ARNT-Sim (bHLH-PAS) family which consists of the AhR dimerisation partners such as AhR nuclear translocator (ARNT), hypoxia inducible factor-1 alpha (HIF-1 $\alpha$ ) (Semenza 2000) and AhR repressor (AhRR) (Tsuchiya et al. 2003). These findings suggest that the AhR-mediated pathway may interact with other nuclear receptor signalling pathways which are associated with the expression of CYP1A1 and CYP1B1.

It is well known that CYP1A1 and CYP1B1 are involved in oestrogen metabolism and Angus and colleagues demonstrated that oestradiol can also induce the transcriptional expression of CYP1A1 and CYP1B1 (Angus et al. 1999). The induction of CYP1B1 by oestradiol has been demonstrated in breast cancer oestrogen-dependent (i.e. MCF7) and oestrogen-independent (i.e. MDA-MB-435) cells. An increased CYP1B1 mRNA level was found in the MCF7 cells but not in the MDA-MB-435 cells, indicating that the induction of CYP1B1 by oestradiol may be transcriptionally regulated by oestrogen receptor as previously demonstrated (Tsuchiya et al. 2004, Tsuchiya et al. 2005). In addition, Beischlag and colleagues (Beischlag and Perdew 2005) demonstrated that AhR agonist TCDD can increase the expression of CYP1A1 and that it activates inhibitory crosstalk between ER and the AhR-ARNT complex.

In the current project, expression of AhR, ARNT and HIF-1 $\alpha$  mRNA was found in all the test cells, which is similar to CYP1B1 mRNA but not CYP1A1 mRNA expression. Allan and co-workers found that decreased CYP1A1 expression under hypoxic conditions.

These findings may have been caused by competition for dimerisation with ARNT, leading to inhibition of target genes of AhR (i.e. CYP1A1), as dimerisation of AhR/ARNT and HIF-1 $\alpha$ /ARNT is required for transcription initiation in their target genes including CYP1A1 and CYP1B1 (Allen et al. 2005). These cell lines could be useful models to clarify their relationship in breast cancer cells.

Previous studies have failed to detect CYP1A2 mRNA expression in normal human breast or breast tumour tissues (Hellmold et al. 1998, Modugno et al. 2003). However, a recent study using immunohistochemistry detected CYP1A2 protein expression in 51 out of the 393 breast tumours tested (Haas et al. 2006). Although multiple bands were observed in the present study, no specific product at the correct size was identified for CYP1A2 mRNA in either the normal breast or breast cancer cell lines. The lack of CYP1A2 mRNA expression in the normal breast MCF10A cell line is in agreement with the recent study by Thomas and colleagues (Thomas et al. 2006). The lack of CYP1A2 mRNA expression is surprising as CYP1A2 is involved in the metabolism of oestradiol (Tsuchiya et al. 2005, Yager and Davidson 2006) and the catalysis of a wide range of procarcinogens e.g. heterocyclic amines, nitrosamine, aflatoxin B1 and arylamine (Kim and Guengerich 2005). It should be noted that CYP1A2 mainly catalyses flutamide (a nonsteroidal antiandrogenic anticancer drug) which is commonly used for prostate cancer treatment (Goda et al. 2006).

CYP2A6 mRNA expression was detected in (10/15) normal breast tissues (Hellmold et al. 1998) whereas Iscan and co-workers reported no mRNA expression of CYP2A6 in normal breast tissues or breast tumour tissues (Iscan et al. 2001). Using real-time quantitative RT-PCR analysis, Bieche and colleagues (Bièche et al. 2004) demonstrated that the CYP2A6 mRNA level in ER-positive breast tumours is higher than in either ER-negative breast tumours or normal breast tissue. As was observed in present investigation, the expression level of CYP2A6 was higher in the breast cancer ER-positive Caucasian ZR-75-1 and

T47D than in the ER-negative Afro-Caribbean MDA-MB-157. However, it is important to note that CYP2A6 expression was not observed in the invasive Caucasian MDA-MB-231 or the non-invasive Afro-Caribbean MDA-MB-468 breast cancer cells. No CYP2A6 mRNA was found in the normal breast MCF10A cells either, which correlates well with results from a study by Thomas and colleagues (Thomas et al. 2006).

The importance of CYP2A6 in the metabolism of anticancer drugs (i.e. letrozole) and procarcinogens (i.e. nitrosamine and food mutagens) has been emphasised, and it has been reported that this cytochrome P450 enzyme may play a role in the development of breast cancer (von Weymarn et al. 2006). CYP2A6 metabolises a third-generation aromatase inhibitor, letrozole, which is superior to tamoxifen for first-line treatment in breast cancer (Simpson et al. 2004). Therefore, the findings in the present study suggest that expression of CYP2A6 mRNA in the opposite oestrogen receptor status/invasive phenotype of Caucasian and Afro-Caribbean breast cancer cell lines could be of benefit in the design of more effective drug regimens in cancer patients.

The expression of CYP2B6 mRNA was found in (12/14) breast tumours and (13/14) normal breast tissues (Iscan et al. 2001), and Hellmond and co-workers detected CYP2B6 mRNA expression in (9/15) normal breast tissues (Hellmold et al. 1998). In the present study, CYP2B6 mRNA was observed in five out of the seven breast cancer cell lines but not in the normal breast MCF10A cells. The lack of CYP2B6 mRNA expression in the MCF7 and MCF10A cells is in agreement with the studies of Soulez and Parker (Soulez and Parker 2001), and Thomas and colleagues (Thomas et al. 2006) respectively. The presence of CYP2B6 mRNA in the ZR-75-1 cells is in agreement with the research of Soulez and Parker (Soulez and Parker 2001). The absence of CYP2B6 mRNA in the MDA-MB-231 cells, classified as invasive and ER-negative Caucasian breast cancer cells,

is important to note as CYP2B6 converts the prodrug cyclophosphamide to an active phospharamide mustard and acrolein (Zanger et al. 2007).

Targeting drugs at CYP2B6 is an attractive concept for cancer research as this would lead to local activation of the prodrug and higher concentrations of the active metabolites at the tumour site. MetXia is a recombinant retroviral vector encoding human CYP2B6 which is designed to activate cyclophosphamide. MetXia is currently in Phase I trial and being investigated in advanced breast cancer patients (Braybrooke et al. 2005).

The CYP2B6 gene is regulated by CAR (Honkakoski and Negishi 2000) and a positive correlation between CYP2B6 and CAR mRNA expression has been demonstrated in human liver (Chang et al. 2003). A study by Lamba and colleagues showed that ethnic and gender differences influence the expression of CYP2B6 and CAR (Lamba et al. 2003). The expression level of CYP2B6 mRNA was significantly higher in females than in males (p<0.001) and CYP2B6 activity in Hispanic females higher than in Caucasian (p<0.022) or African-American (p<0.038) females. These findings can be related to female hormones including oestrogen. However, Min and colleagues (Min et al. 2002) demonstrated inhibitory crosstalk between ER and CAR in human hepatoma HepG2 cell line and that CAR inhibits ER-mediated transactivation of endogenous and exogenous oestrogen. Recent research by the same group found that ER-mediated transactivation can be regulated by CAR which is dependent on the oestrogen response elements (EREs) and oestrogen receptor status of target cells (Min 2006).

In research presented here, a direct correlation between CYP2B6 and CAR mRNA was observed in the ER-positive Caucasian T47D and the ER-negative Afro-Caribbean MDA-MB-157 and MDA-MB-468 breast cancer cells. Further investigation of the expression of CYP2B6 protein in these cell lines is warranted prior to determination in clinical samples. The expected product size of CAR is 232 bp following nested PCR, but multiple bands are seen in lanes 5, 7, 8 and 10 (Figure 4.25). The presence of non-specific products may be due to too much template (first round PCR product) and/or insufficient primer. This problem can be resolved by diluting the concentration of template. However, during the validation of CAR in this research, the PCR product was confirmed by restriction digestion and the expected products were clearly seen (Figure 3.36).

There are four members of the CYP2C subfamily CYP2C8, CYP2C9, CYP2C18 and CYP2C19. They are highly homologous genes and are clinically relevant in anticancer drug treatments. In cancer chemotherapy, CYP2C8 catalyses the metabolism of paclitaxel (Taniguchi et al. 2005) whereas CYP2C9 and CYP2C19 catalyse the bioactivation of cyclophophamide and ifosfamide (Goldstein 2001, Schmidt et al. 2004) as well as the metabolism of tamoxifen. CYP2C18 is involved in the activation of ifosfamide (Zhu-Ge and Yu 2004). Modugno and colleagues reported that the expression of CYP2C mRNA was not detected in normal breast tissues or breast tumours (Modugno et al. 2003). In contrast, previous studies showed that CYP2C mRNA expression was found in normal breast as well as breast tumour tissues (Huang et al. 1996, Iscan et al. 2001). In addition, CYP2C protein was expressed in normal breast tissues (Hellmold et al. 1998). These findings disagreed with those of Albin and colleagues who did not detect CYP2C in normal breast or breast tumours (Albin et al. 1993). All four CYP2C isoforms were investigated in the current study. Unfortunately, no CYP2C9 or CYP2C19 mRNA was detected in the eight cell lines, while the expression of CYP2C8 and CYP2C18 mRNA was detected in the ER-positive Caucasian and ER-negative Afro-Caribbean breast cancer cell lines. These preliminary findings suggest potential ethnic differences in the relationship between the expression of CYP2C8/18 mRNA and ER status should be taken into consideration in the development of novel cancer therapeutics. CYP2C18 mRNA

173

expression was also observed in the normal breast MCF10A cells, a result which differs from the finding of Thomas and colleagues (Thomas et al. 2006).

CYP2D6 catalyses the N-demesthylation and 4-hydroxylation of tamoxifen which have potent antioestrogen effects. Tamoxifen is widely used in the treatment of breast cancer (Coller et al. 2002). Previous studies demonstrated that CYP2D6 mRNA was expressed in normal breast as well as breast tumour samples (Huang et al. 1996, Hellmold et al. 1998, Iscan et al. 2001, Bièche et al. 2007). In addition, Hellmond and colleagues demonstrated CYP2D6 protein expression in normal breast (Hellmold et al. 1998). These findings are in disagreement with those of Modugno and co-workers, who did not detect CYP2D6 mRNA expression in breast tumour and normal adjacent breast tissue using quantitative RT-PCR (Modugno et al. 2003). The present study is the first to examine the expression of CYP2D6 mRNA in breast cancer cell lines. An initial experiment using the primer sequence designed for CYP2D6 by Primer3 (Table 2.6) failed to detect CYP2D6 mRNA in the test cell lines. However, the previously published CYP2D6 primer set (Table 2.7) was employed to confirm the expression of CYP2D6 mRNA and this primer set successfully demonstrates a cDNA product of CYP2D6 mRNA in the eight cell lines.

In the preliminary work, CYP2D6 mRNA was detected in all four ER-positive breast cancer MCF7, T47D, ZR-75-1 and ZR-75-30 cells, suggesting that a local tamoxifen metabolism mediated by CYP2D6 in these cells may be of particular interest in the development of hormonal therapy for breast cancer. It is known that approximately 35% of ER-positive breast cancer patients do not respond to tamoxifen therapy and some patients who do respond to tamoxifen eventually become resistant to this treatment (Osborne 1998, Jin et al. 2005).

174

In addition, preliminary results showed that the presence of CYP2D6 mRNA is detected with the appearance of smearing bands in the two Afro-Caribbean breast cancer cells lines MDA-MB-468 and ZR-75-30 (Figure 4.27, lanes 2 and 9). This is likely to be related to the presence of a splice variant of this gene as currently more than eighty alleles have been identified for the CYP2D6 gene. Variant alleles of the CYP2D6 gene may lead to increased, decreased or absent functionality of the CYP2D6 gene (Klein et al. 2007). The activity of CYP2D6 is extremely variable, and inter-ethnic variations in drug metabolisms are well recognised. Poor metabolisers (PMs) lacking CYP2D6 activity have been found in 7% of Caucasians and 1-3% of other ethnic groups, whereas ultra-rapid metabolisers (UMs) displaying unusually high activity of CYP2D6 have been observed in 29% of people from North Africa and the Middle East (de Leon et al. 2006a). Unfortunately, information on the specific country of origin of the breast cancer cells used in the current study is not available. However, these findings suggest that the breast cancer cell line models may be of particular relevance in the development of cancer therapy for Afro-Caribbean breast cancer patients.

Unlike the MDA-MB-468 and ZR-75-30 cells, CYP2D6 mRNA was observed with smearing in the normal breast MCF10A cells derived from a Caucasian woman. This data is in agreement with the original study (Thomas et al. 2006) using the same primers but Thomas and co-workers detected a fainter expression. This difference may be due to the use of an annealing temperature and concentration of MgCl<sub>2</sub>. The current project used a lower annealing temperature  $(55^{\circ}C)$  and 1.5 mM MgCl<sub>2</sub> whereas Thomas and co-workers employed an annealing temperature at  $60^{\circ}C$  and 5 mM MgCl<sub>2</sub>.

CYP2F1 is predominantly expressed in human lung, with low level or no expression being detected in the liver or other extrahepatic tissues (Carr et al. 2003, Tournel et al. 2007a). CYP2F1 is involved in the activation of some pneumotoxicants including naphthalene and

3-methylindole (Kartha and Yost 2008). The current project is the first to examine expression of CYP2F1 mRNA in a series of breast cancer cell lines. No CYP2F1 mRNA was observed in the normal breast MCF10A cells whereas the original study demonstrated weak expression of CYP2F1 in sub-confluent but not confluent MCF10A cells (Thomas et al. 2006). This finding is in disagreement with preliminary work from the present investigation (see discussion in following section; CYP2S1).

CYP2F1 was detected only in the ER-positive Caucasian ZR-75-1 cells and the ERnegative Afro-Caribbean MDA-MB-468 cells. Recently, Tournel and colleagues (Tournel et al. 2007b) demonstrated inter-ethnic variations in CYP2F1 genetic polymorphism, and results from their studies suggest that a higher frequency of poor metabolisers occur in Afro-Caribbean populations (i.e. Gabonese and Senegalese) compared to French Caucasians and Tunisians. Although this association was derived from healthy volunteers, it may be of interest to assess inter-ethnic variation in cancer patients including those with breast cancer.

CYP2J2 is expressed in various normal tissues (i.e. liver, small intestine, colon and kidney) but higher levels occur in the coronary endothelial cells and cardiac myocytes (Wu et al. 1996). CYP2J2 is involved in the metabolism of arachidonic acids to form epoxyeicosatrienoic acids (EETs) which are which are potent endogenous vasodilators and inhibitors of vascular inflammation (Node et al. 1999, Scarborough et al. 1999). The physiological function of CYP2J2 in xenobiotic metabolism remains unknown.

Preliminary work showed high expression of CYP2J2 mRNA in all breast cancer cell lines but at a very low level in the invasive Caucasian breast cancer MDA-MB-231 cells. Weak expression of CYP2J2 mRNA was detected in the normal breast MCF10A cells which is in agreement with the original investigation (Thomas et al. 2006) using this primer set. CYP2J2 mRNA in breast cancer cell lines has not been previously reported, but Jiang and co-workers detected higher levels of CYP2J2 mRNA and protein in breast cancer than in their corresponding normal tissues (Jiang et al. 2005). These findings suggest that CYP2J2 mRNA is specifically expressed in breast cancer cells and may be involved in breast carcinogenesis. It is important to further investigate the function of CYP2J2 in these cell lines prior to using them as models in breast cancer research.

Many research groups reported the expression of CYP2S1 mRNA and protein in several extrahepatic tissues (Choudhary et al. 2005, Karlgren et al. 2005, Saarikoski et al. 2005, Rivera et al. 2007), but it has not been previously reported in normal breast tissues or breast tumour samples. One previous study has demonstrated CYP2S1 mRNA in sub-confluent normal breast MCF10A cells but not in confluent cultures (Thomas et al. 2006), and this finding is in disagreement with the present study. The MCF10A cells used in this project were grown under the same conditions as the previous study (Thomas et al. 2006), and approximately 70-80% confluent MCF10A culture was employed to determine CYP2S1 mRNA. The observed differences in CYP2S1 mRNA expression in MCF10A between the two research groups could be attributable to differences in PCR conditions (i.e. RNA isolation method, primer sequences and MgCl<sub>2</sub> concentration) and primer sequences used. The present study is the first to report expression of CYP2S1 mRNA in a series of breast cancer cell lines, and it was detected in all cancer cell lines except for the ER-positive Afro-Caribbean ZR-75-30 cells.

CYP2S1 has several characteristics similar to members of the CYP1 family (i.e. CYP1A1 and CYP1B1). A recent study has reported that induction of CYP2S1 mRNA by dioxin and hypoxia is mediated by AhR/ARNT and HIF-1 $\alpha$ /ARNT complex pathway respectively (Rivera et al. 2007). In addition, CYP2S1 is known to be induced by retinoic acid (Smith et al. 2003), naphthalene (Karlgren et al. 2005) and PAHs (Rivera et al. 2002). There is no

information available on crosstalk between these receptors and CYP2S1 expression in breast cancer cells. However, the current study demonstrated the presence of AhR, ARNT, HIF-1 $\alpha$  mRNA in all the test cell lines and this could facilitate a better understanding of the role of CYP2S1 in cancer development and cancer treatment, following confirmation of CYP2S1 protein expression in these cell lines.

Previous studies demonstrated that CYP2S1 protein was associated with poor prognosis of colorectal tumours (Kumarakulasingham et al. 2005), and an increased expression of CYP2S1 protein was found in metastasis disease compared with primary ovarian cancer (Downie et al. 2005). In the present study, CYP2S1 mRNA was detected in the ER-positive/negative and the non-invasive/invasive phenotype.

When detecting CYP2S1 by RT-PCR, a faint band was observed above 216 bp in lane 5 (Figure 4.10). The presence of this non-specific amplification product may have been caused by excessive template cDNA. Importantly though, the expected product size of 216 bp representing CYP2S1 mRNA was seen as a clear band, and this was confirmed by restriction digest (Figure 3.18) and DNA sequencing (Figure D.1) during the validation step in this study.

Another novel exptrahepatic cytochrome P450, CYP2U1, is highly expressed in human thymus and brain (Karlgen et al. 2004). Chuang and colleagues demonstrated that CYP2U1 plays a role in the metabolism of arachidonic acid and other long chain fatty acids (Chuang et al. 2004). CYP2U1 converts arachidonic acid into 19- and 20hydroxyeicosatetraenoic acids (HETEs). The 20-HETE is potent vasoconstrictor in the cerebral circulation, and this indicates that CYP2U1 may control the brain blood flow (Karlgren et al. 2005). In addition, an association between the expression of CYP2U1 and tumour tissues has been shown by two studies and the findings suggest the same trend. Currently, the role of CYP2U1 in the metabolism of xenobiotic compounds remains unknown. CYP2U1 protein in ovarian and colorectal tumours was higher than in normal ovary (Downie et al. 2005) and colon tissues (Kumarakulasingham et al. 2005) respectively. This may indicate that CYP2U1 plays a role in the carcinogenic process in certain types of tumour.

To date, CYP2U1 expression in breast cancer cells has not been demonstrated. Preliminary results from the current study showed that CYP2U1 mRNA was detected in both the normal breast and breast cancer cell lines. CYP2U1 mRNA in the MCF10A cells confirmed the findings of a previous study (Thomas et al. 2006). With the exception of the ZR-75-30 cells, this enzyme was expressed at a lower level in MCF10A than in breast cancer cell lines. Preliminary results suggest that CYP2U1 mRNA in the breast cancer cells relative to the normal breast cells shows a similar trend in ovarian and colorectal cancer as mentioned above. The breast cancer cell line models could potentially be exploited in the development of novel cancer therapeutics and CYP2U1 could be used as a prognostic biomarker for breast cancer. This warrants further investigation.

The human CYP3A subfamily comprises four enzymes, CYP3A4, CYP3A5, CYP3A7 and CYP3A43. CYP3A4 plays an important role in the metabolism of steroid hormones (i.e. oestrogen and glucocorticoid) (Guengerich 1999) as well as xenobiotic compounds (i.e. paclitaxel, cyclophosphamide and doxorubicin) (McFadyen et al. 2004, Michael and Doherty 2005). The CYP3A genes are induced by glucocorticoids, phenobarbital and vitamin D (Pascussi et al. 2003). They are regulated by pregnane X receptor (PXR) which binds DNA as heterodimer with retinoid X receptor (RXR) and activates gene transcription. However, Pascussi and colleagues report that CYP2B6, CYP2C8 and CYP2C9 are also regulated by PXR. These observations suggest that the interaction

among PXR, CAR and VDR may occur after heterodimerisation with RXR as they share DNA response elements, partners and target genes (Pascussi et al. 2003).

Many research groups have investigated the expression of CYP3A in breast tumour tissues, but their findings seem to be inconsistent. For example, Iscan and colleagues found no CYP3A4 or CYP3A5 or CYP3A7 mRNA in normal breast or breast tumours (Iscan et al. 2001), whereas Huang and co-workers reported that CYP3A4 or CYP3A5 mRNA was higher in non-tumour breast tissues than in breast tumour samples (Huang et al. 1996). Hellmond and colleagues demonstrated the presence of CYP3A mRNA in normal breast too but they did not detect CYP3A protein in these tissues (Hellmold et al. 1998). These finding correlate well with those of Albin and co-workers who detected no CYP3A4 protein in breast tumours or normal adjacent breast tissues (Albin et al. 1993). Modugno and colleagues demonstrated CYP3A5 mRNA in normal breast as well as in breast tumour tissues, but they did not detect CYP3A5 mRNA in normal breast as well as in breast tumour tissues (Modugno et al. 2003). The CYP3A43 gene is the most recently discovered in the CYP3A subfamily (Domanski et al. 2001) and the expression of CYP3A43 in breast cells has not been previously reported.

The present study failed to detect any members of the CYP3A subfamily in the panel of breast cancer cell lines. The lack of CYP3A mRNA expression is surprising as the CYP3A enzymes are involved in the metabolism of oestradiol and the activation of procarcinogens (e.g. aflatoxin B1, PAHs and NNK), CYP3A4 in particular (Guengerich 1999). However, a weak expression of CYP3A7 mRNA was found in the normal breast MCF10A cells, a finding which is in accordance with the results of Thomas and co-workers who investigated CYP3A mRNA in the MCF10A cells (Thomas et al. 2006).

180

CYP3A7 is a human foetal liver P450 isoform which plays a role in the hydroxylation of retinoic acid and the  $16\alpha$ -hydroxylation of steroids, suggesting it may be relevant to human normal development and hormone dependent cancer (Daly 2006). However, a previous study showed that a high level of mRNA expression was detected in well differentiated hepatocellular carcinoma (Tsunedomi et al. 2005) and its protein expression was found to be associated with primary ovarian cancer when compared to normal ovary tissues (Downie et al. 2005).

Transcriptional expression of the CYP3A genes is regulated by PXR. The induction of CYP3A4 by PXR has been shown to play a crucial role in endometrial cancer. PXR ligands induce PXR expression, resulting in altered regulation of PXR target genes including CYP3A4 (Masuyama et al. 2003, Masuyama et al. 2005). In addition, a recent report claimed that down-regulation of PXR in the presence of the anticancer drug paclitaxel increased the inhibition of cell growth and apoptosis, leading to a higher sensitivity to anticancer drugs and the overcoming of drug resistance in the treatment of endometrial cancer (Masuyama et al. 2007). Although the present study did not detect CYP3A mRNA in the breast cancer cell lines used, it should be noted that PXR mRNA was found in the MCF7, T47D, ZR-75-1 and MDA-MB-157 cells. These findings suggest that the regulation of CYP3A genes in breast cancer cells may be involved in other signalling pathways rather than the PXR signalling pathway.

CYP24A1 (1,25-(OH)<sub>2</sub>D<sub>3</sub>-24-hydroxylase) is involved in the inactivation of vitamin D<sub>3</sub> and is regulated by vitamin D receptor (VDR). Recently, overexpression of CYP24A1 mRNA has been demonstrated in a variety of tumour tissues such as breast, prostate and lung (Cross et al. 2003). In addition, CYP24A1 mRNA has been shown to be associated with poor prognosis of oesophageal cancer by Mimori and co-workers (Mimori et al. 2004). These findings suggested that CYP24A1 expression plays a role in the carcinogenic

181

process and that CYP24A1 is a candidate oncogene (Anderson et al. 2006). Consequently, an inhibitor of CYP24A1 (EB1089) which has been demonstrated to have an antiproliferative effect on breast cancer has been developed and is currently in clinical trials (Sundaram et al. 2006).

Using real-time quantitative RT-PCR analysis, Fisher and co-workers detected the expression of CYP24A1 wild-type and splice variants in the normal breast MCF10A and breast cancer MCF7 cells; though an association between splice variants and CYP24A1 activity was not clearly established (Fisher et al. 2007). The detection of CYP24A1 mRNA in the MCF10A and MCF7 cells correlates well with results from the present study. CYP24A1 mRNA was detected in all the test cell lines except for the ER-positive ZR-75-30 and ER-negative MDA-MB-468 Afro-Caribbean breast cancer cell lines. It is important to note that ER status differences in CYP24A1 expression may be of particular relevance to the use of an inhibitor of CYP24A1 in breast cancer treatment. Regarding qualitative comparison, a correlation between CYP24A1 and VDR mRNA was observed in this research. The absence of CYP24A1 mRNA (Figure 4.16) agrees with the lack of VDR mRNA (Figure 4.23) in the ZR-75-30 and MDA-MB-468 cells.

In addition to metabolism of vitamin  $D_3$  by cytochrome P450 enzymes, the conversion of the vitamin  $D_3$  precursor into the active metabolite 1,25-(OH)<sub>2</sub> $D_3$  is mediated by CYP2R1 and CYP27A1 which are predominantly expressed in human liver, and by CYP27B1 which is mainly expressed in the kidney (Cheng et al. 2003). Vitamin  $D_3$  is believed to exhibit potent anticancer properties as it inhibits proliferation and stimulates apoptosis of cells. For this reason, enzymes involved in the activation and inactivation of vitamin  $D_3$ are of particular relevance to cancer therapy. To date no information on the physiological role of CYP2R1 in xenobiotic metabolism has been reported. CYP2R1 mRNA is highly expressed in human liver and testis but lower levels are found in other tissues (Karlgren et al. 2005).

The preliminary work presented in this chapter is the first evidence of CYP2R1 mRNA expression in a panel of breast cancer cell lines compared with the normal breast cells. CYP2R1 mRNA in the normal breast MCF10A cells was also observed in this work and is in agreement with the original study (Thomas et al. 2006) using these primers. CYP2R1 mRNA was highly expressed in almost all breast cancer cell lines but lower levels were observed in the Afro-Caribbean breast cancer MDA-MB-468 and ZR-75-30 cells. Regarding the breast cancer cells derived from Caucasian patients, CYP2R1 mRNA seems to be expressed at similar levels both in ER-positive and ER-negative phenotype, suggesting that CYP2R1 may be of interest as a potential target for chemotherapy in either an oestrogen-independent or an oestrogen-dependent breast tumour once protein expression is confirmed in these cell lines.

Members of the CYP26 family (i.e. CYP26A1 and CYP26B1) specifically catalyse the 4hydroxylation of all *trans*-retinoic acid (ATRA) (Sonneveld et al. 1998). CYP26A1 is overexpressed in various tumours including breast tumours (Mira-Y-Lopez et al. 2000) and these findings have led to the use of a CYP26A1 inhibitor (R116010) in oestrogenindependent mouse breast tumours (Njar 2002). Previous studies demonstrated CYP26A1 mRNA in ER-positive T47D (Sonneveld et al. 1998) and MCF7 (van Heusden et al. 1998, Mira-Y-Lopez et al. 2000) and these findings are in agreement with the present study. The current study also detected CYP26A1 in all the cancer cell lines except the ZR-75-1 cells, suggesting that CYP26A1 may be specifically expressed in breast cancer cells and that the test cell lines could be used as models to investigate novel anticancer drugs. Earlier studies have shown that CYP26B1 is predominantly expressed in human adult brain, particularly the cerebellum and cephalic tissues (White et al. 2000, Trofimova-Griffin and Juchau 2002). However, there is no information on CYP26B1 mRNA in normal breast or breast tumour tissues. The current investigation is the first to screen a panel of breast cancer cell lines and a normal breast cell line for the presence of CYP26B1 mRNA. The presence of CYP26B1 mRNA was detected in the normal breast MCF10A cells and all breast cancer cell lines except the ZR-75-30 cells. It is important to note that CYP26B1 mRNA expression was elevated in the normal breast and ER-negative breast cancer cells compared to the ER-positive breast cancer cell lines, suggesting that oestrogen may affect the expression of CYP26B1 mRNA.

The CYP26 genes are regulated by nuclear receptor retinoic acid receptor (RAR) and retinoid X receptor (RXR) which bind to RA response elements and regulate transcription of target genes (Nezzar et al. 2007). RXR is a common heterodimeric partner for several nuclear receptors (i.e. CAR, PXR, PPAR $\alpha$  and VDR) and crosstalk between these receptors leads to unexpected expression of cytochrome P450 enzymes (Section 1.7.9). In the present study, the expression of RAR and RXR mRNA was detected in both the normal breast and breast cancer cell lines; however, no correlation between RAR and RXR mRNA was observed regarding ER status, invasive phenotype or ethnicity. These findings suggest that any applications of the inhibition of CYP24 or CYP26 in cancer therapy must consider a significant function of their corresponding nuclear receptors and possible crosstalk between nuclear receptors in the same superfamily.

Transcriptional expression of some cytochrome P450 genes is regulated by their corresponding nuclear receptors (Section 1.7). However, the existence of intracellular crosstalk between these receptors appears to have important consequences on cytochrome P450 gene regulation as the receptors share heterodimeric partner, ligands, DNA response elements and target genes. A study of coexpression of cytochrome P450s and their corresponding nuclear receptors in breast cancer cells may lead to a better understanding of how the event alters the cancer biology.

CYP4X1 mRNA is widely expressed in several human tissues (i.e. breast, ovary, uterus, liver and kidney, trachea and aorta). Savas and colleagues demonstrated that peroxisome proliferator activated receptor-alpha (PPAR $\alpha$ ) is involved in the activation of human CYP4X1, as investigated through used of human hepatoma HepG2 cells that stably express a mouse PPAR $\alpha$  mutant. A physiological role of CYP4X1 is currently unknown but the consistently high levels of CYP4X1 in the aorta and trachea may suggest that CYP4X1 plays a role in the generation of arachidonic acid metabolites, compounds with vasoactive properties (Savas et al. 2005). Preliminary results showed that CYP4X1 mRNA was expressed in both the normal breast and breast cancer cell lines. The detection of CYP4X1 mRNA in the normal breast MCF10A cells is in agreement with a previous study (Thomas et al. 2006) using these primers, but it is more likely to be overexpressed compared with those of Thomas and co-workers. The expression of CYP4X1 mRNA has not been previously investigated in breast cancer cells. CYP4X1 mRNA was expressed in the seven breast cancer cell lines but at very high levels in the ER-positive Caucasian breast cancer T47D and the ER-negative Afro-Caribbean breast cancer MDA-MB-468 cells. This finding warrants further investigation.

Using real-time quantitative RT-PCR, Savas and co-workers reported that CYP4Z1 mRNA was higher in normal breast tissue than other human normal tissues (Savas et al. 2005). The study of Rieger et al. (2004) demonstrated that CYP4Z1 mRNA was overexpressed in breast tumours compared with the corresponding normal breast tissue. In addition, a previous study demonstrated CYP4Z1 in the normal breast MCF10A cells and this is in agreement with results from the present study (Thomas et al. 2006). In breast cancer cell

lines, CYP4Z1 mRNA was detected in the MCF7 (Rieger et al. 2004, Savas et al. 2005) and T47D cells (Savas et al. 2005). The detection of CYP4Z1 mRNA in the T47D cells is in agreement with results from the present study, though this is not the case for the MCF7 cells. This discrepancy may be attributable to the different conditions used for cell culture and the sensitivity of the RT-PCR method. However, results from studies examination the expression of CYP4Z1 protein in breast cancer cells generally appear to be inconsistent. Rieger and colleagues detected CYP4Z1 protein in MCF7 cells (Rieger et al. 2004) but Savas and co-workers reported no CYP4Z1 protein expression in MCF7 (Savas et al. 2005).

The present study also demonstrated CYP4Z1 mRNA in the ZR-75-1 and MDA-MB-468 cells and this has not been previously reported. These findings suggest that CYP4Z1 mRNA is more likely to be specifically expressed in the non-invasive phenotype of breast cancer cells. A higher level of CYP4Z1 expression was observed in the ER-negative Afro-Caribbean MDA-MB-468 breast cancer cell line than in the ER-positive Caucasian T47D and ZR-75-1 breast cancer cells and this warrants further investigation.

Prior to further elucidation of the sequence, the present investigation confirmed the expression of CYP4Z1 mRNA in all the cDNA samples, using previously published primers (Thomas et al. 2006) that had been designed to cross between exon 9 and exon 12 boundaries (Figures 4.14 and 4.15). This primer set successfully demonstrates a single product of CYP4Z1 mRNA in the test cells. The presence of CYP4Z1 mRNA in the normal breast MCF10A cells is in agreement with the original study (Thomas et al. 2006) using this primer set. The presence of CYP4Z1 mRNA in the MCF10A, T47D, ZR-75-1 and MDA-MB-468 cells using primers designed by Primer3 is also in agreement with the original study (Figure 4.13). In addition, CYP4Z1 mRNA was observed with a faint band in the ZR-75-30 cells (Figure 4.14).

In summary, Chapter 4 has examined the profile of cytochrome P450 mRNA expression in a panel of breast cancer cell lines compared to that of a normal breast cell line. One of the target genes in this study, CYP4Z1, is a novel isoform of the CYP4 family and has been associated with metastatic disease and poor prognosis of specific types of tumours (Downie et al. 2005). Overexpression of CYP4Z1 has also been found in breast cancer (Rieger et al. 2004), though the mechanism of gene regulation and substrate-specificity are still unclear. The presence of two bands at 159 bp and ~250 bp in the CYP4Z1 RT-PCR warrants furthers investigation and this will be carried out in Chapter 5.

# **Chapter 5**

# **Characterisation of CYP4Z1 mRNA in Breast Cancer Cells**

## 5.1 Introduction

The previous two chapters of this study (Sections 3.3.3.4 and 4.3.6) detected CYP4Z1 amplification products by RT-PCR. Results with these primers showed a band of ~250 bp in addition to a band of the correct product size (159 bp). A similar pattern of two bands was found in four of the eight cell lines (i.e. MCF10A, T47D, ZR-75-1 and MDA-MB-468). Purified DNA of both fragments was further investigated by restriction digestion and DNA sequencing. It is important to note that the results of the restricted fragment sizes (Figure 3.21) and the BLAST analysis of DNA sequences did not distinguish between the fragment of 159 bp and ~250 bp. Further elucidation of the sequence is required to identify the source of the additional band. The reproducibility of the presence of this PCR product ruled out a non-specific product, and raises the possibility that alternate splicing of the gene is producing variations in the size of the mRNA.

Alternative splicing is one of the regulatory mechanisms that contribute to variation in the incorporation of coding regions (exons) into messenger RNA, leading to the production of protein. It is well-known that alternative splicing is of great importance to genetics because a splicing error can disrupt the open reading frame of mRNA, and this will generate different mature RNA encoding distinct protein products from a single gene. There are four common types of gene splicing event (i.e. exon skipping, intron retention, alternative 5' site and alternative 3' site; Figure 5.1) and these can occur in the genes after the transcriptional process (Ferreira et al. 2007). In human cells, alternative splicing has been estimated in 40-60% of all genes, and more than 70% of alternative splicing generates significant functional diversity (Modrek and Lee 2002, Stamm et al. 2005).

188



**Figure 5.1**: Patterns of alternative splicing. Constitutive and alternative exons are indicated by blue and grey boxes respectively and introns are shown by straight lines (Adapted from Ferreira et al. 2007).

Splicing errors in human genes appear to be associated with genetic polymorphisms which result in the alteration of signalling/enzymatic activities in several key genes involved in drug metabolism including cytochrome P450 enzymes (Stamm et al. 2005). CYP4F3 plays an important role in the metabolism of leukotriene B4 and arachidonic acid. Christmas and colleagues (Christmas et al. 2001) reported that alternative splicing events of CYP4F3 influence substrate-specificity, and tissue-specific alternative splicing of pre-mRNA has also been demonstrated. There are two isoforms of alternative splice variants of human CYP4F3. CYP4F3A was predominantly expressed in neutrophils, bone marrow and peripheral blood leukocytes while CYP4F3B was preferentially expressed in liver and kidney. With regard to catalytic activity, higher activities for arachidonic acid or leukotriene B4 were found in CYP4F3B than in CYP4F3A, indicating that alternative splice variants of CYP4F3 generate functional diversity of enzymes.

An alternative splicing of the CYP1A1 gene that results in deletion of 87 bp of exon 6 (exon 6 del CYP1A1) was detected in human brain but not in liver. This splice variant does not metabolise polycyclic aromatic hydrocarbon to genotoxic metabolites that causes formation of DNA adducts in the human brain, and therefore differs from wild-type CYP1A1. These observations suggest that splice variants play an important role in the metabolism of procarcinogens (Kommaddi et al. 2007). This effect may contribute to other CYP1A1 substrates including anticancer drugs.

Evidence of a link between alternative splice variants of some cytochrome P450 genes and a variety of cancers has emerged. Leung and co-workers detected overexpression of CYP1A1 protein and its splice variant (CYP1A1v) in ovarian cancer cells and found they can metabolise procarcinogens such as  $17\beta$ -oestradiol and benzo(a)pyrene. This suggests that the splice variant of CYP1A1 may be involved in both the initiation and progression of ovarian cancer (Leung et al. 2005). An alternative splicing of the CYP27B1 gene has been shown to be associated with breast cancer cells (Cordes et al. 2007) and endometrial cancer cells (Becker et al. 2007). These findings suggest that splice variants may be involved in the initiation and progression of specific types of tumours.

In the CYP4Z1 gene, no information on its alternative splicing has previously been reported although Rieger and colleagues found a truncated version of CYP4Z1 a pseudogene CYP4Z2P with 96% identity to CYP4Z1. Currently, there are two studies that have demonstrated a specific amplification product of CYP4Z1 mRNA using qualitative RT-PCR. Figure 5.2 highlights the structure of the CYP4Z1 gene and shows the positions of the primers used in this study and previous studies using a physical mapping of the CYP4Z1 genome (GenBank accession number: NC\_000001; Human Genome Sequencing Consortium International 2004). Thomas and co-workers (Thomas et al. 2006) employed specific primers against CYP4Z1 mRNA, designed to cross between exon 9 and exon 12

190



**Figure 5.2:** Schematic view of the structure of the CYP4Z1 gene showing the positions of the primers in this study and previous studies. *Abbreviations: E; exon, I; intron* 

boundaries, whereas Savas and colleagues (Savas et al. 2005) used primers located crossing the exon 7 to exon 8 boundaries of the CYP4Z1 gene. These two separate studies detected CYP4Z1 mRNA in the normal breast MCF10A cells (Thomas et al. 2006), human normal tissues (i.e. breast, liver and kidney) and the breast cancer cell lines (i.e. MCF7 and T47D) (Savas et al. 2005). Both found a single band of the CYP4Z1 amplification product.

By contrast, the present study designed the CYP4Z1 primer set F1&R1 (Figure 5.3), crossing from exon 5 to exon 6 boundaries and detected the presence of two bands (159 bp and ~250 bp). These observations would suggest an alternative splicing event particularly intron retention in CYP4Z1. Prior to elucidation of the sequence, the previous chapter (Section 4.3.6) confirmed the expression of CYP4Z1 mRNA in all the cDNA samples using the previously published CYP4Z1 primer (Thomas et al. 2006). The primer set successfully demonstrates a single PCR product of CYP4Z1 mRNA in the test cells. The presence of CYP4Z1 mRNA in the normal breast MCF10A cells is in agreement with the original study (Thomas et al. 2006) using this primer set. The aims of this chapter were to identify the mRNA sequences (exon 5-6) and further elucidate the two bands at 159 bp and ~250 bp.

# 5.2 Materials and Methods

### 5.2.1 Qualitative and nested RT-PCR of CYP4Z1 mRNA

To further elucidate the sequence between exon 5 and exon 6 boundaries, the purified DNA of 159 bp and ~250 bp fragments (i.e. T47D, ZR-75-1 and MDA-MB-468), cDNA samples (i.e. MCF10A, T47D, ZR-75-1 and MDA-MB-468) and the first PCR products were used as template DNA for PCR reactions. It is important to note that the MCF10A cells were excluded from the experiments which required the use of purified DNA



**Figure 5.3:** The sequence of exon 5-intron 5-exon 6 boundaries of the CYP4Z1 gene showing the position of the primer set F1&R1 (Table 2.6). Exon sequences are shown in upper cases and intron sequences in lower case letters. The sequences in exon 5 and exon 6 are indicated in yellow and green respectively. Abbreviations: F1; forward primer 1, R1; reverse primer 1

fragments as templates because CYP4Z1 mRNA in this cell line was expressed at a very low level and it was very difficult to purify a DNA sample. Templates were subjected to the PCR protocols with four different pairs of primers (Table 5.1). The first protocol employed a set of primers F1&r2 that are located in the exon 5 (46 bp) and the intronretained (79 bp), and used the purified DNA (Figures 5.4 and 5.5) and cDNA samples as template (Figure 5.7). To amplify the intron 6-retained, the second PCR protocol was a nested PCR, using a pair of primers Ff2&r2 covering the exon 5 (10 bp) and the intronretained (79 bp) and the first PCR product of the first protocol as template (Figure 5.6). The third protocol used primers F3&R3 that are located in the beginning of exon 5 (118 bp) and the middle of exon 6 (123 bp), and employed the cDNA samples as template (Figure 5.8). The final protocol was a nested PCR using primers F3&r2 that cover the beginning of exon 5 (118 bp) and the intron-retained (79 bp), and employing the first PCR product of the third PCR protocol as template DNA (Figure 5.9). The specificity of the primer sequences used was confirmed by BLAST analysis. The first PCR and nested PCR were performed under the conditions previously described (Tables 2.3, 2.4 and 2.8).

#### 5.2.2 Purified DNA and DNA sequencing

PCR product was purified using the high pure PCR product purification kit, and the purified DNA samples were sent to the Rowett Research Institute for sequence analysis using automated dideoxy DNA sequencing (Section 2.10, Appendix D; Figure D.3). The acceptable  $A_{260}/A_{280}$  ratio of DNA extracted sample was between 1.8 and 2.1.

Protocol	Name	Primer	Position	Sequence (5' to 3')	Length	$T_m (^{o}C)$	$T_a (^{o}C)$	Product size (bp)
1	CYP4Z1	Forward (F1)	Exon 5	TCATGAAGTGTGCCTTCAGC	20 20	64.1	55	125
2	(F1&r2)	Reverse (r2)	Intron 5		20	60.4	55	80
	(F/f2&r2)	Forward (F/12) Reverse (r2)	Intron 5	ctggaatagggaagatggaa	20	60.8 60.4	22	89
3	CYP4Z1	Forward (F3)	Exon 5	GGGAGGAACACATTGCCCA	19	68.3	60	241
	(F3&R3)	Reverse (R3)	Exon 6	GAAAAGATTTGGCCTTGAGAGC	22	64.9		
4	CYP4Z1	Forward (F3)	Exon 5	GGGAGGAACACATTGCCCA	19	68.3	55	197
	(F3&r2)	Reverse (r2)	Intron 5	ctggaatagggaagatggaa	20	60.4		

**Table 5.1:** Primer sequences for CYP4Z1 used to elucidate the sequence between exon 5 and exon 6.

Note: Exon sequences are shown in capital letters and intron sequences in lower case letters

Characterisation of CYP4Z1

## 5.3 Results

### 5.3.1 Identification of CYP4Z1 splice variants in the cell lines

Initial experiments were performed using the primer set CYP4Z1 F1&R1, and the results showed fragments of 159 bp and ~250 bp in the amplification product of CYP4Z1 RT-PCR (Figures 3.20 and 4.13). Restriction digestion and DNA sequencing confirmed both of these products as CYP4Z1. Prediction of PCR products from whole genomic sequences (Figure 5.2) was a PCR product of 159 bp generated from an intron free mRNA (exon 5 (46 bp)/exon 6 (113 bp)) and a product of ~250 bp obtained from a retained intron mRNA (exon 5 (46 bp)-intron 5 (79 bp)-exon 6 (113 bp)).

The first PCR protocol (primer set CYP4Z1 F1&r2) was designed to investigate the exon 5 and a retained intron using the purified 159 bp and ~250 bp DNA fragments. First of all, purified 159 bp and ~250 bp DNA fragments from the T47D cells were amplified at different annealing temperatures (i.e. 55, 58 and 60°C) under the same PCR conditions. The purpose of this was to investigate the effect of annealing temperature on the specificity of primers. After varied temperature annealing, a single product of retained intron (125 bp) was detected in both purified fragments, and 55°C was selected as the optimum annealing temperature for CYP4Z1 F1&r2 amplification as DNA bands appeared sharp and clear (Figure 5.4). This annealing temperature would be employed for experiments with MDA-MB-468 and ZR-75-1 cells. Similarly, the expected size of 125 bp was found in the purified DNA of 159 bp and ~250 bp from the MDA-MB-468 and ZR-75-1 cells (Figure 5.5). However, the 125 bp product was expected only for the purified DNA of 250 bp to confirm intron retention event in the CYP4Z1 gene and not for the purified DNA of 159 bp. These results were different to those expected (Figure 5.5A).

To further identify the complete intron 5 sequences, the second PCR protocol was performed using the first PCR products (125 bp) obtained from T47D, MDA-MB-468 and

ZR-75-1 cells as templates, and the primer set CYP4Z1 F/f2&r2 used as nested primers. A product size of 89 bp was expected for nested PCR obtained from the purified DNA of ~250 bp fragment. The results showed the presence of the expected product size (89 bp) but the product size of primary template (125 bp) was also detected in all PCR products. It should be noted that a fragment of ~250 bp was detected in the purified 159 bp DNA fragment derived from the ZR-75-1 cells (Figure 5.6). These observations suggest that there may have been too much template in the PCR reaction.



**Figure 5.4:** The first protocol (a pair of primers F1&r2) used purified DNA as templates. (A) The position of the primers F1&r2 in the CYP4Z1 gene and the expected product size. (B) Illustration of the effect of annealing temperature on the primer F1&r2 RT-PCR product in the T47D cells. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lanes 3-5: purified DNA of 159 bp, Lanes 6-8: purified DNA of 250 bp, Lanes 3&6: 55°C, Lanes 4&7: 58°C, Lanes 5&8: 60°C



**Figure 5.5:** The first protocol (a pair of primers F1&r2) used purified DNA as templates. (A) The position of the primers F1&r2 in the CYP4Z1 gene and the expected product size. (B) CYP4Z1 F1&r2 RT-PCR in the ZR-75-1 and MDA-MB-468 cells . Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lanes 3-4: ZR-75-1, Lanes 5-6: MDA-MB-468, Lanes 3&5: purified DNA of 159 bp, Lanes 4&6: purified DNA of 250 bp.



**Figure 5.6:** The second protocol (a pair of primers F/f2&r2) used the first round PCR product of the first protocol as templates. (A) The position of the primers F/f2&r2 in the CYP4Z1 gene and the expected product size. (B) Nested PCR CYP4Z1 F1&r2 RT-PCR using the primer set CYP4Z1 F/f2&r2. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lanes 3&6: MDA-MB-468, Lanes 4&7:T47D, Lanes 5&8: ZR-75-1, Lanes 3-5: purified DNA of 159 bp, Lanes 6-8: purified DNA of 250 bp.

In addition, the original cDNA samples isolated from the T47D, MDA-MB-468 and ZR-75-1 cells were also used with the first PCR protocol (primer set CYP4Z1 F1&r2). The expected product size of this protocol was 125 bp and this was detected, but there were also faint DNA fragments present at 159 bp, ~250 bp and ~450 bp in all cDNA used (Figure 5.7). The occurrence of non-specific amplification products may be due to low specificity of primers, although the BLAST analysis showed that the primers were specific for the CYP4Z1 gene.



**Figure 5.7:** The first protocol (a pair of primers F1&r2) used cDNA samples as templates. (A) The position of the primers F1&r2 in the CYP4Z1 gene and the expected product size. (B) CYP4Z1 F1&r2 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: T47D, Lane 4: ZR-75-1, Lane 5: MDA-MB-468.

The third PCR protocol (a primer pair F3&R3) was designed to further identify the sequence between exon 5 and exon 6 boundaries which were at different positions to the original primers F1&R1. This primer set was amplified with the cDNA isolated from MCF10A, T47D, MDA-MB-468 and ZR-75-1 cells. The expected product sizes for intron free mRNA [exon 5 (118 bp)/exon 6 (123 bp)] and retained intron mRNA [exon 5 (118 bp)/exon 6 (123 bp)] and retained intron mRNA [exon 5 (118 bp)-intron 5 (79 bp)-exon 6 (123 bp)] were 241 bp and 320 bp fragments. However, the presence of 241 bp and 320 bp fragments was detected and an additional band of 520 bp band was also found in all cDNA samples (Figure 5.8). It should be noted that a band of 241 bp detected in the MCF10A cells was expressed at a very low level when compared to

Characterisation of CYP4Z1

those of the T47D, MDA-MB-468 and ZR-75-1 cells. Although the primer set failed to detect a single band of PCR product, it behaved in the same pattern as primer set F1&R1.



**Figure 5.8:** The third protocol (a pair of primers F3&R3) used cDNA samples as templates. (A) The position of the primers F3&R3 in the CYP4Z1 gene and the expected product size. (B) CYP4Z1 F3&R3 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MDA-MB-468, Lane 5: T47D, Lane 6: ZR-75-1.

To further elucidate the retained intron sequence, the fourth PCR protocol was performed using the primary PCR products obtained from the third protocol as templates and primers F3&r2. The product size of 197 bp was expected for nested PCR but a band of 197 bp and the first PCR products used as templates (241 bp and 320 bp) were detected as well as a faint additional band of 520 bp (Figure 5.9).



**Figure 5.9:** The fourth protocol (a pair of primers F3&r2) used the first round PCR product of the third protocol as templates. (A) The position of the primers F3&r2 in the CYP4Z1 gene and the expected product size. (B) Nested PCR CYP4Z1 F3&R3 RT-PCR using the primer set CYP4Z1 F3&r2. Nested PCR CYP4Z1 F3&R3 RT-PCR using the primer set CYP4Z1 F3&r2 using different templates. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MDA-MB-468, Lane 5: T47D, Lane 6: ZR-75-1.

# 5.4 Discussion

The investigation in this chapter provides the first evidence of alternative splicing premRNA by intron retention in the CYP4Z1 gene. As mentioned in Section 5.1, alternative splicing events in cytochrome P450 enzymes, for example the CYP4F3 (Christmas et al. 2001) and CYP1A1 (Kommaddi et al. 2007) genes, are associated with some aspects of enzymatic activity such as substrate-specificity, catalytic activity and functional protein diversity. Tissue-specific alternative splice variants have also been shown. In addition, alternative splicing events in cytochrome P450s such as CYP1A1 (Leung et al. 2005) and CYP27B1 (Becker et al. 2007, Cordes et al. 2007) have been demonstrated to play a key role in the carcinogenic process, suggesting that this could be a potential target for the development of cancer therapeutics. Alternative splicing in the CYP4Z1 gene has not been previously documented. The current study is the first to present the sequence between exon 5 and exon 6 boundaries and this does show potential alternative splicing intron retention in the normal breast MCF10A cells and the breast cancer T47D, MDA-MB-468 and ZR-75-1 cells.

The present study initially employed specific primers against CYP4Z1 mRNA designed to cross between the exon 5 and exon 6 boundaries that generated the fragments of 159 bp and ~250 bp in the amplification product of CYP4Z1 RT-PCR (Figures 3.20 and 4.13). These observations were required to further elucidate the sequence because restriction digestion and DNA sequencing confirmed that both of these products were indicated as *Homo sapiens* CYP4Z1 mRNA (GenBank accession number: NM\_178134). In addition, the result obtained from digestion of purified ~250 bp DNA fragment with *Rsa*I i.e. 112 bp and 120 bp fragments (Figure 3.21B) was unexpected. The recognition site for *Rsa*I is GT//AC. In the physical mapping of CYP4Z1 genome (Figure 5.3), G and g are located at the end of exon 5 and intron 5 respectively, and TAC is shown at the beginning of exon 6, which might explain the two restricted products [i.e. 126 bp (exon 5 (46 bp) + intron 5 (79 bp) + exon 6 (1 bp)) + exon 6 (112 bp) = 238 bp fragment]. These findings suggested that the retained intron may be located at the junction of intron 5 and exon 6.

In PCR reactions, the four different primer sets covering the exon 5 to exon 6 boundaries were used to further elucidate the sequences. The primer sets F1&r2 (the first protocol) and F3&r2 (the fourth protocol) were designed to detect the sequence in exon 5 and intron 5 whereas primer pairs F/f2&r2 (the second protocol) cover only the retained intron region.
*Characterisation of CYP4Z1* The primers F3&R3 (the third protocol) are located in the beginning of exon 5 and the middle of exon 6.

The primer set in the first protocol (F1&r2) was initially amplified with the purified DNA of 159 bp and ~250 bp fragments. A single band of 125 bp was detected in purified DNA of both fragments from the T47D, MDA-MB468 and ZR-75-1 cells (Figures 5.6 and 5.7). Originally, the 125 bp product was expected only for the purified fragment of ~250 bp as it could prove retained intron mRNA, but not in the purified fragment of 159 bp. This may be due to unknown sequences in this splice variant, perhaps duplicated coding sequence in exon 5. For clarification, further experiments were performed using the primer set in the second protocol (F/f2&r2) to detect the nucleotide sequences of intron 5. The presence of the expected product size of 89 bp was detected as well as the first PCR product (125 bp) (Figure 5.6), suggesting that there may have been too much template DNA or non-specificity of primers. The pair of primers F1&r2 were also amplified with the original cDNA, but non-specific products were detected (Figure 5.7). Varying annealing temperatures to confirm specificity of primers could be used to address this problem.

The third PCR protocol with the primer set F3&R3 was use to amplify cDNA samples. Non-specific amplification of PCR products was detected (Figure 5.8) though nested PCR using the primer set F3&r2 was performed to increase primer specificity and detect retained intron (Figure 5.7). It is essential to note that the primers used failed to detect a single band of expected product size, but the presence of predicted retained intron products was found in all experiments (Figures 5.4-5.9).

Currently, there are two available databases to search full genomes of the CYP4Z1 genes; the NCBI (http://www.ncbi.nlm.nih.gov/sites/entrez) and Genatlas (<u>http://www.genatlas</u>. org). The NCBI database (NC\_000001) which was used to identify the intron 5 sequences in the CYP4Z1 gene during this research was markedly different in sequences from those of the Genatlas database (Figure 5.10). At a glance in the Genatlas database, there are two regions of the CYP4Z1 gene which demonstrate duplicate repeat sequences, one between exon 5 and intron 5 (101 nucleotides, Figure 5.11A) and the other between intron 5 and intron 5/exon 6 (180 nucleotides, Figure 5.11B). This may help to explain the non-specific products obtained in this chapter because the duplicated sequences also cover the primer sequence used in this research. The forward primer F1 is found in exon 5 and intron 5 whereas the reverse primer r2 is located in two positions in intron 5. The forward primer F/f2 is found in the junction at exon 5/intron 5 and intron 5, and eight nucleotides of the reverse primer R1 were found in intron 5 (Figure 5.12). Obviously, the sequences in exon 5-intron 5-exon 6 in the CYP4Z1 gene appear to be complicated. Further analysis including advanced molecular analysis of genomic/proteomic data is required to achieve a better understanding of the alternative splicing event in the CYP4Z1 gene.

Introns with a high GC content are associated with lower excision rates, increased susceptibility to transcription factor binding and higher transcriptional activity (Goodall and Filipowicz 1991, Galante et al. 2004). The GC content within intron 5 of the CYP4Z1 genome (the NCBI database) is approximately 42% (33 sequences of GC in a total 79 sequences). Moreover, average lengths of retained and non-retained introns are ~115 bp and 3.3 kb respectively, and several studies have confirmed that retained introns are usually shorter than non-retained introns (Stamm et al. 2000, Galante et al. 2004, Chen et al. 2006). Intron 5 of the CYP4Z1 gene has only 79 nucleotide sequences (Figure 5.3). Therefore, the specific region of this gene is more likely to have retained intron.

Data presented in this chapter provides the first evidence of CYP4Z1 mRNA expression in normal breast and breast cancer cell lines, and also of a retained intron 5. Future work could identify the sequences of intron 5 by DNA sequencing and elucidate the distribution

205

*Characterisation of CYP4Z1* of intron retention events on CYP4Z1 mRNA by cloning PCR products of interest in the cell lines used. The biological significance of this event warrants further investigation. This chapter provides information on basal mRNA expression of the CYP4Z1 gene. The next chapter will provide information on the regulation of CYP4Z1 mRNA by investigating the potential nuclear receptors that regulate CYP4Z1 expression in induced cells and determining if the proposed alternative spliced forms are differentially expressed.

### The NCBI database

AACAAATGGGAGGAACACATTGCCCAAAACTCACGTCTGGAGCTCTTTCAACATGTCTCCCTGATG ACCCTGGACAGCATCATGAAGTGTGCCTTCAGCCACCAGGGCAGCATCCAGTTGGACAG aaggaaggtaattgtttgccaataactgtgtcacccactaacatgttgttccatcttccctattccag TACCCTGGACTCATACCTGAAAGCAGTG TTCAACCTTAGCAAAATCTCCCAACCAGCGCATGAACAATTTTCTACATCACAACGACCTGGTTTTC AAATTCAGCTCTCAAGGCCAAATCTTTTCTAAATTTAACCAAGGAACTTCATCAGGTCCACAG

### The Genatlas database

**Figure 5.10:** Sequence comparison of the CYP4Z1 gene from the NCBI and Genatlas database. Exon sequences are shown in upper cases and intron sequences in lower case letters. The sequences in exon 5 and exon 6 are indicated in yellow and green respectively.



**Figure 5.11:** *Highly repeated DNA sequences found in the CYP4Z1 gene using the Genatlas database. (A) Similar sequences between the exon 5 and intron 5 (101 nucleotides) defined by black box. (B) Similar sequences between the intron 5 and intron5/ exon 6 (180 nucleotides) defined by pink box. Exon sequences are shown in upper cases and intron sequences in lower case letters. The sequences in exon 5 and exon 6 are indicated in yellow and green respectively.* 



**Figure 5.12:** Repeated DNA sequences in the CYP4Z1 gene using the Genatlas database associated with primer sequences used. Exon sequences are shown in upper cases and intron sequences in lower case letters. The sequences in exon 5 and exon 6 are indicated in yellow and green respectively.

# **Chapter 6**

# **Regulation of CYP4Z1 expression in Breast Cancer Cell Lines**

## 6.1 Introduction

In Chapter 5, evidence was found of alternative pre-mRNA splicing of the CYP4Z1 gene in the normal breast MCF10A cells and the breast cancer T47D, MDA-MB-468 and ZR-75-1 cells. Although experiments have not been successful in fully characterising the alternative splicing by intron retention, the expected product size of the retained intron 5 (79 bp) was present in all the test cells. Usually, the molecular basis of eukaryotic transcription is controlled by a complex regulatory system as gene expression can be regulated at a variety of steps in the pathway from DNA to RNA and RNA to protein. Transcription is the process by which a monocistronic mRNA is transcribed from DNA and it is controlled by the level of RNA polymerase. The resulting pre-mRNA is capped at the 5' end and the 3' end is polyadenylated. Introns are removed by splicing before the mature mRNA is released from the nucleus to be translated by ribosomes in the cytoplasm into cellular protein (Figure 6.1) (Turner 2005).

Most members of cytochrome P450 family 4 catalyse the ω-hydroxylation of fatty acids, steroid hormones (i.e progesterone, testosterone and growth hormones) and arachidonic acids (i.e. leukotrienes and prostanoid) and these enzymes appear to be implicated in vascular tone and blood pressure regulation (Hsu et al. 2007). Currently, the cytochrome P450 subfamilies 4A and 4F are the most widely characterised and they are mainly expressed in human liver and kidney. Information on expression, function and genetic variation of these genes has been well established in animal models (i.e. rat, mouse and rabbit) (Hsu et al. 2007). The expression of CYP4A genes is induced by peroxisome proliferators (i.e. hypolipidemic drug fibrates and Wy14643 or pirinixic acid). The

209

induction of CYP4A expression by peroxisome proliferators is mediated by the peroxisome proliferator activated receptor alpha (PPARα) (Waxman 1999, Okita and Okita 2001), which requires heterodimerisation with the retinoid X receptor (RXR) for transcriptional regulation of CYP4A (Waxman 1999, Okita and Okita 2001). PPARα is believed to play an important role in the regulation of lipid homeostasis as well as lipid metabolism. However, crosstalk between PPARα and other nuclear receptors such as retinoic acid receptor (RAR), vitamin D receptor (VDR), constitutive androstane receptor (CAR) and pregnane X receptor (PXR) may occur as they share a dimmeric partner (RXR) and substrates (i.e. steroid hormones) (Honkakoski and Negishi 2000).



Figure 6.1: A schematic view of eukaryotic gene expression (Adapted from Turner 2005).

The CYP4Z1 gene is a novel cytochrome P450 enzyme that belongs to family 4 and the physiological functions of the CYP4Z1 gene are yet to be clearly characterised. The tissue localisation of CYP4Z1 has been studied in an attempt to identify its functions. Reiger and co-workers reported that CYP4Z1 mRNA and protein are preferentially expressed in breast tumour tissues but low levels are seen in other tumour types (i.e. uterus, prostate, lung and kidney) (Rieger et al. 2004). No significant difference in CYP4Z1 mRNA was found in

the breast tumour types and progression stages. In addition, Downie and colleagues demonstrated that overexpression of CYP4Z1 is associated with a poor prognosis of primary ovarian cancer (Downie et al. 2005). These two research groups suggest that the expression of CYP4Z1 is involved in the carcinogenic process of human breast and ovary.

A study by Savas and co-workers demonstrated that CYP4Z1 mRNA in the breast cancer T47D cells is markedly induced by progesterone or the synthetic glucocorticoid dexamethasone. The induction of CYP4Z1 mRNA by these compounds was suppressed by the antagonist of progesterone and glucocorticoid receptor RU486. In addition, CYP4Z1 mRNA in the breast cancer progesterone receptor negative MCF7 cells is elevated by dexamethasone but not progesterone. However, there was no significant alteration of CYP4Z1 mRNA by the PPARα agonist pirinixic acid in either the T47D or MCF7 cells (Savas et al. 2005). These findings suggest that the CYP4Z1 gene has cell-type specificity for the induction of mRNA expression.

The physiological role of the splicing variant in the CYP4Z1 gene found in this study is unknown, but it is likely to be involved in the carcinogenic property of the test cell lines. Therefore, the regulation of gene transcription and substrate-specificity of this gene with the occurrence of the alternative splice variant in CYP4Z1 are of interest for further investigation. The main aims of this chapter were to (a) profile mRNA expression of the candidate nuclear receptors (i.e. progesterone receptor; PGR, glucocorticoid receptor; GCR and peroxisome proliferator-activated receptor alpha; PPAR $\alpha$ ) for the CYP4Z1 gene, (b) determine the inducibility of the two forms of CYP4Z1 mRNA, and (c) ascertain whether the induction of CYP4Z1 mRNA is regulated at the transcriptional or translational level.

Regulation of CYP4Z1 gene

# 6.2 Materials and Methods

#### 6.2.1 Cell culture and CYP4Z1 induction in cultured cells

#### 6.2.1.1 Induction of CYP4Z1 mRNA by the nuclear receptor agonists

All four CYP4Z1-expressing cell lines (i.e MCF10A, T47D, MDA-MB-468 and ZR-75-1) and two representative non-CYP4Z1-expressing cell lines (i.e. MCF7 and MDA-MB-231) were used to determine the inducibility of CYP4Z1 mRNA by agonists of the progesterone receptor (PGR), glucocorticoid receptor (GCR) and peroxisome proliferator-activated receptor alpha (PPAR $\alpha$ ). The cells were seeded at  $5 \times 10^5$  cells in a 25 cm<sup>2</sup> flask, and were grown under appropriate conditions (Section 2.2.3). Cells at ~60% confluence were treated with growth medium containing one of the following nuclear receptor agonists for 21 hours; either 2  $\mu$ M progesterone, 1  $\mu$ M dexamethasone or 50  $\mu$ M PPAR $\alpha$  agonist (pirinixic acid) as previously described (Savas et al. 2005) with the exception that chemicals were dissolved in ethanol instead of dimethysulphoxide (DMSO). This amendment was made because a previous study had demonstrated that expression of certain cytochrome P450s, particularly members of the CYP1, CYP2 and CYP3 families can be inhibited or induced by DMSO in *in vitro* studies (Busby et al. 1999). The final concentration of ethanol in the culture medium was 0.1% (v/v). Vehicle control cells were exposed to 0.1% (v/v) ethanol and treated in the same manner as those treated with nuclear receptor agonists. Untreated cells were used as an additional control against which to compare the solvent controls.

#### 6.2.1.2 Evaluation of transcriptional and translational regulation of CYP4Z1

The breast cancer T47D and ZR-75-1 cell lines were used to ascertain whether CYP4Z1 mRNA induction is through transcriptional or translational regulation. This is because CYP4Z1 mRNA in the T47D or ZR-75-1 cells was highly inducible by the nuclear receptor agonists according to results obtained from the initial experiment (Section 6.2.1.1). Cells at 60% confluence were pre-treated with either 10 µg/mL of actinomycin D

or 10  $\mu$ g/mL of cycloheximide for 30 minutes in order to block the transcriptional or translational regulation respectively. Spent medium containing actinomycin D or cycloheximide was removed, and the cells were washed twice with warm (37°C) serum-free medium to minimise the residue of actinomycin D or cycloheximide. Growth media containing the same concentrations of progesterone, dexamethasone or pirinixic acid (Section 6.2.1.1) were used to treat the cells for 21 hours. Untreated and vehicle control cells were included in each experiment. Three independent experiments were performed in triplicate on cultures of the T47D and ZR-75-1 cells.

#### 6.2.2 RNA extraction and cDNA synthesis

Following treatment of the cells with vehicle control, the nuclear receptor agonists or coexposure to actinomycin D/cycloheximide with the nuclear receptor agonists (Section 6.2.1), total RNA was isolated using the RNeasy Plus Mini kit. The RNA concentration and purity were quantified by  $A_{260}/A_{280}$  ratios. Integrity of the isolated total RNA sample was determined by examination of the 28S and 18S rRNA bands on 1% (w/v) agarose gel electrophoresis (Section 2.3). Synthesis of cDNA was performed under the conditions described in Tables 2.2 and 2.3.

#### 6.2.3 Qualitative RT-PCR analysis

To determine the appropriate receptor for mRNA expression in the test cell lines, the eight cDNA samples from the investigation of the target genes in Chapter 4 were employed. These samples were subjected to routine amplification (Tables 2.3 and 2.4) with previous PGR, GCR and PPARα published primers (Table 2.7). Primer specificity was confirmed by BLAST analysis. The PCR products were separated by 1.5% (w/v) agarose gels containing ethidium bromide and visualised by ultra-violet light (Section 2.8). To confirm expression of the target genes, the PCR products were digested with specific restriction To examine the inducibility of the two forms of CYP4Z1 mRNA by the nuclear receptor agonists in the six cell lines, expression of  $\beta_2$ -microglobulin and CYP4Z1 mRNA was assessed in each cell line by the qualitative RT-PCR method (Sections 3.3.2 and 3.3.3.4). The PCR products were visualised on 1.5% (w/v) agarose gels containing ethidium bromide.

#### 6.2.4 Semi-quantitative PCR amplification

In order to determine whether CYP4Z1 mRNA induction is through transcriptional or translational regulation, the PCR products obtained from  $\beta_2$ -microglobulin and CYP4Z1 mRNA in the T47D and ZR-75-1 cells were visualised on 1.5% (w/v) agarose gels containing ethidium bromide. To normalise amplified RNA products and PCR variations, the band intensities of CYP4Z1 mRNA were corrected with the band intensities of  $\beta_2$ -microglobulin mRNA as the internal standard (Section 4.2.3).

#### 6.2.5 Statistical analysis

Results of CYP4Z1/ $\beta_2$ -microglobulin mRNA relative expression for each sample from three independent experiments in the T47D and ZR-75-1 cells were presented as mean ± standard error of the mean (SEM). Statistical analysis was performed with one-way analysis of variance (ANOVA), followed by Tukey's Honesty Significant Difference (HSD) to compare the results from the various experimental groups with their corresponding controls (vehicle control or untreated cells) and the nuclear receptor agonist alone. The differences were considered statistically significant when the *p* value was less than 0.05, 0.01 or 0.001. Data were analysed with SPSS version 13 (SPSS Corporation, Chicago, Illinois, USA).

## 6.3 Results

# 6.3.1 Profile of constitutive mRNA of the candidate nuclear receptors

As stated in Section 6.2.3, the eight cDNA samples used were the same set as those investigated for the target genes in Chapter 4. All RNA samples derived from the eight cell lines were shown to have high integrity and purity (Figure 4.1). Reverse transcription efficiency was verified by the presence of  $\beta_2$ -microglobulin and no major difference in expression was observed (Figure 4.2).

### 6.3.1.1 Progesterone receptor (PGR) RT-PCR

PGR mRNA was detected in four out of the seven breast cancer cell lines as the correct band at 533 bp but no product was found in the normal breast MCF10A cell line. A higher level of expression was observed in the T47D cells than in the MCF7, ZR-75-1 and MDA-MB-157 cells (Figure 6.2). The expression of PGR mRNA was confirmed by restriction digestion with *Alu*I which generated restricted product of the correct size (417 bp and 84 bp, Figure 6.3).



**Figure 6.2:** *PGR RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468.* 



**Figure 6.3:** *Restriction digestion of PGR RT-PCR using AluI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 6.3.1.2 Glucocorticoid receptor (GCR) RT-PCR

A band of 360 bp corresponding to the cDNA product size for GCR mRNA was found in three out of the seven breast cancer cell lines and the normal breast cell line. A higher mRNA level was detected in the ZR-75-1 cells than in the MCF10A cells. Very low levels of expression were observed in both the MDA-MB-157 and MDA-MB-468 cells. However, there is a faint band below 360 bp in the presence of GCR mRNA expression (Figure 6.4). To confirm the GCR mRNA expression, the PCR product was digested with restriction enzyme *Alu*I and the correct fragments were found at 172 bp and 148 bp (Figure 6.5).



**Figure 6.4:** *GCR RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468.* 



**Figure 6.5:** *Restriction digestion of GCR RT-PCR using AluI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.* 

### 6.3.1.3 Peroxisome proliferator-activated receptor alpha (PPARα) RT-PCR

A band of 735 bp was observed for PPAR $\alpha$  mRNA in all of the cell lines except MCF7. PPAR $\alpha$  mRNA was expressed in the MCF10A, T47D, ZR-75-1 and MDA-MB-468 cells at higher levels than in the MDA-MB-231 and MDA-MB-157 cells. Only a very weak level was detected in the ZR-75-30 cells (Figure 6.6). PPAR $\alpha$  mRNA was confirmed by restriction digest with *Rsa*I which generated digested fragments of the correct size (518 bp and 188 bp) (Figure 6.7).



**Figure 6.6:** *PPARαRT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: MCF10A, Lane 4: MCF7, Lane 5: T47D, Lane 6: ZR-75-1, Lane 7: ZR-75-30, Lane 8: MDA-MB-231, Lane 9: MDA-MB-157, Lane 10: MDA-MB-468.* 



**Figure 6.7:** Restriction digestion of PPAR $\alpha$  RT-PCR using RsaI. Lane 1: 1000 bp PCR markers, Lane 2: undigested product, Lane 3: digested product.

# 6.3.2 Induction of CYP4Z1 mRNA by the agonists of candidate nuclear receptors

In Chapter 4, CYP4Z1 mRNA was expressed in three out of the seven breast cancer cells (i.e. T47D, MDA-MB-468 and ZR-75-1) and also in the normal breast MCF10A cells (Figure 4.13). To determine which nuclear receptors were involved in regulating the expression of CYP4Z1, agonists against the PGR, GCR and PPAR $\alpha$  were used. The panel of all eight cell lines were used in this investigation and were exposed to the individual agonists over a period of 21 hours. This time period was selected as it was successfully used in a previous study by Savas and colleagues (Savas et al. 2005) who investigated CYP4Z1 regulation in the T47D and MCF7 cells. All cDNA samples were verified by the housekeeping  $\beta_2$ -microglobulin RT-PCR (319 bp). Little variation of  $\beta_2$ -microglobulin expression was observed between untreated and treated cells in the MCF7 (Figures 6.8A), MDA-MB-231 (Figures 6.9A) and MCF10A cells (Figures 6.10A). However,  $\beta_2$ microglobulin mRNA in the treated cells in the MDA-MB-468 (Figure 6.11A, lane 6), T47D (Figure 6.12A, lane 4) and ZR-75-1 cells (Figure 6.13A, lane 5) was slightly weaker than the other lanes which was taken into account when interpreting the result.

In the non CYP4Z1-expressing cell lines, there was no detectable cDNA product size corresponding to CYP4Z1 mRNA (159 bp or ~250 bp) either in the MCF7 (Figure 6.8B) or MDA-MB-231 cells following exposure to the respective agonists (Figure 6.9B).

MCF10A cell line was barely detectable (Figure 6.10B). However, it is important to note that the presence of a ~250 bp fragment was only found in the cells treated with dexamethasone and a band of 159 bp was not detected. In the MDA-MB-468 cells, it was found that CYP4Z1 mRNA was slightly altered by the three nuclear receptor agonists in comparison to the untreated and vehicle control cells (Figure 6.11B). In addition, CYP4Z1 mRNA in the T47D cells was induced only by dexamethasone (Figure 6.12B) whereas, CYP4Z1 mRNA in the ZR-75-1 cells was elevated by either dexamethasone or pirinixic acid (Figure 6.13B).



**Figure 6.8:** Induction of CYP4Z1 in the MCF7 cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).



**Figure 6.9:** Induction of CYP4Z1 in the MDA-MB-231 cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).



**Figure 6.10:** Induction of CYP4Z1 in the MCF10A cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).



**Figure 6.11:** Induction of CYP4Z1 in the MDA-MB-468 cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).



**Figure 6.12:** Induction of CYP4Z1 in theT47D cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).

223



**Figure 6.13:** Induction of CYP4Z1 in the ZR-75-1 cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated cells, Lane 4: 0.1% (v/v) ethanol, Lane 5: progesterone (2  $\mu$ M), Lane 6: dexamethasone (1  $\mu$ M), Lane 7: pirinixic acid (50  $\mu$ M).

(A)

### 6.3.3 Evaluation of transcriptional and translational regulation of CYP4Z1

To investigate whether the nuclear receptor agonists are able to increase *de novo* CYP4Z1 mRNA synthesis, expression of CYP4Z1 mRNA was investigated in the T47D and ZR-75-1 cells and this was found to increase following exposure to the nuclear receptor agonists. These experiments were performed in the presence or absence of 10  $\mu$ g/mL actinomycin D (an RNA synthesis inhibitor) for 30 minutes, which prevents the induction of CYP4Z1 mRNA, followed by exposure to the agonists (i.e. PGR, GCR and PPAR $\alpha$ ) for 21 hours. If CYP4Z1 mRNA expression was increased due to *de novo* RNA synthesis, a decrease in the level of CYP4Z1 mRNA following exposure to actinomycin D would be expected (Figure 6.14).

To examine whether *de novo* protein synthesis is required for the induction of CYP4Z1 mRNA, the experiments were performed in the presence or absence of 10  $\mu$ g/mL cycloheximide (a protein synthesis inhibitor) for 30 minutes, followed by treatment with the nuclear receptor agonists (i.e. PGR, GCR and PPAR $\alpha$ ) for 21 hours. If CYP4Z1 mRNA was regulated through *de novo* protein synthesis, an increase in the accumulation of CYP4Z1 mRNA after treatment with cycloheximide would be observed (Figure 6.14).

The effect of the RNA and protein synthesis inhibitors on  $\beta_2$ -microglobulin expression and CYP4Z1 mRNA induction by the nuclear receptor agonists in the ZR-75-1 and T47D cells is shown in Figures 6.15 and 6.16 respectively. This data is representative of triplicate experiments. The differences in  $\beta_2$ -microglobulin mRNA expression observed in the samples were taken into account by normalising the CYP4Z1 mRNA levels in each treatment against the equivalent  $\beta_2$ -microglobulin. This data is presented in Figures 6.17, 6.18 and 6.19. There were no significant differences (*p* value > 0.05) in CYP4Z1 mRNA between the untreated cells and cells treated with vehicle control 0.1% (v/v) ethanol, suggesting that 0.1% (v/v) ethanol has no effect on CYP4Z1 mRNA expression.



Figure 6.14: A schematic view of the effect of actinomycin D and cycloheximide on mRNA expression.

In the T47D cells, a significant increase of CYP4Z1 mRNA expression was observed in cells treated with either progesterone (p value < 0.05, Figure 6.17A) or dexamethasone alone (p value < 0.001, Figure 6.18A), when compared with the vehicle control data. The ZR-75-1 cells showed a significant increase of CYP4Z1 mRNA when induced by either dexamethasone (p value < 0.01, Figure 6.18A) or pirinixic acid alone (p value < 0.001, Figure 6.18A).

Co-treatment of the T47D cells with actinomycin D and the individual nuclear receptor agonists resulted in a significant decrease of CYP4Z1 mRNA for progesterone (p value < 0.001, Figure 6.17A), dexamethasone (p value < 0.001 Figure 6.18A) and pirinixic acid (pvalue < 0.01, Figure 6.19A) when compared with cells induced by the nuclear receptor agonist alone. A significant decrease in CYP4Z1 transcription was found when the ZR-75-1 cells were treated with actinomycin D following either dexamethasone (p value < 0.001, Figure 6.18A) or pirinixic acid (p value < 0.01, Figure 6.19A).

Pre-treatment of the T47D cells with cycloheximide followed by one of the individual nuclear receptor agonists resulted in a significant decrease of CYP4Z1 mRNA in the cells co-treated with progesterone (p value < 0.001, Figure 6.17A), dexamethasone (p value < 0.001, Figure 6.18A) or pirinixic acid (p value < 0.05, Figure 6.19A). A significant

increase of CYP4Z1 mRNA was observed in the ZR-75-1 cells when the cells were pretreated with cycloheximide following treatment with progesterone. Cells co-treated with cycloheximide following treatment with either dexamethasone or pirinixic acid showed a significant decrease of CYP4Z1 mRNA.

Considerable variation was observed with the expected CYP4Z1 product size of 159 bp during these experiments and this was also found with the alternative splice variant ~250 bp fragment (CYP4Z1~250 bp). However, the difference between the variations in the expression of two bands was observed. A significant increase of CYP4Z1~250 bp mRNA was found in the T47D cells treated with dexamethasone alone (p value < 0.001, Figure 6.18B) and in the ZR-75-1 cells induced by either dexamethasone (p value < 0.05, Figure 6.18B) or pirinixic acid alone (p value < 0.001, Figure 6.19B) as compared with vehicle control data.

There were no significant differences (*p* value > 0.05) in CYP4Z1~250 bp mRNA when T47D cells were co-treated with actinomycin D and progesterone or pirinixic acid. However, a significant decrease of the CYP4Z1~250 bp fragment was found when the cells were treated with actinomycin D followed by dexamethasone (*p* value < 0.001, Figure 6.18B), compared with cells treated with dexamethasone alone. Pre-treatment of the ZR-75-1 cells with actinomycin followed by one of the individual nuclear receptor agonists resulted in a significant decrease of CYP4Z1~250 bp in the cells co-treated with either dexamethasone or pirinixic acid (*p* value < 0.001) was compared with cells treated with the nuclear receptor agonist alone (Figures 6.18B and 6.19B).

A significant increase of CYP4Z1~250 bp was found in the ZR-75-1 cells treated with cycloheximide followed by progesterone (p value < 0.001, Figure 6.17B), compared to cells treated with vehicle control or progesterone alone. Pre-treatment of the T47D cells

with cycloheximide followed by dexamethasone resulted in a significant decrease of CYP4Z1~250 bp (p value < 0.001, Figure 6.18B) compared to cells treated with dexamethasone alone.

Tables 6.1 and 6.2 summarise the results for the T47D and ZR-75-1 cells demonstrating differences in expression of CYP4Z1 (159 bp and ~250 bp) mRNA against  $\beta_2$ -microglobulin expression following co-treatment with either the nuclear receptor agonists plus or minus actinomycin D or cycloheximide using Tukey's test.



**Figure 6.17:** Effect of RNA and protein synthesis inhibitors on CYP4Z1 induction in the T47D cells. (A)  $\beta_2$ -microglobulin RT-PCR. (B) CYP4Z1 RT-PCR. Lane 1: 1000 bp PCR markers, Lane 2: negative control, Lane 3: untreated, Lane 4: VC, Lane 5: PG, Lane 6: ACTD+PG, Lane 7: CHX+PG, Lane 8: DEX, Lane 9: ACTD+DEX, Lane 10: CHX+DEX, Lane 11: PNX, Lane 12: ACTD+PNX, Lane 13: CHX+PNX.

Abbreviations: VC; 0.1% (v/v) ethanol, PG; progesterone (2  $\mu$ M), DEX; dexamethasone (1  $\mu$ M), PNX; pirinixic acid (50  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL).

**(A)** 





Abbreviations: VC; 0.1% (v/v) ethanol, PG; progesterone (2  $\mu$ M), DEX; dexamethasone (1  $\mu$ M), PNX; pirinixic acid (50  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL).

**(A)** 



**Figure 6.19:** Effect of RNA and protein synthesis inhibitors on progesterone-mediated increase in CYP4Z1 mRNA in the T47D and ZR-75-1 cells. (A) CYP4Z1 (159 bp). (B) CYP4Z1 (250 bp). Data represent the mean $\pm$  SEM of three independent experiments for each treatment showing the relative intensity of CYP4Z1 normalised against  $\beta_2$ -microglobulin mRNA. \*p<0.05, \*\*p<0.01, \*\*\*p<0.001 compared with VC. p<0.05, p<0.01, p<0.01, p<0.01 compared with PG.

Abbreviations: VC; 0.1% (v/v) ethanol, PG; progesterone (2  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL).



**Figure 6.20:** Effect of RNA and protein synthesis inhibitors on dexamethasone-mediated increase in CYP4Z1 mRNA in the T47D and ZR-75-1 cells. (A) CYP4Z1 (159 bp). (B) CYP4Z1 (250 bp). Data represent the mean $\pm$  SEM of three independent experiments for each treatment showing the relative intensity of CYP4Z1 normalised against  $\beta_2$ -microglobulin mRNA. \*p<0.05, \*\*p<0.01, \*\*\*p<0.001 compared with VC.  $\phi$ p<0.05,  $\phi$ p<0.01,  $\phi$ 

Abbreviations: VC; 0.1% (v/v) ethanol, DEX; dexamethasone (1  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL).



**Figure 6.21:** Effect of RNA and protein synthesis inhibitors on pirinixic acid-mediated increase in CYP4Z1 mRNA in the T47D and ZR-75-1 cells. (A) CYP4Z1 (159 bp). (B) CYP4Z1 (250 bp). Data represent the mean  $\pm$  SEM of three independent experiments for each treatment showing the relative intensity of CYP4Z1 normalised against  $\beta_2$ -microglobulin mRNA. \*p<0.05, \*\*p<0.01, \*\*\*p<0.001 compared with VC.  $\Rightarrow p$ <0.05,  $\Rightarrow p$ <0.01,  $\Rightarrow \Rightarrow p$ <0.001 compared with PNX.

Abbreviations: VC; 0.1% (v/v) ethanol, PNX; pirinixic acid (50  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL).

	Untre	eated	V	с	Р	G	ACTI	D+PG	СНХ	+PG	DI	EX	ACTD	+DEX	CHX-	-DEX	PI	NX	ACTD	+PNX	CHX-	PNX
	159bp	250bp																				
Untreated			NS	NS	NS	NS	<b>^</b> ***	NS	NS	NS	↓***	<b>↓</b> ***	<b>^</b> **	NS	<b>^</b> **	NS	NS	NS	<b>^</b> ***	NS	<b>^</b> ***	NS
VC	NS	NS			↓*	NS	<b>^</b> ***	NS	NS	NS	↓***	<b>↓</b> ***	^*	NS	<b>^</b> **	NS	NS	NS	<b>^</b> ***	NS	<b>↑</b> **	NS
PG	NS	NS	<b>↑</b> *	NS			<b>^</b> ***	NS	<b>^</b> ***	NS	↓***	↓***	<b>^</b> ***	NS	<b>^</b> ***	NS	<b>↑</b> **	NS	<b>^</b> ***	NS	<b>^</b> ***	NS
ACTD+PG	↓***	NS	↓***	NS	↓***	NS			→*	NS	↓***	<b>↓</b> ***	NS	NS	NS	NS	↓**	NS	NS	NS	NS	NS
CHX+PG	NS	NS	NS	NS	↓***	NS	^*	NS			↓***	↓***	NS	NS	NS	NS	NS	NS	^*	NS	NS	NS
DEX	<b>^</b> ***	<b>↑</b> ***	<b>^</b> ***			<b>^</b> ***																
ACTD+DEX	↓**	NS	*	NS	+***	NS	NS	NS	NS	NS	***	↓***			NS							
CHX+DEX	↓**	NS	↓**	NS	↓***	NS	NS	NS	NS	NS	↓***	<b>↓</b> ***	NS	NS			NS	NS	NS	NS	NS	NS
PNX	NS	NS	NS	NS	↓**	NS	<b>↑</b> **	NS	NS	NS	<b>↓</b> ***	↓***	NS	NS	NS	NS			<b>↑</b> **	NS	^∗	NS
ACTD+PNX	<b>↓</b> ***	NS	***	NS	↓***	NS	NS	NS	↓*	NS	<b>↓</b> ***	<b>**</b> *	NS	NS	NS	NS	↓**	NS			NS	NS
CHX+PNX	↓***	NS	↓**	NS	↓***	NS	NS	NS	NS	NS	<b>↓</b> ***	<b>↓</b> ***	NS	NS	NS	NS	↓*	NS	NS	NS		

**Table 6.1:** Summary of results for the T47D cells illustrating differences in expression of CYP4Z1 (159 bp and ~250 bp) mRNA/ $\beta_2$ -microglobulin following co-treatment with either the nuclear receptor agonists plus or minus actinomycin D or cycloheximide using Tukey's test.

Abbreviations: VC; 0.1% (v/v) ethanol, PG; progesterone (2  $\mu$ M), DEX; dexamethasone (1  $\mu$ M), PNX; pirinixic acid (50  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL),  $\downarrow$ ; decreasing,  $\uparrow$ ; increasing, NS=not significant, \*p<0.05, \*\*p<0.01, \*\*\*p<0.001

	Untreated		VC		PG		ACTD+PG		CHX+PG		DEX		ACTD+DEX		CHX+DEX		PNX		ACTD+PNX		CHX+PNX	
	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp	159bp	250bp
Untreated			NS	NS	NS	NS	NS	NS	↓**	↓***	↓**	↓*	NS	NS	NS	NS	↓**	<b>↓</b> ***	NS	NS	↓*	↓***
VC	NS	NS			NS	NS	NS	NS	↓**	↓***	↓**	↓*	NS	NS	NS	NS	↓***	<b>↓</b> ***	NS	NS	↓*	↓***
PG	NS	NS	NS	NS			NS	NS	↓*	↓***	↓**	NS	NS	NS	NS	NS	↓**	<b>↓</b> ***	NS	NS	NS	↓***
ACTD+PG	NS	NS	NS	NS	NS	NS			↓**	↓***	↓**	↓**	NS	NS	NS	NS	↓**	<b>↓</b> ***	NS	NS	NS	↓***
CHX+PG	^**	<b>↑</b> ***	<b>^</b> **	<b>^</b> ***	^∗	<b>^</b> ***	<b>^</b> **	<b>^</b> ***			NS	↓*	<b>^</b> ***	<b>^</b> ***	NS	<b>^</b> ***	NS	<b>↓</b> ***	<b>^</b> ***	<b>^</b> ***	NS	↓**
DEX	^**	<b>↑</b> *	^**	^∗	^**	NS	^**	**	NS	↓*			<b>↑</b> ***	<b>↑</b> **	^∗	NS	NS	***	<b>↑</b> ***	<b>↑</b> **	NS	↓***
ACTD+DEX	NS	NS	NS	NS	NS	NS	NS	NS	↓***	↓***	<b>↓</b> ***	↓**			NS	NS	↓***	***	NS	NS	↓**	↓***
CHX+DEX	NS	NS	NS	NS	NS	NS	NS	NS	NS	↓***	↓*	NS	NS	NS			↓*	***	NS	NS	NS	↓***
PNX	<b>^</b> **	<b>↑</b> ***	<b>↑</b> ***	<b>↑</b> ***	<b>^</b> **	<b>↑</b> ***	<b>↑</b> **	<b>↑</b> ***	NS	<b>↑</b> ***	NS	<b>↑</b> ***	<b>↑</b> ***	<b>↑</b> ***	^∗	<b>↑</b> ***			<b>↑</b> ***	<b>↑</b> ***	NS	NS
ACTD+PNX	NS	NS	NS	NS	NS	NS	NS	NS	↓***	↓***	↓***	↓**	NS	NS	NS	NS	↓***	→***			↓**	↓***
CHX+PNX	^*	<b>^</b> ***	^*	<b>^</b> ***	NS	<b>^</b> ***	NS	<b>^</b> ***	NS	<b>↑</b> **	NS	<b>^</b> ***	<b>↑</b> **	<b>^</b> ***	NS	<b>^</b> ***	NS	NS	<b>↑</b> **	<b>↑</b> ***		

**Table 6.2:** Summary of results for the ZR-75-1 cells illustrating differences in expression of CYP4Z1 (159 bp and ~250 bp) mRNA/ $\beta_2$ -microglobulin following co-treatment with either the nuclear receptor agonists plus or minus actinomycin D or cycloheximide using Tukey's test.

Abbreviations: VC; 0.1% (v/v) ethanol, PG; progesterone (2  $\mu$ M), DEX; dexamethasone (1  $\mu$ M), PNX; pirinixic acid (50  $\mu$ M), ACTD; actinomycin D (10  $\mu$ g/mL), CHX; cycloheximide (10  $\mu$ g/mL),  $\downarrow$ ; decreasing,  $\Lambda$ S=not significant, \*p<0.05, \*\*p<0.01, \*\*\*p<0.001

# 6.4 Discussion

The results presented in this chapter provide an in depth look at the mRNA expression profile of candidate nuclear receptors (i.e. PGR, GCR and PPAR $\alpha$ ) in the test cell lines. This should facilitate a better understanding of the association between expression of those receptors and CYP4Z1 mRNA. The CYP4Z1 mRNA was not induced in response to the nuclear receptor agonists (i.e. PGR, GCR and PPAR $\alpha$ ) in the non CYP4Z1-expressing cell lines (i.e. MCF7 and MDA-MB-231). In addition, CYP4Z1 mRNA in the CYP4Z1expressing cell lines, particularly the T47D and ZR-75-1 breast cancer cells, seems to be preferentially induced by different substrates and the pattern of expression differs between the cells lines.

The expression of nuclear receptors PGR, GCR and PPAR $\alpha$  in the cell lines was demonstrated and confirmed (Figures 6.2-6.7). The relationship between the basal expression of CYP4Z1 and those receptors is presented in Table 6.3. It is important to note that a positive correlation between CYP4Z1 mRNA and PPAR $\alpha$  expression was found in the T47D, ZR-75-1 and MDA-MB-468 breast cancer cells. However, an inverse relationship between CYP4Z1 expression and PPAR $\alpha$  mRNA was observed only in the normal breast MCF10A cells. PPAR $\alpha$  is involved in the metabolism of endogenous (i.e. fatty acids) and xenobiotic compounds (i.e. fibrate drugs, long-chain polyunsaturated fatty acids) (Hihi et al. 2002) and is highly expressed in the liver (Motojima and Hirai 2006). In mouse mammary gland, PPAR $\alpha$  expression has been shown to be associated with the regulation of steroid hormones during pregnancy and lactation, suggesting that PPAR $\alpha$ may play a role in mammary gland physiology and pathophysiology. The expression of PPAR $\alpha$  mRNA and protein was found to occur in normal breast and breast tumours (Roberts-Thomson 2000).

Gene Name	MCF10A	MCF7	T47D	ZR-75-1	ZR-75-30	MDAMB-231	MDA-MB-157	MDA-MB-468
<b>CYP4Z1</b> (Figure 4.13)	Faint	-	++	+	-	-	-	+++
<b>PGR</b> (Figure 6.2)	-	++	+++	+	-	-	++	-
GCR (Figure 6.4)	Faint	-	-	++	-	-	Faint	Faint
<b>PPARα</b> (Figure 6.6)	++/ +++	-	++	++/ +++	Faint	Faint	+	+++

Note: +: indicates the level of intensity of the cDNA product by comparing in each gene.

In human breast cancer cell lines, Suchanek and colleagues reported that a higher level of PPAR $\alpha$  mRNA expression was found in the breast cancer ER-negative MDA-MB-231 cells than in the breast cancer ER-positive MCF7 cells (Suchanek et al. 2002). In addition, a previous study used quantitative real-time RT-PCR to demonstrate a relationship between ER mRNA and PPAR $\alpha$  expression in the breast cancer ER-positive ZR-75-1, T47D, MCF7 and BT-483 cells. The findings showed a significant (*p* value < 0.001) inverse relationship between these genes, a high level of ER expression was correlated with a low level of PPAR $\alpha$  mRNA (Faddy et al. 2006). In the current study, the presence of PPAR $\alpha$  mRNA in MDA-MB-231, T47-D and ZR-75-1 is in agreement with the previous studies at a qualitative level (Suchanek et al. 2002, Faddy et al. 2006). However, the current project did not detect the expression of PPAR $\alpha$  mRNA in the MCF7 cells and this does not correlate with the previous two studies. It should be noted that the lack of PPAR $\alpha$  mRNA in the MCF7 cells was correlated with the absence of CYP4Z1expression in this cell line as shown in Figures 4.13 and 6.6 respectively. This suggests that the

regulation of CYP4Z1 mRNA may have a link with the expression of PPARα mRNA. These results are the first to show a positive relationship between CYP4Z1 expression and PPARα mRNA.

The highest level of PPAR $\alpha$  mRNA was detected in the MDA-MB-468 cells and moderate to low levels of PPAR $\alpha$  expression were found in the MCF10A, ZR-75-1 and T47D cells. However, the induction of CYP4Z1 mRNA by PPAR $\alpha$  agonists was clearly shown only in the ZR-75-1 cells (Figure 6.13), not in the MDA-MB-468, T47D or MCF10A cells. These results may suggest that the presence of PPAR $\alpha$  was not directly associated with the induction of CYP4Z1 mRNA.

PPARα needs to form a dimerisation with RXR to activate transcription and a PPARα/RXR heterodimer binds to PPAR response elements (PPREs) in the regulatory region of target genes. Transcriptional activation or repression of target gene requires recruitment of co-activators or co-repressors respectively to the ligand binding domain of the gene. RXR is a heterodimerisating partner for several members of the nuclear receptor super family such as PPARα, PXR, CAR, VDR and RAR (Rastinejad 2001). It should be noted that these nuclear receptors play a crucial role in the regulation of several cytochrome P450 enzymes (i.e. PPARα for CYP4A, CAR for CYP2B6, PXR for CYP3A, VDR for CYP24A1 and RAR for CYP26). Crosstalk between PPARα and other members of the nuclear receptor may occur at an intracellular level, as they share DNA response elements in the regulatory region or ligands (i.e. steroid hormones and some xenobiotic compounds) in target genes (Motojima and Hirai 2006).

Evidence of crosstalk between the PPARα signalling pathway and nuclear hormone receptor signalling pathway has been reported. Keller and colleagues (Keller et al. 1995)

238

demonstrated inhibitory crosstalk between the ER signalling pathway and expression of PPAR $\alpha$ . Due to competition for oestrogen response element (ERE) binding between PPAR $\alpha$  /RXR and oestrogen receptor, transcriptional activation of PPAR $\alpha$  /RXR heterodimerisation is inhibited by ER in target genes. These observations support the findings of Faddy and co-workers (Faddy et al. 2006) who demonstrated an inverse relationship between PPAR $\alpha$  and ER expression in breast cancer cell lines.

In addition, a previous study reported that PPAR $\alpha$  mRNA was induced by glucocorticoids in rat hepatocyte cultures, suggesting that the transcriptional activation of PPAR $\alpha$  was mediated through GCR (Lemberger et al. 1994). In this research, there was a lack of GCR mRNA (Figure 6.4, lane 5) in the T47D cells but when the cells were induced by the GCR agonist dexamethasone, an increase in CYP4Z1 mRNA was observed (Figure 6.12). A moderate level of PPAR $\alpha$  mRNA expression was detected in the T47D cells (Figure 6.6, lane 5). These findings may suggest that CYP4Z1 induction by the GCR agonist in the T47D cells is mediated through the PPAR $\alpha$  signalling pathway. However, information on protein-protein interaction of the PPAR $\alpha$  signalling pathway with other hormone nuclear receptor signalling pathways would be necessary to identify the regulation of CYP4Z1 expression in breast cancer cells.

To determine whether the nuclear receptor agonists' effect on CYP4Z1 mRNA is at the transcriptional or post-transcriptional level, CYP4Z1 mRNA was further investigated in the T47D and ZR-75-1 cells. These two cell lines were selected as they showed increased CYP4Z1 mRNA expression following treatment with the nuclear receptor agonist for 21 hours. Pre-treatment of the T47D and ZR-75-1cells with the RNA synthesis inhibitor actinomycin D followed by one of the individual nuclear receptor agonists resulted in a significant decrease in CYP4Z1 mRNA levels, and inhibition of CYP4Z1 induction by
Moreover, the induction of CYP4Z1 mRNA by all three agonists in the ZR-75-1 cells was not prevented by the protein synthesis inhibitor cycloheximide, ruling out the possibility that protein synthesis is required for CYP4Z1 mRNA induction. These findings suggest that the pre-existing expression of PGR, GCR or PPARα mediated the progesterone, dexamethasone or pirinixic acid respectively for CYP4Z1 mRNA induction. Unlike the ZR-75-1 cells, CYP4Z1 mRNA induction by all three nuclear agonists in the T47D cells was significantly inhibited by cycloheximide (Figures 6.17, 6.18 and 6.19), suggesting that specific proteins are involved in the transcription or stability of CYP4Z1 mRNA by these nuclear receptor agonists.

This research demonstrated that a significant increase of CYP4Z1 mRNA in T47D cells is induced by progesterone and dexamethasone but not pirinixic acid. These observations are in agreement with those of Savas and co-workers who demonstrated CYP4Z1 mRNA induction in the T47D cells by either progesterone or dexamethasone, and suppression of CYP4Z1 induction by RU486 (the antagonist of PGR and GCR) (Savas et al. 2005). However, the present study is the first to show that progesterone or dexamethasone affect CYP4Z1 mRNA at a pre-translational level. The CYP4Z1 mRNA induction profile has not been previously reported. This research demonstrated that CYP4Z1 mRNA in the ZR-75-1 cells is significantly elevated by either dexamethasone or pirinixic acid. These two agonists affect CYP4Z1 mRNA transcription and/or stability, and protein synthesis is not required.

CYP4Z1 mRNA in the ZR-75-1 cells was inducible by either dexamethasone or pirinixic acid, which is similar to CYP4A11 expression in the human hepatoma HepG2 cell line

(Savas et al. 2003). A study by Savas and colleagues (Savas et al. 2005) also demonstrated that CYP4X1 mRNA was induced by pirinixic acid. The CYP4Z1 gene is located in a cluster that contains CYP4A11 and CYP4X1 genes on chromosome 1p33 (Rieger et al. 2004). CYP4Z1 shares 54% and 52% sequence homology with CYP4X1 and CYP4A11 respectively (Savas et al. 2005). Comparative genomics of CYP4Z1 between human and rodent in a study by Nelson and colleagues (Nelson et al. 2004) has indicated that there are no orthologs expressed in mouse or rat, and CYP4Z1 may be human or primate specific. This could suggest overlapping substrate specificity among these genes, and CYP4Z1 could be responsive to peroxisome proliferator chemicals including the PPAR $\alpha$  mediated pathways. PPAR $\alpha$  is activated by a variety of fatty acids and previous studies demonstrated that high-fat diets are associated with an increase in breast cancer incidence (Lu et al. 1995).

In the T47D and ZR-75-1 cells, considerable variation was observed with the expected product size of 159 bp as well as with the alternative splice variant (~250 bp) (Figures 6.15 and 6.16), and there were differences in the induction profile with some treatments. However, this research has demonstrated that higher levels of CYP4Z1 mRNA induction were observed with the fragment of 159 bp than with the fragment of ~250 bp, particularly in the T47D cells (Figures 6.17, 6.18 and 6.19). As shown in Figure 6.1, removal of introns from pre-mRNAs is an essential process in eukaryotic gene expression. Alternative splicing of pre-mRNA generates the production of multiple mature mRNAs obtained from a single gene encoding different protein. Alternative pre-mRNA splicing and its regulation has an important role in human diseases including cancer and this observation is being exploited to identify novel targets for cancer therapeutics and diagnostic markers (Ben-Dov et al. 2008). Alternative splicing events can cause down regulation of a gene by producing a pre-mature stop codon in the mRNA, promoting the degradation of mRNA (Fedor 2008).

241

In conclusion, this chapter provides evidence of potential nuclear receptors regulating the CYP4Z1 gene, and the inducibility of CYP4Z1 in non CYP4Z1-expressing and CYP4Z1 expressing cell lines. Differential regulation of the PCR products in different cell lines and in response to different nuclear receptor agonists suggests alternative splicing may be under specific controls. Further investigation into protein expression in these cell lines, both the wild-type and alternative splice variant, is warranted as this may provide a novel target for cancer therapy.

# Chapter 7

## **General Discussion**

#### 7.1 Introduction

The work presented in this thesis represents the first large scale study to characterise cytochrome P450 mRNA and cytochrome P450-regulatory nuclear receptor mRNA expression in a wide range of human breast cancer cell lines and compare this profile with a normal breast cell line. The relationship between cytochrome P450 profile and specific tumour characteristics (i.e. oestrogen receptor status, invasive phenotype and ethnicity) was investigated in Chapter 4. Of particular interest in the profile obtained was the CYP4Z1 gene as findings from the validation (Chapter 3) and its expression in a panel of the cell lines (Chapter 4) raised the possibility of alternative splicing. The results in Chapter 5 demonstrates intron retention at the boundary of exon 5 and exon 6 in the CYP4Z1 gene using specific primer sets within the regions of both intron and exon. In addition, this study has established *in vitro* models for the study of regulatory mechanisms of the CYP4Z1 gene using the T47D and ZR-75-1 cells to determine whether the CYP4Z1 gene is regulated at a transcriptional or translational level. Alteration of the CYP4Z1 expression pattern after induction and combination treatment with RNA or protein synthesis inhibitor was also determined with regard to the presence of intron retention (Chapter 6).

#### 7.2 Profile of cytochrome P450 mRNA in breast cancer cell lines

As mentioned previously, the results in Chapter 4 provide detailed information on differential expression of cytochrome P450s in human breast tumour cultured cells of different tumour origins. Results showed that mRNA from six cytochrome P450s, CYP1B1, CYP2D6, CYP2J2, CYP2R1, CYP2U1 and CYP4X1, was found in all normal

*Discussion* breast and breast cancer cell lines. However, the CYP2J2 and CYP2U1 mRNA expression profiles are probably the most exciting as lower levels of mRNA expression were observed in the normal breast cells than in almost all of the breast cancer cells. If their protein expression shows the same trend as their mRNA expression, this may suggest that CYP2J2 and CYP2U1 exhibit tumour-specific expression in breast cancer cells and they can be used as prognostic biomarkers. This research also suggests that CYP2J2 and CYP2U1 may represent new targets for the development of cancer therapeutics. To date, the physiological role of CYP2J2 and CYP2U1 in anticancer drug metabolism remains

Although no difference in cytochrome P450 mRNA expression was observed between the different ethnic groups used in this study. Some trends were suggested in particular a potential inverse relationship of cytochrome P450 mRNA (i.e. CYP2A6, CYP2C8, CYP2C18, CYP2F1 and CYP4Z1) between ER status and ethnic groups (i.e. ER-positive Caucasian and ER-negative Afro-Caribbean) was highlighted. In addition, CYP2A6 and CYP2C8 mRNA were found to be primarily expressed in the non-invasive Caucasian and invasive Afro-Caribbean breast cancer cells in these preliminary studies.

unknown.

CYP2A6 and CYP2C8 are involved in the inactivation of anticancer drugs letrozole and paclitaxel respectively. The potent aromatase inhibitor letrozole is used as adjuvant hormonal therapy for postmenopausal women with ER-positive tumours (Goldhirsch et al. 2006). Paclitaxel is widely used as adjuvant chemotherapy in first-line treatment in women with ER-negative tumours (Goldhirsch et al. 2007). In the current investigation, CYP2A6 mRNA was not detected in the ER-positive Afro-Caribbean breast cancer cells and no CYP2C8 mRNA was found in the ER-negative Caucasian breast cancer cells. This variability in letrozole and paclitaxel metabolism between different ethnic groups may help to explain unpredictability of clinical outcome. This research suggests potential similarities in the ER-positive Caucasian and ER-negative Afro-Caribbean breast cancer cell lines which warrants further investigation.

It is known that the hepatic CYP2B6 enzyme plays an important role in the conversion of anticancer prodrug cyclophosphamide to its active form (phosphramide mustard and acrolein) which is then delivered to tumour cells via blood circulation (Stoff-Khalili et al. 2006). Cyclophosphamide is one of the drugs in the TAC regimen which is commonly used for women with metastatic breast cancer who have ER-negative tumours (Ahluwalia et al. 2005). To minimise toxic effects to normal cells, MetXia which encodes human CYP2B6 activating cyclophosphamide in tumours, is injected directly into the malignant cells prior to treatment with cyclophosphamide (Braybrooke et al. 2005). In the present study, it was surprising that CYP2B6 mRNA was detected in the ER-positive Caucasian/ Afro-Caribbean and the ER-negative Afro-Caribbean breast cancer cells but not in the invasive ER-negative Caucasian MDA-MB-231 breast cancer cell line. Future studies should investigate CYP2B6 protein expression in breast cancer cell lines presenting CYP2B6 mRNA. This preliminary study highlights the possibility that if the expression of protein correlates well with mRNA expression, a clinician in future may have to adjust the dosage of cyclophophamide/MetXia.

#### 7.3 Characterisation of CYP4Z1

Results in Chapter 5 indicate the possibility of intron retention between exon 5 and exon 6 in the CYP4Z1 gene. A similar pattern of expression of CYP4Z1 (i.e. 159 bp and ~250 bp) was found in all four cell lines. As described previously, the purified DNA of these fragments was analysed by restriction digest and DNA sequencing to confirm that the additional band was not a PCR artefact.

Discussion

Alternative splicing is associated with alterations of transcriptional regulation for a certain gene. Alternative splice variants of genes can be found in a specific type of tissue and may play a crucial role in transcriptional regulation of certain proteins. Variability in drug metabolising enzymes including cytochrome P450s in a specific tissue has major clinical consequences, and has been associated with adverse drug side effects and susceptibility to cancer (Stamm et al. 2005). A study by Christmas and colleagues (Christmas et al. 2001) demonstrated intron and exon splice junctions in the CYP4F3 gene in the liver and neutrophils. These findings indicate tissue-specific splicing expression (Christmas et al. 1999), and an alteration in the regulation of CYP4F3, affecting its substrate specificity. This may also occur in other members of the cytochrome P450 family 4, including the CYP4Z1 gene, given that they have at least 40% amino acid sequence homology and overlapping properties (Nebert et al. 1989).

The current investigation demonstrated the presence of CYP4Z1 mRNA in the ER-positive Caucasian ZR-75-1 and the ER-negative Afro-Caribbean MDA-MB-468 breast cancer cells and those are novel observations. Particularly as CYP4Z1 mRNA expression appears to be specific to the non-invasive phenotype in both ethnic groups. These preliminary findings may also suggest a relationship between CYP4Z1 and ethnic differences with regard to ER status and stage of breast cancer.

#### 7.4 Regulation of CYP4Z1

As mentioned in Chapter 6, this project is the first study to establish *in vitro* cell models for study of the regulatory mechanisms of CYP4Z1. Breast cancer T47D and ZR-75-1 cell lines were used to evaluate whether CYP4Z1 mRNA induction is regulated at transcriptional or translational level, because CYP4Z1 mRNA in these cells was highly inducible by the nuclear receptor agonists according to results from induction experiments. In T47D, a significant increase of CYP4Z1 mRNA was observed in either the cells induced

by progesterone (*p* value < 0.05) or dexamethasone alone (*p* value < 0.001). ZR-75-1 showed a significant increase of CYP4Z1 expression in the cells treated with either dexamethasone (*p* value < 0.01) or pirinixic acid (*p* value < 0.001). These findings suggest that there are different receptor mechanism involved in CYP4Z1 mRNA induction in T47D and ZR-75-1.

In conjunction, pre-treatment of these two cell lines with the RNA synthesis inhibitor actinomycin D followed by the nuclear receptor agonists resulted in a significant decrease in (*p* value < 0.05) CYP4Z1 mRNA levels and inhibited CYP4Z1 induction by either progesterone, dexamethasone or pirinixic acid, indicating that these nuclear receptor agonists have effects on CYP4Z1 mRNA stability. In contrast, the protein synthesis inhibitor cycloheximide did not inhibit CYP4Z1 mRNA induction by these agonists in the ZR-75-1 cells so protein synthesis is not involved. These results suggest that CYP4Z1 mRNA induction in T47D and ZR-75-1 is mediated through differential cell type specific regulatory mechanisms. The detection of CYP4Z1 induction in the T47D cells is in agreement with a study by Savas and colleagues (Savas et al. 2005). The results in the ZR-75-1 cells are a novel observation, suggesting that the ZR-75-1 cells are suitable to use as a model to investigate the regulation of CYP4Z1.

In terms of alternative splicing in CYP4Z1, comparative analysis of CYP4Z1 mRNA between the fragments of 159 bp and ~250 bp was performed to clarify whether this event changes the regulation of the gene or not. A difference in the induction patterns was observed between two bands. These observations suggest that the alternative splice variant of CYP4Z1 may have a different regulation mechanism. It is essential to further investigate CYP4Z1 protein expression in these cells to clarify the mechanisms involved.

## 7.5 Conclusions

The main points emerging from the current study are:

- (a) The relationships between cytochrome P450 mRNA expression profile and specific tumour characteristics are observed in the current study:
  - CYP2B6 which plays a crucial role in the activation of cyclophosphamide was not detected in the invasive ER-negative Caucasian breast cancer cells.
  - Some of the findings in this study are of a preliminary nature and warrant further investigation particularly as to whether the individual cytochrome P450 protein is expressed. But also as to whether the trends observed in this thesis are confirmed in clinical patient samples (at both mRNA and protein level). The trends in cytochrome P450 expression suggested in this thesis are detailed as follows:
    - An inverse relationship of CYP2A6, CYP2C8, CYP2C18, CYP2F1 and CYP4Z1 mRNA between ethnicity and ER status (ER-positive Caucasian and ER-negative Afro-Caribbean breast cancer cell lines).
    - An inverse relationship of CYP2A6 and CYP2C8 mRNA between ethnicity and invasive phenotype (Caucasian non-invasive and Afro-Caribbean invasive breast cancer cell lines).
    - CYP4Z1 and CYP2F1 mRNA were specific for non-invasive phenotype.
- (b) Two distinct bands obtained from the CYP4Z1 PCR product were elucidated by specific primer sets, and this project does suggest potential intron retention at the boundary of exon 5 and exon 6 in the CYP4Z1 gene.
- (c) Two out of the four CYP4Z1-expressing cell lines, T47D and ZR-75-1, seem to be preferentially induced by different substrates and the pattern of expression differs between the cells lines:
  - CYP4Z1 mRNA in T47D was induced only by the PGR or GCR agonist whereas
    CYP4Z1 mRNA in ZR-75-1 was elevated by either the GCR or PPARα agonist.

Discussion (d) CYP4Z1 induction by progesterone, dexamethasone or pirinixic acid is inhibited by actinomycin D, suggesting that these agonists have effects on CYP4Z1 mRNA transcription or stability. However, CYP4Z1 induction in the ZR-75-1 cells is not prevented by cycloheximide, indicating that CYP4Z1 mRNA induction by these agonists is unlikely to involve protein synthesis. These suggest that the inducibility of CYP4Z1 mRNA is mediated through differential cell type specific regulatory mechanisms and there is evidence for differential regulation of the splice variants (CYP4Z1~250 bp).

## 7.6 Future work

This investigation has provided comprehensive information on the differential cytochrome P450 mRNA expression among breast cancer cell lines with different characteristics, the exploration of alternative splicing in CYP4Z1, and clarification of the regulatory mechanism of the CYP4Z1 gene in the breast cancer cell lines. These findings raise a number of questions though, and further study in these areas is likely to lead to a better understanding of the role of cytochrome P450 enzymes in breast cancer cell.

The expression of CYP2A6, CYP2B6, CYP2C8, CYP2J2, CYP2F1, CYP2U1 and CYP4Z1 mRNA should be investigated by looking at their protein expression in test cell lines using western blotting or immunocytochemistry (ICC) followed by cytochrome P450 expression in clinical samples. It is known that mRNA expression does not always correlate with protein expression, for example the expression of CYP1B1 in normal tissues (i.e. breast, brain and testis) (Murray et al. 1997). Therefore, a demonstration of protein expression in these cell lines is required. If protein expression is detected in these cell lines, they could be established as a model system for the development of novel cancer therapeutics. Alternatively, they could be of benefit in the design of more effective drug regimens for breast cancer patients.

The data presented here provides a clearer understanding of the presence/absence of cytochrome P450 mRNA using qualitative and semi-quantitative RT-PCR. The use of a quantitative PCR method would be useful in further investigations as it would increase the sensitivity of the method.

Detection and evaluation of an intron retention event in the CYP4Z1 gene could be achieved through the establishment of a specific cell line stably expressing the human CYP4Z1 gene, identification of specific region of intron retention and investigation of protein expression. This would provide a better understanding of the presence of abnormal transcripts and a functional gene of CYP4Z1.

In order to investigate whether alternative splicing in CYP4Z1 is specific for breast cancer or other types of cancer, determination of this event in a variety of tumour cell lines would allow a clearer understanding of the involvement of CYP4Z1 expression in the carcinogenic process.

The regulation of CYP4Z1 in the T47D and ZR-75-1 cells is mediated through different receptors. The current study was carried out using a single set of concentrations of nuclear receptor agonists and a 21-hour of treatment as described previously by Savas and co-workers (Savas et al. 2005). Varying concentrations of agonists and incubation time would be a useful approach to elucidate the regulatory mechanisms of CYP4Z1. The present study employed semi-quantitative RT-PCR method; it would be more accurate if quantitative real-time PCR was instead used to determine mRNA level. Correlation between mRNA and protein expression would be required to explain the mechanism of regulation in breast cancer cells.

Transcriptional regulation of CYP4Z1 by nuclear receptors requires further examination to clarify the factors involved in the regulatory mechanisms and establish whether CYP4Z1 induction is connected with the expression of PGR, GCR or PPAR $\alpha$ . Identification of a molecular mechanism of functional cross-talk between nuclear receptors would provide a theoretical basis for development of novel therapeutic strategies and design of novel compounds for treatment of breast cancer.

### References

ADRIANCE, M., INMAN, J., PETERSEN, O., et al. 2005. Myoepithelial cells: good fences make good neighbors. *Breast Cancer Research*, **7**, 190-197.

AHLUWALIA, M. S., DAW, H. A., NORONHA, V., et al. 2005. Adjuvant Docetaxel for Node-Positive Breast Cancer. *The New England Journal of Medicine*, 353, 954-955.

AIMOVA, D., SVOBODOVA, L., KOTRBOVA, V., et al. 2007. The Anticancer Drug Ellipticine Is a Potent Inducer of Rat Cytochromes P450 1A1 and 1A2, Thereby Modulating Its Own Metabolism. *Drug Metabolism and Disposition*, 35, 1926-1934.

AKIYAMA, T. E. and GONZALEZ, F. J. 2003. Regulation of P450 genes by liverenriched transcription factors and nuclear receptors. *Biochimica et Biophysica Acta (BBA)-General Subjects*, 1619, 223-234.

AL-ANIZY, M., HORLEY, N. J., KUO, C.-W. S., et al. 2006. Cytochrome P450 Cyp4x1 is a major P450 protein in mouse brain. *FEBS Journal*, 273, 936–947.

ALBERTELLA, M. R., LOADMAN, P. M., JONES, P. H., et al. 2008. Hypoxia-Selective Targeting by the Bioreductive Prodrug AQ4N in Patients with Solid Tumors: Results of a Phase I Study. *Clinical Cancer Research*, 14, 1096-1104.

ALBIN, N., MASSAAD, L., TOUSSAINT, C., et al. 1993. Main drug-metabolizing enzyme systems in human breast tumors and peritumoral tissues. *Cancer Research*, 53, 3541-3546.

ALEXANDRE, J., REY, E., GIRRE, V., et al. 2007. Relationship between cytochrome 3A activity, inflammatory status and the risk of docetaxel-induced febrile neutropenia: a prospective study. *Annals of Oncology*, 18, 168-172.

ALLEN, J. W., JOHNSON, R. S. and BHATIA, S. N. 2005. Hypoxic inhibition of 3methylcholanthrene-induced CYP1A1 expression is independent of HIF-1alpha. *Toxicology Letters*, 155, 151-159.

ALTUCCI, L., LEIBOWITZ, M. D., OGILVIE, K. M., et al. 2007. RAR and RXR modulation in cancer and metabolic disease. *Nature Reviews Drug Discovery*, 6, 793-810.

ANDERSON, M. G., NAKANE, M., RUAN, X., et al. 2006. Expression of VDR and CYP24A1 mRNA in human tumors. *Cancer Chemotherapy and Pharmacology*, 57, 234-240.

ANGUS, W. G. R., LARSEN, M. C. and JEFCOATE, C. R. 1999. Expression of CYP1A1 and CYP1B1 depends on cell-specific factors in human breast cancer cell lines: role of estrogen receptor status. *Carcinogenesis*, 20, 947-955.

AOYAMA, Y., NOSHIRO, M., GOTOH, O., et al. 1996. Sterol 14-demethylase P450 (P45014DM\*) is one of the most ancient and conserved P450 species. *Journal of Biochemistry*, 119, 926-933.

ARORA, V., CATE, M. L., GHOSH, C., et al. 2002. Phosphorodiamidate Morpholino Antisense Oligomers Inhibit Expression of Human Cytochrome P450 3A4 and Alter Selected Drug Metabolism. *Drug Metabolism and Disposition*, 30, 757-762.

ATHAR, M., BACK, J. H., TANG, X., et al. 2007. Resveratrol: A review of preclinical studies for human cancer prevention. *Toxicology and Applied Pharmacology*, 224, 274-283.

BAER, B. R. and RETTIE, A. E. 2007. CYP4B1: An Enigmatic P450 at the interface between xenobiotic and endobiotic metabolism. *Drug Metabolism and Disposition*, 38, 451-476.

BARNETT, J. A., URBAUER, D. L., MURRAY, G. I., et al. 2007. Cytochrome P450 1B1 Expression in Glial Cell Tumors: An Immunotherapeutic Target. *Clinical Cancer Research*, 13, 3559-3567.

BARNGROVER, D., THOMAS, J. and THILLY, W. 1985. High density mammalian cell growth in Leibovitz bicarbonate-free medium: effects of fructose and galactose on culture biochemistry. *Journal of Cell Science*, 78, 173-189.

BASTIEN, J. and ROCHETTE-EGLY, C. 2004. Nuclear retinoid receptors and the transcription of retinoid-target genes. *Gene*, 328, 1-16.

BECKER, S., CORDES, T., DIESING, D., et al. 2007. Expression of 25 hydroxyvitamin D3-1alpha-hydroxylase in human endometrial tissue. *The Journal of Steroid Biochemistry and Molecular Biology*, 103, 771-775.

BEIGNEUX, A., HOFMANN, A. F. and YOUNG, S. G. 2002. Human CYP7A1 deficiency: progress and enigmas. *The Journal of Clinical Investigation*, 110, 29-31.

BEISCHLAG, T. V. and PERDEW, G. H. 2005. ER{alpha}-AHR-ARNT Protein-Protein Interactions Mediate Estradiol-dependent Transrepression of Dioxin-inducible Gene Transcription. *The Journal of Biological Chemistry*, 280, 21607-21611.

BELL, D. R., PLANT, N. J., RIDER, C. G., et al. 1993. Species-specific induction of cytochrome P-450 4A RNAs: PCR cloning of partial guinea-pig, human and mouse CYP4a cDNAs. *Biochmical Journal*, 294, 173-180.

BELLAMINE, A., WANG, Y., WATERMAN, M. R., et al. 2003. Characterization of the CYP4A11 gene, a second CYP4A gene in humans. *Archives of Biochemistry and Biophysics*, 409, 221-227.

BEN-DOV, C., HARTMANN, B., LUNDGREN, J., et al. 2008. Genome-wide Analysis of Alternative Pre-mRNA Splicing. *The Journal of Biological Chemistry*, 283, 1229-1233.

BERRY, D. A., CIRRINCIONE, C., HENDERSON, I. C., et al. 2006. Estrogen-receptor status and outcomes of modern chemotherapy for patients with node-positive breast cancer. *JAMA*, 295, 1658-1667.

BIÈCHE, I., GIRAULT, I., URBAIN, E., et al. 2004. Relationship between intratumoral expression of genes coding for xenobiotic-metabolizing enzymes and benefit from adjuvant tamoxifen in estrogen receptor alpha-positive postmenopausal breast carcinoma. *Breast Cancer Research*, 6, R252-263.

BIÈCHE, I., NARJOZ, C., ASSELAH, T., et al. 2007. Reverse transcriptase-PCR quantification of mRNA levels from cytochrome (CYP)1, CYP2 and CYP3 families in 22 different human tissues. *Pharmacogenetics and Genomics*, 17, 731-742.

BISHOP-BAILEY, D. and WRAY, J. 2003. Peroxisome proliferator-activated receptors: a critical review on endogenous pathways for ligand generation. *Prostaglandins & Other Lipid Mediators*, 71, 1-22.

BJORKHEM, I., LUTJOHANN, D., DICZFALUSY, U., et al. 1998. Cholesterol homeostasis in human brain: turnover of 24S-hydroxycholesterol and evidence for a cerebral origin of most of this oxysterol in the circulation. *Journal of Lipid Research*, 39, 1594-1600.

BLAKE, M. J., CASTRO, L., LEEDER, J. S., et al. 2005. Ontogeny of drug metabolizing enzymes in the neonate. *Seminars in Fetal and Neonatal Medicine Perinatal Pharmacology*, 10, 123-138.

BOGDANOVIC, N., BRETILLON, L., LUND, E. G., et al. 2001. On the turnover of brain cholesterol in patients with Alzheimer's disease. Abnormal induction of the cholesterol-catabolic enzyme CYP46 in glial cells. *Neuroscience Letters*, 314, 45-48.

BOOCOCK, D. J., FAUST, G. E. S., PATEL, K. R., et al. 2007. Phase I Dose Escalation Pharmacokinetic Study in Healthy Volunteers of Resveratrol, a Potential Cancer Chemopreventive Agent. *Cancer Epidemiology Biomarkers & Prevention*, 16, 1246-1252.

BORUBAN, M. C., YASAR, U., BABAOGLU, M. O., et al. 2006. Tamoxifen inhibits cytochrome P450 2C9 activity in breast cancer patients. *Journal of Chemotherapy*, 18, 421-424.

BRADSHAW, T. D. and WESTWELL, A. D. 2004. The Development of the Antitumour Benzothiazole Prodrug, Phortress, as a Clinical Candidate. *Current Medicinal Chemistry*, 11, 1009-1021.

BRAYBROOKE, J. P., SLADE, A., DEPLANQUE, G., et al. 2005. Phase I Study of MetXia-P450 Gene Therapy and Oral Cyclophosphamide for Patients with Advanced Breast Cancer or Melanoma. *Clinical Cancer Research*, 11, 1512-1520.

BRETILLON, L., DICZFALUSY, U., BJÖRKHEM, I., et al. 2007. Cholesterol-24Shydroxylase (CYP46A1) is specifically expressed in neurons of the neural retina. *Current Eye Research*, 32, 361-366.

BROCKDORFF, B. L., SKOUV, J., REITER, B. E., et al. 2000. Increased expression of cytochrome P450 1A1and 1B1 genes in anti-oestrogen-resistant human breast cancer cell lines. *International Journal of Cancer*, 88, 902-906.

BROWN, J., III, THEISLER, C., SILBERMAN, S., et al. 2004. Differential Expression of Cholesterol Hydroxylases in Alzheimer's Disease. *The Journal of Biological Chemistry*, 279, 34674-34681.

BROWN, J. M. 2007. In *Oxygen Biology and Hypoxia* (Ed, BRUNE) Academic Press, pp. 295, 297-321.

BRUEGGEMEIER, R. W., HACKETT, J. C. and DIAZ-CRUZ, E. S. 2005. Aromatase Inhibitors in the Treatment of Breast Cancer. *Endocrine Reviews*, 26, 331-345.

BRUNO, R. D. and NJAR, V. C. O. 2007. Targeting cytochrome P450 enzymes: A new approach in anti-cancer drug development. *Bioorganic & Medicinal Chemistry*, 15, 5047-5060.

BURDALL, S. E., HANBY, A. M., LANSDOWN, M. R. J., et al. 2003. Breast cancer cell lines: friend or foe? *Breast Cancer Research*, *5*, 89-95.

BUSBY, W. F., JR., ACKERMANN, J. M. and CRESPI, C. L. 1999. Effect of Methanol, Ethanol, Dimethyl Sulfoxide, and Acetonitrile on In Vitro Activities of cDNA-Expressed Human Cytochromes P-450. *Drug Metabolism and Disposition*, 27, 246-249.

BYLUND, J., BYLUND, M. and OLIW, E. H. 2001. cDNA Cloning and Expression of CYP4F12, a Novel Human Cytochrome P450. *Biochemical and Biophysical Research Communications*, 280, 892-897.

BYLUND, J., FINNSTROM, N. and OLIW, E. H. 1999. Gene Expression of a Novel Cytochrome P450 of the CYP4F Subfamily in Human Seminal Vesicles. *Biochemical and Biophysical Research Communications*, 261, 169-174.

CALI, J. J. and RUSSELL, D. W. 1991. Characterization of human sterol 27-hydroxylase. A mitochondrial cytochrome P-450 that catalyzes multiple oxidation reaction in bile acid biosynthesis. *The Journal of Biological Chemistry*, 266, 7774-7778.

CARR, B. A., WAN, J., HINES, R. N., et al. 2003. Characterization of the Human Lung CYP2F1 Gene and Identification of a Novel Lung-specific Binding Motif. *The Journal of Biological Chemistry*, 278, 15473-15483.

CAVALIERI, E., CHAKRAVARTI, D., GUTTENPLAN, J., et al. 2006. Catechol estrogen quinones as initiators of breast and other human cancers: Implications for biomarkers of susceptibility and cancer prevention. *Biochimica et Biophysica Acta (BBA) - Reviews on Cancer*, 1766, 63-78.

CHANG, T. K. H., BANDIERA, S. M. and JIE, C. 2003. Constitutive androstane receptor and pregnane X receptor gene expression in human liver: Interindividual variability and correlation with CYP2B6 mRNA levels. *Drug Metabolism and Disposition*, 31, 7-10.

CHASE, M. B., BAEK, S. J., PURTELL, D. C., et al. 1993. Mapping of the human thromboxane synthase gene (TBAXS1) to chromosome 7q34-q35 by two color fluorescence in situ hybridization. *Genomics*, 16, 771-773.

CHEN, F.-C., CHEN, C.-J., HO, J.-Y., et al. 2006. Identification and evolutionary analysis of novel exons and alternative splicing events using cross-species EST-to-genome comparisons in human, mouse and rat. *BMC Bioinformatics*, 7, 136.

CHEN, S. A., BESMAN, M. J., SPARKES, R. S., et al. 1988. Human aromatase: cDNA cloning, southernblot analysis, and assignment of the gene to chromosome 15. *DNA*, 7, 27-38.

CHEN, Y., GAMMON, M. D., TEITELBAUM, S. L., et al. 2008. Estrogen-biosynthesis Gene CYP17 and Its Interactions with Reproductive, Hormonal and Lifestyle Factors in Breast Cancer Risk: Results from the Long Island Breast Cancer Study Project. *Carcinogenesis*, Epub ahead of print.

CHENG, J. B., MOTOLA, D. L., MANGELSDORF, D. J., et al. 2003. De-orphanization of Cytochrome P450 2R1: A microsomal vitamin D 25-hydroxylase. *The Journal of Biological Chemistry*, 278, 38084-38093.

CHIA, S., BRYCE, C. and GELMON, K. 2005. The 2000 EBCTCG overview: a widening gap. *The Lancet*, 365, 1665-1666.

CHOMCZYNSKI, P. 1992. Solubilization in formamide protects RNA from degradation. *Nucleic Acids Research*, 20, 3791–3792.

CHOUDHARY, D., JANSSON, I., STOILOV, I., et al. 2005. Expression patterns of mouse and human CYP orthologs (families 1-4) during development and in different adult tissues. *Archives of Biochemistry and Biophysics*, 436, 50-61.

CHRISTIE, A. and BUTLER, M., 1, 9. 1994. Growth and Metabolism of a Murine Hybridoma in Cultures Containing Glutamine-based Dipeptides. *Focus*, 16, 9-13.

CHRISTMAS, P., JONES, J. P., PATTEN, C. J., et al. 2001. Alternative Splicing Determines the Function of CYP4F3 by Switching Substrate Specificity. *The Journal of Biological Chemistry*, 276, 38166-38172.

CHRISTMAS, P., URSINO, S. R., FOX, J. W., et al. 1999. Expression of the CYP4F3 Gene. Tissue-Specific Splicing and Alternative Promoters Generate High and Low Km Forms of Leukotriene B4 Omega-Hydroxylase. *The Journal of Biological Chemistry*, 274, 21191-21199.

CHUANG, S. S., HELVIG, C., TAIMI, M., et al. 2004. CYP2U1, a Novel Human Thymus- and Brain-specific Cytochrome P450, Catalyzes {omega}- and ({omega}-1)-Hydroxylation of Fatty Acids. *The Journal of Biological Chemistry*, 279, 6305-6314.

CHUNG, B. C., MATTERSON, K. J., VOUTILAINEN, R., et al. 1986. Human cholesterol side-chain cleavage enzyme, P450scc: cDNA cloning, assignment of the gene to chromosome 15, and expression in the placenta. *Proceedings of the National Academy of Sciences of the United States of America*, 83, 8962-8966.

COLLER, J. K., KREBSFAENGER, N., KLEIN, K., et al. 2002. The influence of CYP2B6, CYP2C9 and CYP2D6 genotypes on the formation of the potent antioestrogen Z-4-hydroxy-tamoxifen in human liver. *British Journal of Clinical Pharmacology*, 54, 157-167.

COMPAGNONE, N. A., BULFONE, A., RUBENSTEIN, J. L., et al. 1995. Expression of the steroidogenic enzyme P450scc in the central and peripheral nervous systems during embryogenesis. *Endocrinology*, 136, 2689-2696.

COON, M. J. 2003. Multiple oxidants and multiple mechanisms in cytochrome P450 catalysis. *Biochemical and Biophysical Research Communications Irwin C. Gunsalus, Versatile and Creative Scientist*, 312, 163-168.

CORDES, T., DIESING, D., BECKER, S., et al. 2007. Expression of splice variants of 1alpha-hydroxylase in MCF-7 breast cancer cells. *Journal of Steroid Biochemistry & Molecular Biology*, 103, 326-329.

CROSS, H. S., KÁLLAY, E., FARHAN, H., et al. 2003. Regulation of extrarenal vitamin D metabolism as a tool for colon and prostate cancer prevention. *Recent Results in Cancer Research*, 164, 413-425.

CURNOW, K. M., TUSIE-LUNA, M. T., PASCOE, L., et al. 1991. The product of the CYP11B2 gene is required for aldosterone biosynthesis in the human adrenal cortex. *Molecular Endocrinology*, *5*, 1513-1522.

CURTIS, E., QUALE, C., HAGGSTROM, D., et al. 2008. Racial and ethnic differences in breast cancer survival: how much is explained by screening, tumor severity, biology, treatment, comorbidities, and demographics? *Cancer*, 112, 171-180.

DALY, A. K. 2006. Significance of the minor cytochrome P450 3A isoforms. *Clinical Pharmacokinetics*, 45, 13-31.

DAVIS, J. M. 2002. *Basic cell culture: a practical approach*, Oxford University Press, New York, USA.

DE LEON, J., ARMSTRONG, S. C. and COZZA, K. L. 2006a. Clinical Guidelines for Psychiatrists for the Use of Pharmacogenetic Testing for CYP450 2D6 and CYP450 2C19. *Psychosomatics*, 47, 75-85.

DE LEON, J., SUSCE, M. T. and MURRAY-CARMICHAEL, E. 2006b. The AmpliChip CYP450 genotyping test: Integrating a new clinical tool. *Molecular Diagnosis & Therapy*, 10, 135-151.

DE LYRA, E. C., DA SILVA, I. A., KATAYAMA, M. L. H., et al. 2006. 25(OH)D<sub>3</sub> and 1,25(OH)<sub>2</sub>D<sub>3</sub> serum concentration and breast tissue expression of 1[alpha]-hydroxylase, 24-hydroxylase and Vitamin D receptor in women with and without breast cancer. *The Journal of Steroid Biochemistry and Molecular Biology*, 100, 184-192.

DEEB, K. K., TRUMP, D. L. and JOHNSON, C. S. 2007. Vitamin D signalling pathways in cancer: potential for anticancer therapeutics. *Nature Reviews. Cancer*, 7, 684-700.

DEHAL, S. S. and KUPFER, D. 1997. CYP2D6 Catalyzes Tamoxifen 4-Hydroxylation in Human Liver. *Cancer Research*, 57, 3402-3406.

DENNY, W. A. 2004. The Design of Drugs that Target Tumour Hypoxia. *Australian Journal of Chemistry*, 57, 821-828.

DEROO, B. J. and KORACH, K. S. 2006. Estrogen receptors and human disease. *The Journal of Clinical Investigation*, 116, 561-570.

DESAI, P., DEKOSKY, S. T. and KAMBOH, M. I. 2002. Genetic variation in the cholesterol 24-hydroxylase (CYP46) gene and the risk of Alzheimer's disease. *Neuroscience Letters*, 328, 9-12.

DIESEL, B., RADERMACHER, J., BUREIK, M., et al. 2005. Vitamin D3 Metabolism in Human Glioblastoma Multiforme: Functionality of CYP27B1 Splice Variants, Metabolism of Calcidiol, and Effect of Calcitriol. *Clinical Cancer Research*, 11, 5370-5380.

DING, S., LAKE, B. G., FRIEDBERG, T., et al. 1995. Expression and alternative splicing of the cytochrome P-450 CYP2A7. *The Biochemical Journal*, 306, 161–166.

DING, X. and KAMINSKY, L. S. 2003. Human extrahepatic cytochrome P450: Function in Xenobiotic Metabolism and Tissue-Selective Chemical Toxicity in the Respiratory and Gastrointestinal Tracts. *Annual Review of Pharmacology and Toxicology*, 43, 149-173.

DOMANSKI, T. L., FINTA, C., HALPERT, J. R., et al. 2001. cDNA Cloning and Initial Characterization of CYP3A43, a Novel Human Cytochrome P450. *Molecular Pharmacology*, 59, 386-392.

DOTZLAW, H., LEYGUE, E., WATSON, P., et al. 1999. The Human Orphan Receptor PXR Messenger RNA Is Expressed in Both Normal and Neoplastic Breast Tissue. *Clinical Cancer Research*, *5*, 2103-2107.

DOWNIE, D., McFADYEN, M. C. E., ROONEY, P. H., et al. 2005. Profiling Cytochrome P450 Expression in Ovarian Cancer: Identification of Prognostic Markers. *Clinical Cancer Research*, 11, 7369-7375.

DRAGAN, C. A., HARTMANN, R. W. and BUREIK, M. 2006. A fission yeast-based test system for the determination of IC50 values of anti-prostate tumor drugs acting on CYP21. *Journal of Enzyme Inhibition and Medicinal Chemistry*, 21, 547-556.

DULBECCO, R. and FREEMAN, G. 1959. Plaque production by the polyoma virus. *Virology*, **8**, 396-397.

DUMITRESCU, R. G. and COTARLA, I. 2005. Understanding breast cancer risk - where do we stand in 2005? *Journal of Cellular and Molecular Medicine*, 9, 208-221.

ECHCHGADDA, I., SONG, C. S., OH, T., et al. 2007. The Xenobiotic-Sensing Nuclear Receptors Pregnane X Receptor, Constitutive Androstane Receptor, and Orphan Nuclear Receptor Hepatocyte Nuclear Factor 4{alpha} in the Regulation of Human Steroid-/Bile Acid-Sulfotransferase. *Molecular Endocrinology*, 21, 2099-2111.

EDWARDS, R. J., ADAMS, D. A., WATTS, P. S., et al. 1998. Development of a Comprehensive Panel of Antibodies against the Major Xenobiotic Metabolising Forms of Cytochrome P450 in Humans. *Biochemical Pharmacology*, 56, 377-387.

EGGERTSEN, G., OLIN, M., ANDERSSON, U., et al. 1996. Molecular Cloning and Expression of Rabbit Sterol 12alpha -Hydroxylase. *The Journal of Biological Chemistry*, 271, 32269-32275.

EL-RAYES, B. F., ALI, S., HEILBRUN, L. K., et al. 2003. Cytochrome P450 and Glutathione Transferase Expression in Human Breast Cancer. *Clinical Cancer Research*, 9, 1705-1709.

ENGELS, F. K., TEN TIJE, A. J., BAKER, S. D., et al. 2004. Effect of cytochrome P450 3A4 inhibition on the pharmacokinetics of docetaxel. *Clinical Pharmacology & Therapeutics*, 75, 448-454.

EVANS, B. R., KARCHNER, S. I., ALLAN, L. L., et al. 2008. Repression of Aryl Hydrocarbon Receptor (AHR) Signaling by AHR Repressor: Role of DNA Binding and Competition for AHR Nuclear Translocator. *Molecular Pharmacology*, 73, 387-398.

FADDY, H. M., ROBINSON, J. A., LEE, W. J., et al. 2006. Peroxisome proliferatoractivated receptor [alpha] expression is regulated by estrogen receptor [alpha] and modulates the response of MCF-7 cells to sodium butyrate. *The International Journal of Biochemistry & Cell Biology*, 38, 255-266.

FAN, Y.-S., SASI, R., LEE, C. S., et al. 1992. Localization of the human CYP17 gene (cytochrome P450-17-alpha) to 10q24.3 by fluorescence in situ hybridization and simultaneous chromosome banding. *Genomics*, 14, 1110-1111.

FARHAN, H., WAHALA, K. and CROSS, H. S. 2003. Genistein inhibits Vitamin D hydroxylases CYP24 and CYP27B1 expression in prostate cells. *The Journal of Steroid Biochemistry and Molecular Biology*, 84, 423-429.

FEDOR, M. J. 2008. Alternative Splicing Minireview Series: Combinatorial Control Facilitates Splicing Regulation of Gene Expression and Enhances Genome Diversity. *The Journal of Biological Chemistry*, 283, 1209-1210.

FERNANDEZ, S. V., RUSSO, I. H. and RUSSO, J. 2006. Estradiol and its metabolites 4hydroxyestradiol and 2-hydroxyestradiol induce mutations in human breast epithelial cells. *International Journal of Cancer*, 118, 1862-1868.

FERNANDEZ-SALGUERO, P., HOFFMAN, S. M., CHOLERTON, S., et al. 1995. A genetic polymorphism in coumarin 7-hydroxylation: sequence of the human CYP2A genes and identification of variant CYP2A6 alleles. *American Journal of Human Genetics*, 57, 651-660.

FERREIRA, E. N., GALANTE, P. A., CARRARO, D. M., et al. 2007. Alternative splicing: a bioinformatics perspective. *Molecular BioSystems*, *3*, 473-477.

FISCHER, M., KNOLL, M., SIRIM, D., et al. 2007. The Cytochrome P450 Engineering Database: a navigation and prediction tool for the cytochrome P450 protein family. *Bioinformatics*, 23, 2015-2017.

FISHER, D., SEIFERT, M., BECKER, S., et al. 2007. 25-hydroxyvitamin D3 1alphahydroxylase splice variants in breast cell lines MCF-7 and MCF-10. *Cancer Genomics Proteomics*, 4, 295-300.

FITZGERALD, P., TENG, M., CHANDRARATNA, R. A. S., et al. 1997. Retinoic Acid Receptor {alpha} Expression Correlates with Retinoid-induced Growth Inhibition of Human Breast Cancer Cells Regardless of Estrogen Receptor Status. *Cancer Research*, 57, 2642-2650.

FORRESTER, L., NEAL, G., JUDAH, D., et al. 1990. Evidence for Involvement of Multiple Forms of Cytochrome P-450 in Aflatoxin B1 Metabolism in Human Liver. *Proceedings of the National Academy of Sciences*, 87, 8306-8310.

FRESHNEY, R. I. 2005. *Culture of animal cells: a manual of basic technique*, Wiley-Liss, New York, USA.

FURUYA, H., MEYER, U. A., GELBOIN, H. V., et al. 1991. Polymerase chain reaction directed identification, cloing and quantification of human CYP2C18 mRNA. *Molecular Pharmacology*, 40, 374-382.

GAFVELS, M., OLIN, M., CHOWDHARY, B. P., et al. 1999. Structure and Chromosomal Assignment of the Sterol 12[alpha]-Hydroxylase Gene (CYP8B1) in Human and Mouse: Eukaryotic Cytochrome P-450 Gene Devoid of Introns. *Genomics*, 56, 184-196.

GALANTE, P. A., SAKABE, N. J., KIRSCHBAUM-SLAGER, N., et al. 2004. Detection and evaluation of intron retention events in the human transcriptome. *RNA*, 10, 757-765.

GELLNER, K., EISELT, R., HUSTERT, E., et al. 2001. Genomic organization of the human CYP3A locus: identification of a new, inducible CYP3A gene. *Pharmacogenetics*, 11, 111-121.

GERMAIN, P., ALTUCCI, L., BOURGUET, W., et al. 2003. Nuclear receptor superfamily: Principles of signaling. *Pure and Applied Chemistry*, 75, 1619-1664.

GERVASINI, G., GARCIA-MARTIN, E., LADERO, J., et al. 2007. Genetic variability in CYP3A4 and CYP3A5 in primary liver, gastric and colorectal cancer patients. *BMC Cancer*, 7, 118.

GIBSON, L. J., DAWSON, C. K., LAWRENCE, D. H., et al. 2007. Aromatase inhibitors for treatment of advanced breast cancer in postmenopausal women. *Cochrane database of systematic reviews (Online)*, 24, CD003370.

GIBSON, P., GILL, J. H., KHAN, P. A., et al. 2003. Cytochrome P450 1B1 (CYP1B1) Is Overexpressed in Human Colon Adenocarcinomas Relative to Normal Colon: Implications for Drug Development. *Molecular Cancer Therapeutics*, 2, 527-534.

GODA, R., NAGAI, D., AKIYAMA, Y., et al. 2006. Detection of a new N-oxidized metabolite of flutamide, N-[4-nitro-3-(trifluoromethyl)phenyl]hydroxylamine, in human liver microsomes and urine of prostate cancer patients. *Drug Metabolism and Disposition*, 34, 828-835.

GOLDHIRSCH, A., COATES, A., GELBER, R., et al. 2006. First--select the target: better choice of adjuvant treatments for breast cancer patients. *Annals of Oncology*, 17, 1772-1776.

GOLDHIRSCH, A., WOOD, W. C., GELBER, R. D., et al. 2007. Progress and promise: highlights of the international expert consensus on the primary therapy of early breast cancer 2007. *Annals of Oncology*, 18, 1133-1144.

GOLDSTEIN, J. A. 2001. Clinical relevance of genetic polymorphisms in the human CYP2C subfamily. *British Journal of Clinical Pharmacology*, 52, 349–355.

GOLDSTEIN, J. A. and DE MORAIS, S. M. 1994. Biochemistry and molecular biology of the human CYP2C subfamily. *Pharmacogenetics*, *4*, 285-299.

GONZALEZ, F. J. and GELBOIN, H. 1994. Role of human cytochromes P450 in the metabolic activation of chemical carcinogens and toxins. *Drug Metabolism Reviews*, 26, 165-183.

GONZÁLEZ-SANCHO, J. M., LARRIBA, M. J., ORDÓÑEZ-MORÁN, P., et al. 2006. Effects of 1alpha,25-dihydroxyvitamin D3 in human colon cancer cells. *Anticancer Research*, 26, 2669-2681.

GOODALL, G. J. and FILIPOWICZ, W. 1991. Different effects of intron nucleotide composition and secondary structure on pre-mRNA splicing in monocot and dicot plants. *The EMBO Journal*, 10, 2635-1644.

GORDON, L. A., MULLIGAN, K. T., MAXWELL-JONES, H., et al. 2003. Breast cell invasive potential relates to the myoepithelial phenotype. *International Journal of Cancer*, 106, 8-16.

GOTTFRIED, E., REHLI, M., HAHN, J., et al. 2006. Monocyte-derived cells express CYP27A1 and convert vitamin D3 into its active metabolite. *Biochemical and Biophysical Research Communications*, 349, 209-213.

GOUGH, A. C., SMITH, C. A. D., HOWELL, S. M., et al. 1993. Localization of the CYP2D Gene Locus to Human Chromosome 22q13.1 by Polymerase Chain Reaction, in Situ Hybridization, and Linkage Analysis. *Genomics*, 15, 430-432.

GRAY, I. C., NOBILE, C., MURESU, R., et al. 1995. A 2.4-Megabase Physical Map Spanning the CYP2C Gene Cluster on Chromosome 10q24. *Genomics*, 28, 328-332.

GREENE, A., ATHREYA, B., LEHR, H., et al. 1967. Viability of cell cultures following extended preservation in liquid nitrogen. *Proceedings of the Society for Experimental Biology and Medicine*, 124, 1302-1307.

GRIBBEN, J. G., RYAN, D. P., BOYAJIAN, R., et al. 2005. Unexpected Association between Induction of Immunity to the Universal Tumor Antigen CYP1B1 and Response to Next Therapy. *Clinical Cancer Research*, 11, 4430-4436.

GRUVBERGER-SAAL, S. K., BENDAHL, P.-O., SAAL, L. H., et al. 2007. Estrogen Receptor {beta} Expression Is Associated with Tamoxifen Response in ER{alpha}-Negative Breast Carcinoma. *Clinical Cancer Research*, 13, 1987-1994.

GUENGERICH, F. P. 1999. Cytochrome P-450 3A4: regulation and role in drug metabolism. *Annual Review of Pharmacology and Toxicology*, 39, 1-17.

GUENGERICH, F. P. 2004. Cytochrome P450: what have we learned and what are the future issues? *Drug Metabolism Reviews*, 36, 159-197.

GUENGERICH, F. P. and SHIMADA, T. 1991. Oxidation of toxic and carcinogen chemicals by human cytochrome P450 enzymes. *Chemical Research in Toxicology*, 4, 391-407.

GUENGERICH, F. P., WU, Z. L. and BARTLESON, C. J. 2005. Function of human cytochrome P450s: characterization of the orphans. *Biochemical and Biophysical Research Communications*, 338, 465-469.

GUIDICE, J.-M. L., MAREZ, D., SABBAGH, N., et al. 1997. Evidence for CYP2D6 Expression in Human Lung. *Biochemical and Biophysical Research Communications*, 241, 79-85.

GUPTA, V. and WAGNER, B. J. 2003. Expression of the Functional Glucocorticoid Receptor in Mouse and Human Lens Epithelial Cells. *Investigative Ophthalmology & Visual Science*, 44, 2041-2046.

HAAS, S., PIERL, C., HARTH, V., et al. 2006. Expression of xenobiotic and steroid hormone metabolizing enzymes in human breast carcinomas. *International Journal of Cancer*, 119, 1785-1791.

HAHN, C. N., BAKER, E., LASLO, P., et al. 1993. Localization of the human vitamin D 24-hydroxylase gene (CYP24) to chromosome 20q13.2-->q13.3. *Cytogenetics and Cell Genetics*, 62, 192-193.

HAKKI, T. and BERNHARDT, R. 2006. CYP17- and CYP11B-dependent steroid hydroxylases as drug development targets. *Pharmacology & Therapeutics*, 111, 27-52.

HAM, R. G. 1965. Clonal growth of mammalian cells in a chemically defined, synthetic medium. *Proceedings of the National Academy of Sciences of the United States of America*, 53, 288-293.

HAMADEH, H. K. and AFSHARI, C. 2004. *Toxicogenomics: Principles and Applications*, John Wiley and Sons Inc, Hoboken, NJ, USA.

HANDSCHIN, C. and MEYER, U. A. 2003. Induction of Drug Metabolism: The Role of Nuclear Receptors. *Pharmacological Reviews*, 55, 649-673.

HANNA, I. H., DAWLING, S., ROODI, N., et al. 2000. Cytochrome P450 1B1 (CYP1B1) pharmacogenetics: association of polymorphisms with functional differences in estrogen hydroxylation activity. *Cancer Research*, 60, 3440-3444.

HARLAN, L. C., CLEGG, L. X., ABRAMS, J., et al. 2006. Community-Based Use of Chemotherapy and Hormonal Therapy for Early-Stage Breast Cancer: 1987-2000 10.1200/JCO.2005.03.5840. *Journal of Clinical Oncology*, 24, 872-877.

HE, X. Y., SHEN, J., HU, W. Y., et al. 2004. Identification of Val(117) and Arg(372) as critical amino acid residues for the activity difference between human CYP2A6 and CYP2A13 in coumarin 7-hydroxylation. *Archives of Biochemistry and Biophysics*, 427, 143–153.

HELLMOLD, H., RYLANDER, T., MAGNUSSON, M., et al. 1998. Characterization of Cytochrome P450 Enzymes in Human Breast Tissue from Reduction Mammaplasties. *The Journal of Clinical Endocrinology and Metabolism*, 83, 886-895.

HENDERSON, B. E. and FEIGELSON, H. S. 2000. Hormonal carcinogenesis. *Carcinogenesis*, 21, 427-433.

HIGASHI, E., FUKAMI, T., ITOH, M., et al. 2007. Human CYP2A6 Is Induced by Estrogen via Estrogen Receptor. *Drug Metabolism and Disposition*, 35, 1935-1941.

HIGASHI, Y., YOSHIOKA, H., YAMANE, M., et al. 1986. Complete nucleotide sequence of two steroid 21-hydroxylase genes tandemly arranged in human chromosome: a pseudogene and a genuine gene. *Proceedings of the National Academy of Sciences*, 83, 2841–2845.

HIHI, A. K., MICHALIK, L. and WAHLI, W. 2002. PPARs: transcriptional effectors of fatty acids and their derivatives. *Cellular and Molecular Life Sciences*, 59, 790-798.

HONKAKOSKI, P. and NEGISHI, M. 2000. Regulation of cytochrome P450 (CYP) genes by nuclear receptors. *The Biochemical Journal*, 347, 321-337.

HSU, M.-H., SAVAS, U. Z., GRIFFIN, K. J., et al. 2007. Human Cytochrome P450 Family 4 Enzymes: Function, Genetic Variation and Regulation. *Drug Metabolism Reviews*, 39, 515-538.

HUANG, Z., FASCO, M., FIGGE, H., et al. 1996. Expression of cytochromes P450 in human breast tissue and tumors. *Drug Metabolism and Disposition*, 24, 899-905.

IKEYA, K., JAISWAL, A. K., OWENS, R. A., et al. 1989. Human CYP1A2: sequence, gene structure, comparison with the mouse and rat orthologous gene, and differences in liver 1A2 mRNA expression. *Molecular Endocrinology*, **3**, 1399-1408.

INGELMAN-SUNDBERG, M. 2005a. Genetic polymorphisms of cytochrome P450 2D6 (CYP2D6): clinical consequences, evolutionary aspects and functional diversity. *The Pharmacogenomics Journal*, 5, 6-13.

INGELMAN-SUNDBERG, M. 2005b. The human genome project and novel aspects of cytochrome P450 research. *Toxicology and Applied Pharmacology*, 207, 52-56.

INGELMAN-SUNDBERG, M., SIM, S. C., GOMEZ, A., et al. 2007. Influence of cytochrome P450 polymorphisms on drug therapies: Pharmacogenetic, pharmacoepigenetic and clinical aspects. *Pharmacology & Therapeutics*, 116, 496-526.

INOUE, K., INAZAWA, J., NAKAGAWA, H., et al. 1992. Assignment of the human cytochrome P-450 nifedipine oxidase gene (CYP3A4) to chromosome 7 at band q22.1 by fluorescence in situ hybridization. *The Japanese Journal of Human Genetics*, 37, 133-138.

ISCAN, M., KLAAVUNIEMI, T., COBAN, T., et al. 2001. The expression of cytochrome P450 enzymes in human breast tumours and normal breast tissue. *Breast Cancer Research and Treatment*, 70, 47-54.

ITO, O., NAKAMURA, Y., TAN, L., et al. 2007. Expression of cytochrome P-450 4 enzymes in the kidney and liver: regulation by PPAR and species-difference between rat and human. *Molecular and Cellular Biochemistry*, 299, 1-3.

IWANARI, M., NAKAJIMA, M., KIZU, R., et al. 2002. Induction of CYP1A1, CYP1A2, and CYP1B1 mRNAs by nitropolycyclic aromatic hydrocarbons in various human tissuederived cells: chemical-, cytochrome P450 isoform-, and cell-specific differences. *Archives of Toxicology*, 76, 287-298.

JAISWAL, A. K., NEBERT, D. W. and GONZALEZ, F. J. 1986. Human P(3)450: cDNA and complete amino acid sequence. *Nucleic Acids Research*, 14, 6773-6774.

JIANG, J.-G., CHEN, C.-L., CARD, J. W., et al. 2005. Cytochrome P450 2J2 Promotes the Neoplastic Phenotype of Carcinoma Cells and Is Up-regulated in Human Tumors. *Cancer Research*, 65, 4707-4715.

JIANG, J.-G., NING, Y.-G., CHEN, C., et al. 2007. Cytochrome P450 Epoxygenase Promotes Human Cancer Metastasis. *Cancer Research*, 67, 6665-6674.

JIN, Y., DESTA, Z., STEARNS, V., et al. 2005. CYP2D6 Genotype, Antidepressant Use, and Tamoxifen Metabolism During Adjuvant Breast Cancer Treatment. *Journal of the National Cancer Institute*, 97, 30-39.

JOHN, M. S. B. and STIRLING, D. 2003. PCR Protocols, Humana Press, Totowa, NJ.

JOHNSRUD, E. K., KOUKOURITAKI, S. B., DIVAKARAN, K., et al. 2003. Human Hepatic CYP2E1 Expression during Development. *The Journal of Pharmacology and Experimental Therapeutics*, 307, 402-407.

JONAT, W., HILPERT, F. and KAUFMANN, M. 2007. Aromatase inhibitors: a safety comparison. *Expert Opinion on Drug Safety*, 6, 165-174.

JONES, G., STRUGNELL, S. A. and DELUCA, H. F. 1998. Current Understanding of the Molecular Actions of Vitamin D. *Physiological Reviews*, 78, 1193-1231.

JONES, L. W., HAYKOWSKY, M. J., SWARTZ, J. J., et al. 2007. Early Breast Cancer Therapy and Cardiovascular Injury. *Journal of the American College of Cardiology*, 50, 1435-1441.

JONES, P., QUI, J. and RICKWOOD, D. 1994. *RNA: Isolation and Analysis*, Bios Scientific Publishers Ltd, Oxford, UK.

KAINZ, P. 2000. The PCR plateau phase - towards an understanding of its limitations. *Biochimica et Biophysica Acta (BBA) - Gene Structure and Expression*, 1494, 23-27.

KALSOTRA, A., TURMAN, C. M., KIKUTA, Y., et al. 2004. Expression and characterization of human cytochrome P450 4F11: Putative role in the metabolism of therapeutic drugs and eicosanoids. *Toxicology and applied pharmacology*, 199, 295-304.

KARLGEN, M., BACKLUND, M., JOHANSSON, I., et al. 2004. Characterization and tissue distribution of a novel human cytochrome P450-CYP2U1. *Biochemical and Biophysical Research Communications*, 315, 679-685.

KARLGREN, M., GOMEZ, A., STARK, K., et al. 2006. Tumor-specific expression of the novel cytochrome P450 enzyme, CYP2W1. *Biochemical and Biophysical Research Communications*, 341, 451-458.

KARLGREN, M. and INGELMAN-SUNDBERG, M. 2007. Tumour-specific expression of CYP2W1: its potential as a drug target in cancer therapy. *Expert Opinion on Therapeutic Targets*, 11, 61-67.

KARLGREN, M., MIURA, S.-I. and INGELMAN-SUNDBERG, M. 2005. Novel extrahepatic cytochrome P450s. *Toxicology and Applied Pharmacology*, 207, 57-61.

KARTHA, J. S. and YOST, G. S. 2008. Mechanism-Based Inactivation of Lung-Selective Cytochrome P450 CYP2F Enzymes. *Drug Metabolism and Disposition*, 36, 155-162.

KAWAJIRI, K. and FUJII-KURIYAMA, Y. 2007. Cytochrome P450 gene regulation and physiological functions mediated by the aryl hydrocarbon receptor. *Archives of Biochemistry and Biophysics Highlight Issue: Enzymology of Drug Metabolism and Toxicology*, 464, 207-212.

KELLER, H., GIVEL, F., PERROUD, M., et al. 1995. Signaling cross-talk between peroxisome proliferator-activated receptor/retinoid X receptor and estrogen receptor through estrogen response elements. *Molecular Endocrinology*, 9, 794-804.

KEMMIS, C. M., SALVADOR, S. M., SMITH, K. M., et al. 2006. Human Mammary Epithelial Cells Express CYP27B1 and Are Growth Inhibited by 25-Hydroxyvitamin D-3, the Major Circulating Form of Vitamin D-3. *The Journal of Nutrition*, 136, 887-892.

KHAN, S., LIU, S., STONER, M., et al. 2007. Cobaltous chloride and hypoxia inhibit aryl hydrocarbon receptor-mediated responses in breast cancer cells. *Toxicology and Applied Pharmacology*, 223, 28-38.

KHORCHIDE, M., LECHNER, D. and CROSS, H. S. 2005. Epigenetic regulation of Vitamin D hydroxylase expression and activity in normal and malignant human prostate cells. *The Journal of Steroid Biochemistry and Molecular Biology*, 93, 167-172.

KIM, D. and GUENGERICH, F. P. 2005. Cytochrome P450 activation of arylamines and heterocyclic amines. *Annual Review of Pharmacology and Toxicology*, 45, 27-49.

KIM, E.-Y., IWATA, H., SUDA, T., et al. 2005. Aryl hydrocarbon receptor (AHR) and AHR nuclear translocator (ARNT) expression in Baikal seal (Pusa sibirica) and association with 2,3,7,8-TCDD toxic equivalents and CYP1 expression levels. *Comparative Biochemistry and Physiology Part C: Toxicology & Pharmacology*, 141, 281-291.

KLEIN, K., TATZEL, S., RAIMUNDO, S., et al. 2007. A Natural Variant of the Heme-Binding Signature (R441C) Resulting in Complete Loss of Function of CYP2D6. *Drug Metabolism and Disposition*, 35, 1247-1250.

KLINGENBERG, M. 1958. Pigments of rat liver microsomes. *Archives of Biochemistry and Biophysics*, 75, 376-386.

KOLARS, J. C., SCHMIEDLIN-REN, P., SCHUETZ, J. D., et al. 1992. Identification of rifampin-inducible P450IIIA4 (CYP3A4) in human small bowel enterocytes. *The Journal of Clinical Investigation*, 90, 1871-1878.

KÖLBLE, K. 1993. Regional mapping of short tandem repeats on human chromosome 10: cytochrome P450 gene CYP2E, D10S196, D10S220, and D10S225. *Genomics*, 18, 702-704.

KOMMADDI, R. P., TURMAN, C. M., MOORTHY, B., et al. 2007. An alternatively spliced cytochrome P4501A1 in human brain fails to bioactivate polycyclic aromatic hydrocarbons to DNA-reactive metabolites. *Journal of Neurochemistry*, 102, 867-877.

KUMARAKULASINGHAM, M., ROONEY, P. H., DUNDAS, S. R., et al. 2005. Cytochrome P450 Profile of Colorectal Cancer: Identification of Markers of Prognosis. *Clinical Cancer Research*, 11, 3758-3765.

LACROIX, M. and LECLERCQ, G. 2004. Relevance of breast cancer cell lines as models for breast tumours: an update. *Breast Cancer Research and Treatment*, 83, 249-289.

LAMBA, V., LAMBA, J., YASUDA, K., et al. 2003. Hepatic CYP2B6 Expression: Gender and Ethnic Differences and Relationship to CYP2B6 Genotype and CAR (Constitutive Androstane Receptor) Expression. *The Journal of Pharmacology and Experimental Therapeutics*, 307, 906-922.

LAU, K.-M., MOK, S. C. and HO, S.-M. 1999. Expression of human estrogen receptoralpha and -beta, progesterone receptor, and androgen receptor mRNA in normal and malignant ovarian epithelial cells. *Proceedings of the National Academy of Sciences of the United States of America*, 96, 5722-5727.

LEE, A. J., CONNEY, A. H. and ZHU, B. T. 2003. Human Cytochrome P450 3A7 Has a Distinct High Catalytic Activity for the 16 $\alpha$ -Hydroxylation of Estrone but not 17 $\beta$ -Estradiol. *Cancer Research*, 63, 6532–6536.

LEE, S. S. T., BUTERS, J. T. M., PINEAU, T., et al. 1996. Role of CYP2E1 in the Hepatotoxicity of Acetaminophen. *The Journal of Biological Chemistry*, 271, 12063-12067.

LEFEVRE, C., BOUADJAR, B., FERRAND, V., et al. 2006. Mutations in a new cytochrome P450 gene in lamellar ichthyosis type 3. *Human Molecular Genetics*, 15, 767-776.

LEIBOVITZ, A. 1963. The growth and maintenance of tissue-cell cultures in free gas exchange with the atmosphere. *American Journal of Hygiene*, 78, 173-180.

LEMBERGER, T., STAELS, B., SALADIN, R., et al. 1994. Regulation of the peroxisome proliferator-activated receptor alpha gene by glucocorticoids. *The Journal of Biological Chemistry*, 269, 24527-24530.

LEONG, C. O., GASKELL, M., MARTIN, E. A., et al. 2003. Antitumour 2-(4-aminophenyl)benzothiazoles generate DNA adducts in sensitive tumour cells in vitro and in vivo. *British Journal of Cancer*, 88, 470-477.

LEUNG, Y.-K., LAU, K.-M., MOBLEY, J., et al. 2005. Overexpression of Cytochrome P450 1A1 and Its Novel Spliced Variant in Ovarian Cancer Cells: Alternative Subcellular Enzyme Compartmentation May Contribute to Carcinogenesis. *Cancer Research*, 65, 3726-3734.

LI, A., JIAO, X., MUNIER, F. L., et al. 2004. Bietti crystalline corneoretinal dystrophy is caused by mutations in the novel gene CYP4V2. *American Journal of Human Genetics*, 74, 817-826.

LI, Y., MEZEI, O. and SHAY, N. F. 2007. Human and Murine Hepatic Sterol-12-{alpha}-Hydroxylase and Other Xenobiotic Metabolism mRNA Are Upregulated by Soy Isoflavones. The *Journal of Nutrition*, 137, 1705-1712.

LI-HAWKINS, J., LUND, E. G., BRONSON, A. D., et al. 2000. Expression Cloning of an Oxysterol 7alpha -Hydroxylase Selective for 24-Hydroxycholesterol. *The Journal of Biological Chemistry*, 275, 16543-16549.

LOAIZA-PEREZ, A. I., KENNEY, S., BOSWELL, J., et al. 2004. Aryl hydrocarbon receptor activation of an antitumor aminoflavone: Basis of selective toxicity for MCF-7 breast tumor cells. *Molecular Cancer Therapeutics*, 3, 715-725.

LODGE, J., LUND, P. and MINCHIN, S. 2007. *Gene cloning*, Taylor & Francis Group, Abingdon, UK.

LU, H. and WAXMAN, D. J. 2005. Antitumor Activity of Methoxymorpholinyl Doxorubicin: Potentiation by Cytochrome P450 3A Metabolism. *Molecular Pharmacology*, 67, 212-219.

LU, J., JIANG, C., FONTAINE, S., et al. 1995. ras may mediate mammary cancer promotion by high fat. *Nutrition and Cancer*, 23, 283-290.

LU, M., MIRA-Y-LOPEZ, R., NAKAJO, S., et al. 2005. Expression of estrogen receptor [alpha], retinoic acid receptor [alpha] and cellular retinoic acid binding protein II genes is coordinately regulated in human breast cancer cells. *Oncogene*, 24, 4362-4369.

LUND, E. G., GUILEYARDO, J. M. and RUSSELL, D. W. 1999. cDNA cloning of cholesterol 24-hydroxylase, a mediator of cholesterol homeostasis in the brain. *Proceedings of the National Academy of Sciences of the United States of America*, 96, 7238-7243.

LUPBERGER, J., KREUZER, K.-A., BASKAYNAK, G., et al. 2002. Quantitative analysis of beta-actin, beta-2-microglobulin and porphobilinogen deaminase mRNA and their comparison as control transcripts for RT-PCR. *Molecular and Cellular Probes*, 16, 25-30.

MA, J., RAMACHANDRAN, S., FIEDOREK, J., FREDERICK T., et al. 1998. Mapping of the CYP2J Cytochrome P450 Genes to Human Chromosome 1 and Mouse Chromosome 4. *Genomics*, 49, 152-155.

MACLEAN, G., ABU-ABED, S., DOLLE, P., et al. 2001. Cloning of a novel retinoic-acid metabolizing cytochrome P450, Cyp26B1, and comparative expression analysis with Cyp26A1 during early murine development. *Mechanisms of Development*, 107, 195-201.

MACLEAN, G., LI, H., METZGER, D., et al. 2007. Apoptotic Extinction of Germ Cells in Testes of Cyp26b1 Knockout Mice. *Endocrinology*, 148, 4560-4567.

MAECKER, B., SHERR, D. H., VONDERHEIDE, R. H., et al. 2003. The shared tumorassociated antigen cytochrome P450 1B1 is recognized by specific cytotoxic T cells. *Blood*, 102, 3287-3294.

MAN, Y. and SANG, Q. 2005. Myoepithelial cell layer disruption and human breast cancer invasion. *Breast Cancer Research*, **7**, P3.06.

MARTIN, C., ROSS, M., CHAPMAN, K. E., et al. 2004. CYP7B generates a selective estrogen receptor beta agonist in human prostate. *The Journal of Clinical Endocrinology and Metabolism*, 89, 2928-2935.

MARTIN, R. 1996. Gel electrophoresis: nucleic acids, BIOS Scientific, Oxford, UK.

MASSON, L. F., SHARP, L., COTTON, S. C., et al. 2005. Cytochrome P-450 1A1 Gene Polymorphisms and Risk of Breast Cancer: A HuGE Review. *American Journal of Epidemiology*, 161, 901-915.

MASUDA, S., STRUGNELL, S. A., KNUTSON, J. C., et al. 2006. Evidence for the activation of 1[alpha]-hydroxyvitamin D2 by 25-hydroxyvitamin D-24-hydroxylase: Delineation of pathways involving 1[alpha],24-dihydroxyvitamin D2 and 1[alpha],25-dihydroxyvitamin D2. *Biochimica et Biophysica Acta (BBA) - Molecular and Cell Biology of Lipids*, 1761, 221-234.

MASUYAMA, H., HIRAMATSU, Y., KODAMA, J.-I., et al. 2003. Expression and Potential Roles of Pregnane X Receptor in Endometrial Cancer. *The Journal of Clinical Endocrinology and Metabolism*, 88, 4446-4454.

MASUYAMA, H., NAKATSUKASA, H., TAKAMOTO, N., et al. 2007. Down-Regulation of Pregnane X Receptor Contributes to Cell Growth Inhibition and Apoptosis by Anticancer Agents in Endometrial Cancer Cells. *Molecular Pharmacology*, 72, 1045-1053.

MASUYAMA, H., SUWAKI, N., TATEISHI, Y., et al. 2005. The Pregnane X Receptor Regulates Gene Expression in a Ligand- and Promoter- Selective Fashion. *Molecular Endocrinology*, 19, 1170-1180.

MATHIJSSEN, R. H. J. and VAN SCHAIK, R. H. N. 2006. Genotyping and phenotyping cytochrome P450: Perspectives for cancer treatment. *European Journal of Cancer*, 42, 141-148.

MATTHEWS, J. and GUSTAFSSON, J. A. 2006. Estrogen receptor and aryl hydrocarbon receptor signaling pathways. *Nuclear Receptor Signaling*, 4:e016.

MATTHEWS, J., WIHLEN, B., THOMSEN, J., et al. 2005. Aryl Hydrocarbon Receptor-Mediated Transcription: Ligand-Dependent Recruitment of Estrogen Receptor {alpha} to 2,3,7,8-Tetrachlorodibenzo- p-Dioxin-Responsive Promoters. *Molecular and Cellular Biology*, 25, 5317-5328. McDONNELL, D. P. 2004. The molecular determinants of estrogen receptor pharmacology. *Maturitas*, 48, 7-12.

McFADYEN, M. C. E., BREEMAN, S., PAYNE, S., et al. 1999. Immunohistochemical Localization of Cytochrome P450 CYP1B1 in Breast Cancer with Monoclonal Antibodies Specific for CYP1B1. *The Journal of Histochemistry and Cytochemistry*, 47, 1457-1464.

McFADYEN, M. C. E., CRUICKSHANK, M. E., MILLER, I. D., et al. 2001. Cytochrome P450 CYP1B1 over-expression in primary and metastatic ovarian cancer. *British Journal of Cancer*, 85, 242-246.

McFADYEN, M. C. E., MELVIN, W. T. and MURRAY, G. I. 2004. Cytochrome P450 enzymes: Novel options for cancer therapeutics. *Molecular Cancer Therapeutics*, *3*, 363-371.

McFADYEN, M. C. E., ROONEY, P. H., MELVIN, W. T., et al. 2003. Quantitative analysis of the Ah receptor/cytochrome P450 CYP1B1/CYP1A1 signalling pathway. *Biochemical Pharmacology*, 65, 1663-1674.

McKAY, J. A., MELVIN, W. T., AH-SEE, A. K., et al. 1995. Expression of cytochrome P450 CYP1B1 in breast cancer. *FEBS Letters*, 374, 270-272.

McKEOWN, S. R., COWEN, R. L. and WILLIAMS, K. J. 2007. Bioreductive Drugs: from Concept to Clinic. *Clinical Oncology Importance of Radiobiology to Cancer Therapy: Current Practice and Future Perspectives - Radiobiology & Cancer Therapy*, 19, 427-442.

McKINNON, R. A., BURGESS, W. M., GONZALEZ, F. J., et al. 1994. Species-specific expression of CYP4B1 in rabbit and human gastrointestinal tissues. *Pharmacogenetics*, 4, 260-270.

MEYER, U. A. 2007. Endo-Xenobiotic Crosstalk and the Regulation of Cytochromes P450. *Drug Metabolism Reviews*, 39, 639 - 646.

MICHAEL, M. and DOHERTY, M. M. 2005. Tumoral Drug Metabolism: Overview and Its Implications for Cancer Therapy. *Journal of Clinical Oncology*, 23, 205-229.

MIKI, Y., SUZUKI, T., KITADA, K., et al. 2006. Expression of the Steroid and Xenobiotic Receptor and Its Possible Target Gene, Organic Anion Transporting Polypeptide-A, in Human Breast Carcinoma. *Cancer Research*, 66, 535-542.

MILES, J. S., McLAREN, A. W., FORRESTER, L. M., et al. 1990. Identification of the human liver cytochrome P-450 responsible for coumarin 7-hydroxylase activity. *The Biochemical Journal*, 267, 365-371.

MILES, J. S., SPURR, N. K., GOUGH, A. C., et al. 1988. A novel human cytochrome P4S0 gene (P450IIB): chromosomal localization and evidence for alternative splicing. *Nucleic Acids Research*, 16, 5783-5795.

MIMORI, K., TANAKA, Y., YOSHINAGA, K., et al. 2004. Clinical significance of the overexpression of the candidate oncogene CYP24 in esophageal cancer. *Annals of Oncology*, 15, 236-241.

MIMURA, M., BABA, T., YAMAZAKI, H., et al. 1993. Characterization of cytochrome P-450 2B6 in human liver microsomes. *Drug Metabolism and Disposition*, 21, 1048-1056.

MIN, G. 2006. Different Modulation of ER-Mediated Transactivation by Xenobiotic Nuclear Receptors Depending on the Estrogen Response Elements and Estrogen Target Cell Types. *Annals of the New York Academy of Sciences*, 1091, 244-257.

MIN, G., KIM, H., BAE, Y., et al. 2002. Inhibitory Cross-talk between Estrogen Receptor (ER) and Constitutively Activated Androstane Receptor (CAR). CAR inhibits ERmediated signaling pathway by squelching p160 coactivators. *The Journal of Biological Chemistry*, 277, 34626-34633.

MIRA-Y-LOPEZ, R., ZHENG, W. L., KUPPUMBATTI, Y. S., et al. 2000. Retinol conversion to retinoic acid is impaired in breast cancer cell lines relative to normal cells. *Journal of Cellular Physiology*, 185, 302-309.

MIYATA, A., HARA, S., YOKOYAMA, C., et al. 1994. Molecular Cloning and Expression of Human Prostacyclin Synthase. *Biochemical and Biophysical Research Communications*, 200, 1728-1734.

MIYOSHI, Y., TAGUCHI, T., KIM, S. J., et al. 2005. Prediction of response to docetaxel by immunohistochemical analysis of CYP3A4 expression in human breast cancers. *Breast Cancer*, 12, 11-15.

MODREK, B. and LEE, C. 2002. A genomic view of alternative splicing. *Nature Genetics*, 30, 13-19.

MODUGNO, F., KNOLL, C., KANBOUR-SHAKIR, A., et al. 2003. A Potential Role for the Estrogen-metabolizing Cytochrome P450 Enzymes in Human Breast Carcinogenesis. *Breast Cancer Research and Treatment*, 82, 191-197.

MOHAMMED, K. and SHERVINGTON, A. 2007. Can CYP1A1 siRNA be an effective treatment for lung cancer? *Cellular & Molecular Biology Letters*, Epub ahead of print.

MONTELLANO, P. R. O. D. 1995. Cytochrome P450: Structure, Mechanism, and Biochemistry, Plenum Press, New York, USA.

MOORE, D. D., KATO, S., XIE, W., et al. 2006. International Union of Pharmacology. LXII. The NR1H and NR1I Receptors: Constitutive Androstane Receptor, Pregnene X Receptor, Farnesoid X Receptor {alpha}, Farnesoid X Receptor beta, Liver X Receptor {alpha}, Liver X Receptor beta, and Vitamin D Receptor. *Pharmacological Reviews*, 58, 742-759.

MOORE, G. E., GERNER, R. E. and FRANKLIN, H. A. 1967. Culture of normal human leukocytes. *The Journal of the American Medical Association*, 199, 519-524.

MOORE, L. B., PARKS, D. J., JONES, S. A., et al. 2000. Orphan Nuclear Receptors Constitutive Androstane Receptor and Pregnane X Receptor Share Xenobiotic and Steroid Ligands. *The Journal of Biological Chemistry*, 275, 15122-15127.

MOREAU, A., MAUREL, P., VILAREM, M.-J., et al. 2007a. Constitutive androstane receptor-vitamin D receptor crosstalk: Consequence on CYP24 gene expression. *Biochemical and Biophysical Research Communications*, 360, 76-82.

MOREAU, A., VILAREM, M. J., MAUREL, P., et al. 2007b. Xenoreceptors CAR and PXR Activation and Consequences on Lipid Metabolism, Glucose Homeostasis, and Inflammatory Response. *Molecular Pharmacology*.

MOREIRA, V. M. A., VASAITIS, T. S., NJAR, V. C. O., et al. 2007. Synthesis and evaluation of novel 17-indazole androstene derivatives designed as CYP17 inhibitors. *Steroids*, 72, 939-948.

MORNET, E., DUPONT, J., VITEK, A., et al. 1989. Characterization of two genes encoding human steroid 11 $\beta$ -hydroxylase (P450(11)  $\beta$ ). *The Journal of Biological Chemistry*, 264, 20961-20967.

MORRIS, G. J., NAIDU, S., TOPHAM, A. K., et al. 2007. Differences in breast carcinoma characteristics in newly diagnosed African-American and Caucasian patients. *Cancer*, 110, 876-884.

MOTOJIMA, K. and HIRAI, T. 2006. Peroxisome proliferator-activated receptor a plays a vital role in inducing a detoxification system against plant compounds with crosstalk with other xenobiotic nuclear receptors. *FEBS Journal*, 273, 292–300.

MOULDER, S. and HORTOBAGYI, G. 2008. Advances in the Treatment of Breast Cancer. *Clinical Pharmacology & Therapeutics*, 83, 26-36.

MU, Y.-M., YANASE, T., NISHI, Y., et al. 2000. A nuclear receptor system constituted by RAR and RXR induces aromatase activity in MCF-7 human breast cancer cells. *Molecular and Cellular Endocrinology*, 166, 137-145.

MURRAY, G., TAYLOR, M., McFADYEN, M., et al. 1997. Tumor-specific expression of cytochrome P450 CYP1B1. *Cancer Research*, 57, 3026-3031.

MURRAY, G. I., MELVIN, W. T., GREENLEE, W. F., et al. 2001. Regulation, function, and tissue-specific expression of cytochrome P450 CYP1B1. *Annual Review of Pharmacology and Toxicology*, 41, 297-316.

NAGAMANI, M. and URBAN, R. J. 2003. Expression of messenger ribonucleic acid encoding steroidogenic enzymes in postmenopausal ovaries. *Journal of the Society for Gynecologic Investigation*, 10, 37-40.

NAGASAWA, H., UTO, Y., KIRK, K. L., et al. 2006. Design of hypoxia-targeting drugs as new cancer chemotherapeutics. *Biological & Pharmaceutical Bulletin*, 29, 2335-2342.

NEBERT, D. W. and JONES, J. E. 1989. Regulation of the mammalian cytochrome P1-450 (CYP1A1) gene. *The International Journal of Biochemistry*, 21, 243-252.

NEBERT, D. W. and NELSON, D. R. 2005. In *Encyclopedia of Life Sciences* John Wiley & Sons, Ltd., USA.

NEBERT, D. W., NELSON, D. R., ADESNIK, M., et al. 1989. The P450 superfamily: updated listing of all genes and recommended nomenclature for the chromosomal loci. *DNA*, 8, 1-13.

NEBERT, D. W., NELSON, D. R., COON, M. J., et al. 1991. The P450 superfamily: update on new sequences, gene mapping, and recommended nomenclature. *DNA*, 10, 1-14.

NELSON, D. 2003. Cytochrome P450s in humans. (online) Available from <u>http://drnelson.utmem.edu/P450lect.html</u>. (Accessed 1 March 2008).

NELSON, D. R. 1999. A Second CYP26 P450 in Humans and Zebrafish: CYP26B1. *Archives of Biochemistry and Biophysics*, 371, 345-347.

NELSON, D. R., KOYMANS, L., KAMATAKI, T., et al. 1996. P450 superfamily: update on new sequences, gene mapping, accession numbers and nomenclature. *Pharmacogenetics*, 6, 1-42.

NELSON, D. R., ZELDIN, D. C., HOFFMAN, S. M., et al. 2004. Comparison of cytochrome P450 (CYP) genes from the mouse and human genomes, including nomenclature recommendations for genes, pseudogenes and alternative-splice variants. *Pharmacogenetics*, 14, 1-18.

NEWTON, C. R. and GRAHAM, A. 1997. PCR, BIOS Scientific Ltd., Oxford, UK.

NEZZAR, H., CHIAMBARETTA, F., MARCEAU, G., et al. 2007. Molecular and metabolic retinoid pathways in the human ocular surface. *Molecular Vision*, 11, 1641-1650.

NHAMBURO, P. T., GONZALEZ, F. J., McBRIDE, O. W., et al. 1989. Identification of a new P450 expressed in human lung: complete cDNA sequence, cDNA-directed expression and chromosome mapping. *Biochemistry*, 28, 8060-8066.

NIE, D., CHE, M., ZACHAREK, A., et al. 2004. Differential Expression of Thromboxane Synthase in Prostate Carcinoma: Role in Tumor Cell Motility. *The American Journal of Pathology*, 164, 429-439.

NJAR, V. C. 2002. Cytochrome p450 retinoic acid 4-hydroxylase inhibitors: potential agents for cancer therapy. *Mini Reviews in Medicinal Chemistry*, 2, 261-269.

NJAR, V. C. O., GEDIYA, L., PURUSHOTTAMACHAR, P., et al. 2006. Retinoic acid metabolism blocking agents (RAMBAs) for treatment of cancer and dermatological diseases. *Bioorganic & Medicinal Chemistry*, 14, 4323-4340.

NODE, K., HUO, Y., RUAN, X., et al. 1999. Anti-inflammatory properties of cytochrome P450 epoxygenase-derived eicosanoids. *Science*, 285, 1276-1279.

O'HALLORAN, D., GUYERS, K. and HENDERSON, J. 2004. *Notes on Anatomy and Oncology*, Churchill Livingstone, London, UK.

OKITA, R. T. and OKITA, J. R. 2001. Cytochrome P450 4A fatty acid omega hydroxylases. *Current Drug Metabolism*, 2, 265-281.

OLSON, S. H., BANDERA, E. V. and ORLOW, I. 2007. Variants in Estrogen Biosynthesis Genes, Sex Steroid Hormone Levels, and Endometrial Cancer: A HuGE Review. *American Journal of Epidemiology*, 165, 235-245.

OMURA, T. and SATO, R. 1964. The Carbon Monoxide-binding Pigment of Liver Microsomes. I. Evidence for its hemoprotein nature. *The Journal of Biological Chemistry*, 239, 2370-2378.

ONICA, T., NICHOLS, K., LARIN, M., et al. 2008. Dexamethasone-Mediated Up-Regulation of Human CYP2A6 Involves the Glucocorticoid Receptor and Increased Binding of Hepatic Nuclear Factor 4{alpha} to the Proximal Promoter. *Molecular Pharmacology*, 73, 451-460.

OSANAI, M. and PETKOVICH, M. 2005. Expression of the Retinoic Acid-Metabolizing Enzyme CYP26A1 Limits Programmed Cell Death. *Molecular Pharmacology*, 67, 1808-1817.

OSBORNE, C. K. 1998. Tamoxifen in the Treatment of Breast Cancer. *The New England Journal of Medicine*, 339, 1609-1618.

O'SHAUGHNESSY, J. 2005. Extending Survival with Chemotherapy in Metastatic Breast Cancer. *Oncologist*, 10, 20-29.

PALMER, C. N. A., RICHARDSON, T. H., GRIFFIN, K. J., et al. 1993. Characterisation of a cDNA encoding a human kidney cytochrome P450 fatty acid ω-hydroxylase and the cognate enzyme expressed in Escherichia coli. *Biochimica et Biophysica Acta*, 1172, 161-166.

PASCUSSI, J. M., GERBAL-CHALOIN, S., DROCOURT, L., et al. 2003. The expression of CYP2B6, CYP2C9 and CYP3A4 genes: a tangle of networks of nuclear and steroid receptors. *Biochimica et Biophysica Acta (BBA) - General Subjects*, 1619, 243-253.

PASCUSSI, J. M., ROBERT, A., NGUYEN, M., et al. 2005. Possible involvement of pregnane X receptor-enhanced CYP24 expression in drug-induced osteomalacia. *The Journal of Clinical Investigation*, 115, 177-186.

PAYNE, A. H. and HALES, D. B. 2004. Overview of Steroidogenic Enzymes in the Pathway from Cholesterol to Active Steroid Hormones. *Endocrine Reviews*, 25, 947-970.

PELKONEN, O., MAENPAA, J., TAAVITSAINEN, P., et al. 1998. Inhibition and induction of human cytochrome P450 (CYP) enzymes. *Xenobiotica*, 28, 1203-1253.

PEREZ, E. A. 2007a. Safety of aromatase inhibitors in the adjuvant setting. *Breast Cancer Research and Treatment*, 105, 75-89.

PEREZ, E. A. 2007b. Safety profiles of tamoxifen and the aromatase inhibitors in adjuvant therapy of hormone-responsive early breast cancer. *Annals of Oncology*, 18, viii26-35.

PICADO-LEONARD, J. and MILLER, W. L. 1987. Structure of the human P450c17 gene is closely related to the P450c21 gene. *American Journal of Human Genetics*, 41, A234.

PISHVAIAN, M. J., FELTES, C. M., THOMPSON, P., et al. 1999. Cadherin-11 Is Expressed in Invasive Breast Cancer Cell Lines. *Cancer Research*, 59, 947-952.

PIVER, B., FER, M., VITRAC, X., et al. 2004. Involvement of cytochrome P450 1A2 in the biotransformation of trans-resveratrol in human liver microsomes. *Biochemical Pharmacology*, 68, 773-782.

POCH, M. T., CUTLER, N. S., YOST, G. S., et al. 2005. Molecular Mechanisms Regulating Human CYP4B1 Lung-Selective Expression. *Drug Metabolism and Disposition*, 33, 1174-1184.

POOLE, E. M., BIGLER, J., WHITTON, J., et al. 2006. Prostacyclin Synthase and Arachidonate 5-Lipoxygenase Polymorphisms and Risk of Colorectal Polyps. *Cancer Epidemiology, Biomarkers & Prevention*, 15, 502-508.

PURNAPATRE, K., KHATTAR, S. K. and SAINI, K. S. 2008. Cytochrome P450s in the development of target-based anticancer drugs. *Cancer Letters*, 259, 1-15.

PUTTI, T. C., EL-REHIM, D. M., RAKHA, E. A., et al. 2005. Estrogen receptor-negative breast carcinomas: a review of morphology and immunophenotypical analysis. *Modern Pathology*, 18, 26-35.

RASTINEJAD, F. 2001. Retinoid X receptor and its partners in the nuclear receptor family. *Current Opinion in Structural Biology*, 11, 33-38.

RAY, W. J., BAIN, G., YAO, M., et al. 1997. CYP26, a Novel Mammalian Cytochrome P450, Is Induced by Retinoic Acid and Defines a New Family. *The Journal of Biological Chemistry*, 272, 18702-18708.

REDDY, M. and GIVEN-WILSON, R. 2004. Screening for breast cancer. *Surgery* (*Oxford*) *Breast*, 22, 155-160.

REKHA, G. K. and SLADEK, N. E. 1997. Multienzyme-mediated stable and transient multidrug resistance and collateral sensitivity induced by xenobiotics. *Cancer Chemotherapy and Pharmacology*, 40, 215-224.

REMÉNYI, A., SCHÖLER, H. R. and WILMANNS, M. 2004. Combinatorial control of gene expression. *Nat Struct Mol Biol*, 11, 812-815.

RESZKA, E., WASOWICZ, W. and GROMADZINSKA, J. 2006. Genetic polymorphism of xenobiotic metabolising enzymes, diet and cancer susceptibility. *The British Journal of Nutrition*, 96, 609-619.

RIEGER, M. A., EBNER, R., BELL, D. R., et al. 2004. Identification of a Novel Mammary-Restricted Cytochrome P450, CYP4Z1, with Overexpression in Breast Carcinoma. *Cancer Research*, 64, 2357-2364.

RIVERA, S. P., SAARIKOSKI, S. T. and HANKINSON, O. 2002. Identification of a Novel Dioxin-Inducible Cytochrome P450. *Molecular Pharmacology*, 61, 255-259.

RIVERA, S. P., WANG, F., SAARIKOSKI, S. T., et al. 2007. A Novel Promoter Element Containing Multiple Overlapping Xenobiotic and Hypoxia Response Elements Mediates Induction of Cytochrome P4502S1 by Both Dioxin and Hypoxia. *The Journal of Biological Chemistry*, 282, 10881-10893.

ROBERTS-THOMSON, S. J. 2000. Peroxisome proliferator-activated receptors in tumorigenesis: Targets of tumour promotion and treatment. *Immunology and Cell Biology*, 78, 436–441.

ROCHEFORT, H., GLONDU, M., SAHLA, M., et al. 2003. How to target estrogen receptor-negative breast cancer? *Endocrine-Related Cancer*, 10, 261-266.

RODRIGUEZ-ANTONA, C. and INGELMAN-SUNDBERG, M. 2006. Cytochrome P450 pharmacogenetics and cancer. *Oncogene*, 25, 1679-1691.

ROOS, P. H. and BOLT, H. M. 2005. Cytochrome P450 interactions in human cancers: new aspects considering CYP1B1. *Expert opinion on Drug Metabolism & Toxicology*, 1, 187-202.

ROTH, A., LOOSER, R., KAUFMANN, M., et al. 2008. Regulatory Crosstalk between Drug Metabolism and Lipid Homeostasis: CAR and PXR Increase Insig-1 Expression. *Molecular Pharmacology*, 73, 1282-9.

ROUSSEAU, C., PETTERSSON, F., COUTURE, M.-C., et al. 2003. The N-terminal of the estrogen receptor (ER[alpha]) mediates transcriptional cross-talk with the retinoic acid receptor in human breast cancer cells. *The Journal of Steroid Biochemistry and Molecular Biology*, 86, 1-14.

ROY, P. and WAXMAN, D. J. 2006. Activation of oxazaphosphorines by cytochrome P450: Application to gene-directed enzyme prodrug therapy for cancer. *Toxicology in Vitro*, 20, 176-186.

ROZEN, S. and SKALETSKY, H. J. 2000. *Primer3 on the WWW for general users and for biologist programmers*, Humana Press, Totowa, NJ.

ROZMAN, D., STROMSTEDT, M., TSUI, L.-C., et al. 1996. Structure and Mapping of the Human Lanosterol 14[alpha]- Demethylase Gene (CYP51) Encoding the Cytochrome P450 Involved in Cholesterol Biosynthesis; Comparison of Exon/Intron Organization with other Mammalian and Fungal CYP Genes. *Genomics*, 38, 371-381.

RUBIN, M., FENIG, E., ROSENAUER, A., et al. 1994. 9-cis Retinoic Acid Inhibits Growth of Breast Cancer Cells and Down-Regulates Estrogen Receptor RNA and Protein. *Cancer Research*, 54, 6549-6556.

RYLANDER, T., NEVE, E. P. A., INGELMAN-SUNDBERG, M., et al. 2001. Identification and tissue distribution of the novel human cytochrome P450 2S1 (CYP2S1). *Biochemical and Biophysical Research Communications*, 281, 529-535.

RYLANDER-RUDQVIST, T., WEDREN, S., GRANATH, F., et al. 2003. Cytochrome P450 1B1 gene polymorphisms and postmenopausal breast cancer risk. *Carcinogenesis*, 24, 1533-1539.

SAARIKOSKI, S. T., RIVERA, S. P., HANKINSON, O., et al. 2005a. CYP2S1: A short review. *Toxicology and Applied Pharmacology*, 207, 62-69.

SAARIKOSKI, S. T., WIKMAN, H. A. L., SMITH, G., et al. 2005b. Localization of Cytochrome P450 CYP2S1 Expression in Human Tissues by In Situ Hybridization and Immunohistochemistry. *The Journal of Histochemistry and Cytochemistry*, 53, 549-556.

SAFE, S., WANG, F., PORTER, W., et al. 1998. Ah receptor agonists as endocrine disruptors: antiestrogen activity and mechanisms. *Toxicology Letters*, 102-103, 343-347.

SAFE, S. and WORMKE, M. 2003. Inhibitory aryl hydrocarbon receptor-estrogen receptor alpha cross-talk and mechanisms of action. *Chemical Research in Toxicology*, 16, 807-816.

SAFE, S., WORMKE, M. and SAMUDI, O. I. 2000. Mechanisms of inhibitory aryl hydrocarbon receptor-estrogen receptor crosstalk in human breast cancer cells. *Journal of Mammary Gland Biology Neoplasia*, 5, 295-306.

SAIDI, S., HOLLAND, C., CHARNOCK-JONES, D. S., et al. 2006. In vitro and in vivo effects of the PPAR-alpha agonists fenofibrate and retinoic acid in endometrial cancer. *Molecular Cancer*, **5**, 13.

SAKAI, H., IWATA, H., KIM, E.-Y., et al. 2006. Constitutive Androstane Receptor (CAR) as a Potential Sensing Biomarker of Persistent Organic Pollutants (POPs) in Aquatic Mammal: Molecular Characterization, Expression Level, and Ligand Profiling in Baikal Seal (Pusa sibirica). *Toxicological Sciences*, 94, 57-70.

SAMBROOK, J. and RUSSELL, D. W. 2001. *Molecular Cloning: A Laboratory Manual,* Cold Spring Harbor Laboratory Press, Cold Spring Harbor, New York.

SARRIO, D., RODRIGUEZ-PINILLA, S. M., HARDISSON, D., et al. 2008. Epithelial-Mesenchymal Transition in Breast Cancer Relates to the Basal-like Phenotype. *Cancer Research*, 68, 989-997.

SAVAS, U., HSU, M.-H., GRIFFIN, K. J., et al. 2005. Conditional regulation of the human CYP4X1 and CYP4Z1 genes. *Archives of Biochemistry and Biophysics*, 436, 377-385.

SAVAS, U., HSU, M.-H. and JOHNSON, E. F. 2003. Differential regulation of human CYP4A genes by peroxisome proliferators and dexamethasone. *Archives of Biochemistry and Biophysics*, 409, 212-220.

SCARBOROUGH, P. E., MA, J., QU, W., et al. 1999. P450 subfamily CYP2J and their role in the bioactivation of arachidonic acid in extrahepatic tissues. *Drug Metabolism Reviews*, 31, 205-234.

SCHIRNER, M. and SCHNEIDER, M. R. 1997. Inhibition of metastasis by cicaprost in rats with established SMT2A mammary carcinoma growth. *Cancer Detection and Prevention*, 21, 44-50.

SCHMIDT, R., BAUMANN, F., KNÜPFER, H., et al. 2004. CYP3A4, CYP2C9 and CYP2B6 expression and ifosfamide turnover in breast cancer tissue microsomes. *British Journal of Cancer*, 90, 911-916.

SCHMITTGEN, T. D. and ZAKRAJSEK, B. A. 2000. Effect of experimental treatment on housekeeping gene expression: validation by real-time, quantitative RT-PCR. *Journal of Biochemical and Biophysical Methods*, 46, 69-81.

SCHROTH, W., ANTONIADOU, L., FRITZ, P., et al. 2007. Breast Cancer Treatment Outcome With Adjuvant Tamoxifen Relative to Patient CYP2D6 and CYP2C19 Genotypes. *Journal of Clinical Oncology*, 25, 5187-5193.

SCHUSTER, I. and BERNHARDT, R. 2007. Inhibition of cytochrome P450: existing and new promising therapeutic targets. *Drug Metabolism Reviews*, 39, 481-499.

SCRIPTURE, C. D. and FIGG, W. D. 2006. Drug interactions in cancer therapy. *Nature Reviews Cancer*, 6, 546-558.

SCRIPTURE, C. D., SPARREBOOM, A. and FIGG, W. D. 2005. Modulation of cytochrome P450 activity: implications for cancer therapy. *The Lancet Oncology*, 6, 780-789.

SEMENZA, G. L. 2000. HIF-1: mediator of physiological and pathophysiological responses to hypoxia. *Journal of Applied Physiology*, 88, 1474-1480.

SETCHELL, K. D. R., SCHWARZ, M., O'CONNELL, N. C., et al. 1998. Identification of a New Inborn Error in Bile Acid Synthesis: Mutation of the Oxysterol 7alpha - Hydroxylase Gene Causes Severe Neonatal Liver Disease. *The Journal of Clinical Investigation*, 102, 1690-1703.

SHAO, W. and BROWN, M. 2004. Advances in estrogen receptor biology: prospects for improvements in targeted breast cancer therapy. *Breast Cancer Research*, 6, 39 - 52.
SHARMA, R. K., DOIG, M. V., LEWIS, D., et al. 1989. Role of hepatic and renal cytochrome P-450-IVA1 in the metabolism of lipic substrates. *Biochemical pharmacology*, 38, 3621-3629.

SHIMADA, T. 2006. Xenobiotic-metabolizing enzymes involved in activation and detoxification of carcinogenic polycyclic aromatic hydrocarbons. *Drug Metabolism and Pharmacokinetics*, 21, 257-276.

SHIMADA, T., HAYES, C. L., YAMAZAKI, H., et al. 1996. Activation of chemically diverse procarcinogens by human cytochrome P-450 1B1. *Cancer Research*, 56, 2979-2984.

SHIMADA, T., YAMAZAKI, H., MIMURA, M., et al. 1994. Interindividual variations in human liver cytochrome P-450 enzymes involved in the oxidation of drugs, carcinogens and toxic chemicals: studies with liver microsomes of 30 Japanese and 30 Caucasians. *The Journal of Pharmacology and Experimental Therapeutics*, 270, 414-423.

SIMPSON, D., CURRAN, M. P. and PERRY, C. M. 2004. Letrozole: a review of its use in postmenopausal women with breast cancer. *Drugs*, 64, 1213-1230.

SINGH, S., CHAKRAVARTI, D., EDNEY, J. A., et al. 2005. Relative imbalances in the expression of estrogen-metabolizing enzymes in the breast tissue of women with breast carcinoma. *Oncology Reports*, 14, 1091-1096.

SMITH, G., HARRISON, D. J., EAST, N., et al. 1993. Regulation of cytochrome P450 gene expression in human colon and breast xenografts. *British journal of cancer*, 68, 57-63.

SMITH, G., WOLF, C. R., DEENI, Y. Y., et al. 2003. Cutaneous expression of cytochrome P450 CYP2S1: individuality in regulation by therapeutic agents for psoriasis and other skin diseases. *Lancet*, 36, 1336-1343.

SOMMERS, C. L., BYERS, S. W., THOMPSON, E. W., et al. 1994. Differentiation state and invasiveness of human breast cancer cell lines. *Breast Cancer Research and Treatment*, 31, 325-335.

SONNEVELD, E., VAN DEN BRINK, C. E., VAN DER LEEDE, B. M., et al. 1998. Human retinoic acid (RA) 4-hydroxylase (CYP26) is highly specific for all-trans-RA and can be induced through RA receptors in human breast and colon carcinoma cells. *Cell Growth & Differentiation*, 9, 629-637.

SONODA, J., PEI, L. and EVANS, R. M. 2008. Nuclear receptors: Decoding metabolic disease. *FEBS Letters Metabolic Disease*, 582, 2-9.

SOULE, H. D., MALONEY, T. M., WOLMAN, S. R., et al. 1990. Isolation and Characterization of a Spontaneously Immortalized Human Breast Epithelial Cell Line, MCF-10. *Cancer Research*, 50, 6075-6086.

SOULE, H. D. and McGRATH, C. M. 1986. A simplified method for passage and longterm growth of human mammary epithelial cells. *In Vitro Cellular & Developmental Biology*, 22, 6-12.

SOULEZ, M. and PARKER, M. 2001. Identification of novel oestrogen receptor target genes in human ZR75-1 breast cancer cells by expression profiling. *Journal of Molecular Endocrinology*, 27, 259-274.

SPINK, D., HAYES, C., YOUNG, N., et al. 1994. The effects of 2,3,7,8tetrachlorodibenzo-p-dioxin on estrogen metabolism in MCF-7 breast cancer cells: evidence for induction of a novel 17 beta-estradiol 4-hydroxylase. *Journal of Steroid Biochemistry*, 51, 251-258.

SPINK, D., LINCOLN, D., DICKERMAN, H., et al. 1990. 2,3,7,8-Tetrachlorodibenzo-pdioxin causes an extensive alteration of 17 beta-estradiol metabolism in MCF-7 breast tumor cells. *Proceedings of the National Academy of Sciences of the United States of America*, 87, 6917-6921.

SPINK, D., SPINK, B., CAO, J., et al. 1998. Differential expression of CYP1A1 and CYP1B1 in human breast epithelial cells and breast tumor cells. *Carcinogenesis*, 19, 291-298.

SPINK, D. C., KATZ, B. H., HUSSAIN, M. M., et al. 2003. Estrogen regulates Ah responsiveness in MCF-7 breast cancer cells. *Carcinogenesis*, 24, 1941-1950.

SPINK, D. C., SPINK, B. C., CAO, J. Q., et al. 1997. Induction of cytochrome P450 1B1 and catechol estrogen metabolism in ACHN human renal adenocarcinoma cells. *The Journal of steroid biochemistry and molecular biology*, 62, 223-232.

STAMM, S., BEN-ARI, S., RAFALSKA, I., et al. 2005. Function of alternative splicing. *Gene*, 344, 1-20.

STAMM, S., ZHU, J., NAKAI, K., et al. 2000. An alternative-exon database and its statistical analysis. *DNA and Cell Biology*, 19, 739-756.

STARK, K., BYLUND, J., TORMA, H., et al. 2005. On the mechanism of biosynthesis of 19-hydroxyprostaglandins of human seminal fluid and expression of cyclooxygenase-2, PGH 19-hydroxylase (CYP4F8) and microsomal PGE synthase-1 in seminal vesicles and vas deferens. *Prostaglandins Other Lipid Mediat*, 75, 47-64.

STARK, K., TORMA, H., CRISTEA, M., et al. 2003. Expression of CYP4F8 (prostaglandin H 19-hydroxylase) in human epithelia and prominent induction in epidermis of psoriatic lesions. *Archives of Biochemistry and Biophysics*, 409, 188-196.

ST-ARNAUD, R., MESSERLIAN, S., MOIR, J. M., et al. 1997. The 25-hydroxyvitamin D 1-alpha-hydroxylase gene maps to the pseudovitamin D-deficiency rickets (PDDR) disease locus. *The Journal of Bone and Mineral Research*, 12, 1552-1559.

STERLING, K. M. J. and CUTRONEO, K. R. 2004. Constitutive and inducible expression of cytochromes P4501A (CYP1A1 and CYP1A2) in normal prostate and prostate cancer cells. *Journal of Cellular Biochemistry*, 91, 423-429.

STIBOROVA, M., ARLT, V. M., HENDERSON, C. J., et al. 2008. Role of hepatic cytochromes P450 in bioactivation of the anticancer drug ellipticine: Studies with the hepatic NADPH:Cytochrome P450 reductase null mouse. *Toxicology and Applied Pharmacology*, 226, 318-327.

STIBOROVA, M., BIELER, C. A., WIESSLER, M., et al. 2001. The anticancer agent ellipticine on activation by cytochrome P450 forms covalent DNA adducts. *Biochemical Pharmacology*, 62, 1675-1684.

STOFF-KHALILI, M. A., DALL, P. and CURIEL, D. T. 2006. Gene therapy for carcinoma of the breast. *Cancer Gene Therapy*, 13, 633-647.

STROMSTEDT, M., WATERMAN, M. R., HAUGEN, T. B., et al. 1998. Elevated Expression of Lanosterol 14{alpha}-Demethylase (CYP51) and the Synthesis of Oocyte Meiosis-Activating Sterols in Postmeiotic Germ Cells of Male Rats. *Endocrinology*, 139, 2314-2321.

SU, T., BAO, Z., ZHANG, Q.-Y., et al. 2000. Human cytochrome P450 CYP2A13: predominant expression in the respiratory tract and its high efficacy metabolic activation of a tobacco-specific carcinogen, 4-(methylnitrosamino)-1(3-pyridyl)-1-butanone. *Cancer Research*, 60, 5074-5079.

SUCHANEK, K. M., MAY, F. J., ROBINSON, J. A., et al. 2002. Peroxisome proliferatoractivated receptor alpha in the human breast cancer cell lines MCF-7 and MDA-MB-231. *Molecular Carcinogenesis*, 34, 165-171.

SUNDARAM, S., BECKMAN, M. J., BAJWA, A., et al. 2006. QW-1624F2-2, a synthetic analogue of 1,25-dihydroxyvitamin D3, enhances the response to other deltanoids and suppresses the invasiveness of human metastatic breast tumor cells. *Molecular Cancer Therapeutics*, 5, 2806-2814.

SUZUKI, T., MIKI, Y., NAKAMURA, Y., et al. 2005. Sex steroid-producing enzymes in human breast cancer. *Endocrine-Related Cancer*, 12, 701-720.

SWALES, K. E., KORBONITS, M., CARPENTER, R., et al. 2006. The Farnesoid X Receptor Is Expressed in Breast Cancer and Regulates Apoptosis and Aromatase Expression. *Cancer Research*, 66, 10120-10126.

TAIMI, M., HELVIG, C., WISNIEWSKI, J., et al. 2004. A Novel Human Cytochrome P450, CYP26C1, Involved in Metabolism of 9-cis and All-trans Isomers of Retinoic Acid. *The Journal of Biological Chemistry*, 279, 77-85.

TAKATA, Y., MASKARINEC, G. and LE MARCHAND, L. 2007. Breast density and polymorphisms in genes coding for CYP1A2 and COMT: the Multiethnic Cohort. *BMC Cancer*, 7, 30.

TAKEYAMA, K., KITANAKA, S., SATO, T., et al. 1997. 25-Hydroxyvitamin D<sub>3</sub> 1alphahydroxylase and vitamin D synthesis. *Science*, 277, 1827-1830.

TANAKA, TERADA and MISAWA. 2000. Cytochrome P450 2E1: its clinical and toxicological role. *Journal of Clinical Pharmacy and Therapeutics*, 25, 165-175.

TANG, W., NORLIN, M. and WIKVALL, K. 2007. Regulation of human CYP27A1 by estrogens and androgens in HepG2 and prostate cells. *Archives of Biochemistry and Biophysics*, 462, 13-20.

TANG, Y. M., WO, Y.-Y. P., STEWART, J., et al. 1996. Isolation and Characterization of the Human Cytochrome P450 CYP1B1 Gene. *The Journal of Biological Chemistry*, 271, 28324-28330.

TANIGUCHI, R., KUMAI, T., MATSUMOTO, N., et al. 2005. Utilization of Human Liver Microsomes to Explain Individual Differences in Paclitaxel Metabolism by CYP2C8 and CYP3A4. *Journal of Pharmacological Sciences*, 97, 83-90.

THOMAS, J. H. 2007. Rapid Birth-Death Evolution Specific to Xenobiotic Cytochrome P450 Genes in Vertebrates. *PLoS Genetics*, 3, e67.

THOMAS, R. D., GREEN, M. R., WILSON, C., et al. 2006. Cytochrome P450 expression and metabolic activation of cooked food mutagen 2-amino-1-methyl-6-phenylimidazo[4,5-b]pyridine (PhIP) in MCF10A breast epithelial cells. *Chemico-Biological Interactions*, 160, 204-216.

THUM, T. and BORLAK, J. 2000. Gene expression in distinct regions of the heart. *Lancet*, 355, 979-983.

TIAN, W. D., WELLS, A. V., CHAMPION, P. M., et al. 1995. Measurements of CO Geminate Recombination in Cytochromes P450 and P420. *The Journal of Biological Chemistry*, 270, 8673-8679.

TITCOMB, M., GOTTARDIS, M., PIKE, J., et al. 1994. Sensitive and specific detection of retinoid receptor subtype proteins in cultured cell and tumor extracts. *Molecular Endocrinology*, 8, 870-877.

TORANZOS, G. A. 1997. *Environmental Applications of Nucleic Acid Amplification Techniques*, Technomic Publishing Company, Inc., Pennsylvania, U.S.A.

TOTAH, R. A. and RETTIE, A. E. 2005. Cytochrome P450 2C8: substrates, inhibitors, pharmacogenetics, and clinical relevance. *Clinical Pharmacology & Therapeutics*, 77, 341-352.

TOURNEL, G., CAUFFIEZ, C., BILLAUT-LADEN, I., et al. 2007a. Molecular analysis of the CYP2F1 gene: Identification of a frequent non-functional allelic variant. *Mutation Research/Fundamental and Molecular Mechanisms of Mutagenesis*, 617, 79-89.

TOURNEL, G., CAUFFIEZ, C., LECLERC, J., et al. 2007b. CYP2F1 genetic polymorphism: Identification of interethnic variations. *Xenobiotica*, 37, 1433 - 1438.

TOWNSEND, K., BANWELL, C. M., GUY, M., et al. 2005. Autocrine Metabolism of Vitamin D in Normal and Malignant Breast Tissue. *Clinical Cancer Research*, 11, 3579-3586.

TRASK, B., FERTITTA, A., CHRISTENSEN, M., et al. 1993. Fluorescence in Situ Hybridization Mapping of Human Chromosome 19: Cytogenetic Band Location of 540 Cosmids and 70 Genes or DNA Markers. *Genomics*, 15, 133-145.

TROFIMOVA-GRIFFIN, M. E. and JUCHAU, M. R. 2002. Developmental expression of cytochrome CYP26B1 (P450RAI-2) in human cephalic tissues. *Developmental Brain Research*, 136, 175-178.

TSUCHIYA, Y., NAKAJIMA, M., ITOH, S., et al. 2003. Expression of Aryl Hydrocarbon Receptor Repressor in Normal Human Tissues and Inducibility by Polycyclic Aromatic Hydrocarbons in Human Tumor-Derived Cell Lines. *Toxicological Sciences*, 72, 253-259.

TSUCHIYA, Y., NAKAJIMA, M., KYO, S., et al. 2004. Human CYP1B1 Is Regulated by Estradiol via Estrogen Receptor. *Cancer Research*, 64, 3119-3125.

TSUCHIYA, Y., NAKAJIMA, M. and YOKOI, T. 2005. Cytochrome P450-mediated metabolism of estrogens and its regulation in human. *Cancer Letters*, 227, 115-124.

TSUNEDOMI, R., IIZUKA, N., HAMAMOTO, Y., et al. 2005. Patterns of expression of cytochrome P450 genes in progression of hepatitis C virus-associated hepatocellular carcinoma. *International Journal of Oncology*, 27, 661-667.

TURNER, P. C. 2005. Molecular biology, Taylor & Francis, New York, USA.

TURUNEN, M. M., DUNLOP, T. W., CARLBERG, C., et al. 2007. Selective use of multiple vitamin D response elements underlies the 1 alpha,25-dihydroxyvitamin D3-mediated negative regulation of the human CYP27B1 gene. *Nucleic Acids Research*, 35, 2734-2747.

VAN HEUSDEN, J., WOUTERS, W., RAMAEKERS, F. C., et al. 1998. The antiproliferative activity of all-trans-retinoic acid catabolites and isomers is differentially modulated by liarozole-fumarate in MCF-7 human breast cancer cells. *British Journal of Cancer*, 77, 1229-1235.

VILLANI, M. G., APPIERTO, V., CAVADINI, E., et al. 2004. Identification of the Fenretinide Metabolite 4-Oxo-Fenretinide Present in Human Plasma and Formed in Human Ovarian Carcinoma Cells through Induction of Cytochrome P450 26A1. *Clinical Cancer Research*, 10, 6265-6275.

VOGEL, V. G., COSTANTINO, J. P., WICKERHAM, D. L., et al. 2006. Effects of tamoxifen vs raloxifene on the risk of developing invasive breast cancer and other disease outcomes: the NSABP Study of Tamoxifen and Raloxifene (STAR) P-2 trial. *JAMA*, 295, 2727-2741.

VON WEYMARN, L. B., CHUN, J. A. and HOLLENBERG, P. F. 2006. Effects of benzyl and phenethyl isothiocyanate on P450s 2A6 and 2A13: potential for chemoprevention in smokers. *Carcinogenesis*, 27, 782-790.

VYHLIDAL, C. A., ROGAN, P. K. and LEEDER, J. S. 2004. Development and Refinement of Pregnane X Receptor (PXR) DNA Binding Site Model Using Information Theory: Insights Into PXR-Mediated Gene Regulation. *The Journal of Biological Chemistry*, 279, 46779-46786.

WANG, H., FAUCETTE, S., MOORE, R., et al. 2004. Human Constitutive Androstane Receptor Mediates Induction of CYP2B6 Gene Expression by Phenytoin. *The Journal of Biological Chemistry*, 279, 29295-29301.

WANG, J., PITARQUE, M. and INGELMAN-SUNDBERG, M. 2006. 3'-UTR polymorphism in the human CYP2A6 gene affects mRNA stability and enzyme expression. *Biochemical and Biophysical Research Communications*, 340, 491-497.

WANG, W., PORTER, W., BURGHARDT, R., et al. 1997. Mechanism of inhibition of MDA-MB-468 breast cancer cell growth by 2,3,7,8-tetrachlorodibenzo-p-dioxin. *Carcinogenesis*, 18, 925-933.

WANG, J. and YEN, A. 2004. A Novel Retinoic Acid-Responsive Element Regulates Retinoic Acid-Induced BLR1 Expression. *Molecular and Cellular Biology*, 24, 2423-2443.

WAPNIR, I. L., ANDERSON, S. J., MAMOUNAS, E. P., et al. 2006. Prognosis After Ipsilateral Breast Tumor Recurrence and Locoregional Recurrences in Five National Surgical Adjuvant Breast and Bowel Project Node-Positive Adjuvant Breast Cancer Trials. *Journal of Clinical Oncology*, 24, 2028-2037.

WATERMAN, M. R. and LEPESHEVA, G. I. 2005. Sterol 14[alpha]-demethylase, an abundant and essential mixed-function oxidase. *Biochemical and Biophysical Research Communications*, 338, 418-422.

WAXMAN, D. J. 1999. P450 Gene Induction by Structurally Diverse Xenochemicals: Central Role of Nuclear Receptors CAR, PXR, and PPAR. *Archives of Biochemistry and Biophysics*, 369, 11-23.

WEIGELT, B., PETERSE, J. L. and VAN'T VEER, L. J. 2005. Breast cancer metastasis: markers and models. *Nature Reviews Cancer*, 5, 591-602.

WESTLIND, A., MALMEBO, S., JOHANSSON, I., et al. 2001. Cloning and Tissue Distribution of a Novel Human Cytochrome P450 of the CYP3A Subfamily, CYP3A43. *Biochemical and Biophysical Research Communications*, 281, 1349-1355.

WHITE, J. A., BECKETT, B., SCHERER, S. W., et al. 1998. P450RAI (CYP26A1) Maps to Human Chromosome 10q23-q24 and Mouse Chromosome 19C2-3. *Genomics*, 48, 270-272.

WHITE, J. A., RAMSHAW, H., TAIMI, M., et al. 2000. Identification of the human cytochrome P450, P450RAI-2, which is predominantly expressed in the adult cerebellum and is responsible for all-trans-retinoic acid metabolism. *Proceedings of the National Academy of Sciences*, 97, 6403-6408.

WIJESURIYA, S. D., ZHANG, G., DARDIS, A., et al. 1999. Transcriptional regulatory elements of the human gene for cytochrome P450c21 (steroid 21-hydroxylase) lie within intron 35 of the linked C4B gene. *The Journal of Biological Chemistry*, 274, 38097–38106.

WRIGHTON, S. A. and STEVENS, J. C. 1992. The human hepatic cytochromes P450 involved in drug metabolism. *Critical Reviews in Toxicology*, 22, 1-21.

WU, M., CHEN, S. and WU, X. 2006. Differences in cytochrome P450 2C19 (CYP2C19) expression in adjacent normal and tumor tissues in Chinese cancer patients. *Medical Science Monitor*, 12, BR174-178.

WU, S., MOOMAW, C. R., TOMER, K. B., et al. 1996. Molecular Cloning and Expression of CYP2J2, a Human Cytochrome P450 Arachidonic Acid Epoxygenase Highly Expressed in Heart. *The Journal of Biological Chemistry*, 271, 3460-3468.

WU, Z., MARTIN, K. O., JAVITT, N. B., et al. 1999. Structure and functions of human oxysterol 7{alpha}-hydroxylase cDNAs and gene CYP7B1. *Journal of Lipid Research*, 40, 2195-2203.

XIE, W. and EVANS, R. M. 2001. Orphan Nuclear Receptors: The Exotics of Xenobiotics. *The Journal of Biological Chemistry*, 276, 37739-37742.

YAGER, J. D. and DAVIDSON, N. E. 2006. Estrogen Carcinogenesis in Breast Cancer. *The New England Journal of Medicine*, 354, 270-282.

YAGER, J. D. and LEIHR, J. G. 1996. Molecular Mechanisms of Estrogen Carcinogenesis. *Annual Review of Pharmacology and Toxicology*, 36, 203-232.

YANG, X., SOLOMON, S., FRASER, L. R., et al. 2007. Constitutive regulation of CYP1B1 by the aryl hydrocarbon receptor (AhR) in pre-malignant and malignant mammary tissue. *Journal of Cellular Biochemistry*, Epub ahead of print.

YASPAN, B. L., BREYER, J. P., CAI, Q., et al. 2007. Haplotype Analysis of CYP11A1 Identifies Promoter Variants Associated with Breast Cancer Risk. *Cancer Research*, 67, 5673-5682.

YEE, S. W., CAMPBELL, M. J. and SIMONS, C. 2006. Inhibition of Vitamin D3 metabolism enhances VDR signalling in androgen-independent prostate cancer cells. *The Journal of Steroid Biochemistry and Molecular Biology*, 98, 228-235.

YOKOYAMA, C., YABUKI, T., INOUE, H., et al. 1996. Human gene encoding prostacyclin synthase (PTGIS): genomic organization, chromosomal localization, and promotor activity. *Genomics*, 36, 296-304.

YUN, C. H., SHIMADA, T. and GUENGERICH, F. P. 1991. Purification and characterization of human liver microsomal P-4502A6. *Molecular pharmacology*, 40, 679-685.

ZANGER, U. M., KLEIN, K., SAUSSELE, T., et al. 2007. Polymorphic CYP2B6: molecular mechanisms and emerging clinical significance. *Pharmacogenomics*, 8, 743-759.

ZEHNDER, D., BLAND, R., WILLIAMS, M. C., et al. 2001. Extrarenal Expression of 25-Hydroxyvitamin D3-1{{alpha}}-Hydroxylase. *The Journal of Clinical Endocrinology and Metabolism*, 86, 888-894.

ZEIGLER-JOHNSON, C., FRIEBEL, T., WALKER, A. H., et al. 2004. CYP3A4, CYP3A5, and CYP3A43 Genotypes and Haplotypes in the Etiology and Severity of Prostate Cancer. *Cancer Research*, 64, 8461-8467.

ZHANG, X., D'AGOSTINO, J., WU, H., et al. 2007. CYP2A13: Variable Expression and Role in Human Lung Microsomal Metabolic Activation of the Tobacco-Specific Carcinogen 4-(Methylnitrosamino)-1-(3-pyridyl)-1-butanone. *The Journal of Pharmacology and Experimental Therapeutics*, 323, 570-578.

ZHANG, X. and HARDWICK, J. P. 2000. Regulation of CYP4F2 leukotriene B4 omegahydroxylation by retinoic acid in HepG2 cells. *Biochemical and Biophysical Research Communications*, 279, 864-871.

ZHANG, Z., BURCH, P. E., COONEY, A. J., et al. 2004. Genomic Analysis of the Nuclear Receptor Family: New Insights Into Structure, Regulation, and Evolution From the Rat Genome. *Genome Research*, 14, 580-590.

ZHAO, X. and IMIG, J. D. 2003. Kidney CYP450 enzymes: biological actions beyond drug metabolism. *Current Drug Metabolism*, 4, 73-84.

ZHOU, C., ASSEM, M., TAY, J. C., et al. 2006. Steroid and xenobiotic receptor and vitamin D receptor crosstalk mediates CYP24 expression and drug-induced osteomalacia. *The Journal of Clinical Investigation*, 116, 1703-1712.

ZHU-GE, J. and YU, Y. N. 2004. Enzyme activity analysis of CYP2C18 with exon 5 skipped. *Acta Pharmacologica Sinica*, 25, 1065-1069.

## Appendix A

Table A.1: List of	<sup>r</sup> chemicals	and reagents.
--------------------	------------------------	---------------

Supplier	Chemical/reagent (Catalogue number)
Fisher Scientific Ltd.	<ul> <li>Agarose (BPE1356-100)</li> </ul>
(Leicestershire, UK)	<ul> <li>Ethylenediaminetetraacetic acid (EDTA) (BPE120-500)</li> <li>Tris base (T/P/630/53)</li> </ul>
Invitrogen Ltd.	• 0.4% (w/v) trypan blue stain (15250-061)
(Paisley, UK)	<ul> <li>DMEM/Ham's F12 with Glutamax<sup>TM</sup>I (31331-093)</li> <li>Dulbecco's Modified Eagle's Medium (DMEM) with Glutamax<sup>TM</sup>I (61965-026)</li> </ul>
	• Epidermal growth factor (13247-051)
	<ul> <li>Foetal bovine serum (FBS) (10108-165)</li> <li>Horse serum (26050-070)</li> </ul>
	<ul> <li>Insulin, human recombinant Zinc (12585-014)</li> </ul>
	• Leiboitz's (L15) with Glutamax <sup>TM</sup> I (31415-086)
	• Low melting point agarose gel (15517-014)
	<ul> <li>Mixtures of enicillin (10,000 U/mL) and streptomycin (10,000</li> </ul>
	µg/mL) solution (15140-122)
	• RPMI 1640 with (61870-010)
	Sodium pyruvate MEM (11360-039)
	<ul> <li>Trypsin-EDTA (0.05% w/v trypsin with 0.53 mM EDTA•4Na) (25300-054)</li> </ul>
Promega Ltd.	PCR markers 50-1000 bp (G3161)
(Southamton, UK)	
Qiagen Ltd. (Crawley, UK)	<ul> <li>RNeasy Plus Mini Column kit (74134)</li> </ul>
Ouadratech Diagnosis Ltd.	<ul> <li>Cholera toxin (101B)</li> </ul>
(Surrey, UK)	
Roche Diagnostics Ltd.	■ <i>Taq</i> DNA polymerase (11418432001)
(Lewes, UK)	<ul> <li>Deoxynucleotide triphosphates (dNTPs) (11581295001)</li> </ul>
	High pure PCR product purification kit (11732668001)
Sigma Aldrich Ltd.	<ul> <li>3 M sodium acetate buffer solution (S7899)</li> </ul>
(Poole, UK)	<ul> <li>Actinomycin D (A9415)</li> </ul>
	<ul> <li>Boric acid (B6768)</li> </ul>
	<ul> <li>Cycloheximide (C1988)</li> </ul>
	<ul> <li>Dexamethasone (D4902)</li> </ul>
	<ul> <li>Diethyl pyrocarbonate (DEPC) (D5758)</li> </ul>
	<ul> <li>Dimethylsulphoxide (DMSO) (D5879)</li> </ul>
	• Ethanol (E7023)
	• Ethidium bromide (E1385)
	• Hydrocortisol solution (H6909)
	<ul> <li>Isopropanol (19516)</li> <li>Malance Marine Landersia sime (M MLV) researce transmisters</li> </ul>
	• Moloney Murine Leukemia virus (M-MLV) reverse transcriptase (M1302)
	Phosphate buffered saline (PBS) tablets (P4417)
	<ul> <li>Pirinixic acid (C7081)</li> </ul>
	<ul> <li>Restriction endonuclease: <i>Alu</i>I (R6885), <i>Eco</i>RI (R6265), <i>Hae</i>III (R56280), <i>MspI</i> (R4506), <i>PstI</i> (R7023), <i>RsaI</i> (R4756), <i>SspI</i> (R6759)</li> </ul>
	and $xno1$ (K65/9)
	<ul> <li>p-mercaptoethanol (M6250)</li> <li>Progesterone (P8783)</li> </ul>
Sigma-Genosys Ltd.	<ul> <li>All the specific oligonucleotide primers</li> </ul>
(Haverhill, UK)	<ul> <li>Oligo P(dT)<sub>13</sub> primers</li> </ul>

### **Appendix B**

#### **B.1** Preparation of RNase-free solution

Ribonuclease (RNase) is a major concern in molecular biology experiments as it is a nuclease that catalyses the hydrolysis of RNA into smaller components, leading to a short lifespan for any RNA. Diethyl pyrocarbonate (DEPC), a histidine-specific alkylating agent, is an effective nuclease inhibitor. DEPC treated (RNase-free) solution is routinely used for handling RNA, effectively reducing the risk of RNA being degraded by environmental RNases. Typically a 0.1% solution is used to inactive the environmental RNases by the covalent modifications of the histidine residues.

All solutions used in this study for PCR or molecular biology techniques were pretreated 0.1% (v/v) DEPC. A 0.1 mL aliquot of DEPC was added into 100 mL of the solution to be treated. As DEPC is extremely viscous so the mixture was stirred for 2 hours to bring the DEPC into solution prior to autoclaving for 15 minutes at 15  $lb/in^2$  to inactivate traces of DEPC

# Appendix C

Table C.1:	<b>Components</b>	of the	digestion	buffers.
------------	-------------------	--------	-----------	----------

Buffer for restriction enzymes	Component (1x)
Buffer SA (For <i>Alu</i> I)	<ul> <li>33 mM Tris-acetate</li> <li>66 mM Potassium acetate</li> <li>10 mM Magnesium acetate</li> <li>0.5 mM dithiothreitol (DTT)</li> </ul>
	■ pH 7.9
Buffer SB	• 10 mM Ins-HCl
(For BamHI)	<ul> <li>5 mM Magnesium chloride</li> <li>1 mM 2-Mercaptoethanol</li> <li>pH 8.0</li> </ul>
Buffer SH	• 50mM Tris-HCl
(For <i>Eco</i> RI, <i>Pst</i> I and <i>Ssp</i> I)	<ul> <li>100 mM NaCl</li> <li>10 mM MgCl<sub>2</sub></li> <li>1 mM dithioerythritol</li> <li>pH 7.5</li> </ul>
Buffer SL	• 10 mM Tris-HCl
(For <i>Rsa</i> I and <i>Msp</i> I)	<ul> <li>10 mM MgCl<sub>2</sub></li> <li>1 mM dithioerythritol (DTE)</li> <li>pH 7.5</li> </ul>
Buffer SM	• 10 mM Tris-HCl
(For <i>Hae</i> III)	<ul> <li>50 mM NaCl</li> <li>10 mM MgCl<sub>2</sub></li> <li>1 mM dithioerythritol (DTE)</li> <li>pH 7.5</li> </ul>

### **Appendix D**

PCR product CYP2S1 mRNA	1 365	GTAGCGATGCT GGAAGGGACTTTTGAT GGCCAT GGGGTTTTCTTCTCCAACGGGGAGCGG 60
PCR product C <b>YP2S1 mRNA</b>	61 425	T GGAGGCAGCT GAGGAAGTTTACCATGCTT GCT CT GCGGGACCTGGGCATGGGGAAGCGA 120
PCR product C <b>YP2S1 mRNA</b>	121 485	GAAGGCGAGGAGCTGATCCAGGCGGAGGCCCGGTGTCTGGTGGAGACATTCCAGGGGACA 180
PCR product CYP2S1 mRNA	181 545	GAA 183     GAA 547

**Figure D.1:** Sequence alignment of the CYP2S1 PCR product against NM\_030622 (published human CYP2S1 mRNA) [a 100% homology (183 nucleotides)].

PCR product	GTTGGACAGTACCCTGAGAACTCATACCTGAAAAGCAGTGTTCAACCTTAAGCAAAATCT	66 670
CTP421 MRNA (159 bp)		104
CVD471 mDND (150 bm)		720
PCP product		129
CYP4Z1 mRNA (159 bp)	Alige 744	

**Figure D.2:** Sequence alignment of the CYP4Z1 PCR product (159 bp) against NM\_178134 (published human CYP4Z1 mRNA) [a 94% homology (118/125 nucleotides)].

PCR product		TGGTTGGAGATTTTGCTAAGGTTGAACACTGCTTTCAGGT	б4
CYP4Z1 mRNA (250 bp)	TAGAAAATT-GTTCATGCG0	TGGTTGGAGATTTTGCTAAGGTTGAACACTGCTTTCAGGT	635
PCR product	ATGAGTCCAGGGTACTGT	82	
CYP4Z1 mRNA (250 bp)	ATGAGTCCAGGGTACTGT	617	

**Figure D.3:** Sequence alignment of the CYP4Z1 PCR product (250 bp) against NM\_178134 (published human CYP4Z1 mRNA) [a 98% homology (77/78 nucleotides)].

PCR product	GAGGCACACAAACTCC	ACTCCCAGCACATTCATCACGCGGAACCGAGATGATGTATCCAG	249
ERα mRNA (208 bp)	GAGGCACACAAACTCC'	ICTCCCTGCAGATTCATCATGCGGAACCGAGATGATGTAGCCAG	1647
PCR product	CAGCATGACGAAGAT	264	
ERa mRNA (208 bp)	CAGCATGTCGAAGAT	1632	

**Figure D.4:** Sequence alignment of the ER $\alpha$  PCR product (208 bp) against NM\_000125 (published human ER $\alpha$  mRNA) [a 92% homology (69/75 nucleotides)].

Appendices